UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 or 15(d) of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported) November 26, 2012

AVALONBAY COMMUNITIES, INC.

(Exact Name of Registrant as Specified in Its Charter)

Maryland

(State or Other Jurisdiction of Incorporation)

1-12672 (Commission File Number) 77-0404318 (I.R.S. Employer Identification No.)

671 N. Glebe Road, Suite 800, Arlington, Virginia (Address of Principal Executive Offices)

22203 (Zip Code)

(703) 329-6300

(Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:

- Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
- o Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
- o Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
- o Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Table of Contents

Item 1.01 Entry into A Material Definitive Agreement.

The Archstone Portfolio Acquisition

On November 26, 2012, we entered into an asset purchase agreement, or the Purchase Agreement, with (i) Equity Residential and its operating partnership, ERP Operating Partnership LP, or, collectively, Equity Residential, (ii) Lehman Brothers Holdings Inc., or Lehman, and (iii) Archstone Enterprise LP, or Archstone, pursuant to which we and Equity Residential will acquire, directly or indirectly, all of the assets and entities owned by, and all of the liabilities of, Archstone (other than certain excluded liabilities).

Pursuant to the Purchase Agreement and separate arrangements between us and Equity Residential governing the allocation of liabilities to be assumed under the Purchase Agreement, our portion of consideration under the Purchase Agreement is approximately \$6.9 billion, and consists of the following:

- the issuance of 14,889,706 shares of our common stock to Archstone, valued at \$1.9 billion, using the closing price for our common stock on the New York Stock Exchange on November 23, 2012 of \$128.54;
- \$669.0 million in cash;
- the assumption of indebtedness with a fair value of approximately \$4.1 billion, consisting of \$3.7 billion principal amount for consolidated borrowings, \$238.3 million principal amount for our proportionate share of debt related to unconsolidated joint ventures, and \$197.5 million representing the amount by which the fair value of the aforementioned debt exceeds the principal face value;
- · an obligation to pay, when presented for redemption from time to time, approximately \$132.2 million in respect of the liquidation value of and accrued dividends on outstanding Archstone preferred units; and
- the assumption of 40% of all other liabilities, known or unknown, of Archstone, other than certain excluded liabilities.

For this consideration, we will acquire a portfolio of direct and indirect interests in apartment communities and net other liabilities that we refer to as the Archstone Portfolio Acquisition.

The Archstone Portfolio Acquisition represents a rare opportunity for us to buy a large portfolio of high-quality apartment communities that is concentrated in our existing high barrier-to-entry markets and is consistent with our strategic objective of more deeply penetrating our chosen markets with a broader range of products and services. For the nine months ended September 30, 2012, 98% of the net operating income, or NOI, from the consolidated apartment communities we expect to acquire in the Archstone Portfolio Acquisition is generated in our existing markets. The Archstone Portfolio Acquisition consists of the following:

- 66 consolidated apartment communities, containing 22,222 apartment homes, of which six communities are under construction and are expected to contain 1,666 apartment homes upon completion;
- · three parcels of land, which are consolidated, and if developed as expected, will contain a total of 968 apartment homes;
- interests in unconsolidated joint ventures in which we expect to be the general partner or managing member, which own 10 apartment communities containing 2,040 apartment homes, of which one community is under construction and is expected to contain 157 apartment homes upon completion; and
- a 40% ownership interest in unconsolidated joint venture arrangements with Equity Residential which will hold assets that we will jointly manage, sell to third parties, and/or subsequently transfer to Equity Residential or to us.

The following table provides the approximate allocation of the Company's investment in the Archstone Portfolio Acquisition:

Archstone Portfolio Acquisition	 Acquisition Value(1)
	(in thousands)
Consolidated stabilized assets	\$ 6,140,323
Development communities under construction	308,819
Land held for future development	49,800
Net equity in unconsolidated joint ventures plus allocated joint venture debt	410,112
Total	\$ 6,909,054

(1) Value is based on the closing price of our common stock on November 23, 2012 and the fair market value of debt we expect to assume.

2

Table of Contents

Under the Purchase Agreement, we will acquire 40% of the assets and liabilities of Archstone and Equity Residential will acquire the remaining 60% of the assets and liabilities of Archstone. We refer herein to 40% and 60% as our respective pro rata shares. We and Equity Residential are jointly and severally liable for most obligations to Lehman under the Purchase Agreement. The Purchase Agreement provides that the closing of the acquisitions thereunder must occur within 120 days after execution of the Purchase Agreement. If we and Equity Residential Fail to close the acquisition, then Equity Residential and we could be liable for payment of a termination fee of \$800.0 million (or \$650.0 million if the Purchase Agreement is terminated in the first 60 days after signing) as discussed Under "Purchase Agreement and Related Arrangements." The Archstone Portfolio Acquisition is also subject to customary closing conditions, which do not include our and Equity Residential's ability to obtain the necessary financing or lender consents for the transaction. Neither we nor Equity Residential could terminate the Purchase Agreement because of a lack of financing or lender consents without incurring the termination fee.

The Archstone Portfolio

We believe the Archstone Portfolio Acquisition will create strategic, portfolio and financial benefits for us, including the following:

- · It will increase our ownership of high-quality apartment communities in our existing markets.
- · It will allow us to more closely align our portfolio allocation with our long-term geographic allocation goals.
- It will expand our portfolio into complementary submarkets with more varied product offerings.
- · It will enable us to more rapidly implement our multi-branding strategy.
- · It will provide enhanced corporate efficiencies due to our increased scale.

High-Quality Portfolio Concentrated in Our Markets. The following table provides the percentage of NOI generated during the nine months ended September 30, 2012, by region for our consolidated apartment communities, as well as communities that were under construction, or for which substantial redevelopment is planned or that occurred during 2012, pro forma for the Archstone Portfolio Acquisition (dollars in thousands);

	Avalor	пВау	Archstone Portfolio(3)			Pro Forma		
Region/Portfolio (1)	NOI YTD 2012(2)	% of Total NOI		NOI YTD 2012(2)	% of Total NOI	NOI YTD 2012(2)	% of Total NOI	
New England	\$ 99,277	19%	\$	7,433	3%	\$ 106,710	14%	
Metro NY/NJ	156,135	29%		31,336	14%	187,471	25%	
Mid-Atlantic	71,401	14%		61,727	29%	133,128	18%	
Pacific Northwest	27,241	5%		5,471	3%	32,712	4%	
Northern California	102,335	19%		31,544	15%	133,879	18%	
Southern California	74,287	14%		72,910	34%	147,197	20%	
Non-Core and Other	447	0%		5,227	2%	5,674	1%	
Total NOI	\$ 531,123	100%	\$	215,648	100%	\$ 746,771	100%	

⁽¹⁾ NOI for the nine months ended September 30, 2012 for consolidated apartment communities, excluding NOI from apartment communities in joint ventures. NOI is a non-GAAP financial measure. For a description of how we define NOI and a reconciliation of NOI to net income, please see "Reconciliation of Non-GAAP Financial Measures" beginning on page F-14 of this report.

- (2) GAAP net income for AvalonBay, the Archstone Portfolio Acquisition, and on a pro forma basis, assuming the acquisition of the portfolio, was \$301,178, \$43,425 and \$344,603, respectively for the nine months ended September 30, 2012.
- (3) NOI for the nine months ended September 30, 2012 for this portfolio includes \$26 for communities for which construction is not complete but leasing activity has begun.

For the nine months ended September 30, 2012, pro forma for the Archstone Portfolio Acquisition, 99% of the NOI from our apartment communities would be concentrated in our existing high barrier-to-entry markets.

In connection with the Archstone Portfolio Acquisition, we will acquire interests in certain non-core assets or communities in non-core markets for us, including interests in assets in Texas, Florida and Germany. Some of these interests will be held and managed in a joint venture with Equity Residential, pending disposition or assignment to Equity Residential or us.

Achieving Geographic Portfolio Allocation Goals. We believe that the Archstone Portfolio Acquisition will allow us to bring our overall portfolio more closely in line with our long-term goals for regional allocation. In particular, our current Southern California portfolio is expected to grow from approximately 10,400 apartment homes at September 30, 2012 to approximately 19,200 apartment homes pro forma for the Archstone Portfolio Acquisition. Other geographic allocation adjustments expected to be achieved through the Archstone Portfolio Acquisition include a reduction in our relative concentration in the New England and Metro New York/New Jersey regions (where relatively few of the Archstone Portfolio Acquisition apartment communities to be acquired by us are located) and an increase in our mid-Atlantic region presence.

Complementary Submarket and Product Positioning. Many of the apartment communities to be acquired by us in the Archstone Portfolio Acquisition are located in submarkets that are different than, but complementary to, the submarkets where our current apartment communities are located within each of our six main market regions.

3

Table of Contents

For example, in the Mid-Atlantic region, our existing apartment communities are more concentrated in suburban locations such as the Rockville/Gaithersburg corridor in suburban Maryland and the Tysons Corner and Fairfax submarkets in Northern Virginia. By contrast, the Archstone Portfolio Acquisition apartment communities in the Washington, D.C. market are concentrated in more urban, higher density submarkets such as the Rosslyn/Ballston corridor in Arlington, Virginia and the Wisconsin/Connecticut Avenue NW corridor in Washington D.C. We believe that many of the apartment communities to be acquired by us in the Archstone Portfolio Acquisition are also positioned to appeal to a somewhat different consumer segment than our existing assets based on the different characteristics of the Archstone Portfolio Acquisition apartment communities, which are generally more heavily weighted towards smaller apartments in high-rise buildings. We believe the Archstone Portfolio Acquisition will provide us with a more diversified portfolio with broader product offerings within each region where we currently do business.

Increased Scale Supports our Multi-Brand Strategy. We recently introduced a three brand strategy. We are currently in the process of rolling out our two new brands (AVA and Eaves by Avalon) at certain of our existing apartment communities, and refining our core Avalon brand. We believe that the Archstone Portfolio Acquisition will allow us to extend our three brand platform across additional assets in an efficient manner, bringing these brands to a meaningful market presence sooner than we otherwise would be able to do so.

Enhanced Corporate Efficiencies Due to Increased Scale. We expect that the increased scale of our apartment community portfolio from the Archstone Portfolio Acquisition will require a lower proportionate increase in our corporate and property management overhead and general and administrative expenses. We believe that our corporate support functions, including our customer care center in Virginia Beach and our marketing, customer insight, and market research groups in our corporate headquarters in Arlington, Virginia, are positioned to allow for absorption of additional assets to provide additional operating leverage.

The table below provides details by region for the operating apartment communities as of September 30, 2012 in which we expect to acquire a direct or indirect interest in connection with the Archstone Portfolio Acquisition.

Community	Location	Number of Homes	Average SF Per Home(1)	Year Built(2)	per C H	venue Occupied Iome egion)(3)
New England	0: 1/4	22.4	705	1077		
Archstone Quincy	Quincy, MA	224	705	1977		
Archstone Bear Hill	Waltham, MA	324	1,208	1999		
Legacy Place(4)	Dedham, MA	285	1,075	2011		
Subtotal — New England		833	1,027		\$	2,270
Metro New York / New Jersey						
Archstone Clinton North	New York, NY	339	536	2008		
Archstone Clinton South	New York, NY	288	557	2007		
Archstone Midtown West	New York, NY	550	716	1998		
Kips Bay(4)	New York, NY	208	733	1997		
Archstone Meadowbrook Crossing	Westbury, NY	396	1,014	2006		
Subtotal — Metro New York / New Jersey		1,781	724		\$	3,338

4

Table of Contents

Community Mid-Atlantic	Location	Number of Homes	Average SF Per Home(1)	Year Built(2)	Revenue per Occupied Home (by Region)(3)
Archstone Ballston Place	Arlington, VA	383	871	2001	
Archstone Ballston Square	Arlington, VA	714	877	1992	
Archstone Courthouse Place	Arlington, VA	564	849	1999	
Crystal House I(5)	Arlington, VA	426	880	1969	

C + 1 H H(f)	A 1' 4 X7A	401	012	1065		
Crystal House II(5)	Arlington, VA	184	913 839	1965 1987		
Oakwood Arlington(6)	Arlington, VA					
Archstone Charter Oak	Reston, VA	262	1,097	1970		
Archstone Reston Landing	Reston, VA	400	995	2000		
Archstone Tysons Corner	Vienna, VA	217	967	1980		
Oakwood Gaithersburg(6)	Gaithersburg, MD	136	878	1988		
Archstone Wheaton Station	Wheaton, MD	243	884	2005		
Archstone Russett	Laurel, MD	238	1,154	1999		
Grosvenor Tower(4)	North Bethesda, MD	236	972	1987		
Archstone Glover Park	Washington, DC	120	869	1953		
Archstone Tunlaw Gardens	Washington, DC	166	816	1944		
Brandywine(7)(8)	Washington, DC	305	1,280	1954		
The Albemarle	Washington, DC	228	1,123	1966		
The Consulate	Washington, DC	268	843	1978		
The Statesman	Washington, DC	281	678	1961		
Oakwood Philadelphia(6)	Philadelphia, PA	80	831	1945		
Subtotal — Mid-Atlantic		5,852	929		\$	2,101
Pacific Northwest			_			
Archstone Redmond Campus	Redmond, WA	422	1,017	1991		
Archstone Redmond Lakeview	Redmond, WA	166	849	1987		
Archstone Kirkland at Carillon Point(4)	Kirkland, WA	130	1,344	1990		
Subtotal — Pacific Northwest		718	1,037		\$	1,633
Northern California			<u> </u>		<u>-</u>	
Archstone Walnut Creek	Walnut Creek, CA	510	746	1987		
Archstone Walnut Creek Station	Walnut Creek, CA	360	700	1989		
Archstone Walnut Ridge	Walnut Creek, CA	106	764	2000		
Archstone San Bruno	San Bruno, CA	300	891	2004		
Archstone San Bruno II	San Bruno, CA	185	846	2007		
Archstone San Bruno III	San Bruno, CA	187	1,237	2005		
Archstone West Valley	San Jose, CA	789	639	1970		
Archstone Mountain View at Middlefield	Mountain View, CA	402	651	1969		
Archstone Willow Glen	San Jose, CA	412	928	2002		
Sunnyvale(4)	Sunnyvale, CA	192	1,063	1991		
Subtotal — Northern California	Sullily vale, CA	3,443	790	1991	\$	1,734
			790		Φ	1,734
Southern California(9)	Danadana CA	347	975	2006		
Archstone Del Mar Station	Pasadena, CA	96		1972		
Archstone Old Town Pasadena	Pasadena, CA	120	692			
Archstone Pasadena	Pasadena, CA		854	2004		
Archstone La Mesa	La Mesa, CA	168	830	1989		
Archstone La Jolla Colony	San Diego, CA	180	761	1987		
Archstone Oak Creek	Agoura Hills, CA	336	1,084	2004		
Archstone Calabasas	Calabasas, CA	600	844	1988		
Archstone Los Feliz	Los Angeles, CA	263	767	1989		
Oakwood Toluca Hills(6)	Los Angeles, CA	1,151	691	1973		
Archstone Marina Bay(4)(10)	Marina Del Rey, CA	205	815	1967		
Venice on Rose(4)	Venice, CA	70	1,207	2011		
Archstone Simi Valley	Simi Valley, CA	500	860	2007		
Archstone Studio City	Studio City, CA	450	736	1987		
Archstone Studio City II	Studio City, CA	101	834	1991		
		5				

Table of Contents

Community	Location	Number of Homes	Average SF Per Home(1)	Year Built(2)	per C H	venue Occupied Ome egion)(3)
Archstone Studio City III	Studio City, CA	276	955	2002	(by R	egion)(5)
Archstone Thousand Oaks	Thousand Oaks, CA	154	873	1992		
Archstone Thousand Oaks Plaza	Thousand Oaks, CA	148	949	2002		
Archstone Warner Center	Woodland Hills, CA	522	882	2007		
Archstone Woodland Hills	Woodland Hills, CA	883	655	1971		
Archstone Santa Monica on Main	Santa Monica, CA	133	921	2007		
Archstone Long Beach(11)	Long Beach, CA	206	705	1985		
Archstone Terracina(11)	Ontario, CA	736	915	1988		
Archstone Seal Beach	Seal Beach, CA	549	706	1971		
Archstone Vanoni Ranch	Ventura, CA	316	936	2005		
Archstone Ventura Colony(12)	Ventura, CA	272	807	1989		
Subtotal — Southern California		8,782	816		\$	1,601
Non Core Markets						
Archstone Boca Town Center(4)	Boca Raton, FL	252	1,064	1988		
Archstone Lexington	Flower Mound, TX	222	983	2000		
Archstone Memorial Heights	Houston, TX	556	781	1996		
Subtotal — Non Core Markets		1,030	894		\$	1,351
Total		22,439	853		\$	1,894

- (1) This information is taken primarily from the most recent filing by Archstone Inc. with the SEC on November 19, 2012, which we have not independently verified, and it may contain inaccuracies and/or reflect calculation methodologies that differ from those we use to present information about our apartment communities. Regional totals represent weighted averages.
- (2) Represents the date that construction of the apartment community was completed, and does not consider any subsequent capital expenditures to redevelop the applicable apartment community or for other purposes.
- (3) Represents the weighted average monthly revenue per occupied home for the respective region for the three months ended September 30, 2012.
- (4) This apartment community is owned by a joint venture in which Archstone owns a 28.6% interest.
- (5) Interest in this apartment community is subject to a ground lease which expires in November 2060.
- (6) This apartment community has been leased by Archstone to a third party under a master lease that has been extended to July 2017.
- (7) This apartment community is managed by a third party.
- (8) This apartment community is owned by a joint venture in which Archstone owns a 26.1% interest.
- (9) Subsequent to September 30, 2012, Archstone acquired an indirect interest in Studio 4121, a 205 apartment home community located in Los Angeles, CA.
- (10) This apartment community includes the ownership of 218 boat slips in the adjoining marina.
- (11) Represents an apartment community that was under contract for disposition as of, or subsequent to, September 30, 2012.
- (12) Subsequent to September 30, 2012, Archstone sold Archstone Ventura Colony.

6

Table of Contents

The table below provides certain details with respect to properties under construction or planned for development in which we expect to acquire a direct or indirect interest in connection with the Archstone Portfolio Acquisition. In addition, Archstone controls certain other land parcels primarily through purchase options. Some of these option contracts may be assigned to us at closing in addition to the assets listed below.

			Expected Apartment
Property	Location	Start Date	Homes(1)
Under Construction and/or in Lease-up(2)			
Archstone West Valley Expansion	San Jose, CA	Q3 2012	84
Archstone First+M Phase I	Washington, DC	Q3 2010	469
Parkland Gardens	Arlington, VA	Q2 2012	227
Memorial Heights Phase I	Houston, TX	Q3 2012	318
Archstone Berkeley on Addison	Berkeley, CA	Q3 2012	94
Archstone Toscano	Houston, TX	Q2 2011	474
Archstone Santa Clarita(3)	Santa Clarita, CA	Q1 2012	157
Land Held, In Planning and Owned(4)			
Archstone First+M Phase II	Washington, DC	TBD	434
Oakwood Toluca Hills Land	Toluca Hills, CA	TBD	150
Huntington Beach(5)	Huntington Beach, CA	TBD	384

- (1) This information is taken from the most recent filing by Archstone Inc. with the SEC on November 19, 2012, which we have not independently verified, and it may contain inaccuracies and/or reflect calculation methodologies that differ from those we use to present information about our apartment communities.
- (2) Total expected investment for consolidated apartment communities under construction and/or in lease-up is \$470.1 million with \$161.2 million remaining to invest.
- (3) Development property is being constructed in an unconsolidated joint venture in which Archstone has a 21.6% interest.
- (4) Commencement of construction of projects in planning is subject to regulatory approval, acquisition of financing and/or suitable market conditions.
- (5) Land is owned by a consolidated joint venture in which Archstone has a 95.0% interest.

Adjustments to the Anticipated Archstone Portfolio Acquisition

The properties that will be part of the overall transaction involving Archstone and the assets and liabilities to be held by certain joint ventures we expect to form with Equity Residential are subject to final adjustment at closing between us and Equity Residential pursuant to arrangements separate from the Purchase Agreement. These arrangements are designed to account for ongoing changes in the status of the assets and liabilities up until closing. For example, some of the properties and other assets may be sold and liabilities may be settled before closing between us and Equity Residential. As a result, the final allocation between us and Equity Residential of the properties, other assets and liabilities in connection with the overall transaction involving Archstone could change and impact the amount of consideration to be paid by us under the Purchase Agreement.

Assumed Indebtedness

We will assume approximately \$3.7 billion principal amount of consolidated indebtedness in connection with the Archstone Portfolio Acquisition. Of this amount, Fannie Mae and Freddie Mac (the Government Sponsored Enterprises, or GSEs) are the lenders in connection with approximately (i) \$2.8 billion of pooled debt, (ii) \$463.4 million of tax-exempt bond financing for which the GSEs provide credit enhancement and liquidity and (iii) \$104.4 million of other single asset loans (collectively, the GSE Indebtedness). We have entered into a commitment letter with each of Fannie Mae and Freddie Mac whereby they have provided us with their consents to the transaction contemplated by the Purchase Agreement and our and Equity Residential's assumption of the allocable portions of the loans made to Archstone. In connection with such

7

Table of Contents

Portfolio Acquisition to \$953.0 million. The following table sets forth the consolidated indebtedness we will assume as of September 30, 2012, and on a pro forma basis giving effect to the repayment of indebtedness as discussed above.

Community / Debt Facility	Stated Interest Rate	Principal Final Maturity Date	Balance Outstanding at 9/30/12(1) (in thousands)	Pro forma Balance Outstanding at 9/30/12(1) (in thousands)
Tax-exempt bonds			(iii tiiousaiius)	(in thousands)
Fixed Rate				
Meadowbrook	4.61%	Nov-2036	\$ 62,200	\$ 62,200
Variable rate				
Clinton (North)	SIFMA + 1.53%	Nov-2038	143,409	143,409
Clinton (South)	SIFMA + 1.53%	Nov-2038	118,532	118,532
Midtown West	SIFMA + 1.13%	May-2029	99,537	99,537
San Bruno	SIFMA + 1.35%	Dec-2037	61,875	61,875
Calabasas	SIFMA + 1.48%	Apr-2028	40,073	40,073
		·	463,426	463,426
Conventional loans				
Fixed Rate				
Fannie Mae Pool 6(2)	6.19%	Nov-2015	940,923	940,923
Fannie Mae Pool 2(2)	6.26%	Nov-2017	692,192	692,192
First and M	5.57%	May-2053	116,166	116,166
San Bruno II	5.37%	Apr-2021	31,700	31,700
Meadowbrook	4.70%	Nov-2036	22,493	22,493
Lexington	5.55%	Mar-2016	17,079	17,079
			1,820,553	1,820,553
Variable rate				<u> </u>
Fannie Mae Pool 9(2)	LIBOR + 1.27%	Nov-2014	636,756	_
Freddie Mac Pool	LIBOR + 0.96%	Nov-2014	412,724	304,375
South San Francisco(3)	DMBS + 1.00%	Apr-2013	76,706	´ —
Calabasas	DMBS + 1.44%	Aug-2018	57,781	57,781
San Bruno III	LIBOR + 2.60%	May-2013	47,000	47,000
Wheaton Station(3)	DMBS + 1.00%	Apr-2013	44,539	_
La Mesa(3)	DMBS + 1.00%	Apr-2013	24,755	_
Oakwood Gaithersburg	FRMB + 3.79%	Jan-2017	14,876	_
Parkland Gardens	LIBOR + 2.25%	May-2017	12,935	_
Toscano	LIBOR + 6.00%	May-2016	31,938	_
Memorial Heights	LIBOR + 2.50%	May-2017	2,143	_
			1,362,153	409,156
			\$ 3,708,332	\$ 2,755,335

⁽¹⁾ Balances are for consolidated debt assumed and do not include our share of the principal amount of debt held by unconsolidated joint ventures of approximately \$238.3 million. Balances are also presented exclusive of amounts held in principal reserve funds, that are held for the repayment of the respective borrowing. The proforma balance outstanding reflects expected repayments of indebtedness as of the closing of the Archstone Portfolio Acquisition.

8

Table of Contents

Purchase Agreement and Related Arrangements

Pursuant to the Purchase Agreement, the acquisition is subject to customary closing conditions, including, among other things, the accuracy of the other parties' representations and warranties and compliance with covenants, subject in each case to significant materiality standards as well as the absence of a material adverse change with respect to the business, financial condition or results of operations of Archstone. There can be no assurance that any condition to the closing of the transaction will be satisfied or waived, if permitted, or that there will not be events, developments or changes that can cause the closing not to occur. Therefore, there can be no assurance with respect to the timing of the closing of the Archstone Portfolio Acquisition or whether the transaction will be completed on the currently contemplated terms, other terms or at all. There is no closing condition relating to our or Equity Residential's ability to obtain the necessary financing or lender consents for the transaction. If we extend the closing beyond the 90th day following the execution of the Purchase Agreement, the purchase price will increase by an amount equal to \$1.0 million per day thereafter. The amount of this increase would be paid 60% by Equity Residential and 40% by us, with each party's additional payment reduced by the amount of any accrued but unpaid dividends on the stock consideration payable by that party for the period prior to closing. The Purchase Agreement also provides that we, Equity Residential or Lehman may terminate the Purchase Agreement if the closing has not occurred by the 120th day following the execution of the Purchase Agreement, so long as the failure of the closing to occur on or before this date was not due to a breach by the terminating party of its respective obligations under the Purchase Agreement and, in the case of us and Equity Residential we pay the termination fee discussed below if, at the time, all of the conditions to closing are satisfied.

⁽²⁾ Borrowings are cross-defaulted.

⁽³⁾ Borrowings are cross-collateralized and cross-defaulted as part of a pooled financing with respect to the apartment communities.

Under the terms of the Purchase Agreement, if Lehman terminates the Purchase Agreement in the event that (1) we or Equity Residential breach any covenant or agreement under the Purchase Agreement, where the effect of the breach causes the closing conditions to be incapable of being satisfied, and, if the breach is capable of being cured, the breach is not cured by us or Equity Residential within 20 business days after receipt of notice from Lehman, (2) any event occurs that makes it impossible to satisfy a condition precedent to Lehman's obligation to close the transaction and Lehman and Archstone are not then in breach of representations, warranties, covenants or other agreements thereunder that would result in the conditions precedent to our obligations and Equity Residential's obligations not being satisfied, or (3) we or Equity Residential breach our obligation to close the acquisition at a time when all closing conditions have been satisfied or waived, then we and Equity Residential will be jointly and severally obligated to pay a termination fee of \$650.0 million. If we or Equity Residential extend the closing beyond the 60th day following the execution of the Purchase Agreement and such events occur, the termination fee increases to \$800.0 million. The termination fee will be Lehman's sole and exclusive remedy for a breach under the Purchase Agreement. Neither we nor Equity Residential can terminate the Purchase Agreement or fail to close when all the closing conditions have been satisfied without incurring the termination fee.

Although we and Equity Residential are jointly and severally liable for liabilities assumed in the Archstone Portfolio Acquisition as well as the termination fee, if it is payable, we have separately agreed with Equity Residential that such liabilities, fees and expenses will generally be apportioned 40% to us and 60% to Equity Residential. Among the matters that we and Equity Residential have agreed to apportion 40% to us and 60% to Equity Residential are obligations that arise from the redemption of preferred equity obligations issued by two Archstone entities pursuant to the terms of those preferred equity obligations. Certain liabilities to be assumed pursuant to the Purchase Agreement are allocated solely to us or solely to Equity Residential, generally because the liability relates to one or more properties to be acquired by us or Equity Residential. If the termination of the Purchase Agreement is a result of a breach by either us or Equity Residential, then the breaching party shall be solely responsible for, and shall indemnify the non-breaching party against, the fees and expenses, including the termination fee, payable to Lehman resulting from a failure to consummate the acquisition, and any costs and expenses incurred by the non-breaching party in connection with enforcing its rights against the breaching party.

Assets and liabilities of Archstone that are not acquired or assumed directly or indirectly wholly by either us or Equity Residential will be acquired through joint venture arrangements to be entered into by and between us and Equity Residential prior to or contemporaneously with the closing of the acquisition. One set of assets and liabilities to be

9

Table of Contents

acquired through joint venture arrangements consists of properties which we will jointly evaluate how best to manage after the closing, which may result in (i) in the ultimate transfer to either us or Equity Residential when all necessary consents have been obtained and other transfer conditions satisfied, (ii) in continued joint ownership for a period, or (iii) in sale to a third party or parties. A second set of assets and liabilities to be acquired through joint venture arrangements consists of properties subject to contractual commitments to third parties to make certain payments in the event that taxable gain is recognized with respect to that property, which will be acquired in a transaction that should not recognize taxable gain. We then intend to contribute these properties, without the recognition of taxable gains, to limited partnerships or other entities controlled solely by either us or Equity Residential in exchange for passive interests in such limited partnership or other entities, which will eventually be the only assets of the joint venture. The properties designated for contribution to us are included in the operating properties table included above.

The foregoing description of the Purchase Agreement and the Archstone Portfolio Acquisition does not purport to be complete and is subject to, and qualified in its entirety by, the full text of the Purchase Agreement, a copy of which is attached hereto as Exhibit 2.1 and the terms of which are incorporated herein by reference.

Registration Rights Agreement and Shareholders Agreement

In connection with the closing of the Archstone Portfolio Acquisition, we will enter into a registration rights agreement with Lehman, or the Registration Rights Agreement, pursuant to which we will be required to register under the Securities Act of 1933, as amended, or the Securities Act, the resale of the 14,889,706 shares of common stock we will issue to Archstone at the closing of the Archstone Portfolio Acquisition. Under the Registration Rights Agreement we will be obligated to file a resale shelf registration statement within 10 days following the closing pursuant to which, after the expiration of Lehman's lock-up described below on April 26, 2013, it will be able to sell, generally without restrictions, shares received in the transaction. We have the right to suspend sales under the shelf registration statement in limited circumstances for up to 90 days in any twelve-month period. We are obligated to maintain the effectiveness of the registration statement until the earlier of the fifth anniversary of the closing of the Archstone Portfolio Acquisition or when Lehman owns less than \$250.0 million of the shares it originally received under the Purchase Agreement.

The Registration Rights Agreement gives Lehman the right to conduct two underwritten offerings of our shares each year. We will be obligated to cooperate with, and assist in, those offerings and to enter into a 30-day lock-up with Lehman's underwriters. Following the first anniversary of the closing, we will have the right once in any twelve-month period to delay an underwritten offering requested by Lehman so that we may undertake our own underwritten offering. If we effect an underwritten offering of our shares, Lehman has agreed to enter into a 30-day lock-up with the underwriters so long as it owns over 5% of our outstanding shares.

Further, pursuant to a shareholders agreement we will enter into between us and Lehman, or the Shareholders Agreement, Lehman has agreed to a 150-day lock-up, starting from the date of the Purchase Agreement, with respect to the shares of common stock acquired in connection with the Archstone Portfolio Acquisition. Under the Shareholders Agreement, so long as Lehman owns more than 5% of our common stock, Lehman has agreed it will not (i) acquire beneficial ownership of any additional shares of our common stock; (ii) participate in any voting or similar arrangement with a third party; (iii) enter into, propose or facilitate any change in control transaction (or other extraordinary transaction involving us); or (iv) otherwise act, alone or in concert with others, to seek to control, or influence, our board of directors or our management or policies.

Additionally under the Shareholders Agreement, for one year starting from the date of the closing of the Archstone Portfolio Acquisition, Lehman will vote all of its shares of our common stock in accordance with the recommendation of our board of directors on any matter other than an extraordinary transaction. After the first year, and for so long as Lehman holds more than 5% of our common stock, Lehman will vote all of its shares of our common stock (i) in accordance with the recommendations of our board of directors with respect to any election of directors, compensation and equity plan matters, and any amendment to our charter to increase our authorized capital stock; (ii) on all matters proposed by other shareholders, either proportionately in accordance with the votes of the other shareholders or, at its election, in accordance with the recommendation of our board of directors; and (iii) on all other matters, in its sole and absolute discretion.

The foregoing description of the Registration Rights Agreement and Shareholders Agreement does not purport to be complete and is subject to, and qualified in its entirety by reference to the full text of the Registration

10

Table of Contents

Prior to the issuance of our shares of common stock to Archstone, we will waive the stock ownership limit in our charter, which generally prohibits a stockholder from holding more than 9.8% of the issued and outstanding shares of any class or series of our stock and will instead permit Lehman to hold the 14,889,706 shares of our common stock received in the Archstone Portfolio Acquisition, plus the shares they owned as of the date of the waiver. We will grant this waiver based on our belief that such waiver will not jeopardize our ability to qualify as a REIT.

Bridge Loan Facility

In connection with the Archstone Portfolio Acquisition, on November 26, 2012, we received a commitment letter from Goldman Sachs Lending Partners LLC for a senior unsecured bridge loan facility, or the Bridge Loan Facility, in an aggregate principal amount of up to \$2.2 billion, less the sum of the net proceeds from an equity offering, an offering of certain debt securities and/or any other bank or similar term loan financing and the aggregate principal amount of certain debt that we assume in connection with the Archstone Portfolio Acquisition for which lender consents have been obtained.

The Bridge Loan Facility would mature 364 days from the date of funding and any loans provided under the Bridge Loan Facility are to be used to fund the Archstone Portfolio Acquisition and related costs, fees and expenses, including the repayment of certain debt we assume in connection with the Archstone Portfolio Acquisition. The loans under the Bridge Loan Facility must be funded on or before the date the Archstone Portfolio Acquisition closes, and will only be funded, if requested by us, upon the satisfaction of certain customary conditions to funding for this type of facility, including a condition that we shall have used commercially reasonable efforts to raise capital through an equity offering. Because the conditions to the lender's obligations under the Bridge Loan Facility are substantially similar to the conditions to our obligations under the Purchase Agreement, other than closing conditions that are customary for lending transactions such as the Bridge Loan Facility, we do not believe that there would be circumstances under which we would be obligated to complete the Archstone Portfolio Acquisition without also being able to obtain the funds to be borrowed under the Bridge Loan Facility.

Loans under the Bridge Loan Facility will bear interest at a rate per annum equal to either (i) the reserve adjusted LIBOR rate or (ii) the base rate, plus an applicable margin based on the credit ratings of our unsecured and unsubordinated long-term indebtedness. The Bridge Loan Facility will also require us to pay Goldman Sachs Lending Partners LLC certain customary fees.

The above summary of the Bridge Loan Facility is based on the commitment letter we received from Goldman Sachs Lending Partners LLC, the terms of which are subject to the final documentation between us and Goldman Sachs Lending Partners LLC.

Forward-Looking Statements

This report contains statements that are forward-looking statements within the meaning of Section 27A of the Securities Act and Section 21E of the Securities Exchange Act of 1934, as amended, or the Exchange Act, including, without limitation, information regarding our expectations, goals and intentions regarding the future, statements regarding the anticipated closing the Archstone Portfolio Acquisition, potential debt and equity financings to finance the Archstone Portfolio Acquisition, and the expected effect of the Archstone Portfolio Acquisition on the Company. You can identify forward-looking statements by the use of the words "believe," "expect," "anticipate," "intend," "estimate," "assume," "plan," "project," "may," "shall," "will," "outlook" and other similar expressions that predict or indicate future events and trends and which do not relate to historical matters. We cannot assure the future results or outcome of the matters described in these statements; rather, these statements merely reflect our current expectations of the approximate outcomes of the matters discussed. You should not rely on forward-looking statements because they involve known and unknown risks, uncertainties and other factors, some of which are beyond our control. Some of the known risks, uncertainties and other factors include:

- · We may fail to consummate the Archstone Portfolio Acquisition or may not consummate it on the terms described herein because we are unable to secure financing, we cannot assume the indebtedness we expect to assume in connection with the acquisition or otherwise;
- · We and/or Equity Residential may fail to perform under the Purchase Agreement or may not perform on the terms prescribed;

11

Table of Contents

- · We may need to utilize cash on hand and debt financings, in order to close the Archstone Portfolio Acquisition, the sufficiency of which cannot be assured, or seek alternative sources of financing to close the Archstone Portfolio Acquisition, and such alternative sources of financing may not be available on favorable terms or at all:
- · Assuming indebtedness in connection with the Archstone Portfolio Acquisition may have an adverse effect on our financial condition and results of operations;
- The intended benefits of the Archstone Portfolio Acquisition may not be realized, which could have a negative impact on the market price of our common stock after the Archstone Portfolio Acquisition.
- The governance provision of our joint ventures with Equity Residential could adversely affect our flexibility in dealing with such joint venture assets and liabilities.
- · We expect to assume substantially all liabilities related to the Archstone Portfolio Acquisition, and may be responsible for liabilities that were not known when we entered into the Purchase Agreement.
- · We intend to liquidate certain assets acquired in connection with the Archstone Portfolio Acquisition at, or shortly after, the expected closing of the Archstone Portfolio Acquisition, but we may be unable to achieve the expected proceeds of these acquisitions or may be unable to liquidate these assets at all.
- Our historical and pro forma condensed consolidated financial information may not be representative of our results as a combined company.
- · Our qualification as a real estate investment trust, or a REIT, will depend in part on the nature of the assets and rights to income we acquire as part of the Archstone Portfolio Acquisition and joint ventures with Equity Residential.
- The Archstone Portfolio Acquisition will significantly increase the size of our real estate portfolio and related personnel and operating and financial needs, and we may not be successful in integrating the Archstone Portfolio into our business.
- Other risks and uncertainties indicated from time to time in the Company's filings with the Securities and Exchange Commission, including the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2011 under the heading "Risk Factors" and under the heading "Management's Discussion and Analysis of Financial Condition and Results of Operations Forward-Looking Statements" and in subsequent current reports on Form 8-K and quarterly reports on Form 10-Q.

These forward-looking statements are based on management's present expectations and beliefs about future events. As with any projection or forecast, these statements are inherently susceptible to uncertainty and changes in circumstances. The Company is under no obligation to, and expressly disclaims any obligation to, update or alter its

forward-looking statements whether as a result of such changes, new information, subsequent events or otherwise.

Item 2.02. Results of Operations and Financial Condition.

The information set forth below in "Item 8.01 — Other Information" is incorporated herein by reference.

The information contained in this Item 2.02 shall not be deemed "filed" with the Securities and Exchange Commission nor incorporated by reference in any registration statement filed by the Company under the Securities Act of 1933, as amended.

Item 2.03 Creation of a Direct Financial Obligation or an Obligation under an Off-Balance Sheet Arrangement of a Registrant.

The information set forth above in "Item 1.01 — Entry into a Material Definitive Agreement" is incorporated herein by reference.

12

Table of Contents

Item 3.02 Unregistered Sales of Equity Securities.

The information set forth above in "Item 1.01 — Entry into a Material Definitive Agreement" is incorporated herein by reference. The issuance by the Company of common stock constituting a portion of the total consideration pursuant to the Purchase Agreement will be made in reliance upon the exemption from securities registration afforded by Section 4(2) of the Securities Act of 1933, as amended, or another applicable exemption.

Item 7.01. Regulation FD Disclosure.

On November 26, 2012, the Company issued a press release announcing the Archstone Portfolio Acquisition. A copy of the Company's press release is furnished as Exhibit 99.1 and is incorporated herein by reference.

The information disclosed under this Item 7.01, including Exhibit 99.1 hereto, is being furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, and shall not be deemed incorporated by reference into any filing under the Securities Act of 1933, as amended, except as expressly set forth by specific reference in such filing.

Item 8.01 Other Events

Financial Highlights and Outlook

Operating Results and Outlook. Between January 1, 2012, and September 30, 2012, our portfolio of established communities achieved an increase in NOI of 8.1% compared to the same period of 2011. Throughout 2012, apartment fundamentals have remained strong, driven by a combination of a decline in the homeownership rate, modest employment growth and limited supply of new multifamily rental product. For the nine months ended September 30, 2012, our established apartment communities achieved an increase in rental revenue of 6.0% as compared to the prior year period, primarily from an increase of 5.8% in average rental rates and a 0.2% improvement in economic occupancy, as we have historically defined that term. On October 24, 2012 we updated our guidance for full year 2012 earnings per share — diluted, or EPS, to be in the range of \$4.47 and \$4.52, and Funds from Operations — diluted, or FFO, to be in the range of \$5.45 to \$5.50. FFO is a non-GAAP financial measure. For a description of how we define FFO and a reconciliation of FFO to net income, please see "Reconciliation of Non-GAAP Financial Measures" beginning on page F-18 of this report.

On November 12, 2012, we issued a press release regarding the financial impact of Hurricane Sandy to us, in which we estimated the out-of-pocket costs to repair damages to be in the range of \$5.0 to \$7.0 million after considering insurance reimbursements. We now estimate that the aggregate out-of-pocket costs to repair damages associated with Hurricane Sandy will be in the range of \$3.5 to \$5.0 million, after considering insurance reimbursements. A portion of these out-of-pocket costs will be capitalized, such that we expect the impact of these costs on reported earnings to be between \$1.0 and \$1.5 million, with the majority of the costs expected to be incurred during the fourth quarter of 2012.

Current Operating Environment. We anticipate market fundamentals impacting our business to remain favorable for the remainder of 2012 and into 2013. Our expectations are based in part on our outlook for employment conditions, where we anticipate moderate but accelerated job and population growth, particularly in the age groups that have historically demonstrated a higher propensity to rent. Our assessment of our portfolio's potential performance for 2013 also considers the individual demand/supply characteristics of each submarket in which we operate. Over the past several years supply within the majority of our markets has been below long-term trend levels, although it is expected to increase closer to long-term levels over the next several years. We anticipate that the properties we expect to acquire as part of the Archstone Portfolio Acquisition will show operating trends in line with our established communities.

Acquisition Capitalization Rate. As part of our due diligence process, we analyzed the anticipated capitalization rate we expect to derive from the Archstone Portfolio Acquisition. We define capitalization rate as NOI as a percentage of the purchase price after making adjustments to historical NOI to reflect anticipated growth in rental rates, as well as reserves for capital expenditures, property management fees, and other market conventions to estimate future stabilized NOI. We believe that the weighted average capitalization rate for the combined Archstone portfolio, including properties to be acquired both by us and by Equity Residential, is in the high 4% range. Our definition of NOI and calculation of the capitalization rate may not be comparable with those of other REITs because of differences in accounting policies, forward adjustments to historical NOI, assumed property management fees, reserves and other items. We also caution you not to place undue reliance on our estimates of NOI

13

Table of Contents

to be generated by the apartment communities we expect to acquire as part of the Archstone Portfolio Acquisition, because such estimates are based principally on data made available to us in the acquisition diligence process, combined with our experience in operating our own comparable apartment communities. Actual results could differ materially from expectations, as discussed in "Forward-Looking Statements" in this report.

Pro Forma Impact of the Archstone Portfolio Acquisition. The following table presents our EPS and FFO per share for the year ended December 31, 2011 and the nine months ended September 30, 2012, on a historical basis and pro forma for the Archstone Portfolio Acquisition:

For the Year ended		For the Nine Months		
	12/	31/11	Eı	nded 9/30/12
	Historical	Pro Forma	Historical	Pro forma

EPS	\$ 4.87	\$ 2.11	\$ 3.13	\$ 2.74
FFO per share	\$ 4.57	\$ 5.04	\$ 4.05	\$ 4.34

The pro forma adjustments assume that the Archstone Portfolio Acquisition occurred on January 1, 2011, and reflect management's estimates of costs that would have been incurred during the respective periods for property management and additional general and administrative expenses for the Archstone Portfolio Acquisition. Assuming that the Archstone Portfolio Acquisition closed on January 1, 2011, and excluding one-time transaction costs, the transaction would have been approximately 7% accretive to our FFO per share for the nine months ended September 30, 2012. The actual impact of the Archstone Portfolio Acquisition in future periods will be affected by the timing of the transaction's closing, as well as the amount and timing of one-time transaction costs. We currently anticipate that we will incur \$153.8 million in expensed transaction costs related to the Archstone Portfolio Acquisition. We caution you not to place undue reliance on pro forma performance metrics because the contribution to EPS and FFO per share from the apartment communities we expect to acquire as part of the Archstone Portfolio Acquisition will depend on a number of factors that we cannot predict with certainty. Our actual experience operating the apartment communities we expect to acquire as part of the Archstone Portfolio Acquisition, following the closing of the transaction, including rental rates, economic occupancy, operating and capital expenditures and other items, may change our expectations with respect to the impact of the transaction on our EPS and FFO per share.

Dividend Outlook. Based on our estimates for our existing apartment communities and our expectations for apartment communities we anticipate acquiring as part of the Archstone Portfolio Acquisition, we expect to increase our common stock dividend for the first quarter of 2013, by a range of between 8% and 12% as compared to the dividend we previously declared and announced for the fourth quarter of 2012. This increase is supported by our current operating platform and recent operating trends, the expectation for continued favorable apartment fundamentals in 2013 and the contribution from apartment communities we expect to acquire as part of the Archstone Portfolio Acquisition. The payment of any dividend, however, is subject to the approval of our board of directors and there can be no assurance that economic conditions generally and/or their impact on our operating results and payout ratio will not lead us to change our expectation.

Item 9.01 Financial Statements and Exhibits.

(a) Financial Statements. The following is required financial information relating to the Archstone Portfolio:

ARCHSTONE PORTFOLIO

Independent Auditors' Report	F-1
Combined Statements of Revenue and Certain Expenses for the Nine Months Ended September 30, 2012 (unaudited) and for the Year Ended December 31, 2011	F-2
Notes to Combined Statements of Revenue and Certain Expenses	F-3

(b) Pro Forma Financial Information.

ARCHSTONE PORTFOLIO

Unaudited Pro Forma Condensed Consolidated Financial Statements	F-
Unaudited Pro Forma Condensed Consolidated Balance Sheet as of September 30, 2012	F-
Unaudited Pro Forma Condensed Consolidated Statements of Operations for the Nine Months Ended September 30, 2012 and for the Year Ended December 31,	
<u>2011</u>	F-
Notes to Unaudited Pro Forma Condensed Consolidated Financial Statements	F-1
Reconciliation of Non-GAAP Financial Measures	F-1

(d) Exhibits.

The Exhibit Index appearing immediately after the signature page of this Form 8-K is incorporated herein by reference.

14

Table of Contents

INDEPENDENT AUDITORS' REPORT

The Partners

Archstone Enterprise LP:

We have audited the accompanying combined statement of revenue and certain expenses of the Archstone Portfolio (the "Properties") for the year ended December 31, 2011. This combined financial statement is the responsibility of the management of Archstone Enterprise LP. Our responsibility is to express an opinion on the combined financial statement based on our audit.

We conducted our audit in accordance with auditing standards generally accepted in the United States of America. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statement is free of material misstatement. An audit includes consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Properties' internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statement, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audit provides a reasonable basis for our opinion.

The accompanying combined statement of revenue and certain expenses was prepared for the purpose of complying with the rules and regulations of the Securities and Exchange Commission and for the inclusion in Form 8-K of AvalonBay Communities, Inc., as described in Note 1 to the combined financial statement. It is not intended to be a complete presentation of the Properties' revenue and expenses.

In our opinion, the combined statement of revenue and certain expenses referred to above presents fairly, in all material respects, the combined revenue and certain expenses, as described in Note 1, of the Properties for the year ended December 31, 2011, in conformity with U.S. generally accepted accounting principles.

/s/ KPMG LLP Denver, Colorado November 26, 2012

ARCHSTONE PORTFOLIO

Combined Statements of Revenue and Certain Expenses

	Nine Months Ended September 30, 2012 (unaudited)		 Year Ended December 31, 2011
Revenue			
Rental revenue	\$	282,051,298	\$ 331,043,138
Other revenue		64,729,294	78,499,077
Total revenue		346,780,592	409,542,215
Certain expenses			
Operating		39,748,154	46,181,145
Utilities		15,796,861	19,911,736
Maintenance		5,087,823	6,363,345
Real estate taxes and insurance		48,132,141	50,348,082
Ground lease		1,413,844	1,740,947
Management fees		9,233,591	10,523,146
Total certain expenses		119,412,414	135,068,401
Revenue in excess of certain expenses	\$	227,368,178	\$ 274,473,814

See accompanying notes.

F-2

Table of Contents

ARCHSTONE PORTFOLIO

Combined Statements of Revenue and Certain Expenses

1. Basis of Presentation

The accompanying combined statements of revenue and certain expenses (the "Statements") include the operations of a portfolio of 68 properties throughout the United States (the "Properties"), collectively containing 22,167 apartment homes, which have been combined for purposes of the Statements as each of the Properties are expected to be acquired from the same seller. On November 26, 2012, an announcement was made describing the signing of a purchase and sale agreement and the expected sale of the Properties to AvalonBay Communities, Inc., ("AVB"), a Maryland corporation. The specific properties and terms of the sale are outlined in the Asset Purchase Agreement made as of November 26, 2012.

The Statements have been prepared for the purpose of complying with the provisions of Rule 3-14 of Regulation S-X promulgated by the Securities and Exchange Commission (the "SEC"). The Statements include the combined historical revenue and certain expenses of the Properties, exclusive of items which may not be comparable to the proposed future operations of the Properties. Material amounts that would not be directly attributable to future operating results of the Properties are excluded, and the Statements are not intended to be a complete presentation of the Properties' revenue and expenses. Items excluded consist of depreciation and amortization, interest expense and income taxes.

The unaudited interim Statement for the nine months ended September 30, 2012 was prepared on the same basis as the Statement for the year ended December 31, 2011 and reflects all adjustments (consisting of only normal recurring adjustments), which are, in the opinion of management, necessary for a fair presentation of the combined revenue and certain expenses for the period presented. The combined revenue and certain expenses for the nine months ended September 30, 2012 are not necessarily indicative of the expected results for the entire fiscal year ending December 31, 2012.

The Statements also include 100% of the revenue and certain expenses of properties in which the seller has a direct or indirect ownership interest of less than 100%.

2. Summary of Significant Accounting Policies

Basis of Accounting

The Statements are prepared on the accrual basis of accounting in accordance with U.S. generally accepted accounting principles (GAAP).

Estimates

The preparation of the Statements in conformity with GAAP requires management to make estimates and assumptions that affect the reported amounts of combined revenue and expenses during the reporting periods. Actual results could differ from those estimates.

Revenue Recognition

The Properties generally lease apartment units under operating leases with terms of one year or less. Rental income related to apartment leases is recognized on a straight-line basis, which is not materially different than if it were recorded when due from residents and recognized monthly as it was earned.

Utility reimbursements from residents are recorded on a gross basis and included in other revenue in the accompanying Statements.

Certain properties are subject to residential master leases ("Master Leases") entered into with a third-party operator in 2005 and have seven-year terms, expiring July 2012, subject to the counterparty's right to terminate individual leases under certain circumstances. Upon termination of each lease certain lease transition provisions will apply. In July 2012, the Properties entered into agreements with the third party operator to extend four of the master leases for an additional five years to July 31, 2017. Each of the master leases is cancellable by either party after two years; provided that the Properties also have the right to sell the communities subject to the master lease at any time, without penalty, which will cause the master lease to terminate 90 days after the sale. In 2012 (unaudited), an affiliate of the Properties assumed management of four of

Table of Contents

the communities for which master leases were not extended. After giving effect to the new lease terms and considering only the four extended leases, the estimated aggregate annual contractual base rent due under the remaining leases is \$6.4 million (unaudited) for the period from October 1, 2012 to December 31, 2012, \$25.7 million (unaudited) for each of 2013-2016, and \$15.0 million (unaudited) for 2017. The base rent is subject to annual adjustments on January 1 of each year generally equal to the percentage change in the average same store net operating income ("NOI") for certain other specified properties managed by an affiliate of the Properties. Rental income related to these leases is recognized in the period earned over the lease term in accordance with FASB ASC 840, *Leases*.

Repairs and Maintenance

Repairs and maintenance costs are expensed as incurred, while significant improvements, renovations and replacements are capitalized.

Advertising Costs

All advertising costs are expensed as incurred and included as operating expenses in the accompanying Statements. For the nine-month period ending September 30, 2012 (unaudited) and the year ended December 31, 2011, advertising expenses were approximately \$2,338,000 and \$2,754,000, respectively.

3. Related Party Transactions

An affiliate of the Properties performed the property management function and charged the Properties total management fees in the amounts of approximately \$584,000 and \$22,000 for the nine-month period ending September 30, 2012 (unaudited) and the year ended December 31, 2011, respectively, which are included in management fees in the accompanying Statements.

An affiliate of the Properties incurred costs to manage the Properties and allocated a proportionate share of the costs to the Properties based on a percentage of revenue in the amounts of \$8,649,237 and \$10,501,057 for the nine-month period ending September 30, 2012 (unaudited) and the year ended December 31, 2011, respectively, and such amounts are included in management fees in the accompanying Statements.

An affiliate of the Properties allocated insurance expense under a master policy to the Properties and various affiliated properties. Insurance expense was \$7,666,047 and \$10,175,535 for the nine months ended September 30, 2012 (unaudited) and the year ended December 31, 2011, respectively, and such amounts are included in real estate taxes and insurance in the accompanying Statements.

4. Ground Leases

Two of the Properties are subject to a non-cancelable ground lease in effect as of December 31, 2011. The effective minimum annual payments, excluding contingent rent based on a percentage of gross income, are as follows:

Year	 Amount
2012	\$ 42,000
2013	42,000
2014	42,000
2015	42,000
2016	42,000
Thereafter	1,844,500
Total	\$ 2,054,500

F-4

Table of Contents

AVALONBAY COMMUNITIES, INC.

UNAUDITED PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

As of and For the Nine Months Ended September 30, 2012 and For the Year Ended December 31, 2011

On November 26, 2012, AvalonBay Communities, Inc. (the "Company") entered into an asset purchase agreement (the "Purchase Agreement") with (i) Equity Residential and its operating partnership, ERP Operating Partnership LP (collectively "EQR"), (ii) Lehman Brothers Holdings Inc. and certain of its affiliates ("Lehman") and (iii) Archstone Enterprise LP ("Archstone") pursuant to which the Company and EQR will acquire, directly or indirectly, all of the assets and entities owned by, and certain liabilities of, Archstone (the "Acquisition"). Pursuant to the Purchase Agreement and a separate agreement between the Company and EQR, the Company's portion of consideration for the Acquisition is as follows: (i) 14,889,706 shares of the Company's common stock, (ii) \$669.0 million in cash from the Company, (iii) assumption of indebtedness with a principal amount of \$3.7 billion and a fair value of approximately \$3.9 billion, (iv) an obligation to pay, when presented for redemption, approximately \$132.2 million in respect of the liquidation value and accrued dividends on outstanding Archstone preferred units and (v) 40% of certain other Archstone liabilities. Through the Acquisition, the Company expects to acquire a full or partial ownership interest in a portfolio of properties (the "Archstone Portfolio"), consisting of 68 operating apartment communities, containing 22,167 apartment homes, seven development properties and three land holdings. The remaining assets and liabilities of Archstone that are not acquired or assumed directly or indirectly by either the Company or EQR will be acquired through one or more joint ventures to be formed between the Company and EQR. In addition, the Company anticipates incurring approximately \$241.8 million in transaction costs and costs associated with funding the Acquisition.

The Acquisition, including transaction and funding related costs, is expected to be funded through:

- · approximately \$1.8 billion of net proceeds from this offering;
- the assumption, by the Company, of indebtedness with a fair value of approximately \$3.9 billion (as of September 30, 2012) of which the Company anticipates repaying \$953 million; and

the issuance, by the Company, of 14,889,706 shares of the Company's common stock, par value \$0.01 per share.

The accompanying unaudited Pro Forma Condensed Consolidated Balance Sheet of the Company is presented as if the Acquisition had occurred on September 30, 2012. The accompanying unaudited Pro Forma Condensed Consolidated Statements of Operations for the nine months ended September 30, 2012 and the year ended December 31, 2011 are presented as if the Acquisition had occurred on January 1, 2011.

The Acquisition will be accounted for using the acquisition method of accounting. The total consideration will be allocated to the assets ultimately acquired and the liabilities ultimately assumed at their respective fair values. The allocations of the purchase price reflected in these unaudited Pro Forma Condensed Consolidated Financial Statements have not been finalized and are based upon preliminary estimates of fair values, which is the best available information at the current time. The final determination of the fair values of these assets and liabilities, which cannot be made prior to the completion of the Acquisition, will be based on the actual valuations of tangible and intangible assets and liabilities that exist as of the date the transaction is completed. Consequently, amounts preliminarily allocated to identifiable tangible and intangible assets and liabilities could change significantly from those used in the accompanying unaudited Pro Forma Condensed Consolidated Financial Statements and could result in a material change in depreciation and amortization of tangible and intangible assets and liabilities. Additionally, proceeds assumed to satisfy our purchase obligation are predicated on anticipated issuances of equity securities and debt by the Company. There can be no assurance that such transactions will occur on the terms estimated or at all.

These unaudited Pro Forma Condensed Consolidated Financial Statements are prepared for informational purposes only, and are based on assumptions and estimates considered appropriate by the Company's management. However, they are not necessarily indicative of what the Company's consolidated financial condition or results of operations actually would have been assuming the Acquisition had occurred as of the dates indicated, nor do they purport to represent the consolidated financial position or results of operations for future periods. The final valuation of assets and liabilities, allocation of the purchase price, timing of completion of the Acquisition and other changes to the Archstone Portfolio's tangible and

F-5

Table of Contents

intangible assets and liabilities that occur prior to completion of the Acquisition, as well as the ability to obtain loan servicer consents or satisfy other closing conditions, could cause material differences in the information presented.

These unaudited Pro Forma Condensed Consolidated Financial Statements should be read in conjunction with the Company's historical consolidated financial statements reported on Form 10-Q for the nine months ended September 30, 2012 and Form 10-K for the year ended December 31, 2011, as included in the Company's previous filings with the Securities and Exchange Commission.

F-6

Table of Contents

AVALONBAY COMMUNITIES, INC

PRO FORMA CONDENSED CONSOLIDATED BALANCE SHEET

As of September 30, 2012 (Unaudited) (Dollars in thousands)

	Historical 9-30-12 (A)		Acquisition of Archstone Portfolio (B)		Pro Forma Adjustments		 Pro Forma 9-30-12
ASSETS		,		,			
Real estate:							
Net real estate investments	\$	7,878,648	\$	6,498,942(C)	\$	_	\$ 14,377,590
Cash and cash equivalents		664,133		(1,807,915)(K)		1,807,915(K)	664,133
Cash in escrow		49,851		_		_	49,851
Investments in unconsolidated real estate entities		139,405		36,107(C),(D)		_	175,512
Deferred financing costs, net		33,557		32,100(F)		_	65,657
Other assets		202,972		68,665(E)			 271,637
Total assets	\$	8,968,566	\$	4,827,899	\$	1,807,915	\$ 15,604,380
LIABILITIES AND EQUITY		<u> </u>					
Unsecured notes, net	\$	1,899,208	\$	_	\$	_	\$ 1,899,208
Variable rate unsecured credit facility		_		_		_	_
Mortgage notes payable		1,908,872		2,949,317(F)		_	4,858,189
Accrued expenses and other liabilities		432,408		118,477(G)			550,885
Total liabilities		4,240,488		3,067,794		<u> </u>	7,308,282
Redeemable noncontrolling interests	· ·	7,203		_		_	 7,203
Equity:							
Preferred stock, \$0.01 par value; \$25 liquidation preference		_		_		_	_
Common stock, \$0.01 par value		977		149(H)		145(J)	1,271
Additional paid-in capital		4,980,937		1,913,774(H)		1,807,770(J)	8,702,481
Accumulated earnings less dividends		(153,811)		(153,818)(I)		_	(307,629)
Accumulated other comprehensive loss		(110,787)					(110,787)
Total stockholders' equity		4,717,316		1,760,105		1,807,915	8,285,336
Noncontrolling interests		3,559		_		_	3,559
Total equity		4,720,875		1,760,105		1,807,915	8,288,895
Total liabilities and equity	\$	8,968,566	\$	4,827,899	\$	1,807,915	\$ 15,604,380

Table of Contents

AVALONBAY COMMUNITIES, INC.

PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS

For the Nine Months Ended September 30, 2012

(Unaudited)

(Dollars in thousands, except per share data)

		Historical 9-30-12 (AA)	Acquisition of Archstone Portfolio (BB)		rchstone Pro Forma ortfolio Adjustments		 Pro Forma 9-30-12
Revenue:		, ,		, ,			
Rental and other income	\$	773,424	\$	346,781	\$	(22,070)(CC)	\$ 1,098,135
Management, development and other fees		7,852		<u> </u>		<u> </u>	7,852
Total revenue	_	781,276		346,781		(22,070)	1,105,987
Expenses:							
Operating expenses		280,477		119,412		(10,349)(CC)	389,540
Interest expense, net		100,804		_		57,139(DD)	157,943
Loss on extinguishment of debt, net		1,179		_		_	1,179
Depreciation expense		193,434		_		106,206(EE)	299,640
General and administrative expense		26,398		_		1,605(FF)	28,003
Impairment loss — land holdings		_		_		_	_
Total expenses		602,292		119,412		154,601	876,305
Equity in income (loss) of unconsolidated entities		9,801		_		(7,273)(CC)	2,528
Gain on sale of land		280		_		_	280
Gain on acquisition of unconsolidated real estate entity		14,194		_		_	14,194
Income from continuing operations		203,259		227,369		(183,944)	246,684
Net income attributable to noncontrolling interests		334				_	334
Net income from continuing operations attributable to common							
stockholders	\$	203,593	\$	227,369	\$	(183,944)	\$ 247,018
Earnings per common share — basic (from continuing operations)	\$	2.12					\$ 1.97
Earnings per common share — diluted (from continuing							
operations)	\$	2.11					\$ 1.96
Weighted Average Shares — Basic		95,742,676		14,889,706(H)	14,500,000(J)	125,132,382
Weighted Average Shares — Diluted		96,401,558		14,889,706(H)	14,500,000(J)	125,791,264

See accompanying notes to unaudited Pro Forma Condensed Consolidated Financial Statements.

F-8

Table of Contents

AVALONBAY COMMUNITIES, INC.

PRO FORMA CONDENSED CONSOLIDATED STATEMENT OF OPERATIONS

For the Year Ended December 31, 2011

(Unaudited)

(Dollars in thousands, except per share data)

Revenue:	Historical 12-31-11 (AA)	Acquisition of Archstone Portfolio (BB)	Pro Forma Adjustments	Pro Forma 12-31-11
Rental and other income	\$ 959,055		\$ (8,333)(CC)	\$ 1,360,264
Management, development and other fees	9,656	_	·	9,656
Total revenue	968,711	409,542	(8,333)	1,369,920
Expenses:			<u> </u>	
Operating expenses	361,401	135,068	(3,483)(CC)	492,986
Interest expense, net	168,179	_	73,260(DD)	241,439
Loss on extinguishment of debt, net	1,940	_	_	1,940
Depreciation expense	246,666	_	374,940(EE)	621,606
General and administrative expense	29,371	_	2,140(FF)	31,511
Impairment loss — land holdings	14,052	_	_	14,052
Total expenses	821,609	135,068	446,857	1,403,534
Equity in income (loss) of unconsolidated entities	5,120	_	(7,064)(CC)	(1,944)
Gain on sale of land	13,716	_	_	13,716
Gain on acquisition of unconsolidated real estate entity	_	_	_	_
Income from continuing operations	165,938	274,474	(462,254)	(21,842)
Net income attributable to noncontrolling interests	252			252

Net income from continuing operations attributable to common stockholders	\$ 166,190	\$	274,474	\$	(462,254)	\$	(21,590)
Earnings per common share — basic (from continuing	 ,	<u> </u>	,	_			
operations)	\$ 1.84					\$	(0.18)
Earnings per common share — diluted (from continuing							
operations)	\$ 1.84					\$	(0.18)
Weighted Average Shares — Basic	89,922,465		14,889,706(H)		14,500,000(J)		119,312,171
Weighted Average Shares — Diluted	90,777,462		14,889,706(H)		14,500,000(J)		120,167,168

See accompanying notes to unaudited Pro Forma Condensed Consolidated Financial Statements.

F-9

Table of Contents

AVALONBAY COMMUNITIES, INC.

NOTES TO PRO FORMA CONDENSED CONSOLIDATED FINANCIAL STATEMENTS

(Unaudited)

1. Basis of Pro Forma Presentation

AvalonBay Communities, Inc. (the "Company," which term, unless the context otherwise requires, refers to AvalonBay Communities, Inc. together with its consolidated subsidiaries), is a Maryland corporation that elected to be taxed as a real estate investment trust ("REIT") under the Internal Revenue Code of 1986 (the "Code"). The Company focuses on the development, acquisition, ownership and operation of apartment communities in high barrier to entry markets of the United States. These markets are located in the New England, Metro New York/New Jersey, Mid-Atlantic, Pacific Northwest, and Northern and Southern California regions of the country.

On November 26, 2012, the Company entered into an asset purchase agreement (the "Purchase Agreement") with (i) Equity Residential and its operating partnership, ERP Operating Partnership LP (collectively "EQR"), (ii) Lehman Brothers Holdings Inc. and certain of its affiliates ("Lehman") and (iii) Archstone Enterprise LP ("Archstone") pursuant to which the Company and EQR will acquire, directly or indirectly, all of the assets and entities owned by, and certain liabilities of, Archstone (the "Acquisition"). Pursuant to the Purchase Agreement and separate agreement between the Company and EQR, the Company's portion of consideration of the Acquisition is as follows: (i) 14,889,706 shares of the Company's common stock, (ii) \$669.0 million in cash from the Company, (iii) assumption of indebtedness with a principal amount of \$3.7 billion and a fair value of approximately \$3.9 billion, (iv) an obligation to pay, when presented for redemption, approximately \$132.2 million in respect of the liquidation value and accrued dividends on outstanding Archstone preferred units and (v) 40% of certain other Archstone liabilities. Through the Acquisition, the Company expects to acquire a full or partial ownership interest in a portfolio of properties (the "Archstone Portfolio"), consisting of 68 operating apartment communities, containing 22,167 apartment homes, seven development properties and three land holdings. The remaining assets and liabilities of Archstone that are not acquired or assumed directly or indirectly by either the Company or EQR will be acquired through one or more joint ventures to be formed between the Company and EQR. In addition, the Company anticipates incurring approximately \$241.8 million in transaction costs and costs associated with funding the Acquisition.

The Acquisition, including transaction and funding related costs, is expected to be funded through:

- · approximately \$1.8 billion of net proceeds from this offering;
- the assumption, by the Company, of indebtedness with a fair value of approximately \$3.9 billion (as of September 30, 2012) of which the Company anticipates repaying \$953 million; and
- the issuance, by the Company, of 14,889,706 shares of the Company's common stock, par value \$0.01 per share.

2. Adjustments to unaudited Pro Forma Condensed Consolidated Balance Sheet

- (A) Represents the historical condensed consolidated balance sheet of the Company as of September 30, 2012, as contained in the unaudited historical condensed consolidated financial statements and notes thereto filed on Form 10-Q.
- (B) Represents adjustments related to the Acquisition, which is expected to close in the first quarter of 2013. The preliminary estimated fair value of assets to be received and consideration to be given is as follows (dollars in thousands, except per share data):

Preliminary estimated fair value of real estate assets	\$ 6,498,942
Preliminary estimated fair value of real estate assets held in unconsolidated joint ventures, at share	410,112
Total preliminary estimated fair value of real estate assets	\$ 6,909,054
Equity to be issued (14,889,706 shares at \$128.54 per share)(1)	\$ 1,913,923
Cash to be paid	669,000
Repayment of debt	952,997
Assumption of debt(2)	2,949,317
Assumption of unconsolidated joint venture debt, at share(2)	241,829
Assumption of Archstone perferred equity(3)	132,176
Assumption of other net liabilities	49,812
Total consideration to be given	\$ 6,909,054

F-10

Table of Contents

⁽¹⁾ Estimated issuance price per share is based on the closing price of the Company's common stock on the New York Stock Exchange ("NYSE") on November 23, 2012. Purchase price will be adjusted based on the share price of the Company's common stock at closing, consistent with the requirements of ASC 805, Business Combinations.

- (2) Includes mark-to-market adjustments to reflect fair value of \$194.0 million for consolidated debt and \$3.5 million for unconsolidated debt (at share).
- (3) Obligation related to Archstone preferred equity holders assumed though joint venture with EQR as discussed below.

The Company intends to account for the Acquisition under the acquisition method of accounting and recognize the estimated fair value of acquired assets and assumed liabilities on the date of acquisition. The fair values of these assets and liabilities have been preliminarily determined in accordance with Accounting Standards Codification ("ASC") section 820-10, *Fair Value Measurements*. A final determination of the fair values of the assets acquired and liabilities assumed in connection with the Acquisition, which cannot be made prior to the completion of the Acquisition, will be based on the actual valuation of the tangible and intangible assets and liabilities that exist as of the date of completion of the Acquisition.

(C) The Company's preliminary purchase price allocations based on estimated fair value to real estate assets are as follows (dollars in thousands):

Land	\$ 1,860,518
Buildings and improvements	3,985,070
FF&E	61,403
Construction-in-progress, including land and land held for development	358,619
In-place lease intangibles	 233,332
Total consolidated	6,498,942
Unconsolidated interests in joint ventures	410,112
Total	\$ 6,909,054

(D) Represents the expected acquisition of (i) a 28.6% interest in a joint venture that owns nine apartment communities; (ii) a 26.1% interest in a joint venture that owns one apartment community; (iii) a 21.6% interest in a joint venture that owns one development community; and (iv) joint ventures expected to be formed with EQR to acquire those remaining assets and liabilities of Archstone that are not acquired or assumed directly by either the Company or EQR. The Company anticipates that each of these joint ventures will be unconsolidated subsequent to the Acquisition.

The Company is still in the process of negotiating the terms and conditions of its joint venture arrangements with EQR, which will hold assets that we will jointly manage, sell to third parties and/or subsequently transfer to EQR or to the Company. The Company expects that the structure of these arrangements will result in an approximate 40% ownership in such joint ventures. The preliminary estimated fair value of each of these joint ventures is as follows (dollars in thousands):

		Estimated Fair Value							
	Ownership %		Real Estate		Debt		Other		Net (at Share)
Interest in JV	28.6%	\$	175,832	\$	(105,532)	\$	_	\$	70,300
Interest in JV	26.1%		26,937		(4,674)		_		22,263
Interest in JV	21.6%		5,893		(1,568)		_		4,325
Interest in JVs with EQR	TBD		201,450		(130,055)		(132,176)(1)		(60,781)
		\$	410,112	\$	(241,829)	\$	(132,176)	\$	36,107

F-11

Table of Contents

- Under the terms of the Purchase Agreement, the Company and EQR have agreed to redeem the outstanding preferred equity obligations of Archstone pursuant to the terms of those obligations when presented for redemption. The Company anticipates that these obligations will be apportioned approximately 40% to the Company through a joint venture arrangement with EQR. As of September 30, 2012, our portion of the total amount due upon redemption of the preferred equity obligations, which includes the par value of the preferred obligations plus any accrued dividends, is approximately \$132.2 million. Therefore, we have included this obligation in the overall valuation of the net assets (liabilities) to be acquired through joint venture arrangements with EQR. In addition, we have included an estimate of preferred dividends to be paid in determining the proforma adjustment for equity in income (loss) for unconsolidated joint ventures in (CC) below.
- (E) Represents the preliminary estimated fair value of non-real estate assets anticipated to be acquired in the Acquisition, including accounts receivable, prepaid expenses and escrows associated with assumed debt.
- (F) Represents the debt that the Company expects to assume of approximately \$2.75 billion as of September 30, 2012 with an annual weighted average interest rate of approximately 4.9% and a weighted average maturity of approximately 9.1 years. This debt is collateralized by the respective real estate assets.

The estimated fair value of the debt assumed is approximately \$2.95 billion. The difference between the stated mortgage loan amounts and their fair value will be amortized as a reduction of interest expense over the terms of the respective loans. In connection with the assumption of mortgage loans, the Company expects to incur approximately \$32.1 million of assumption and loan substitution costs.

- (G) Represents the preliminary estimated fair value of other liabilities anticipated to be assumed in the Acquisition, including accounts payable, accrued expenses and accrued interest associated with debt assumed.
- (H) Represents the private placement of 14,889,706 shares of common stock. The fair market value of the shares will be determined at issuance, net of issuance costs. The estimated issuance price per share below is based on the closing price of the Company's common stock on the NYSE on November 23, 2012. The shares of common stock expected to be issued upon closing of the Acquisition are valued as follows (dollars in thousands, except per share data):

Number of shares issued	14,889,706
Estimated issuance price, per share	\$ 128.54
Gross value of shares issued	\$ 1,913,923

If the price per share of the Company's common stock were to increase by 5% or decrease by 5%, the value of the shares issued would increase by \$95.7 million or decrease by \$95.7 million, respectively.

(I) Represents estimated transaction costs expected to be incurred for the Acquisition of approximately \$153.8 million, consisting primarily of transfer taxes, legal and other professional fees and separation costs. Amounts exclude costs related to equity or debt financing, as disclosed elsewhere in these unaudited Pro Forma

Condensed Consolidated Financial Statements.

- (J) Represents the estimated net proceeds from the issuance of approximately 14,500,000 shares of the Company's common stock based on \$128.54 per share in this offering, net of expected issuance costs of approximately \$55.9 million. The estimated price per share is equal to the closing price of the Company's common stock on the NYSE on November 23, 2012.
- (K) Represents the net cash activity for the Acquisition, as funded with the proceeds from this offering.

2. Adjustments to unaudited Pro Forma Condensed Consolidated Statements of Operations

(AA) Represents the historical consolidated statements of operations of the Company for the nine months ended September 30, 2012 (unaudited) and for the year ended December 31, 2011 as contained in the historical consolidated financial statements included in previous filings with the Securities and Exchange Commission.

Amounts exclude the historical operations for discontinued operations (primarily real estate asset sales) as previously reported and only reflect those amounts through continuing operations.

F-12

Table of Contents

- (BB) Represents the historical combined statements of revenue and certain expenses for the Archstone Portfolio for the nine months ended September 30, 2012 (unaudited) and the year ended December 31, 2011.
- (CC) Represents the portion of the Archstone Portfolio that the Company anticipates acquiring through interests in unconsolidated joint ventures. The Pro Forma Adjustment for each period also includes the Company's anticipated pro rata share of interest expense on assumed debt and depreciation expense, based on the preliminary estimated fair values of the underlying real estate.

The breakdown of equity in income (loss) for unconsolidated joint ventures is as follows (at pro rata share) (dollars in thousands):

	 Nine Months 9-30-12	Year Ended 12-31-11		
Revenues	\$ 14,914	\$	11,058	
Operating expenses	(7,284)		(5,095)	
Depreciation expense	(5,924)		(4,019)	
Interest expense	(4,930)		(3,610)	
Preferred distributions	(4,049)		(5,398)	
	\$ (7,273)	\$	(7,064)	

(DD) Represents estimated interest expense for each period as follows (dollars in thousands):

	N	ne Months 9-30-12	 Year Ended 12-31-11
Interest expense on assumed debt	\$	95,061	\$ 123,823
Amortization of assumed debt mark-to-market		(41,879)	(55,839)
Amortization of debt assumption and loan substitution costs		3,957	5,276
	\$	57,139	\$ 73,260

(EE) Represents the estimated depreciation expense related to acquired real estate and the estimated amortization expense related to intangible assets for in-place leases, as follows (dollars in thousands):

	Useful Lives (Yrs)	Nine Months 9-30-12	 Year Ended 12-31-11
Land	N/A	\$ _	\$ _
Buildings and improvements	30	99,627	132,836
FF&E	7	6,579	8,772
In-place lease intangibles	1	_	233,332
		\$ 106,206	\$ 374,940

(FF) Represents incremental costs that the Company expects to incur on an on-going basis as a result of the Acquisition. These incremental amounts relate primarily to corporate support functions. These estimates should be reviewed in conjunction with the Combined Statements of Revenue and Certain Expenses as included elsewhere in this report, where the historical management fees incurred by the Archstone Portfolio are disclosed.

F-13

Table of Contents

AVALONBAY COMMUNITIES, INC.

RECONCILIATION OF NON-GAAP FINANCIAL MEASURES

(Unaudited)

Funds from Operations Attributable to Common Stockholders ("FFO")

FFO is considered by the Company to be an appropriate supplemental measure of its operating and financial performance. In calculating FFO, the Company excludes gains or losses related to dispositions of previously depreciated property and excludes real estate depreciation, which can vary among owners of identical assets in similar condition based on historical cost accounting and useful life estimates. FFO can help one compare the operating performance of a real estate company between periods or as compared to different companies. The Company believes that in order to understand its operating results, FFO should be examined with net income as presented in our unaudited Pro Forma Condensed Consolidated Financial Statements.

Consistent with the definition adopted by the Board of Governors of the National Association of Real Estate Investment Trusts® ("NAREIT"), the Company calculates FFO as net income or loss computed in accordance with GAAP, adjusted for:

- · gains or losses on sales of previously depreciated operating communities;
- extraordinary gains or losses (as defined by GAAP);
- · cumulative effect of change in accounting principle;
- · impairment write-downs of depreciable real estate assets;
- · write-downs of investments in affiliates due to a decrease in the value of depreciable real estate assets held by those affiliates;
- depreciation of real estate assets; and
- · adjustments for unconsolidated partnerships and joint ventures.

FFO does not represent net income attributable to common stockholders in accordance with GAAP, and therefore it should not be considered an alternative to net income, which remains the primary measure of performance. In addition, FFO as calculated by other REITs may not be comparable to the Company's calculation of FFO.

The following is a reconciliation of pro forma net income attributable to common stockholders to pro forma FFO for the nine months ended September 30, 2012 and the year ended December 31, 2011 (all unaudited, dollars in thousands, except per share data):

	I		e nine months ender ptember 30, 2012	d				or the year ended ecember 31, 2011	
	Company Historical		Archstone Acquisition/ Pro Forma		Company Pro Forma	Company Historical		Archstone Acquisition/ Pro Forma	Company Pro Forma
Net income from continuing operations	_		_			 _		_	_
attributable to common stockholders	\$ 203,593	\$	43,425	\$	247,018	\$ 166,190	\$	(187,780)	\$ (21,590)
Discontinued operations	97,919				97,919	275,432			275,432
Net income attributable to common									
stockholders	301,512		43,425		344,937	441,622		(187,780)	253,842
Depreciation — real estate assets, including discontinued operations and joint venture adjustments	199,593		112,130		311,723	256,986		378,959	635,945
Distributions to noncontrolling interests,	,				,	,		,	,
including discontinued operations	21		_		21	27		_	27
Gain on sale of unconsolidated entities holding									
previously depreciated real estate assets	(1,471)		_		(1,471)	(3,063)		_	(3,063)
Gain on sale of operating communities	(95,049)		_		(95,049)	(281,090)		_	(281,090)
Gain on acquisition of unconsolidated real									
estate entity	(14,194)				(14,194)				
FFO attributable to common stockholders	\$ 390,412	\$	155,555	\$	545,967	\$ 414,482	\$	191,179	\$ 605,661
Weighted average common shares outstanding		_					_		
— diluted	96,401,558		29,389,706		125,791,264	90,777,462		29,389,706	120,167,168
EPS per common share — diluted	\$ 3.13			\$	2.74	\$ 4.87			\$ 2.11
FFO per common share — diluted	\$ 4.05			\$	4.34	\$ 4.57			\$ 5.04
			F-14						

Table of Contents

FFO also does not represent cash generated from operating activities in accordance with GAAP, and therefore should not be considered an alternative to net cash flows from operating activities, as determined by GAAP, as a measure of liquidity. Additionally, it is not necessarily indicative of cash available to fund cash needs. The FFO information presented above should be read in conjunction with the Company's historical statements of cash flows and discussion of liquidity and capital resources as reported on Form 10-Q for the nine months ended September 30, 2012 and Form 10-K for the year ended December 31, 2011, as included in the Company's previous filings with the Securities and Exchange Commission.

Net Operating Income ("NOI")

NOI is considered by the Company to be an important and appropriate supplemental performance measure to net income because it helps both investors and management to understand the core operations of a community or communities prior to the allocation of any corporate-level or financing-related costs. NOI reflects the operating performance of a community and allows for an easy comparison of the operating performance of individual assets or groups of assets. In addition, because prospective buyers of real estate have different financing and overhead structures, with varying marginal impacts to overhead by acquiring real estate, NOI is considered by many in the real estate industry to be a useful measure for determining the value of a real estate asset or group of assets. We define NOI as total property revenue less direct property operating expenses, including property taxes, and excluding corporate-level income (including management, development and other fees), corporate-level property management and other indirect operating expenses, investments and investment management expenses, expensed development and other pursuit costs, net interest expense, gain (loss) on extinguishment of debt, general and administrative expense, joint venture income (loss), depreciation expense, impairment loss on land holdings, gain on sale of real estate assets and income from discontinued operations.

NOI does not represent cash generated from operating activities in accordance with GAAP. Therefore, NOI should not be considered an alternative to net income as an indication of our performance. NOI should also not be considered an alternative to net cash flow from operating activities, as determined by GAAP, as a measure of liquidity, nor is NOI indicative of cash available to fund cash needs. Reconciliations of pro forma NOI for the nine months ended September 30, 2012 and the year ended December 31, 2011 to pro forma net income for each period are as follows (all unaudited, dollars in thousands):

		September 30, 2012	1		December 31, 2011	
•	Company	Archstone	Company	Company	Archstone	Company
	Historical	Acquisition/	Pro Forma	Historical	Acquisition/	Pro Forma

		Pro Forma			Pro Forma	
Income from continuing operations	\$ 203,259	\$ 43,425	\$ 246,684	\$ 165,938	\$ (187,780)	\$ (21,842)
Discontinued operations	97,919	_	97,919	275,432		275,432
Net income	301,178	43,425	344,603	441,370	(187,780)	253,590
Indirect operating expenses, net of corporate						
income	24,049	_	24,049	30,550	_	30,550
Investment and investment management						
expense	4,526	_	4,526	5,126	_	5,126
Expensed development and other pursuit costs	1,749	_	1,749	2,967	_	2,967
Interest expense, net	100,804	57,139	157,943	168,179	73,260	241,439
Loss on extinguishment of debt, net	1,179	_	1,179	1,940	_	1,940
General and administrative expense	26,398	1,605	28,003	29,371	2,140	31,511
Equity in income of unconsolidated entities	(9,801)	7,273	(2,528)	(5,120)	7,064	1,944
Depreciation expense	193,434	106,206	299,640	246,666	374,940	621,606
Impairment loss — land holdings	_	_	_	14,052	_	14,052
Gain on sale of real estate assets	(95,329)	_	(95,329)	(294,806)	_	(294,806)
Income from discontinued operations	(2,870)	_	(2,870)	5,658	_	5,658
Gain on acquisition of unconsolidated entity	(14,194)	_	(14,194)	_	_	_
Net operating income	\$ 531,123	\$ 215,648	\$ 746,771	\$ 645,953	\$ 269,624	\$ 915,577

F-15

Table of Contents

Earnings Before Interest, Taxes, Depreciation and Amortization ("EBITDA")

EBITDA is defined as net income attributable to common stock holders before interest income and expense (including gain or loss on extinguishment of debt), income taxes, depreciation and amortization from both continuing and discontinued operations. Management generally considers EBITDA to be an appropriate supplemental measure to net income of our operating performance because it helps investors to understand our ability to incur and service debt and to make capital expenditures. EBITDA should not be considered as an alternative to net income (as determined in accordance with GAAP), as an indicator of our operating performance, or to cash flows from operating activities (as determined in accordance with GAAP) as a measure of liquidity. Our calculation of EBITDA may not be comparable to EBITDA as calculated by other companies. Reconciliations of pro forma EBITDA for the nine months ended September 30, 2012 and the year ended December 31, 2011 to pro forma net income from continuing operations attributable to common stockholders for each period are as follows (all unaudited, dollars in thousands):

	For the nine months ended September 30, 2012			d		For the year ended December 31, 2011					
	Company Historical		Archstone Acquisition/ Pro Forma		Company Pro Forma		Company Historical		Archstone Acquisition/ Pro Forma		Company Pro Forma
Net income from continuing operations											
attributable to common stockholders	\$ 203,593	\$	43,425	\$	247,018	\$	166,190	\$	(187,780)	\$	(21,590)
Discontinued operations	97,919		_		97,919		275,432		_		275,432
Net income attributable to common			<u></u>								
stockholders	301,512		43,425		344,937		441,622		(187,780)		253,842
Interest expense, net	101,983		57,139		159,122		170,119		73,260		243,379
Interest expense, discontinued operations	735		_		735		4,443		_		4,443
Depreciation expense	193,434		106,206		299,640		246,666		374,940		621,606
Depreciation expense, discontinued operations	895		_		895		3,603		_		3,603
EBITDA	\$ 598,559	\$	206,770	\$	805,329	\$	866,453	\$	260,420	\$	1,126,873
EBITDA — Continuing operations	\$ 499,010	\$	206,770	\$	705,780	\$	582,975	\$	260,420	\$	843,395
EBITDA — Discontued operations	99,549				99,549		283,478				283,478
EBITDA	\$ 598,559	\$	206,770	\$	805,329	\$	866,453	\$	260,420	\$	1,126,873

F-16

Table of Contents

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the Registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

AVALONBAY COMMUNITIES, INC.

November 26, 2012

By: /s/Thomas J. Sargeant
Name: Thomas J. Sargeant
Title: Chief Financial Officer

Table of Contents

EXHIBIT INDEX

2.1	Asset Purchase Agreement, dated November 26, 2012, by and among AvalonBay Communities, Inc., Equity Residential and its operating partnership, ERP Operating Partnership LP, Lehman Brothers Holdings Inc., and Archstone Enterprise LP *
4.1	Form of Registration Rights Agreement to be entered into by and between Lehman Brothers Holdings Inc. and AvalonBay Communities, Inc.
10.1	Form of Shareholders Agreement to be entered into by and among AvalonBay Communities, Inc., Archstone Enterprise LP and Lehman Brothers Holdings Inc.
23.1	Consent of KPMG LLP
99.1	Press Release dated November 26, 2012

^{*} Certain schedules and exhibits omitted pursuant to Item 601(b)(2) of Regulation S-K. The Company agrees to furnish supplementally a copy of any omitted schedule or exhibit to the SEC upon request.

ASSET PURCHASE AGREEMENT

by and among

ERP OPERATING LIMITED PARTNERSHIP,

EQUITY RESIDENTIAL,

AVALONBAY COMMUNITIES, INC.,

LEHMAN BROTHERS HOLDINGS INC.,

and

ARCHSTONE ENTERPRISE LP

November 26, 2012

TABLE OF CONTENTS

		Page
ARTICLE 1 DEF	TNITIONS	2
1.1	Definitions	2
1.2	Additional Definitions	18
ARTICLE 2 PUR	RCHASE AND SALE OF THE PURCHASED INTERESTS	20
2.1	Purchase and Sale	20
2.2	Purchase Price	21
2.3	Purchase Price Adjustment	22
2.4	Initial Closing	25
2.5	Deferred Closings	26
2.6	Asset Transfer Transactions	32
ARTICLE 3 EFF	ECT OF SALE	35
3.1	Required Tax Withholding	35
3.2	No Further Ownership Rights	36
ARTICLE 4 REF	PRESENTATIONS AND WARRANTIES OF LBHI	36
4.1	Organization and Good Standing	36
4.2	Authorization; Validity of Agreements	36
4.3	Consents and Approvals; No Violations	37
4.4	Ownership of Equity Interests in Seller	37
4.5	Brokerage Fees	38
4.6	Purchase for Investment	38
4.7	Legal Proceedings; Claims	38
4.8	Non-Reliance	38
ARTICLE 5 REF	PRESENTATIONS AND WARRANTIES OF SELLER	39
5.1	Organization and Good Standing	39
5.2	Authorization; Validity of Agreements	39
5.3	Consents and Approvals; No Violations	40
5.4	Ownership of Transferred Assets	40
5.5	Brokerage Fees	41
5.6	Financial Statements	41
5.7	Absence of Dividends and Redemptions	41
5.8	Non-Reliance	41
ARTICLE 6 REF	PRESENTATIONS AND WARRANTIES OF LBHI AND SELLER CONCERNING THE ARCHSTONE ENTITIES	42
6.1	Organization and Good Standing	42
6.2	[Intentionally deleted]	42
6.3	Capitalization of the Archstone Entities	42
6.4	Books and Records	44
	i	

6.5 No Undisclosed Liabilities 44

6.6	Absence of Certain Changes or Events	44
6.7	Taxes	45
6.8	Compliance with Laws and Orders	46
6.9	Legal Proceedings; Claims	
		46
6.10	Executive Compensation	47
6.11	Brokerage Fees	47
6.12	No Other Representations and Warranties	47
	RESENTATIONS AND WARRANTIES OF THE EQUITY RESIDENTIAL PARTIES	48
7.1	Organization and Good Standing	48
7.2	Authorization; Validity of Agreements	48
7.3	Consents and Approvals; No Violations	48
7.4	Capitalization	49
7.5	SEC Reports and Financial Statements	50
7.6	Purchase for Investment	50
7.7	Financing	51
	<u> </u>	
7.8	Valid Issuance of Shares	52
7.9	Brokerage Fees	52
7.10	Non-Reliance	52
7.11	Solvency	53
A DELCT E A DEDI	DECENITATION CAND WADDANTIEC OF AVE	-
	RESENTATIONS AND WARRANTIES OF AVB	53
8.1	Organization and Good Standing	53
8.2	Authorization; Validity of Agreements	54
8.3	Consents and Approvals; No Violations	54
8.4	Capitalization	54
8.5	SEC Reports and Financial Statements	55
8.6	Purchase for Investment	56
8.7	Financing	56
8.8	Valid Issuance of Shares	
8.8 8.9		57
	Brokerage Fees	58
8.10	Non-Reliance	58
8.11	Solvency	59
A DELCE E A COM	DITTONO TO BACH BARTURO ON ICATION TO CLOCK	
	DITIONS TO EACH PARTY'S OBLIGATION TO CLOSE	59
9.1	Competition Matters	59
9.2	No Injunction or Illegality	59
ARTICLE 10 CO	NDITIONS TO THE BUYER PARTIES' OBLIGATION TO CLOSE	60
10.1	Conditions to the Initial Classes	60
10.1	Conditions to the initial Closing	60
10.1	Conditions to the Initial Closing Conditions for Each Extension Closing	64
	Conditions for Each Extension Closing	
	Conditions for Each Extension Closing	
10.2	Conditions for Each Extension Closing	64
ARTICLE 11 COM	Conditions for Each Extension Closing ii NDITIONS TO SELLER'S OBLIGATION TO CLOSE	62
10.2 ARTICLE 11 CON 11.1	Conditions for Each Extension Closing ii NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing	64 65 65
ARTICLE 11 COM	Conditions for Each Extension Closing ii NDITIONS TO SELLER'S OBLIGATION TO CLOSE	62
10.2 ARTICLE 11 CON 11.1 11.2	Conditions for Each Extension Closing ii NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing	62 63 63
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEF	Conditions for Each Extension Closing ii NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES	64 65 68 68
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEF 12.1	Conditions for Each Extension Closing ii NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing	64 65 68 69 69
ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters	64 65 68 69 69
ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties	62 63 63 69 69 72 76
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals	62 63 63 68 69 73 76
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO	62 63 63 69 69 72 76
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals	62 63 63 68 69 73 76
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO	64 65 65 68 69 75 76
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period	62 63 63 64 65 75 76 77
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS	62 63 63 64 65 65 75 76 77 77 78
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances	62 63 63 63 63 73 76 77 77 78 78
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS	62 63 63 64 65 65 75 76 77 77 78
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances	62 63 63 63 63 73 76 77 77 78 78
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations	62 63 63 63 63 73 76 77 77 78 78
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations	64 65 65 65 65 75 76 77 77 78 78 78
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses	64 65 65 68 69 75 76 77 77 78 78 78 81 82 82
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units	64 65 65 68 75 76 77 78 78 78 81 82 82
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7	Conditions for Each Extension Closing II NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements	64 65 65 68 75 76 77 78 78 78 81 82 82 83
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8	Conditions for Each Extension Closing III NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters	64 65 65 68 75 76 77 78 78 78 81 82 83 84 85
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement	64 65 65 68 69 75 76 77 78 78 78 81 82 83 84 85
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses	64 65 65 68 75 76 77 78 78 78 81 82 83 84 85
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement	64 65 65 68 69 75 76 77 78 78 78 81 82 83 84 85
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality	64 65 65 66 65 75 76 77 78 78 78 81 82 82 83 83 84 85 86 88
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12	Conditions for Each Extension Closing III NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements	64 65 65 68 66 75 76 77 78 78 78 81 82 82 82 83 84 85 86 88 88
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEN 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements Debt Financial Statements	64 65 65 66 66 67 77 77 78 78 78 81 82 82 82 83 84 85 86 88 88
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements Debt Financing Representative Indemnification	64 65 65 66 66 75 76 77 78 78 78 81 82 82 82 83 84 85 86 88 88 88 88 88 88 88 88 88 88 88 88
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements Debt Financing Representative Indemnification Employee Matters	64 65 65 65 65 65 75 76 77 78 78 78 81 82 82 83 84 85 86 86 86 87 87 88 88 88 88 88 88 88 88 88 88 88
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15 13.16	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements Debt Financing Representative Indemnification Employee Matters Mutual Release	64 65 65 65 65 65 75 76 77 77 78 78 81 82 82 83 84 85 86 88 88 89 90 91 91
ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15 13.16 13.17	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements Debt Financing Representative Indemnification Employee Matters Mutual Release Certain Intellectual Property Matters	64 65 65 65 66 75 76 77 77 78 78 78 81 82 82 83 84 85 86 88 88 89 90 90 90
10.2 ARTICLE 11 CON 11.1 11.2 ARTICLE 12 CEH 12.1 12.2 12.3 12.4 12.5 12.6 12.7 ARTICLE 13 ADI 13.1 13.2 13.3 13.4 13.5 13.6 13.7 13.8 13.9 13.10 13.11 13.12 13.13 13.14 13.15 13.16	NDITIONS TO SELLER'S OBLIGATION TO CLOSE Conditions for Initial Closing Conditions for Each Extension Closing RTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES Conduct of the Archstone Entities Prior to the Initial Closing Access and Operational Matters Sales and Marketing of Designated Properties Acquisition Proposals Suspension of Archstone IPO Termination of Confidentiality Agreement Cure Period DITIONAL COVENANTS Further Assurances Communications with Governmental Authorities No Control of Operations Regulatory Authorizations Severance Costs; Bonuses Series O Preferred Units and other Preferred Units Public Announcements Tax Matters Asset Acquisition Statement Transaction Expenses Confidentiality 3-14 Financial Statements Debt Financing Representative Indemnification Employee Matters Mutual Release	64 65 65 65 65 65 75 76 77 77 78 78 81 82 82 83 84 85 86 88 88 89 90 91 91

97

ARTICLE 14 SURVIVAL; INDEMNIFICATION

14.1	Survival		97
14.2	Indemnification		98
14.3	Indemnification Procedures		100
14.4	Existing Claims		103
14.5	Litigation Cooperation		103
14.6	Third Party Recovery		104
ARTICLE 15 TI	ERMINATION		104
15.1	Termination of Agreement		104
15.2	Effect of Termination		106
15.3	Remedies		107
		iii	
15.4	Specific Performance		109
ARTICLE 16 M	ISCELLANEOUS PROVISIONS		110
16.1	Notices		110
16.2	Entire Agreement		112
16.3	Amendments and Waivers		113
16.4	Governing Law		113
16.5	Jurisdiction and Venue		113
16.6	WAIVER OF TRIAL BY JURY		114
16.7	Binding Effect		115
16.8	Interpretation		115
16.9	Severability		116
16.10	Counterparts		116
16.11	Third Parties		117
16.12	Assignment		117
16.13	Joint and Several Obligations		117
16.14	Cash Payments		117
16.15	Schedules and Exhibits		117
16.16	Time Periods		117
16.17	No Recourse Against Non-Parties		118
		iv	

SCHEDULES

~	
Schedule II	Lehman Entities
Schedule 1.1(a)	Seller and LBHI Knowledge Parties
Schedule 1.1(b)	Certain Promissory Notes
Schedule 2.3.1(a)	Dividends
Schedule 2.5.1	Deferred Closing Assets
Schedule 2.6	Asset Transfer Actions
Schedule 4.3	LBHI Required Consents
Schedule 4.4.1	Ownership of Equity Interests in Seller
Schedule 4.5	Lehman Entity Brokerage Fees
Schedule 5.3	Seller Required Consents
Schedule 5.5	Seller and Archstone Entity Brokerage Fees
Schedule 5.6.1	Financial Statements
Schedule 6.3.1	Capitalization of the Primary Archstone Entities
Schedule 6.3.2(a)	Capitalization of the Archstone Entities
Schedule 6.3.2(b)	Capitalization of the Archstone Entities - Other
Schedule 6.5	No Undisclosed Liabilities
Schedule 6.6	Absence of Certain Changes or Events
Schedule 6.6.4	Compliance With Interim Operating Covenants
Schedule 6.7.1	Taxes
Schedule 6.7.2	Corporations; Pass-Throughs; REITs
Schedule 6.7.3	Tax Protection Agreements
Schedule 6.7.4	Archstone Entities and Subsidiaries Treated as Partnerships
Schedule 6.7.5	No Section 1374 Liability
Schedule 6.9.2	Legal Proceedings; Claims
Schedule 6.10	Executive Compensation
Schedule 6.11	Archstone Entity Brokerage Fees
Schedule 7.4.2	Capitalization of EQR
Schedule 8.4.2	Capitalization of AVB
Schedule 12.1	Interim Actions
Schedule 12.3	Asset Sale Procedures
Schedule 13.5.2	Employee Plans
Schedule 13.15.3	Individuals to be Terminated at Closing
Schedule 14.2.2	Certain Indemnification Matters

Transferred Interests

Schedule I

Exhibit A Assumption Agreement Exhibit B-1 Registration Rights Agreement - ERPOP Registration Rights Agreement - AVB Exhibit B-2 Shareholders Agreement - ERPOP Exhibit C-1 Shareholders Agreement - AVB Exhibit C-2 Deferred Closing Asset Escrow Agreement Exhibit D Exhibit E Bill of Sale Exhibit F Mutual Release

ASSET PURCHASE AGREEMENT

vi

This Asset Purchase Agreement (this "<u>Agreement</u>") is made effective as of November 26, 2012, by and among ERP Operating Limited Partnership, an Illinois limited partnership ("<u>ERPOP</u>"), Equity Residential, a Maryland real estate investment trust ("<u>EQR</u>", and together with "<u>ERPOP</u>", the "<u>Equity Residential Parties</u>"), AvalonBay Communities, Inc., a Maryland corporation ("<u>AVB</u>"), Lehman Brothers Holdings Inc., a Delaware corporation ("<u>LBHI</u>"), and Archstone Enterprise LP, a Delaware limited partnership ("<u>Seller</u>"). ERPOP, EQR, AVB, LBHI and Seller are sometimes referred to herein as the "<u>Parties</u>" and each, a "<u>Party</u>."

WHEREAS, LBHI and the entities set forth on Schedule II (collectively, the "Lehman Entities"), (a) directly or indirectly, own (i) 100% of the preferred Equity Interests in Seller, and (ii) approximately 95% of the common Equity Interests in Seller, and (b) have the sole right to control the disposition of all of the assets and liabilities of Seller;

WHEREAS, Seller is the direct record and beneficial owner of all of the Equity Interests of the Primary Archstone Entities;

WHEREAS, each of ERPOP and AVB (each, a "Buyer Party") or one or more Buyer Designees desires to purchase (a) from Seller, the Transferred Assets, and (b) from Seller or the applicable Archstone Entity, the Transferred Subsidiary Assets, as further described and defined herein;

WHEREAS, ERPOP and one or more of its Buyer Designees intend that certain of the Transferred Assets and Transferred Subsidiary Assets will be acquired by them as "replacement property" in qualifying exchanges pursuant to Section 1031 of the Code;

WHEREAS, (a) Seller wishes to sell, assign, transfer, convey and deliver to the applicable Buyer Party or Buyer Designee (or cause the applicable Archstone Entity to sell, assign, transfer, convey and deliver to the applicable Buyer Party or Buyer Designee), and LBHI desires to cause Seller or the applicable Archstone Entity to sell, assign, transfer, convey and deliver to the applicable Buyer Party or Buyer Designee, (i) all of Seller's right and title to and interest in the Transferred Assets, free and clear of all Liens (other than any restrictions on transfer by any applicable Buyer Party or Buyer Designee imposed under applicable securities Laws and Permitted Liens), and (ii) all of Seller's or the applicable Archstone Entity's right and title to and interest in the Transferred Subsidiary Assets, free and clear of all Liens (other than any restrictions on transfer by any applicable Buyer Party or Buyer Designee imposed under applicable securities Laws and Permitted Liens), and (b) one or more of the Buyer Parties or Buyer Designees desire to assume all of the Liabilities of Seller (other than the Excluded Liabilities), upon the terms and subject to the conditions set forth in this Agreement;

WHEREAS, the Contemplated Transactions are made under and pursuant to the provisions of the Modified Third Amended Joint Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors dated December 6, 2011 that contemplate the wind-down, sale and liquidation of assets of LBHI and its Affiliates; and

WHEREAS, the Parties have taken all actions, including the receipt of all necessary board, partner, shareholder and other similar approvals, to approve this Agreement and the Contemplated Transactions.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound hereby, agree as follows:

ARTICLE 1 DEFINITIONS

- 1.1 <u>Definitions.</u> When used in this Agreement, the following terms in all of their tenses, cases and correlative forms shall have the meanings assigned to them in this <u>Section 1.1</u> or elsewhere in this Agreement. Accounting terms not otherwise defined in this Agreement shall be interpreted in accordance with GAAP.
- "Affiliate" of any particular Person means any other Person which, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with such specified Person. For purposes of this definition, "control" of any Person means possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract, or otherwise.
- "Allocated Deferred AVB Share Value" means, with respect to the applicable Retained Deferred Closing Asset, 0.20, *multiplied* by (a) the number of AVB Common Shares set forth on Schedule 2.5.1 (or as reduced in accordance with the definition of "Retained Escrowed Stock Consideration") allocated to such Retained Deferred Closing Asset, *times* (b) \$136.00.
- "Allocated Deferred EQR Share Value" means, with respect to the applicable Retained Deferred Closing Asset, 0.20, multiplied by (a) the number of EQR Common Shares set forth in Schedule 2.5.1 (or as reduced in accordance with the definition of "Retained Escrowed Stock Consideration") allocated to such Retained Deferred Closing Asset, times (b) \$58.75.
- "Allocated Share Value" means the sum of (A) the Allocated Deferred EQR Share Value for each Retained Deferred Closing Asset, and (B) the Allocated Deferred AVB Share Value for each Retained Deferred Closing Asset.

"Antitrust Division" means the Antitrust Division of the Department of Justice of the United States.

"Antitrust Laws" means any of the HSR Act, the Sherman Act, as amended, the Clayton Act, as amended, the Federal Trade Commission Act, as amended, and any other United States federal or state or foreign statutes, rules, regulations, orders, decrees, administrative or judicial doctrines or other Laws that are designed to prohibit, restrict or regulate mergers and acquisitions and/or actions having the purpose or effect of lessening competition, monopolization or restraining trade.

2

"Archstone" means Archstone, a Maryland real estate investment trust.

"Archstone Entities" means, collectively, each of the Subsidiaries of Seller, including the Primary Archstone Entities and Archstone.

"Asset Document Escrow Release Instructions" means the joint instructions provided pursuant to Section 2.5.2(d).

"Assumed Liabilities" means all Liabilities of Seller (including any Liabilities resulting from the Contemplated Transactions), other than any Excluded Liabilities.

"Assumption Agreements" means one or more bills of sale, assignment and assumption agreements, in substantially the form of Exhibit A hereto, pursuant to which (i) Seller transfers any tangible and intangible personal property constituting part of the Transferred Assets (other than any personal property relating to any Deferred Closing Assets), and (ii) a Buyer Designee assumes, and agrees to pay and discharge when due and to indemnify Seller against, the Assumed Liabilities set forth in such agreement.

"AVB Equity Consideration" means 14,889,706 shares of common stock of AVB, par value \$0.01 per share.

"AVB Equity Offering" means any public or private offering of AVB Common Shares or other securities of AVB by AVB, on or following the date hereof and prior to the Initial Closing and with respect to which all or a portion of the proceeds from such offering will be used by AVB to fund the AVB Cash Purchase Price Portion.

"AVB Fundamental Representations" means the representations and warranties contained in Section 8.1, 8.2, 8.3, 8.4, 8.6, 8.8, 8.9, 8.10 and 8.11.

"Bankruptcy Code" means Chapter 11 of Title 11 of the United States Code.

"Bankruptcy Event" means, with respect to any Person, the occurrence of any of the following:

- (a) a court of competent jurisdiction enters a decree or order for relief in an involuntary case under applicable bankruptcy, insolvency, or other similar Law now or hereinafter in effect, or appoints a receiver, liquidator, assignee, custodian, trustee, sequestrator, or similar official for such Person or for any substantial part of such Person's property, or ordering the winding up or liquidation of its affairs;
- (b) such Person commences a voluntary case under any applicable bankruptcy, insolvency, or other similar Law now or hereinafter in effect, or consents to the entry of an order for relief in an involuntary case under any such Law, or consents to the appointment of or taking possession by a receiver, liquidator, assignee, trustee, custodian, sequestrator, or other similar official for such Person or for any substantial part of such Person's property, or makes any general assignment for the benefit of creditors, or

3

- fails generally to, or admits in writing its inability, to pay debts as they become due; or

 (c) an insolvency case under any applicable bankruptcy, insolvency, or other similar Law now or hereinafter in effect, is commenced against such

 Person or for the appointment of or taking possession by a receiver, liquidator, assignee, trustee, custodian, sequestrator, or other similar official for
 such Person or for any substantial part of such Person's property and a stay is not instituted against such case or such case has not been dismissed
 within sixty (60) days.
- "Business" means the business of the Archstone Entities as of the date of this Agreement.

"Business Day" means each day, other than a Saturday, Sunday or other day on which banks in New York, New York are required by Law to close.

"Buyer Designees" means (a) one or more Subsidiaries of ERPOP (other than EQR, which under no circumstance shall constitute a Buyer Designee) or AVB or both ERPOP and AVB, or (b) an intermediary, exchange accommodation titleholder or other third party intended to facilitate the consummation of the acquisition of one or more of the Transferred Assets or the Transferred Subsidiary Assets as part of an exchange transaction pursuant to Section 1031 of the Code, in each case, that have been designated by ERPOP or AVB in writing as the Persons to purchase and take ownership of the applicable Transferred Assets or Transferred Subsidiary Assets.

"Buyer Closing Mechanics" means the following actions and/or events: (I) the Buyer Parties have executed the Initial Closing Date Escrow Agreement and deposited with the Escrow Agent in accordance with the Initial Closing Date Escrow Agreement (x) all executed counterparts of the Transaction Documents and all other deliverables that the Buyer Parties or any Buyer Designee is required to deliver pursuant to Section 11.1.5 and Section 11.1.6 and (y) the Cash Purchase Price, ERPOP Equity Consideration (other than the Escrowed Deferred Stock Consideration), and otherwise complied with all their obligations under the Initial Closing Date Escrow Agreement, and (II) the Buyer Parties have executed the Deferred Closing Asset Escrow Agreement and deposited with the Escrow Agent in accordance with the Deferred Closing Asset Escrow Agreement (x) all the Escrowed Deferred Closing Asset Transfer Documents that the Buyer Parties or any Buyer Designee is a party to and (y) the Escrowed Deferred Stock Consideration, and otherwise complied with all their obligations under the Deferred Closing Asset Escrow Agreement.

"Buyer Party Fundamental Representations" means, collectively, the AVB Fundamental Representations and the ERPOP Fundamental Representations.

"Buyer Related Parties" means, collectively, the Buyer Parties, any of their respective former, current and future direct or indirect equity holders, controlling persons, shareholders, directors, officers, employees, agents, Affiliates, members, financing sources (including the AVB Lenders and the ERPOP Lenders), managers, general or limited partners or assignees.

"Code" means the Internal Revenue Code of 1986, as amended.

"Commercially Reasonable Efforts" means the commercially reasonable efforts that a reasonably prudent Person desirous of achieving the contemplated result would use in similar circumstances to ensure that such result is achieved as expeditiously as possible; provided that a Party's commercially reasonable efforts shall not be deemed to include causing any action to be taken that is beyond such Party's authority under the Organizational Documents of any applicable entity; and provided further that no obligation to use commercially reasonable efforts shall be deemed to require any Party hereto or its Affiliates to take any action that would violate any of the Organizational Documents of any of the Archstone Entities.

"Consent" means any written, unconditional notice, approval, consent, license, permission or waiver of any Person, other than a Governmental Authority.

"Consideration and Assumption Escrow Release Instructions" means the joint instructions provided pursuant to Section 2.5.2(c).

"Contemplated Transactions" means the sale of the Transferred Assets and the Transferred Subsidiary Assets, the assumption of the Assumed Liabilities and the performance of the related covenants contemplated by this Agreement or any of the other Transaction Documents.

"Contract" means any agreement, contract, instrument, lease or sublease (including any lease and sublease of real property), license, franchise, trust, note, bond, deed, mortgage, indenture, sale or purchase order, delivery order, change order, understanding, arrangement, instrument, or commitment, obligation, promise, or undertaking of any kind (whether written or oral) that is legally binding.

"<u>Data Room</u>" means (i) the electronic data room established by Seller, the Lehman Entities, and/or the Archstone Entities or their respective Representatives for the purpose of making information, agreements, documents and other due diligence materials regarding LBHI, the Archstone Entities or Seller available to EQR, ERPOP, AVB or other Persons in connection with the Contemplated Transactions, and (ii) sensitive materials (e.g., employment information) that were delivered directly to Buyer Parties or their respective Representatives, in each case, as of November 18, 2012.

"Deferred Closing Asset Arrangement Agreement" means the agreements and obligations set forth on Schedule 2.5.1 hereto.

"Deferred Closing Asset Consent" means any Consent required to be obtained in connection with the consummation of the Contemplated Transactions with respect to any Deferred Closing Asset.

"Deferred Closing Asset Escrow Agreement" means the Deferred Closing Asset Escrow Agreement, substantially in the form attached as Exhibit D hereto.

"Derivative Instruments" means any and all (i) interest rate swaps, interest rate caps, basis swaps, credit derivative transactions, forward rate transactions, commodity swaps, commodity options, forward commodity contracts, equity or equity index swaps or options, bond or bond price or bond index swaps or options or forward bond or forward bond price or forward

5

bond index transactions, interest rate options, forward foreign exchange transactions, cap transactions, floor transactions, collar transactions, currency swaps, cross-currency rate swaps, currency options, spot contracts or any other similar transactions or any combination of any of the foregoing (including any options to enter into any of the foregoing), whether or not any such transaction is governed by or subject to any master agreement, and (ii) transactions of any kind, and the related confirmations, which are subject to the terms and conditions of, or governed by, any form of master agreement published by the International Swaps and Derivatives Association, Inc., the International Foreign Exchange Master Agreement, or any other master agreement, including any such obligations or liabilities under any such agreement.

"Disclosure Schedules" or "Schedules" means the schedules accompanying this Agreement. Any matter set forth in the Disclosure Schedules with respect to any particular section of the Agreement shall refer only to the section of the Disclosure Schedules corresponding to the applicable Agreement section number, except that any item disclosed in the Disclosure Schedules shall be deemed disclosed under any other applicable section of the Disclosure Schedules if it is reasonably apparent on its face that such disclosure is likewise applicable to such other section of the Disclosure Schedules. The inclusion of any information in the Disclosure Schedules (or any information provided subsequent thereto) shall not be deemed to be an admission or acknowledgment, in and of itself, that such information is required by the terms hereof to be disclosed, is material, has resulted in or would result in a Material Adverse Effect or is outside the Ordinary Course of Business.

"<u>Disqualified Equity Interest</u>" means, with respect to any Person, any Equity Interest of such Person that by its terms (or by the terms of any security into which it is convertible or for which it is exchangeable or exercisable) or upon the happening of any event (a) matures or is mandatorily redeemable for cash or other property (other than other Equity Interests) pursuant to a sinking fund obligation or otherwise, or (b) is redeemable for cash or other property (other than other Equity Interests) at the option of the holder thereof, in whole or in part.

"Effect" means any change, effect, event, occurrence, development, circumstance, condition or alteration in status thereof.

"Employee Benefit Plan" means each material "employee benefit plan" (as such term is defined in Section 3(3) of ERISA) and each other material employee benefit plan, program or policy that is maintained, sponsored or contributed to by the Archstone Entities on behalf of employees located within the United States and elsewhere.

"EQR Equity Offering" means any public or private offering of EQR Common Shares or other securities of EQR by ERPOP or EQR, on or following the date hereof and prior to the Initial Closing, in each case, with respect to which all or a portion of the proceeds from such offering will be used by ERPOP to fund the ERPOP Cash Purchase Price Portion.

"Equity Interest" means (a) the equity ownership rights in a business entity, whether a corporation, company, joint stock company, limited liability company, general or limited partnership, joint venture, bank, association, trust, trust company, land trust, business trust, sole proprietorship or other business entity or organization, and whether in the form of capital stock, ownership unit, limited liability company interest, membership interest, limited or general

6

- "ERISA" means the Employee Retirement Income Security Act of 1974, as amended.
- "ERPOP Equity Consideration" means 34,468,085 common shares of beneficial interest of EQR, par value \$0.01 per share.
- "ERPOP Fundamental Representations" means the representations of ERPOP contained in Section 7.1, 7.2, 7.3, 7.4, 7.6, 7.8, 7.9, 7.10 and 7.11.
- "Escrowed AVB Equity Consideration" means the portion of the Escrowed Deferred Stock Consideration that consists of AVB Equity Consideration.
- "Escrowed ERPOP Equity Consideration" means the portion of the Escrowed Deferred Stock Consideration that consists of ERPOP Equity Consideration.
- "Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor Law, and the rules and regulations issued pursuant to that statute or any successor Law.
- "Executive Employment Arrangement" means all stock purchase, stock option, severance, employment, change-in-control, bonus, incentive, deferred compensation, employee loan and all other employee benefit plans, agreements, programs or policies providing compensation or benefits to any senior management employee of any of Seller, Archstone Communities LLC or the Archstone Entities, other than an Employee Benefit Plan.
- "Excluded Liabilities" means (i) any Liabilities of Seller with respect to which Seller, LBHI or any other Lehman Entity has expressly accepted or retained Liability under this Agreement, including pursuant to Sections 13.5.2, 13.8, 13.10.2 and 14.2.2, and (ii) any Liabilities owing or payable to any Lehman Entity or other direct or indirect owner of any Equity Interest in Seller, and (iii) all promissory notes and related arrangements with respect to amounts owing or payable by Seller to any trustee, director, officer or employee of the Archstone Entities, including the arrangements set forth on Schedule 1.1(b). For the avoidance of doubt, any reference in the definition of Excluded Liabilities to an owner of an indirect interest in Seller or any Lehman Entity shall specifically exclude any owner of a direct or indirect interest in LBHI.
 - "Filing" means any filing or registration with, or a written notice to, a Governmental Authority.
- "<u>First Extension Date</u>" means the date that is sixty (60) days after the date of this Agreement; <u>provided</u>, <u>however</u>, that in the event that (i) on or prior to the date that is five (5) Business Days prior to the 60th day after the date of this Agreement, the Buyer Parties have delivered a written notice in accordance with <u>Section 2.4(a)</u> that the Buyer Parties intend that the

7

Initial Closing shall occur on or prior to the 60th day after the date of this Agreement, and (ii) on or prior to the date for the Initial Closing set forth in the notice, the conditions set forth in Sections 10.1.1, 10.1.2 and 10.1.3 (except to the extent waived by the Buyer Parties, in their sole and absolute discretion) shall not have been satisfied, then the "First Extension Date" shall mean the date that is five (5) Business Days following the date on which the conditions set forth in Sections 10.1.1, 10.1.2 and 10.1.3 have been satisfied (except to the extent waived by the Buyer Parties, in their sole and absolute discretion).

- "FTC" means the Federal Trade Commission of the United States or any successor agency.
- "GAAP" means generally accepted accounting principles, as in effect in the United States from time to time.
- "Governmental Authority" means any government or political subdivision, whether federal, state, local or foreign, or any agency or instrumentality of any such government or political subdivision, or any federal, state, local or foreign court, authority, tribunal, department, bureau or commission, in each case having jurisdiction over the applicable matter.
- "GSEs" means, Fannie Mae (also known as the Federal National Mortgage Association) and Freddie Mac (also known as the Federal Home Loan Mortgage Corporation).
- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, or any successor Law, and regulations and rules issued pursuant to that statute or any successor Law.
- "Indebtedness" means, with respect to any Person on any date of determination (without duplication): (a) the principal of and premium (if any) in respect of indebtedness of such Person for borrowed money, (b) the principal of and premium (if any) in respect of obligations of such Person evidenced by bonds, debentures, notes or other similar instruments, (c) all obligations of such Person in respect of letters of credit or other similar instruments (including reimbursement obligations with respect thereto) (other than obligations with respect to letters of credit securing obligations (other than obligations described in clauses (a) and (b) and (e)) entered into in the Ordinary Course of Business of such Person to the extent that such letters of credit are not drawn upon or, if and to the extent drawn upon, such drawing is reimbursed no later than the third business day following receipt by such Person of a demand for reimbursement following payment on the letter of credit), (d) all obligations of such Person to pay the deferred and unpaid purchase price of property or services (except trade payables and accrued Liabilities incurred in the Ordinary Course of Business), (e) all capitalized lease obligations of such Person, provided that the determination as to what constitutes a capitalized lease obligation shall be determined in accordance with GAAP as in effect on the date of this Agreement, (f) the redemption, repayment or other repurchase amount of such Person with respect to any Disqualified Equity Interests, (g) all Indebtedness of other Persons secured by a Lien on any asset of such Person to the extent of the fair market value of the asset pledged, whether or not such Indebtedness is assumed by such Person, and (h) all Indebtedness of other Persons to the extent guaranteed by such Person.

8

- "Indemnified Archstone Entity Liabilities" means all Liabilities of Seller or any Archstone Entity (including any Liabilities resulting from the Contemplated Transactions), other than any Excluded Liabilities.
- "Initial Closing Date" means the date on which the Initial Closing actually occurs with respect to the Contemplated Transactions (other than the purchase by the Buyer Parties of any of the Deferred Closing Assets as to which there will be an Extension Closing in accordance with Section 2.5).
- "Initial Closing Date Escrow Agreement" means the Initial Closing Date Escrow Agreement, substantially in form and substance as mutually reasonably agreed upon by the Parties.
- "Intellectual Property." means all right, title and interest in or relating to intellectual property, whether protected, created or arising under the Laws of the United States or any other jurisdiction, including: (i) all patents and applications therefor, including all continuations, divisionals, and continuations-in-part thereof and patents issuing

thereon, along with all reissues, reexaminations and extensions thereof; (ii) all trademarks, service marks, trade names, service names, brand names, trade dress rights, logos, corporate names, trade styles, logos and other source or business identifiers and general intangibles of a like nature, together with the goodwill associated with any of the foregoing, along with all applications, registrations, renewals and extensions thereof; (iii) all Internet domain names; (iv) all copyrights and all mask works, databases and design rights, whether or not registered or published, all registrations and recordations thereof and all applications in connection therewith, along with all reversions, extensions and renewals thereof; (v) all trade secrets; and (vi) all Contracts granting any right relating to or under the foregoing.

"IRS" means the United States Internal Revenue Service or any successor agency.

"Joint Venture" means each Person in which any of the Primary Archstone Entities or any Subsidiaries of any of the Primary Archstone Entities holds an Equity Interest and/or a voting interest (other than the Primary Archstone Entities and the Subsidiaries of any of the Primary Archstone Entities).

"Knowledge" means, with respect to each of LBHI and Seller, the actual knowledge of each of the Persons set forth on Schedule 1.1(a), without any obligation of inquiry. The fact that a document or other information is set forth in the Data Room does not imply that any individual has knowledge of such document or information.

"Law" means, with respect to any Person, any federal, state, county, municipal, local, multinational or foreign statute, treaty, law, common law, ordinance, rule, regulation, code, order, writ, stipulation, injunction, judicial decision, decree, ruling, determination, finding, permit, constitutional provision or other legally binding requirement of any Governmental Authority applicable to such Person or any of its respective properties, assets, officers, directors, employees, consultants or agents (in connection with such officer's, director's, employee's, consultant's or agent's activities on behalf of such Person).

9

"LBO Transaction" means the merger of Archstone-Smith Operating Trust with River Trust Acquisition (MD), LLC and certain other transactions contemplated by the Agreement and Plan of Merger, dated as of May 28, 2007, as amended by Amendment No. 1 thereto, dated as of August 5, 2007, by and among the Archstone-Smith Trust, Archstone-Smith Operating Trust, River Holding, LP, River Acquisition (MD), LP and River Trust Acquisition (MD), LLC.

"LCPI" means Lehman Commercial Paper, Inc.

"Lehman Designee" means one or more Subsidiaries or Affiliates of a Lehman Entity that have been designated by LBHI in writing as designees of LBHI for purposes of this Agreement.

"Lehman January IPA" means the Interest Purchase Agreement, dated as of January 20, 2012, by and among LBHI, BIH ASN LLC, Archstone Equity Holdings Inc., Bank of America, N.A., and Banc of America Strategic Ventures, Inc.

"Lehman May IPA" means the Interest Purchase Agreement, dated as of May 24, 2012, by and among LBHI, ACQ SPV II Paper, ACQ SPV II Holdings, BIH ASN LLC, Archstone Equity Holdings Inc., Bank of America, N.A., and Banc of America Strategic Ventures, Inc.

"Lehman IPAs" means, collectively, the Lehman January IPA and the Lehman May IPA.

"Liability" means any liability or obligation of whatever kind or nature, whether known or unknown, asserted or unasserted, absolute or contingent, accrued or unaccrued, liquidated or unliquidated, or due or to become due, including any Indebtedness.

"Lien" means any mortgage, lien, claim, pledge, charge, security interest or other encumbrance of any kind, title defects, easements and restrictions, invalidities or irregularities, deed of trust, hypothecation, collateral assignment, preference, priority or other security arrangement, claim, conditional sale, title retention arrangement, lessor's rights under a capital lease, restriction on transfer or voting, right of first offer, right of first refusal or any other right or arrangement having substantially the same legal or economic effect as any of the foregoing.

"Losses" means any and all charges, judgments, losses, liabilities, damages (including consequential damages), costs, Taxes, penalties, expenses, fees, fines, assessments, sanctions or awards (including (a) amounts paid in settlement, and (b) reasonable costs of investigation and defense, reasonable legal expenses and court costs), but specifically excluding any costs incurred by or allocated to an Indemnified Party with respect to time spent by employees of the Indemnified Party or its Affiliates (except, in each case, to the extent that such damages are payable in connection with a Third Party Claim, in which case such damages shall be considered Losses hereunder for indemnification purposes).

"Management Regulations" means each of (i) the Management Regulations of The Archstone German Fund — fonds commun de placement — fonds d'investissement spécialisé dated August 13, 2009; and (ii) the Management Regulations of The Archstone German Residential Fund II — fonds commun de placement — fonds d'investissement spécialisé dated October 24, 2012.

10

"Material Adverse Effect" means: (a) an Effect that individually or when taken together with all other Effects that exist at the date of determination, has a material adverse effect on the business, financial condition or results of operations of the Archstone Entities, taken as a whole; provided, however, that, solely for purposes of clause (a) above, no Effects resulting from, relating to or arising out of the following shall be deemed to be, constitute or give rise to a Material Adverse Effect, or shall be taken into account when determining whether a Material Adverse Effect has occurred: (i) changes in general economic, financial market, credit, capital, business or political conditions in the United States or any other country or changes generally affecting companies in the industry or industries in which the Business or the Archstone Entities operate, if and only to the extent such Effects do not adversely affect the Business or the Archstone Entities to a greater degree than other companies of comparable size operating in such industry or industries in which the Archstone Entities operate, (ii) changes in interest, currency or exchange rates or the price of any commodity, security or market index, (iii) changes in accounting requirements or principles required by GAAP or any other accounting principles or required by any change in applicable Laws and any restatement of the financial statements of the Archstone Entities as a result thereof or public announcement related thereto, (iv) Effects resulting from any change in regulatory conditions or change in applicable Laws affecting the industry in which the Business or the Archstone Entities operate or changes in the interpretation of such regulatory conditions or applicable Laws by Governmental Authorities, (v) Effects arising out of the commencement, continuation or escalation of a war, sabotage, hostilities or other international or national calamity or act of terrorism, if and only to the extent such Effects do not adversely affect the Business or the Archstone Entities to a greater degree than other companies of comparable size operating in such industry or industries in which the Archstone Entities operate, (vi) Effects arising out of, or related to compliance with the terms of, or the taking of any action required by, this Agreement, the taking of any action requested in a signed writing executed by both of the Buyer Parties, the failure to take any action prohibited by this Agreement, or the failure to take any action requested in an executed writing not to be taken by both of the Buyer Parties, (vii) any Effects resulting from execution of this Agreement, the identity of the Buyer Parties, the announcement or pendency of the Contemplated Transactions, including any Effects attributable or relating to expenses incurred in connection with the Contemplated Transactions, (viii) any Effects resulting from any failure to obtain any Consent necessary to

consummate the Contemplated Transactions, (ix) any failure to meet any internal or public revenue or earnings projections, forecasts, estimates or guidance for any period, whether relating to financial performance or business metrics, including revenues, net operating incomes, cash flows or cash positions, (x) any flood, earthquake or other natural disaster if and only to the extent such Effects do not adversely affect the Business or Archstone Entities to a greater degree than other companies of comparable size operating in such industry or industries in which the Archstone Entities operate, (xi) Effects resulting from the death, disability, resignation or termination of employment (in the case of resignation or termination of employment, for any reason) of any officer or other employee of the Archstone Entities, (xii) any actions or inactions of any GSEs, or (xiii) any matters set forth in the Schedules; and (b) any Effects that individually or when taken together with all other Effects that exist at the date of determination, are, or would reasonably be expected to, have a material adverse effect on the ability of Seller, LBHI or the Archstone Entities to consummate the Contemplated Transactions.

11

"Montgomery County Deferred Closing Asset" means any Deferred Closing Asset located in Montgomery County, Maryland for which the purchase has been postponed in accordance with Section 2.5.

"Mutual Release" means the Mutual Release, substantially in the form attached as Exhibit F hereto.

"Order" means any award, decree, decision, injunction, judgment, order, ruling, writ, assessment, subpoena, or verdict or arbitration award entered, issued, made, or rendered by any court, administrative agency or other Governmental Authority, whether arising from a Proceeding or applicable Law.

"Ordinary Course of Business" means an action taken by a Person only if that action is consistent in nature and scope with the past practices of such Person and is taken in the ordinary course of the normal operations of such Person.

"Organizational Documents" means the Articles or Certificate of Incorporation, Code of Regulations and/or Bylaws of a corporate entity, the Articles of Organization or Certificate of Formation and Operating Agreement or Limited Liability Company Agreement of a limited liability company entity, the declaration of trust or trust agreement for a trust entity and other similar organizational and governing documents, including side letters, for entities other than corporations, limited liability companies or trusts.

"Outside Deferred Closing Date" means the date that is 180 days after the date of this Agreement; provided that (i) LBHI and Seller shall have the right, in their sole discretion, to extend the Outside Deferred Closing Date by up to twenty (20) Business Days in order to cure any default by LBHI or Seller, as provided in Section 15.1.3(a), and (ii) subject to the requirements set forth in Section 2.5.2(g), in the event that an Order issued by any Governmental Authority has been entered that prevents or prohibits the consummation of the transfer of one or more Deferred Closing Assets to the applicable Buyer Party prior to the date that is 180 days after the date of this Agreement, any Party shall have the right, in its sole discretion, to extend the Outside Deferred Closing Date with respect to, and only with respect to, such Deferred Closing Assets in accordance with Section 2.5.2(g) until the date that is three (3) Business Days after the date the transfer of such Deferred Closing Assets may first be lawfully consummated. Any extension of the Outside Deferred Closing Date pursuant to clause (ii) hereof shall only be with respect to any Retained Deferred Closing Asset(s) and shall not apply to any other Deferred Closing Assets.

"Permitted Liens" means (a) statutory Liens for Taxes and other governmental charges that are not yet delinquent or which are being contested in good faith by appropriate Proceedings or for which a reserve has been established; (b) mechanic's, workmen's, repairmen's, materialmen's, warehousemen's, carrier's and other similar statutory Liens arising or incurred in the Ordinary Course of Business with respect to amounts not yet delinquent or which are being contested in good faith or for which a reserve has been established; (c) with respect to real property, any Liens that do not in the aggregate materially impair the use or planned use of such real property; (d) Liens in respect of customer advances in favor of a customer of the Business arising in the Ordinary Course of Business; (e) rights and licenses granted to others in

12

Intellectual Property entered into in the Ordinary Course of Business prior to the execution of this Agreement; (f) with respect to any of the any leased real property, any Lien in respect of any developer's, landlord's or other third party's interest in the such leased real property; (g) Liens made in connection with the financing of a joint venture; (h) Liens made in connection with the construction of new developments to the extent permitted or not prohibited by Section 12.1; (i) Liens existing with respect to Seller or any of the Archstone Entities as of the date hereof or arising after the date hereof as permitted or not prohibited pursuant to Article 12; (j) Liens imposed pursuant to the terms of any Indebtedness permitted to be incurred hereunder after the date hereof related to any Transferred Assets or Transferred Subsidiary Assets; (k) Liens or restrictions on transfer imposed under any applicable securities Laws or the Organizational Documents of the Archstone Entities, Joint Ventures or Seller; and (l) Liens resulting from the breach of, default under, or the failure to obtain any Consent required by any Contract resulting from or in connection with the Contemplated Transactions; provided, however, that in no event shall any Liens with respect to any Liability owing or payable to any Lehman Entity or other direct or indirect equity owner of Seller constitute a "Permitted Lien".

"Person" means any individual, a corporation (including any non-profit corporation), a limited liability company, a general partnership, a limited partnership, a limited liability partnership, a trust, a venture, a business, a union, a society, an association, a firm, a Governmental Authority or any other entity or organization.

"Preferred Units" means, collectively (i) those certain Series O, P, Q-1, Q-2 and R Preferred Units issued by Archstone, and (ii) the Series I Preferred Units issued by Archstone Multifamily Series I Trust LLC.

"Primary Archstone Entities" means each of Archstone Multifamily Parallel Guarantor LLC, Archstone Multifamily Parallel Guarantor II LLC, Archstone Multifamily Guarantor LP, Archstone Multifamily Guarantor (GP) LLC, Archstone Multifamily CM LLC and Archstone Inc.

"Proceeding" means any action, claim, audit or other inquiry, hearing, investigation, suit or other charge or proceeding (whether civil, criminal, administrative, investigative, formal or informal) by or before any Governmental Authority or before an arbitrator or arbitral body or mediator.

"Purchase Price" means, collectively, the Cash Purchase Price, the ERPOP Equity Consideration and the AVB Equity Consideration.

"Quantified Losses" means an amount of Losses that relate to Liabilities that are reasonably capable of being determined without undue time or expense and satisfied through the payment of monetary damages.

 $\label{eq:controller} \begin{tabular}{ll} $`\underline{Registration} \ Rights \ Agreement, substantially in the form attached as \ \underline{Exhibit} \ \underline{B-2} \ hereto. \end{tabular}$

"Registration Rights Agreement - ERPOP" means the Registration Rights Agreement, substantially in the form attached as Exhibit B-1 hereto.

"REIT" means a real estate investment trust under Sections 856 through 860 of the Code.

"REIT Representation Letters" means those certain letter agreements executed and delivered by LBHI on the date hereof with respect to certain representations by LBHI made to enable the board of directors of AVB, or board of trustees of EQR, as applicable, to grant to LBHI or the applicable Lehman Designees an exemption from the capital stock ownership limitations contained in EQR's or AVB's Organizational Documents, as applicable, to the extent permissible under such Organizational Documents.

"Representative" means, with respect to a particular Person, any director, officer, manager, general partner, employee, agent, consultant, advisor, trustee, fiduciary, Affiliate or other representative of the Person, including legal counsel, accountants and financial advisors.

"Retained Escrowed Stock Consideration" means the number of respective EQR Common Shares or number of AVB Common Shares (as applicable) equal to the Allocated Share Value of EQR Common Shares or AVB Common Shares set forth on Schedule 2.5.1 attributable to the applicable Retained Deferred Closing Asset(s); provided however, that if the Allocated Share Value attributable to all Retained Deferred Closing Assets is in excess of \$50,000,000, then the number of EQR Common Shares and number of AVB Common Shares shall be reduced in proportion to the relative value represented by the EQR Common Shares or the AVB Common Shares of such Allocated Share Value attributable to all Retained Deferred Closing Assets, until the Allocated Share Value does not exceed \$50,000,000. Any fractional shares as a result of determining the Retained Escrowed Stock Consideration shall be rounded to the nearest whole share. For these purposes, the EQR Common Shares shall be deemed to have a value of \$58.75 per share and the AVB Common Shares shall be deemed to have a value of \$136.00 per share.

"Sarbanes-Oxley Act" means the Sarbanes-Oxley Act of 2002, as amended.

"SEC" means the United States Securities and Exchange Commission.

"Second Extension Date" means the date that is 90 days after the date of this Agreement; provided, however, that in the event that (i) on or prior to the date that is five (5) Business Days prior to the 90th day after the date of this Agreement, the Buyer Parties have delivered a written notice in accordance with Section 2.4(a) that the Buyer Parties intend that the Initial Closing shall occur on or prior to the 90th day after the date of this Agreement, and (ii) on or prior to the date for the Initial Closing set forth in the notice, the conditions set forth in Sections 10.1.1, 10.1.2 and 10.1.3 (except to the extent waived by the Buyer Parties, in their sole and absolute discretion) shall not have been satisfied, then the "Second Extension Date" shall mean the date that is five (5) Business Days following the date on which the conditions set forth in Sections 10.1.1, 10.1.2 and 10.1.3 have been satisfied (except to the extent waived by the Buyer Parties, in their sole and absolute discretion).

"Seller Closing Mechanics" means the following actions and/or events, provided that the Buyer Parties have performed the Buyer Closing Mechanics and provided that the Buyer Parties shall have timely delivered the Transaction Documents and other deliverables to Seller for review as provided in this Agreement: (i) Seller and LBHI have executed the Initial Closing Date

14

Escrow Agreement and deposited with the Escrow Agent in accordance with the Initial Closing Date Escrow Agreement all executed counterparts of the Transaction Documents and all other deliverables that LBHI, Seller or any Archstone Entity is required to deliver pursuant to Section 10.1.5 and Section 10.1.6, and otherwise complied with all their obligations under the Initial Closing Date Escrow Agreement, and (ii) Seller, LBHI and any applicable Archstone Entities shall have executed the Deferred Closing Asset Escrow Agreement and deposited with the Escrow Agent in accordance with the Deferred Closing Asset Escrow Agreement all the Escrowed Deferred Closing Asset Transfer Documents that Seller, LBHI or any Archstone Entity is a party to, and otherwise complied with all their obligations under the Deferred Closing Asset Escrow Agreement.

"Seller Fundamental Representations" means the representations and warranties set forth in Sections 4.1, 4.2, 4.3, 4.4, 4.5, 4.6, 5.1, 5.2, 5.3, 5.4, 5.5, 5.7, 6.3.1 and 6.6.1.

"Seller Liquidated Damages Amount" means an amount in cash equal to \$650,000,000; provided, however, that, if the date of the Initial Closing is extended to a date following the First Extension Date, then "Seller Liquidated Damages Amount" shall mean an amount in cash equal to \$800,000,000.

"Securities Act" means the Securities Act of 1933, as amended, or any successor Law, and the rules and regulations issued pursuant to that statute or any successor Law.

"Series O Preferred Units" mean those certain Series O Preferred Units issued by Archstone.

"Shareholders Agreement- AVB" means the Shareholders Agreement, substantially in the form attached as Exhibit C-2 hereto.

"Shareholders Agreement - ERPOP" means the Shareholders Agreement, substantially in the form attached as Exhibit C-1 hereto.

"Sponsor Material Adverse Effect" means: (a) an Effect that individually or when taken together with all other Effects that exist at the date of determination, has a material adverse effect on the business, financial condition or results of operations of EQR, ERPOP and AVB, individually and not collectively; provided, however, that, solely for purposes of clause (a) above, no Effects resulting from, relating to or arising out of the following shall be deemed to be, constitute or give rise to a Sponsor Material Adverse Effect, or shall be taken into account when determining whether a Sponsor Material Adverse Effect has occurred: (i) changes in general economic, financial market, credit, capital, business or political conditions in the United States or any other country or changes generally affecting companies in the industry or industries in which EQR, ERPOP or AVB operate, if and only to the extent such Effects do not adversely affect the business of each Sponsor to a greater degree than other companies of comparable size operating in such industry or industries in which EQR, ERPOP or AVB operate, (ii) changes in interest, currency or exchange rates or the price of any commodity, security or market index, (iii) changes in accounting requirements or principles required by GAAP or any other accounting principles or required by any change in applicable Laws and any restatement of the financial statements of EQR, ERPOP or AVB, as applicable, as a result thereof or public announcement related thereto,

15

(iv) Effects resulting from any change in regulatory conditions or change in applicable Laws affecting the industry in which EQR, ERPOP or AVB operate or changes in the interpretation of such regulatory conditions or applicable Laws by Governmental Authorities, (v) Effects arising out of the commencement, continuation or escalation of a war, sabotage, hostilities or other international or national calamity or act of terrorism, if and only to the extent such Effects do not adversely affect the business of EQR, ERPOP or AVB to a greater degree than other companies of comparable size operating in such industry or industries in which EQR, ERPOP or AVB operate, (vi) Effects arising out of, or related to compliance with the terms of, or the taking of any action required by, this Agreement, the taking of any action requested in writing by Seller or a Lehman Entity, the failure to take any action prohibited by this Agreement, or the failure to take any action requested in writing not to be taken by Seller or a Lehman Entity, (vii) any Effects

resulting from execution of this Agreement, the identity of the Buyer Parties, the Lehman Entities or Seller, the announcement or pendency of the Contemplated Transactions, including any Effects attributable or relating to expenses incurred in connection with the Contemplated Transactions, (viii) any Effects resulting from any failure to obtain any Consent necessary to consummate the Contemplated Transactions, (ix) any failure to meet any internal or public revenue or earnings projections, forecasts, estimates or guidance for any period, whether relating to financial performance or business metrics, including revenues, net operating incomes, cash flows or cash positions, (x) any flood, earthquake or other natural disaster if and only to the extent such Effects do not adversely affect EQR, ERPOP or AVB to a greater degree than other companies of comparable size operating in such industry or industries in which EQR, ERPOP or AVB operate, (xi) Effects resulting from the death, disability, resignation or termination of employment (in the case of resignation or terminations of employment, for any reason) of any officer or other employee of EQR, ERPOP or AVB, (xii) any actions or inactions of any GSEs, or (xiii) any matters set forth in Articles 7 or 8 of the Schedules; and (b) any Effects that individually or when taken together with all other Effects that exist at the date of determination, are, or would reasonably be expected to, have a material adverse effect on the ability of EQR, ERPOP and AVB, individually and not collectively, to consummate the Contemplated Transactions.

"Subsidiary" means, with respect to any Person, any other Person (other than any Joint Ventures) of which the specified Person, either (a) owns, directly or indirectly, 50% or more of the Equity Interests of the specified Person, or (b) directly or through or together with any other of its Subsidiaries, owns securities or other ownership interests or Equity Interests having voting power to elect a majority of the board of directors or other governing body or Persons performing similar functions on behalf of such other Person or is the general partner or managing member of such Person.

"Substantial Portion of the Contemplated Transactions" means transactions contemplated by this Agreement relating to or affecting real property assets (excluding real property assets that cannot be transferred as a result of any Tenant Purchase Option Law) with an aggregate gross asset value (including the Indebtedness allocated to or directly or indirectly encumbering such real property assets) as reasonably agreed upon by the Parties in excess of \$1 billion.

"<u>Tax</u>" or "<u>Taxes</u>" means any federal, state, local or foreign net income, gross income, net receipts, gross receipts, profit, severance, property, production, sales, use, license, excise, occupation, franchise, employment, unemployment, disability, payroll, severance, withholding,

16

alternative or add-on minimum, ad valorem, value-added, transfer, stamp, estimated, capital stock, registration, license, social security (or similar) or other tax, custom duty, governmental fee or other governmental charge, fee, levy, impost, tariff or assessment of any kind, together with any interest, fine, penalty, addition to tax or additional amount imposed with respect thereto, whether disputed or not, imposed by a Governmental Authority.

"<u>Tax Return</u>" means any return, report, filing, declaration, report, questionnaire, estimate, statement, information return or other document (including schedules, attachments or any related or supporting information) relating to any Tax, including any amendments thereto.

"Tenant Purchase Option Law" means the requirements of the Laws of Montgomery County, Maryland or Washington, D.C. that provide tenants, tenant organizations, or governmental entities an opportunity to purchase certain residential real property and require notices in connection with certain transfers of direct or indirect ownership of such property and require, among other things, notices of such transfers, including the District of Columbia Tenant Opportunity to Purchase Act, DC Code 42-3404 et seq. and Montgomery County, Maryland Code Chapter 53A.

"Transaction Documents" means this Agreement, the Shareholders Agreement — ERPOP, the Shareholders Agreement - AVB, the Registration Rights Agreement - ERPOP, the Registration Rights Agreement - AVB, the Initial Closing Date Escrow Agreement, the Deferred Closing Asset Escrow Agreement, the REIT Representation Letters, all Assumption Agreements, all Bills of Sale, the Deferred Closing Asset Arrangement Agreement, the Mutual Release, deeds, and other assignment and conveyance documents and any additional documents delivered pursuant to Sections 2.5, 2.6, 10.1.6, and 11.1.6.

"Transfer Taxes" means all sales, use, value added, documentary, stamp duty, gross receipts, registration, transfer gain, conveyance, excise, recording, license and other similar taxes and fees, including any interest, penalties, additions to Tax or additional amounts in respect of the foregoing arising out of or in connection with or attributable to (i) the transfer and sale of the Transferred Assets or the Transferred Subsidiary Assets pursuant to this Agreement, or (ii) the transfer and sale of any assets of Seller, the Archstone Entities or the Joint Ventures during the period beginning on October 1, 2012 and ending on the Initial Closing Date.

"Transferred Assets" means all assets (including the Transferred Interests, but not including the Transferred Subsidiary Assets), tangible or intangible, that are owned, leased, or licensed by Seller, in the possession of Seller or in which Seller otherwise has an interest, whether or not reflected in the books and records thereof, including all right and title to and interest in the Transferred Interests, other than any Transferred Subsidiary Assets.

"Transferred Interests" means all Equity Interests and other interests held by Seller in Archstone Multifamily Parallel Guarantor LLC, Archstone Multifamily Parallel Guarantor II LLC, Archstone Multifamily Parallel Guarantor ILC, Archstone Multifamily Guarantor LLC, Archstone Multifamily Guarantor LP, Archstone Multifamily Guarantor GP) LLC, Archstone Multifamily CM LLC, Archstone (MD Subsidiary) Inc., and Archstone Inc., as set forth on Schedule I hereto.

17

"Transferred Subsidiary Assets" means all assets of the Archstone Entities to be transferred to a Buyer Party or Buyer Designee immediately prior to the Initial Closing in accordance with Section 2.6.

"<u>Voting Agreement</u>" means the Voting Agreement, dated as of March 31, 2009, by and among Archstone Multifamily (Governance) LLC, Archstone Multifamily Parallel (Governance) LLC, Multifamily JV, Parallel Fund I, Parallel Fund II, Multifamily Parallel JV, Archstone Multifamily (GP) LLC, Archstone Multifamily Parallel (GP) LLC, REPE Archstone GP Holdings LLC, as amended by Amendment No. 1 to Voting Agreement, dated as of December 2, 2010.

"WARN" means the Worker Adjustment and Retraining Notification Act of 1988, as amended or any similar foreign, state or local law, regulation, or ordinance.

1.2 <u>Additional Definitions.</u> In addition to the terms set forth in <u>Section 1.1</u>, the following terms are defined in the text of this Agreement in the locations specified below:

Term	Cross-Reference
2012 Bonuses	13.5.1
Acquisition Litigation	14.2.2(d)
Agreement	Preamble
Archstone Indemnitee	13.14.1
Asset Acquisition Statement	13.9
Asset Transfer Documents	2.6.1

Asset Transfer Restriction Order	2.5.1
Audit Inquiry Letter	13.12.4
AVB	Preamble
AVB Cash Purchase Price Portion	2.2.1
AVB Common Shares	8.4.1
AVB Debt Commitment Letter	8.7.2
AVB Debt Financing	8.7.2
AVB Lenders	8.7.2
AVB Preferred Shares	8.4.1
AVB SEC Documents	8.5.1
AVB Shares	8.4.1
Bills of Sale	10.1.6(d)
Buyer Cure Period	15.1.3(b)
Buyer Group	12.2.1
Buyer Party	Recitals
Buyer Indemnified Parties	14.2.2
Cash Purchase Price	2.2.1
Claim Notice	14.3.1
Continuing Employees	13.15.1
Debt Commitment Letters	8.7.2
Debt Financing	8.7.2
Deferred Closing Assets	2.5.1
Designated Properties	12.3

Term	Cross-Reference
Entity Level Transaction	13.10.1(e)
EQR	Preamble
EQR Common Shares	7.4.1
EOR Preferred Shares	7.4.1
EQR Reimbursement Agreement	13.12
EQR SEC Documents	7.5.1
EQR Shares	7.4.1
Equity Residential Parties	Preamble
ERPOP	Preamble
ERPOP Cash Purchase Price Portion	2.2.1
ERPOP Debt Commitment Letter	7.7.2
ERPOP Debt Financing	7.7.2
ERPOP Lenders	7.7.2
Escrow Agent	2.5.2(b)(i)
Escrowed Deferred Closing Asset Transfer Documents	2.5.2(b)(iii)
Event	2.3.1(c)
Escrowed Deferred Stock Consideration	2.5.2(b)(i)
Existing Proceedings	14.5
Expenses Amount	15.3.2
Extension Closing	2.5.1
Extension Closing Date	2.5.1
Extension Closing Date Extension Notice	
	2.5.2(g)(i)
Financial Statements	5.6.1
First Extension Notice	2.4
Indemnified Claim	14.3.1
Indemnified Party	14.3.1
Indemnifying Party	14.3.1
Indemnity Agreement	13.1.4
Indemnity Obligations	13.14.5
Initial Closing	2.4
LBHI	Preamble
Lehman Entities	Recitals
Non-Controlled Subsidiary	6.3.2
Non-Party Affiliates	16.17
Ordinary Course Bonuses	13.5.1
Other Bonuses	13.5.1
Outside Closing Date	2.4
Parties	Preamble
Party	Preamble
Reimbursement Agreements	13.12
Requested Transactions	2.6.3
Retained Deferred Closing Asset	2.5.2(g)
Retained Escrowed Documents	2.5.2(g)(ii)
Retention Agreements	13.5.2
Second Extension Notice	2.4

Term Cross-Reference

Seller	Preamble
Seller Cure Period	15.1.3(a)
Seller Indemnified Parties	14.2.1
Severance Obligations	13.5.1
Subsidiary Interests	6.3.2
Tax Protection Agreement	6.7.3
Tenant Purchase Option Laws	13.1.5(b)
Third Party Claim	14.3.2
Voting Interest Parties	2.1.2

ARTICLE 2 PURCHASE AND SALE OF THE PURCHASED INTERESTS

- 2.1 <u>Purchase and Sale.</u> Upon the terms and subject to the conditions of this Agreement and in exchange for the consideration set forth in <u>Section 2.2</u>:
- (a) Immediately prior to the Initial Closing, but on the same Business Day as the Initial Closing, (i) Seller hereby agrees to sell, assign, transfer, convey 2.1.1 and deliver, to the applicable Buyer Party or Buyer Designee (or cause the applicable Archstone Entity to sell, assign, transfer, convey and deliver to the applicable Buyer Party or Buyer Designee), (ii) LBHI agrees to cause Seller and the applicable Archstone Entities to sell, assign, transfer, convey and deliver, to the applicable Buyer Party or Buyer Designee, and (iii) the applicable Buyer Party or Buyer Designee hereby agrees to purchase from Seller or the applicable Archstone Entity, all of Seller's and each applicable Archstone Entity's right and title to and interest in the Transferred Subsidiary Assets, free and clear of all Liens (other than any restrictions on transfer by any applicable Buyer Party or Buyer Designee imposed under applicable securities Laws and Permitted Liens); (b) at the Initial Closing, (x) Seller hereby agrees to sell, assign, transfer, convey and deliver, to the applicable Buyer Party or Buyer Designee, (v) LBHI agrees to cause Seller to sell, assign, transfer, convey and deliver, to the applicable Buyer Party or Buyer Designee, and (z) the applicable Buyer Party or Buyer Designee hereby agrees to purchase from Seller, all of Seller's right and title to and interest in the Transferred Assets (other than any Deferred Closing Assets), free and clear of all Liens (other than any restrictions on transfer by any applicable Buyer Party or Buyer Designee imposed under applicable securities Laws and the Permitted Liens), in each case, ((a) and (b)), in accordance with and in the sequences set forth in Schedule 2.6. If, and only if, any of ERPOP, AVB or the Buyer Designees fail to consummate any of the transactions contemplated by this Section 2.1.1(b) on the same day as the commencement of the transactions contemplated by Section 2.1.1(a), then, in such circumstance, LBHI and Seller shall have the right to elect, by written notice to the Buyer Parties given within thirty (30) days following the alleged breach hereunder by the Buyer Parties, either to pursue all remedies at Law and equity available

20

against ERPOP and AVB, who shall be jointly and severally liable therefor, or to receive the Seller Liquidated Damages Amount pursuant to Section 15.3; provided, however, that if LBHI and Seller do not elect to pursue such additional remedies at Law or equity and instead to receive the Seller Liquidated Damages Amount, then LBHI and Seller shall not be entitled to pursue, and shall be forever precluded from pursuing, such additional remedies.

- at the Initial Closing, (i) LBHI agrees to cause REPE Archstone GP Holdings, LLC, ACQ SPV I Holdings LLC and ACQ SPV II Paper LLC (collectively, the "<u>Voting Interest Parties</u>") to sell, assign, transfer, convey and deliver, to the applicable Buyer Party or Buyer Designee and (ii) the applicable Buyer Party of Buyer Designee hereby agrees to purchase from the Voting Interest Parties, all of the Voting Interest Parties' right and title to and interest in the voting interests in Archstone Property Holdings LLC.
- at each Extension Closing, as more fully described in <u>Section 2.5</u> and in accordance with the Deferred Closing Asset Escrow Agreement, (a) Seller hereby agrees to sell, assign, transfer, convey and deliver to the applicable Buyer Party or Buyer Designee, (b) LBHI agrees to cause Seller to sell, assign, transfer, convey and deliver, to the applicable Buyer Party or Buyer Designee, and (c) the applicable Buyer Party hereby agrees to purchase from Seller, all right, title and interest in and to, and assume all Liabilities with respect to, the applicable Deferred Closing Assets.

2.2 <u>Purchase Price</u>.

Subject to adjustment in accordance with Section 2.3 and Section 2.5 and as otherwise expressly provided for in this Agreement, in consideration for the Transferred Assets and the Transferred Subsidiary Assets, as applicable, subject to the terms and conditions of this Agreement, at the Initial Closing, (a) (i) ERPOP or the applicable Buyer Designee shall pay, or cause to be paid, to Seller an aggregate amount in cash equal to \$2,016,000,000 (the "ERPOP Cash Purchase Price Portion"), (ii) AVB or the applicable Buyer Designee shall pay, or cause to be paid, to Seller an aggregate amount in cash equal to \$669,000,000 (the "AVB Cash Purchase Price Portion", and, together with the ERPOP Cash Purchase Price Portion, the "Cash Purchase Price"), (iii) ERPOP shall deliver, or cause its Buyer Designees to deliver, to Seller (or, if so directed by Seller, to LBHI or a Lehman Designee) the ERPOP Equity Consideration, and (iv) AVB shall deliver, or cause to be delivered, to Seller (or, if so directed by Seller, to LBHI or a Lehman Designee) the AVB Equity Consideration; and (b) one or more Buyer Designees shall assume the Assumed Liabilities pursuant to this Section 2.2.1 and Section 2.2.2. The Cash Purchase Price shall be paid as provided in Section 3.1 and Section 16.14. For purposes of clarity, ERPOP and AVB shall be jointly and severally liable to pay the Cash Purchase Price, ERPOP Equity Consideration and AVB

21

Equity Consideration to Seller (or LBHI or a Lehman Designee, as applicable) and shall be jointly and severally liable for the Assumed Liabilities.

2.2.2 Upon the terms and subject to the conditions of this Agreement, each applicable Buyer Designee shall assume, by execution and delivery of an Assumption Agreement, (a) effective as of the Initial Closing (with respect to all Transferred Assets and Transferred Subsidiary Assets to be acquired by such Buyer Designee other than any Deferred Closing Assets that are subject to an Extension Closing in accordance with Section 2.5), and shall pay, perform and discharge when due, each of the Assumed Liabilities (other than any Assumed Liabilities related to the Deferred Closing Assets subject to an Extension Closing in accordance with Section 2.5), and (b) effective as of any Extension Closing (with respect to the applicable Deferred Closing Assets subject to an Extension Closing) in accordance with Section 2.5, and shall pay, perform and discharge when due, each of the Assumed Liabilities related to the applicable Deferred Closing Assets. The Buyer Parties shall not assume and shall not be responsible to pay, perform or discharge any of the Excluded Liabilities. For the avoidance of doubt, ERPOP and AVB shall each be jointly and severally liable pursuant to the Assumption Agreements for all Assumed Liabilities.

2.3 Purchase Price Adjustment.

2.3.1

(a) If (i) after the date of this Agreement, EQR declares or makes a dividend or distribution of EQR Common Shares on EQR Common Shares with a record date or, if no record date, an effective date prior to the Initial Closing Date, or AVB declares or makes a dividend or distribution of AVB Common Shares on AVB Common Shares with a record date, or, if no record date, an effective date prior to the Initial Closing, or (ii) either EQR or AVB announces or effects a share split or share combination with a record date or, if no record date, an effective date prior to the Initial Closing Date, then the number of EQR Common Shares or AVB Common Shares, as the case may, to be delivered as part of the Purchase Price shall be adjusted based on the following formula:

 $TS_1 = TS_0 \times OS_1/OS_0$

Where

- TS₀ = the number of EQR Common Shares or AVB Common Shares, as the case may be, deliverable hereunder immediately prior to the adjustment relating to such
- TS₁ = the new number of EQR Common Shares or AVB Common Shares, as the case may be, deliverable hereunder taking such event into account
- OS_0 = the number of EQR Common Shares or AVB Common

22

Shares, as the case may be, outstanding immediately prior to such event

- OS₁ = the number of EQR Common Shares or AVB Common Shares, as the case may be, outstanding immediately after such event.
 - If at any time after the date of this Agreement, either EQR or AVB declares or makes a cash dividend or distribution (other than one or more cash (b) dividends declared in the Ordinary Course of Business) with a record date or, if no record date, an effective date, prior to the Initial Closing Date, the EQR Cash Purchase Price Portion or the AVB Cash Purchase Price Portion, as the case may be, will be increased by an amount equal to the amount of such dividend or distribution that Seller would have received if the Initial Closing Date had occurred immediately prior to the record date or effective date for such dividend (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased EQR Cash Purchase Price Portion or AVB Cash Purchase Price Portion shall be a corresponding increase in the Purchase Price). If at any time after the date of this Agreement, either EQR or AVB declares or makes a dividend or distribution (other than a cash dividend or stock dividend (which is adjusted pursuant to Section 2.3.1(a)) that involves property, rights, securities, debt obligations or any other matter, then the Purchase Price will be increased by the property, rights, securities, debt obligations or other forms that Seller would have received if the Initial Closing Date had occurred immediately prior to the record date or effective date for such dividend or distribution (with such increase in the Purchase Price being payable in the form of the applicable property, rights, securities, debt obligations or other forms that Seller would have received if the Initial Closing Date had occurred immediately prior to the record date or effective date for such dividend or distribution). A cash dividend shall be deemed to be in the Ordinary Course of Business if, and only if, it is (i) in respect of the fourth (4th) quarter of 2012, less than the amount set forth for EQR or AVB, as applicable, on Schedule 2.3.1(a), or (b) in respect of 2013, made in accordance with ordinary course past practices and announced as the expected "run rate" for dividends, in respect of EQR, for the first three (3) calendar quarters of 2013 and, in respect of AVB, for all of 2013.
 - (c) If after the date of this Agreement either the EQR Common Shares or AVB Common Shares are changed, altered or modified by reason of recapitalization, reorganization, reclassification, consolidation, merger or otherwise (an "Event") (other than by reason of the events described in Section 2.3.1(a) or (b)), and such change, alteration or modification has a record date or, if no record date, an effective date prior to the Initial Closing Date, then the ERPOP Equity Consideration or AVB Equity

23

Consideration, as applicable, shall be changed, altered or modified as if the ERPOP Equity Consideration or AVB Equity Consideration were outstanding as of the record date or effective date for such Event.

- 2.3.2 In the event that the Initial Closing Date occurs on a date that is later than the Second Extension Date, then the Cash Purchase Price shall be increased by an amount in cash (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased Cash Purchase Price shall be a corresponding increase in the Purchase Price) equal to (a) \$1,000,000 for each day beginning on the day following the Second Extension Date to and including the Initial Closing Date (provided, however, that in no event shall there be any increase in the Cash Purchase Price with respect to any day during the period (x) beginning on the date that is five (5) Business Days following the date that the Buyer Parties have delivered a written notice in accordance with Section 2.4(a) that the Buyer Parties intend that the Initial Closing shall occur to the extent that, as of such date that is five (5) Business Days following the date on which the Buyer Parties have delivered such notice, the conditions set forth in Section 11.1.1, 11.1.2 and 11.1.3 have been satisfied and the conditions set forth in Sections 10.1.1, 10.1.2 and 10.1.3 (except to the extent waived by the Buyer Parties, in their sole and absolute discretion) shall not have been satisfied, and (y) ending on the date on which the conditions set forth in Sections 10.1.1, 10.1.2 and 10.1.3 have been satisfied), minus (b) an amount in cash equal to any cash dividends or distributions that will be received by Seller (or, if directed by Seller, LBHI or a Lehman Designee) at or following the Initial Closing with respect to the ERPOP Equity Consideration or the AVB Equity Consideration and which accrued with respect to any period ending on or before the Initial Closing Date (but the amount in clause (b) shall not exceed the amount payable as an increase in the Cash Purchase Price Portion pursuant to clause (a)). For purposes of clarity, the amount payable in accordance with this Section 2.3.2, if any, shall be the joint and several obligations of ERPOP and AVB, but may be allocated among the Buyer Parties as determined by the Buyer Parties in their sole and absolute discretion.
- 2.3.3 On the Initial Closing Date or any earlier date selected by Seller, all Derivative Instruments to which Archstone, Seller or any Archstone Entity is bound shall be unwound or terminated, as the case may be, and the Cash Purchase Price shall be positively or negatively adjusted on a dollar-for-dollar basis (i.e., if such adjustment is positive, the Cash Purchase Price shall be increased, and if such adjustment is negative, the Cash Purchase Price shall be decreased (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration,

required to be unwound or terminated to the extent that (x) prior to the Initial Closing, the applicable Derivative Instruments have been assigned by Seller or the applicable Archstone Entity to LBHI or a Lehman Designee and LBHI or the applicable Lehman Designee shall have assumed the obligations of the applicable Archstone Entity thereunder, (y) at the Initial Closing, LBHI has executed and delivered to the Buyer Parties a release in form and substance reasonably satisfactory to the Buyer Parties releasing the Buyer Parties, their Affiliates and the Archstone Entities and their respective Representatives from all Liabilities related to such Derivative Instrument and agreeing to indemnify Buyer Parties, their Affiliates and the Archstone Entities and their respective Representatives from and against all Losses relating to such Derivative Instrument, and (z) at the Initial Closing, the Cash Purchase Price shall have been positively or negatively adjusted on a dollar-for-dollar basis (i.e., if the value of such Derivative Instrument is positive, the Cash Purchase Price shall be decreased, and if the value of such Derivative Instrument is negative, the Cash Purchase Price shall be increased (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased or decreased Cash Purchase Price shall be a corresponding increase or decrease in the Purchase Price)) to reflect the value of such Derivative Instruments used in the preparation of the Archstone Entities' June 30, 2012 financial statements. For purposes of clarity, the amount payable by or to the Buyer Parties in accordance with this Section 2.3.3, as applicable, shall be the joint and several obligations of ERPOP and AVB, but may be allocated among the Buyer Parties or the Buyer Designees as determined by the Buyer Parties in their sole and absolute discretion. For the avoidance of doubt, ERPOP and AVB shall be responsible for any non-compliance, violation o

2.4 <u>Initial Closing</u>. Unless this Agreement shall have been terminated pursuant to <u>Article 15</u> and subject to the satisfaction or waiver of the conditions set forth in <u>Articles 9, 10</u> and <u>11</u> (excluding those conditions which by their nature are to be satisfied as part of the Initial Closing, but subject to the fulfillment or waiver of those conditions), the initial closing of the Contemplated Transactions (the "<u>Initial Closing</u>") shall take place at the offices of Hogan Lovells US LLP, 875 3rd Avenue, New York, New York, 10022 at 9:00 a.m., New York time, on or before the Outside Closing Date. The Buyer Parties may fix the Initial Closing Date as any Business Day prior to the Outside Closing Date (a) upon written notice (which notice may be rescinded in the Buyer Parties' sole discretion) to Seller and LBHI delivered at least five (5) Business Days in advance of the Initial Closing Date selected by Buyer Parties (and if the conditions set forth in <u>Article 9</u> are not satisfied or waived on such date, the Initial Closing Date shall be the first date on which all of the conditions set forth in <u>Article 9</u> are satisfied or waived), or (b) at such other place, time or date as the Parties hereto may agree in writing, in each case, subject to the satisfaction or waiver of the conditions set forth in <u>Articles 9</u>, 10 and 11 (excluding those conditions which by their nature are to be satisfied as part of the Initial Closing, but subject to the fulfillment or waiver of those conditions). The "<u>Outside Closing Date</u>" shall mean the

25

date which is sixty (60) days after the date of this Agreement; provided that the Parties acknowledge and agree that the Buyer Parties may determine, in their sole discretion, to postpone (i) upon notice (the "First Extension Notice") to Seller and LBHI delivered by either of the Buyer Parties or both of them prior to the date that is five (5) Business Days prior to the date which is sixty (60) days following the date of this Agreement, the Outside Closing Date to a date that is no later than ninety (90) days after the date of this Agreement, and (ii) upon notice (the "Second Extension Notice") to Seller and LBHI delivered by either of the Buyer Parties or both of them prior to the date that is five (5) Business Days prior to the date which is ninety (90) days following the date of this Agreement, the Outside Closing Date to a date that is no later than one hundred twenty (120) days after the date of this Agreement; provided, further, that in the event that the Buyer Parties have not delivered either (I) a First Extension Notice or a Second Extension Notice, as applicable, within the time periods specified in clause (i) or (ii), as applicable, (II) a notice setting the Initial Closing Date in accordance with Section 2.4(a) that has not been rescinded, or (III) a notice affirmatively stating the Buyer Parties do not wish to extend the Outside Closing Date, then the First Extension Notice or the Second Extension Notice, as applicable, shall be deemed to have been delivered to Seller and LBHI in accordance with this Section 2.4 and the Outside Closing Date shall be deemed to have been extended to the First Extension Date or the Second Extension Date, as applicable. For the avoidance of doubt, (x) the Buyer Parties may not extend the Outside Closing Date to a date that is beyond 120 days after the date of this Agreement under any circumstances (but LBHI and Seller shall have the right, in their sole discretion, to extend the Outside Closing Date by up to twenty (20) Business Days in order to cure any default by LBHI or Seller as provided in Section 15.1.3(a)); (y) in the event that the Initial Closing does not occur on or before the First Extension Date, then the Seller Liquidated Damages Amount shall increase to an amount in cash equal to \$800,000,000; and (z) if the Initial Closing does not occur on or before the Second Extension Date, then the Purchase Price shall increase as set forth in Section 2.3.2. In the event the Buyer Parties do not fix the Initial Closing Date, the Initial Closing Date shall be the one hundred twentieth (120th) day after the date of this Agreement, and the Buyer Parties shall be in breach of this Agreement if the Initial Closing does not occur on or before the one hundred twentieth (120th) day after the date of this Agreement; provided that the conditions set forth in Articles 9 and 10 have been satisfied on such date (except to the extent waived by the Buyer Parties or conditions which by their nature are to be satisfied as part of the Initial Closing).

2.5 <u>Deferred Closings</u>.

2.5.1 In the event that (A) any Deferred Closing Asset Consent required in connection with any of the assets of the Archstone Entities set forth on Schedule 2.5.1 has not been obtained on or prior to the Initial Closing Date and the Buyer Parties have elected pursuant to the provisions below to postpone the purchase thereof, (B) there is an Order issued by a Governmental Authority in effect on the Initial Closing Date (an "Asset Transfer Restriction Order") preventing or prohibiting the transfer of any asset of the Archstone Entities to the Buyer Parties, or (C) with respect to a real property asset located in Montgomery County, Maryland, a certificate of compliance with respect to the applicable Tenant Purchase Option Law has not been obtained as of the Initial Closing Date and the Buyer Parties have concluded that an exemption from the requirement to obtain a certificate of compliance is not

26

available (all such assets described in clauses (A) through (C) being referred to as "<u>Deferred Closing Assets</u>"), then (i) the purchase by the Buyer Parties of any Montgomery County Deferred Closing Asset and any Deferred Closing Asset with respect to which an Asset Transfer Restriction Order is in effect shall be postponed, and (ii) the Buyer Parties may elect, in their sole discretion, by written notice delivered to Seller within five (5) Business Days prior to the Initial Closing Date, subject to and in accordance with the provisions of this <u>Section 2.5</u>, to postpone the purchase by the Buyer Parties of any of the other Deferred Closing Assets, in each case, to a date following the Initial Closing Date that is no later than the Outside Deferred Closing Date, and which date shall be no later than three (3) Business Days following the date on which (x) the applicable Deferred Closing Asset Consent with respect to the applicable Deferred Closing Asset has actually been received, (y) the Asset Transfer Restriction Order with respect to the applicable Deferred Closing Asset is no longer in effect, or (z) the applicable certificate of compliance is received with respect to the applicable Montgomery County Deferred

Closing Asset, whichever may be applicable (each such closing of the purchase of the applicable Deferred Closing Asset by a Buyer Party, an "<u>Extension Closing</u>" and, the date on which any such Extension Closing actually occurs, an "<u>Extension Closing Date</u>"). For the avoidance of doubt, <u>Schedule 2.5.1</u> shall not be modified except as expressly set forth in <u>Section 9.2</u>.

- 2.5.2 In the event the closing of any Deferred Closing Assets is postponed in accordance with Section 2.5.1, the following provisions shall apply:
 - (a) Immediately prior to the Initial Closing, the Parties shall (i) cause such Deferred Closing Assets to be transferred to Seller or another entity mutually agreed upon by ERPOP, AVB and LBHI in accordance with <u>Schedule 2.6</u> such that the applicable Deferred Closing Assets shall be excluded from the Transferred Assets and the Transferred Subsidiary Assets being transferred to the Buyer Parties at or immediately prior to the Initial Closing, and (ii) cause the actions set forth on <u>Schedule 2.6</u> to be modified so as to enable the Initial Closing to occur without transfer of the Deferred Closing Assets (x) with respect to which the applicable Deferred Closing Asset Consents have not been obtained and for which the Buyer Parties have elected to postpone the purchase thereof as provided herein, (y) with respect to which an Asset Transfer Restriction Order is in effect, or (z) with respect to which a certificate of compliance has not been received for any Montgomery County Deferred Closing Asset, in each case, as of the Initial Closing Date.
 - (b) (i) The applicable Buyer Party shall, at the Initial Closing, deposit into an escrow account with an escrow agent mutually acceptable to the Parties (the "Escrow Agent") a portion of the ERPOP Equity Consideration or the AVB Equity Consideration, as applicable (the "Escrowed Deferred Stock Consideration"), in an amount equal to the amount allocated to the

27

applicable Deferred Closing Asset (which allocated amount is set forth opposite such Deferred Closing Asset on Schedule 2.5.1); provided that there shall be no Escrowed Deferred Stock Consideration deposited with the Escrow Agent with respect to any Montgomery County Deferred Closing Asset, unless the aggregate value of Escrowed Deferred Stock Consideration would be less than \$50 million, in which event the aggregate Escrowed Deferred Stock Consideration to be deposited with the Escrow Agent with respect to all such Montgomery County Deferred Closing Assets shall be equal to the number of EQR Common Shares and/or number of AVB Common Shares (as applicable) allocated to the applicable Montgomery County Deferred Closing Asset on Schedule 2.5.1); provided however, that if the aggregate value of such EQR Common Shares and AVB Common Shares (the "Montgomery County Escrowed Share Value") is in excess of an amount equal to \$50 million less the aggregate value of the Escrowed Deferred Stock Consideration with respect to all other Deferred Closing Assets (such amount the "Montgomery County Escrowed Consideration Limit"), then the number of EQR Common Shares and number of AVB Common Shares shall be reduced in proportion to the relative value represented by the aggregate number of EQR Common Shares and AVB Common Shares allocated to all such Montgomery County Deferred Closing Assets, until the Montgomery County Escrowed Share Value does not exceed the Montgomery County Escrowed Consideration Limit. Any fractional shares as a result of determining the Escrowed Deferred Stock Consideration shall be rounded to the nearest whole share. For purposes of valuing the Escrowed Deferred Stock Consideration shall be rounded to the nearest whole share. For purposes of valuing the Escrowed Deferred Stock Consideration or any component thereof, the EQR Common Shares shall be deemed to have a value of \$58.75 per share and the AVB Common Shares shall be deemed to have a value of \$136.00 per share;

- (ii) the ERPOP Equity Consideration or AVB Equity Consideration, as applicable, payable to Seller (or, if so directed by Seller, to LBHI or a Lehman Designee) at the Initial Closing in accordance with <u>Section 2.2.1</u> shall be reduced by the Escrowed Deferred Stock Consideration delivered to the Escrow Agent in accordance with clause (i) above; and
- (iii) the Parties shall (and LBHI shall cause the applicable Archstone Entities to) deliver to the Escrow Agent executed but undated originals of all documents and other deliverables (including Assumption Agreements effecting the assumption by the Buyer Parties or applicable Buyer Designee of Assumed Liabilities relating to such Deferred Closing Asset) necessary to cause the sale, assignment, transfer, conveyance and delivery of the Deferred Closing Assets to the applicable Buyer Party and the assumption of the Assumed Liabilities relating to such Deferred Closing Assets following the earlier to occur of (x) no later than the third Business

28

Day following the date on which (1) the applicable Deferred Closing Asset Consents have actually been received, (2) the Asset Transfer Restriction Orders relating to the applicable Deferred Closing Assets are no longer in effect, or (3) the applicable certificates of compliance are received with respect to the applicable Montgomery County Deferred Closing Assets, whichever may be applicable, and (y) the Outside Deferred Closing Date (the "Escrowed Deferred Closing Asset Transfer Documents").

(c) Subject to satisfaction or waiver of the conditions set forth in Sections 10.2 and 11.2 and subject further to Section 2.5.2(g), the Parties shall provide joint written instructions to the Escrow Agent to (i) release the Escrowed Deferred Stock Consideration (together with any dividends received with respect thereto) relating to each Deferred Closing Asset from escrow, and (ii) deliver the applicable Escrowed Deferred Stock Consideration (together with any dividends received with respect thereto) to Seller (or, if so directed by Seller, to LBHI or a Lehman Designee), upon the earlier to occur of (x) no later than the third Business Day following the date on which (1) the applicable Deferred Closing Asset Consent relating to the applicable Deferred Closing Asset is actually received, (2) the Asset Transfer Restriction Order relating to the applicable Deferred Closing Asset is no longer in effect, or (3) the applicable certificate of compliance is received with respect to the applicable Montgomery County Deferred Closing Asset, whichever may be applicable, and (y) the Outside Deferred Closing Date, in each case ((x) and (y)), unless Seller or LBHI shall not have delivered the joint written instructions required to be delivered to the Escrow Agent in accordance with the Deferred Closing Asset Escrow Agreement.

Notwithstanding the provisions of this paragraph (c), in the event that, at the time when any Escrowed Deferred Stock Consideration would be released pursuant to clause (x) above with respect to any Deferred Closing Asset, the applicable Extension Closings have not previously occurred with respect to all of the Montgomery County Deferred Closing Assets, then the aggregate value of the Escrowed Deferred Stock Consideration that shall be released at such time shall be equal to the excess (if any) of the aggregate value of the Escrowed Deferred Stock Consideration then being held over the aggregate value of the Escrowed Deferred Stock Consideration allocable to the Deferred Closing Assets as to which a closing has not then occurred (including the aggregate value of the portion of the ERPOP Equity Consideration and the AVB Equity Consideration allocated to the Montgomery County Deferred Closing Assets as to which a closing has not occurred, as set forth on Schedule 2.5.1). Upon any release of Escrowed Deferred Stock Consideration under the circumstances described in this paragraph, (A) the value of the Escrowed AVB Equity Consideration shall be based on

\$136.00 per share and the value of the Escrowed ERPOP Equity Consideration shall be based on \$58.75 per share; and (B) the AVB Common Shares and EQR Common Shares that continue to be held by the Escrow Agent after such release shall have aggregate values, to the extent possible, in proportion to the relative aggregate values represented by the EQR Common Shares and the AVB Common Shares allocable to all Deferred Closing Assets (including any Montgomery County Deferred Closing Assets) as to which an Extension Closing has not then occurred, as set forth on Schedule 2.5.1.

Notwithstanding the provisions of this paragraph (c), in the event that, at the time when any Escrowed Deferred Stock Consideration would be released pursuant to clause (y) above, the applicable Extension Closings have not occurred with respect to all of the Montgomery County Deferred Closing Assets, then the value of the Escrowed Deferred Stock Consideration that shall be released at such time shall be equal to the excess (if any) of the aggregate value of the Escrowed Deferred Stock Consideration then being held over the lesser of \$50 million and twenty percent (20%) of the aggregate value of the portion of the ERPOP Equity Consideration and the AVB Equity Consideration allocated to the Montgomery County Deferred Closing Assets as to which a closing has not occurred, as set forth on Schedule 2.5.1. Upon any release of Escrowed Deferred Stock Consideration under the circumstances described in this paragraph, (I) the value of the Escrowed AVB Equity Consideration shall be based on \$136.00 per share and the value of the Escrowed ERPOP Equity Consideration shall be based on \$58.75 per share; and (II) the shares of Escrowed Deferred Stock Consideration that continue to be held by the Escrow Agent after such release shall consist, to the extent possible, of AVB Common Shares and EQR Common Shares representing the same percentages of the portions of the AVB Equity Consideration and the ERPOP Equity Consideration, as applicable, allocable to the Montgomery County Deferred Closing Assets as to which a closing has not then occurred, as set forth on Schedule 2.5.1.

(d) Subject to Section 2.5.2(g), the Deferred Closing Asset Escrow Agreement shall provide that, subject to satisfaction or waiver of the conditions set forth in Sections 10.2 and 11.2, the Parties shall provide joint written instructions to the Escrow Agent to release the Escrowed Deferred Closing Asset Transfer Documents from escrow and deliver the applicable Escrowed Deferred Closing Asset Transfer Documents to the applicable Buyer Parties or to Seller on the date referenced in subsection (c) above (unless the applicable Deferred Closing Asset Consent shall have been received from the applicable third party but the applicable Buyer Party shall not have delivered the joint written instructions required to be delivered by the Deferred Closing Asset Escrow Agreement). Subject to Section 2.5.2(g), without any further action of the parties hereto, on the Outside Deferred Closing Date, subject to satisfaction or waiver of the conditions set forth in Sections 10.2 and 11.2, any remaining Escrowed Deferred Stock Consideration and Escrowed Deferred Closing Asset

30

Transfer Documents shall be released from escrow in accordance with the terms of the Deferred Closing Asset Escrow Agreement.

- (e) From and after the Initial Closing Date, the Buyer Parties shall use their respective Commercially Reasonable Efforts (but without any payment of money or other transfer of value to any third party, except as required by the terms of any applicable Contract relating to such Deferred Closing Asset) to obtain from the applicable third party (i) the applicable Deferred Closing Asset Consent for the conveyance, assignment or transfer of the applicable Deferred Closing Asset to the applicable Buyer Party or Buyer Designee, or (ii) written confirmation reasonably satisfactory in form and substance to Lehman and the Buyer Designees confirming that such Deferred Closing Asset Consent or approval is not required. Subject to compliance by the Buyer Parties with Section 13.10.1(g), Seller and the Lehman Entities shall use their Commercially Reasonable Efforts to cooperate with the Buyer Parties in obtaining the Deferred Closing Asset Consents and shall take any actions that can be taken only by the owner of such Deferred Closing Asset (and not by the Buyer Parties) and which is reasonably requested by the Buyer Parties with respect to such Deferred Closing Asset Consents within five (5) Business Days following the receipt of such request from the Buyer Parties; provided, however, that the Buyer Parties acknowledge that Seller and LBHI shall have limited organization, infrastructure and other resources to provide assistance to Buyer Parties after the Initial Closing Date.
- (f) The Parties shall enter into, as of the Initial Closing Date, the Deferred Closing Asset Arrangement Agreement(s).
- In the event that an Asset Transfer Restriction Order is in effect on the date that is 180 days after the date of this Agreement with respect to any Deferred Closing Asset or a certificate of compliance with respect to the applicable Tenant Purchase Option Law has not been obtained as of the date that is 180 days after the date of this Agreement with respect to any Montgomery County Deferred Closing Asset (each such asset, a "Retained Deferred Closing Asset"), then the Extension Closing with respect to such Retained Deferred Closing Asset shall be further postponed in accordance with this Section 2.5.2(g) and the following provisions shall apply:
 - (i) The applicable Buyer Party shall deliver to the Seller, LBHI and the Escrow Agent, on or prior to the date that is 180 days after the date of this Agreement written certification that an Asset Transfer Restriction Order is in effect or that a certificate of compliance has not been obtained with respect to a Montgomery County Deferred Closing Asset and a calculation of the Retained Escrowed Stock Consideration (such notice, an "Extension Notice").

- (ii) The Deferred Closing Asset Escrow Agreement shall provide that in the event an Extension Notice has been received by the Escrow Agent, the Escrow Agent shall (X) continue to hold in escrow the Deferred Closing Asset Transfer Documents related to the Retained Deferred Closing Asset (the "Retained Escrowed Documents") and (Y) continue to hold the Retained Escrowed Stock Consideration, in each case, until the parties provide joint written instructions to the Escrow Agent to release the Retained Escrowed Documents and the Retained Escrowed Stock Consideration; provided, however, that, subject to satisfaction or waiver of the conditions set forth in Sections 10.2 and 11.2, on the date that is 180 days after the date of this Agreement, the Escrow Agent shall release all of the Deferred Closing Assets Transfer Documents and Escrowed Deferred Stock Consideration related to all other Deferred Closing Assets, except those for which an Asset Transfer Restriction Order is in effect and with respect to any Montgomery County Deferred Closing Assets as to which a certificate of compliance has not been obtained. The parties shall provide joint written instructions to the Escrow Agent no later than three (3) Business Days from the date on which the transfer of such Retained Deferred Closing Asset may first be lawfully consummated or the date on which a certificate of compliance is obtained, as applicable.
- (h) The Parties acknowledge and agree that in the event that all of the Escrowed Deferred Stock Consideration (other than the Retained Escrowed Stock Consideration) relating to all the Deferred Closing Assets is not released and delivered to Seller (or, if so directed by Seller, to LBHI or a Lehman

Designee) on or before the Outside Deferred Closing Date other than as a result of Seller's failure to deliver the joint instructions required by the Deferred Closing Asset Escrow Agreement, then Seller, LBHI and any applicable Lehman Designee shall have all remedies at Law or equity (including the right to seek specific performance), and other than as set forth in Section 2.5.2(g) for Retained Escrowed Stock Consideration, ERPOP and AVB shall, jointly and severally, indemnify Seller, LBHI and any Lehman Designees for all Losses as a result of the failure of the Escrowed Deferred Stock Consideration to be received by Seller (or, if so directed by Seller, to LBHI or a Lehman Designee).

2.6 <u>Asset Transfer Transactions</u>.

2.6.1 Schedule 2.6 attached hereto sets forth the steps and conditions required to cause each Transferred Subsidiary Asset and Transferred Asset to be transferred to the applicable Buyer Party or Buyer Designee and the allocable portion of the Cash Purchase Price and the ERPOP Equity Consideration or the AVB Equity Consideration to be paid in exchange for each such Transferred Subsidiary Asset. Immediately prior to the Initial Closing (as described in Section 2.1.1), or at any Extension Closing Date (with respect to any Deferred Closing Asset for which an Extension Closing will occur), Seller

32

shall (and LBHI shall cause Seller and the applicable Archstone Entities to) take all applicable actions set forth on Schedule 2.6, including the execution and delivery of all resolutions, bills of sale, deeds, assignments, and other conveyance documents contemplated by the transactions set forth on Schedule 2.6, in form and substance reasonably satisfactory to Seller (the "Asset Transfer Documents"), provided that the Buyer Parties shall deliver proposed forms of all such documents a reasonable period in advance of the Initial Closing and the Asset Transfer Documents shall comply in all respects with Section 2.6.3.

- Subject to Section 2.6.3, the Buyer Parties may (subject to the following conditions) revise Schedule 2.6 at any time prior to the Initial Closing in order to designate other Buyer Designees or to change the steps, conditions and allocable portion of the Cash Purchase Price and the ERPOP Equity Consideration or the AVB Equity Consideration to be paid in exchange for each such Transferred Subsidiary Asset set forth on Schedule 2.6 with respect to any Transferred Subsidiary Asset if such revisions and the Buyer Parties' structure comply in all respects with Section 2.6.3; provided that after giving effect to the revisions to Schedule 2.6, the revised Schedule 2.6 is reasonably similar in terms of the types of transactions and actions required of Seller and the Lehman Entities described therein to the Schedule 2.6 that exists as of the date of this Agreement.
- The Buyer Parties shall have the option, in their sole and absolute discretion, to revise Schedule 2.6 to request that Seller, subject to Section 13.10.1(e) and immediately prior to the Initial Closing, (a) convert or cause the conversion of one or more Subsidiaries that are organized as corporations into limited liability companies (or other entities) and one or more Subsidiaries that are organized as limited partnerships or limited liability companies into other entities, on the basis of organizational documents as reasonably requested by the Buyer Parties, (b) revoke or cause to be revoked the election of one or more Subsidiaries (other than corporations) that have elected to be treated for Tax purposes as a corporation, (c) sell or cause to be sold all of the capital stock, shares of beneficial interests, partnership interests or limited liability interests owned, directly or indirectly, by Seller in one or more Archstone Entities to any Person (who, if not an Affiliate of EQR or AVB or an exchange accommodation titleholder, with respect to any transactions related to any property that the Buyer Parties desire to be treated as replacement property in an exchange pursuant to Section 1031 of the Code, shall be reasonably acceptable to LBHI) at a price and on terms all as designated by the Buyer Parties, or (d) sell or cause to be sold any of the assets of Seller or any Archstone Entity to any Person at a price and on terms all as designated by the Buyer Parties (clauses (a), (b), (c) and (d), together with any other requested revisions to Schedule 2.6 pursuant to and in accordance with Section 2.6.2, the "Requested Transactions"); provided that after giving effect to the revisions to Schedule 2.6, the revised Schedule 2.6 is reasonably similar in terms of the types of transactions and actions required of Seller and the

33

Lehman Entities described therein to the Schedule 2.6 that exists as of the date of this Agreement. Notwithstanding anything to the contrary set forth in this Section 2.6, in no event shall the Asset Transfer Documents or the Requested Transactions be inconsistent with the following: (i) the Requested Transactions shall not delay or prevent the completion of the Contemplated Transactions, (ii) the Requested Transactions shall be implemented as close as possible to the Initial Closing (but after the Buyer Parties shall have waived or confirmed in writing that all conditions to the consummation of the Contemplated Transactions have been satisfied) other than those conditions that, by their nature, are to be satisfied at the Initial Closing, (iii) neither Seller nor any Lehman Entity or Archstone Entity shall be required to take any action in contravention of any Law or Organizational Document, or that imposes any criminal or civil liability, (iv) the consummation of any Requested Transactions shall be contingent upon the receipt by Seller of an irrevocable written notice from Buyer Parties confirming that all of the conditions set forth Articles 9 and 10 have been satisfied (or, with respect to Article 10, at the option of the Buyer Parties, waived) and that the Buyer Parties are ready, willing and able to, and unconditionally obligated to, proceed immediately with the Initial Closing, (v) the Requested Transactions (or the inability to complete the Requested Transactions) and the Asset Transfer Documents shall not affect or modify in any respect the obligations of the Buyer Parties under this Agreement, including payment of the Purchase Price, (vi) the Requested Transactions shall not require Seller or any Person owning a direct or indirect interest in Seller to be a party to any Asset Transfer Documents, except, in the case of Seller, with respect to any asset owned directly by Seller, (vii) the Requested Transactions shall not create any additional representations, warranties, indemnities, agreements or other obligations of Seller (except to the extent that such indemnities, agreements or obligations constitute Assumed Liabilities) or the Lehman Entities not expressly set forth in this Agreement, and (viii) the Requested Transactions shall not cause any asset to become more likely to be subject to a purchase option under any Tenant Purchase Option Law if such asset was not previously subject to such an option under the terms of Schedule 2.6 as of the date hereof. ERPOP and AVB, on a joint and several basis, shall and do hereby agree to indemnify and hold harmless Seller, the Lehman Entities, and their Representatives from and against any and all Losses suffered or incurred by them in connection with or as a result of taking such actions (including any Liability under any Tax Protection Agreement arising from the Contemplated Transactions). Without limiting the foregoing, none of the representations, warranties or covenants of Seller or the Lehman Entities shall be deemed to apply to, or deemed breached or violated by, any of the Requested Transactions. ERPOP and AVB, on a joint and several basis, shall and do hereby agree to indemnify Seller, the Lehman Entities and their Representatives from and against all Losses suffered or incurred by them in connection with or as a result of changes made to Schedule 2.6 after the date hereof (but expressly excluding any legal fees incurred by such Persons in

A Buyer Party or a Buyer Designee may acquire one or more of the Transferred Assets or Transferred Subsidiary Assets as replacement property in an exchange pursuant to Section 1031 of the Code. Subject to compliance by the Buyer Parties with Section 13.10.1(e), each Party shall reasonably cooperate with the other Parties in consummating the transfer of such Transferred Assets or Transferred Subsidiary Assets in an exchange transaction pursuant to a written Exchange Agreement and related documents entered into by the applicable party and a qualified intermediary so long as such cooperation does not (i) create any additional representations, warranties, indemnities, agreements or other obligations of Seller (except to the extent that such indemnities, agreements or obligations constitute Assumed Liabilities) or the Lehman Entities not expressly set forth in this Agreement, or (ii) require Seller or any Person owning a direct or indirect interest in Seller to be a party to any such Exchange Agreement and related documents, except, in the case of Seller, with respect to any asset owned directly by Seller. Any assignment of the rights or obligations of a Party hereunder to a qualified intermediary to effectuate the exchange shall not relieve, release or absolve such party of its obligations to the other Parties under this Agreement. For the avoidance of doubt, neither Seller nor any Lehman Entity shall have any Liability to any Buyer Indemnified Party if such Person does not achieve its desired tax treatment under Section 1031 of the Code.

ARTICLE 3 EFFECT OF SALE

3.1 Required Tax Withholding. The Buyer Parties shall be entitled to deduct and withhold from the Purchase Price otherwise payable to Seller or any of its Subsidiaries pursuant to this Agreement such amounts as the Buyer Parties are required to deduct and withhold with respect to the making of such payment under the Code or any other provision of federal, state, local or foreign tax Law; provided that the Buyer Parties shall not deduct or withhold any amounts from the Purchase Price (a) pursuant to Section 1445 of the Code, if Seller (or any applicable Subsidiary of Seller) has provided either (i) a certificate of non-foreign status executed by a duly authorized Representative of Seller or such Subsidiary, sworn under penalty of perjury and in form and substance required under the Treasury Regulations under Section 1445 of the Code, to the effect that withholding under Section 1445 of the Code is not required in connection with the acquisition of the Transferred Assets and Transferred Subsidiary Assets by the Buyer Parties or Buyer Designees, as applicable, or (ii) a withholding certificate issued by the IRS pursuant to Treasury Regulations Section 1.1445-3 that is reasonably satisfactory to the Buyer Parties stating that such amounts are not required to be withheld; or (b) pursuant to any other Law, without having provided to Seller or its applicable Subsidiary written notice of its intention to withhold and the basis for such withholding fifteen (15) days prior to the Initial Closing. To the extent that amounts are withheld by the Buyer Parties or Buyer Designees, such withheld amounts (x) shall be remitted promptly and in accordance with applicable Law by the Buyer Parties or Buyer Designees to the applicable Governmental Authority, and (y) shall be

35

treated for all purposes of this Agreement as having been paid to Seller or its applicable Subsidiary in respect of which such deduction and withholding was made by the Buyer Parties or Buyer Designees and shall be deemed part of the Purchase Price. The Buyer Parties shall promptly provide to Seller or its applicable Subsidiary written evidence reasonably satisfactory to Seller or its applicable Subsidiary of any such remittance.

3.2 No Further Ownership Rights. At and after the Initial Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at and after any Extension Closing (with respect to the applicable Deferred Closing Asset subject to such Extension Closing), the Purchase Price, upon payment thereof, and the assumption of the Assumed Liabilities, shall be deemed to be in full satisfaction of all rights pertaining to the Transferred Assets and Transferred Subsidiary Assets and no Lehman Entity or Seller shall have any further ownership or other rights with respect thereto, other than the performance of the covenants contemplated by this Agreement or any of the other Transaction Documents.

ARTICLE 4 REPRESENTATIONS AND WARRANTIES OF LBHI

Except as set forth in the Disclosure Schedules, LBHI hereby represents and warrants to the Buyer Parties as follows:

- 4.1 <u>Organization and Good Standing</u>. LBHI is a legal entity that is duly organized, validly existing and in good standing under the laws of the state of Delaware and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as presently conducted. LBHI is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except in such jurisdictions where the failure to be so duly qualified or licensed and in good standing has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect on its ability to consummate the Contemplated Transactions.
- 4.2 <u>Authorization; Validity of Agreements.</u> LBHI has the requisite power and authority to execute and deliver (or cause the applicable Lehman Entity to execute and deliver) this Agreement and each other Transaction Document to which it or any of the Lehman Entities is a party and to perform its obligations hereunder and thereunder. The execution, delivery and performance by LBHI of this Agreement, the execution, delivery and performance by LBHI or any other Lehman Entity of any other Transaction Document to which LBHI or any other Lehman Entity is a party and the performance of LBHI's and the other Lehman Entities' obligations hereunder and thereunder have been duly authorized, and no other proceedings, actions or authorizations on any of their parts or on the part of any holders of Equity Interests in them are necessary in connection with such execution, delivery and performance (other than any such action or authorization that has been taken or obtained). Each of the Transaction Documents to which LBHI or any Lehman Entity is a party has been duly executed and delivered, or will at the Initial Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to the applicable Deferred Closing Asset subject to such Extension

36

Closing), be duly executed and delivered, and constitutes, or will constitute at the Initial Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to the applicable Deferred Closing Asset subject to such Extension Closing), as the case may be, its legal, valid and binding obligation, enforceable against it in accordance with the terms of such Transaction Document, except that (a) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights generally, and (b) general equitable principles.

4.3 <u>Consents and Approvals; No Violations.</u> The execution, delivery and performance by LBHI of this Agreement and any other Transaction Documents to which it is a party, and the performance by the Lehman Entities of their obligations under any Transaction Documents to which any of them are a party, do not and will not (a) violate, contravene or conflict with any provision of any Organizational Documents of the Lehman Entities or any Organizational Documents of Seller; (b) except with respect to Consents required under Contracts set forth in <u>Schedule 4.3</u>, result in a breach of or constitute a default under (with or without due notice or lapse of time or both) any of the terms, conditions or provisions of any Contract to which any Lehman Entity is a party; (c) contravene or conflict with or constitute a violation of any Orders or Laws applicable to or binding upon any Lehman Entity; (d) except for any notifications required under any Tenant Purchase Option Law, require on the part of any Lehman Entity any filing or registration with, notification to, or authorization, consent or approval of, any Governmental Authority; (e) result in the creation or imposition of any Lien on any of the Lehman Entities other than Permitted Liens and restrictions on transfer by the applicable Buyer Party or Buyer Designee imposed under the Organizational Documents of

the Lehman Entities; except, (A) in the case of clauses (c) and (d), for filings pursuant to the HSR Act or with respect to Transfer Taxes and/or the payment of such Transfer Taxes and (B) in the case of clauses (b), (c), (d) and (e) as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

- 4.4 Ownership of Equity Interests in Seller.
- 4.4.1 The beneficial and record owners of the Equity Interests of Seller are set forth in Schedule 4.4.1. Such Persons have good, valid and marketable title to, and will, immediately prior to Initial Closing, be the sole beneficial and record owners of and have good, valid and marketable title to, all of the Equity Interests of Seller set forth opposite the respective names of the Lehman Entities in Schedule 4.4.1. LBHI has all necessary corporate, company or other entity power to cause Seller to perform Seller's obligations under the Transaction Documents. Except for the Equity Interests in Seller and the voting interests in Archstone Property Holdings LLC (which interests in Archstone Property Holdings LLC are owned or held by the Lehman Entities set forth in Schedule 4.4.1), the Lehman Entities do not directly own or hold any Equity Interests, voting interests or other interests in any Archstone Entity.
- 4.4.2 Other than pursuant to the terms and conditions of this Agreement, (a) each Lehman Entity is not a party to any option, warrant, purchase right or other

37

contract that could require (i) any Archstone Entity to issue any Equity Interest in such Archstone Entity to any Person other than an Archstone Entity, or (ii) Seller or any Archstone Entity to sell, transfer or otherwise dispose of the Transferred Assets or Transferred Subsidiary Assets, and (b) each Lehman Entity is not a party to any voting trust, agreement, proxy or other Contract with respect to the voting of any Equity Interest in Seller or with respect to the Transferred Interests or any other Equity Interests in any Archstone Entity other than the Voting Agreement.

- 4.5 <u>Brokerage Fees.</u> Except for the Persons set forth in <u>Schedule 4.5</u> (whose fees shall be paid by LBHI pursuant to <u>Section 13.10.2</u>), there is no investment banker, broker, finder or other intermediary who has been retained by or is authorized to act on behalf of any Lehman Entity who might be entitled to any fee or commission in connection with the Contemplated Transactions.
 - 4.6 <u>Purchase for Investment.</u>
 - 4.6.1 Each Lehman Designee is an "accredited investor" within the meaning of that term as defined in Rule 501(a) promulgated under the Securities Act.
 - 4.6.2 The ERPOP Equity Consideration and AVB Equity Consideration will be acquired for investment for each Lehman Designee's own account and not with a view to the distribution of any part thereof in violation of the Securities Act.
 - 4.6.3 Each Lehman Designee's financial condition is such that it is able to bear the risk of holding the ERPOP Equity Consideration and AVB Equity Consideration for an indefinite period of time.
 - 4.6.4 Each Lehman Designee (either alone or together with its advisors) has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of its investment in the ERPOP Equity Consideration and AVB Equity Consideration.
 - 4.6.5 Each Lehman Designee acknowledges that the ERPOP Equity Consideration and the AVB Equity Consideration have not been registered under the Securities Act or under any state or foreign securities laws.
- 4.7 <u>Legal Proceedings; Claims</u>. No Proceeding is pending nor, to LBHI's Knowledge, has any Proceeding been threatened in writing against any Lehman Entity or Seller, which seeks to restrain, prohibit or otherwise challenge legality or validity of this Agreement or any of the Contemplated Transactions.
- 4.8 Non-Reliance. In connection with the due diligence investigation of EQR, ERPOP and AVB, the Lehman Entities have received and may continue to receive from EQR, ERPOP and AVB certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding EQR, ERPOP and AVB, and their respective business and operations. LBHI, on behalf of itself and such other Lehman Entity

38

and/or other Lehman Designee, hereby acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, with which such Lehman Entity or Lehman Designee is familiar. Other than the representations and warranties set forth herein, such Lehman Entity or Lehman Designee acknowledges that (i) neither the Equity Residential Parties nor AVB make or have made any other representation or warranty, including any representation or warranty as to the prospects, financial or otherwise, of ERPOP, EQR or AVB, including with respect to the nature or extent of any of their Liabilities, the effectiveness or the success of any operations, any documents, projections, material or other information (financial or otherwise) regarding EQR, ERPOP and AVB furnished to Lehman Entities or their Representatives or made available to such Lehman Entities and their Representatives in any data room, management presentations or in any other form in expectation of, or in connection with, the Contemplated Transactions, or in respect of any other matter or thing whatsoever, and (ii) that any projections, estimates or forecasts of future results or events provided by or on behalf of EQR, ERPOP and AVB are subject to uncertainty and to the assumptions used in their preparation. LBHI, on behalf of Seller and the other Lehman Entities, has conducted, to its satisfaction, its own independent investigation of the condition, operations and business of ERPOP, EQR and AVB and has been provided access to and an opportunity to review any and all information respecting ERPOP, EQR and AVB requested by LBHI in order for Seller and the Lehman Entities to make their own determination to proceed with the Contemplated Transactions.

ARTICLE 5 REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in the Disclosure Schedules, LBHI and Seller, jointly and severally, represent and warrant to the Buyer Parties as follows:

- 5.1 Organization and Good Standing. Seller is a legal entity that is duly organized, validly existing and in good standing under the laws of the state of Delaware and has all requisite partnership power and authority to own, lease and operate its properties and to carry on its business as presently conducted. Seller is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except in such jurisdictions where the failure to be so duly qualified or licensed and in good standing has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- 5.2 <u>Authorization; Validity of Agreements</u>. Seller has the requisite partnership power and authority to execute and deliver each of the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution, delivery and performance by Seller of each of the Transaction Documents to which it is a

than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to the applicable Deferred Closing Asset subject to such Extension Closing), be duly executed and delivered by Seller, and constitutes, or will constitute at the Initial Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to the applicable Deferred Closing Asset subject to such Extension Closing), the legal, valid and binding obligation of Seller, enforceable against Seller in accordance with its terms, except that (a) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights generally, and (b) general equitable principles.

5.3 Consents and Approvals; No Violations. Except as set forth in Schedule 5.3, the execution, delivery and performance by Seller of each Transaction Document to which it is a party, and the consummation by Seller of the Contemplated Transactions, do not and will not (a) violate, contravene or conflict with any provision of any Organizational Documents of Seller; (b) except for any notifications required under any Tenant Purchase Option Law, require on the part of Seller any filing or registration with, notification to, or authorization, consent or approval of, any Governmental Authority; (c) result in the creation or imposition of any Lien on Seller other than Permitted Liens and restrictions on transfer by the Buyer Parties or Buyer Designees imposed under the Organizational Documents of Seller; and (d) contravene or conflict with or constitute a violation of any Orders or Laws applicable to or binding upon Seller; except, (A) in the case of clauses (b) and (d), for filings pursuant to the HSR Act or with respect to Transfer Taxes and/or the payment of any such Transfer Taxes, and (B) in the case of clauses (b), (c) and (d), as would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.

5.4 Ownership of Transferred Assets.

- Except as set forth in <u>Schedule 5.4.1</u>, Seller is the sole beneficial and record owner and has good, valid and marketable title to, and will, immediately prior to the Initial Closing, be the sole beneficial and record owner and have good, valid and marketable title to, all of the Primary Archstone Entities, free and clear of all Liens (other than any restrictions on transfer by any applicable Buyer Party or Buyer Designee imposed under applicable securities Laws and the Organizational Documents of the Archstone Entities, and Permitted Liens).
- Other than pursuant to the terms and conditions of this Agreement and the Organizational Documents of any Archstone Entity, (a) Seller is not a party to any option, warrant, purchase right or other contract that could require (i) any Archstone Entity to issue any ownership interest or other securities of such Archstone Entity, or (ii) Seller to sell, transfer or otherwise dispose of the Transferred Assets or Transferred Subsidiary Assets, and (b) Seller is not a party to any voting trust, agreement, proxy or other Contract with respect to the voting of any of the Transferred Interests or any other Equity Interests in any Archstone Entity.

40

5.5 <u>Brokerage Fees.</u> Except for the Persons set forth in <u>Schedule 5.5</u> (whose fees shall be paid by LBHI pursuant to <u>Section 13.10.2</u>), there is no investment banker, broker, finder or other intermediary who has been retained by or is authorized to act on behalf of Seller or any Archstone Entity who might be entitled to any fee or commission from Seller or any Archstone Entity in connection with the Contemplated Transactions.

5.6 <u>Financial Statements</u>.

- 5.6.1 Schedule 5.6.1 contains true, correct and complete copies of (a) the audited combined consolidated financial statements of Seller (or its predecessor) for each of the periods ended December 31, 2009, 2010 and 2011, (b) the unaudited combined consolidated financial statements of Seller for each of the six (6) month periods ended June 30, 2011 and 2012, and (c) the unaudited combined consolidated financial statements of Seller for each of the nine (9) month periods ended September 30, 2011 and 2012 (collectively, the "Financial Statements").
- The Financial Statements and notes thereto (a) have been prepared from the books and records of the entities disclosed in those financial statements, (b) have been prepared in accordance with GAAP, consistently applied throughout the periods presented, and (c) to Seller's Knowledge and LBHI's Knowledge, fairly present in all material respects the financial condition and results of operations of the reported entities for the periods presented.
- 5.7 <u>Absence of Dividends and Redemptions.</u> Between June 30, 2012 and the date of this Agreement, Seller has not (a) declared, set aside or paid any dividend or made any distribution with respect to its Equity Interests, or (b) redeemed, purchased, or otherwise acquired any of its Equity Interests.
- Non-Reliance. In connection with the due diligence investigation of EQR, ERPOP and AVB, Seller has received and may continue to receive from EQR, ERPOP and AVB certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding EQR, ERPOP and AVB, and their respective business and operations. Seller hereby acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, with which Seller is familiar. Other than the representations and warranties set forth herein, Seller acknowledges that (i) neither the Equity Residential Parties nor AVB make or have made any other representation or warranty, including any representation or warranty as to the prospects, financial or otherwise, of ERPOP, EQR or AVB, including with respect to the nature or extent of any of their Liabilities, the effectiveness or the success of any operations, any documents, projections, material or other information (financial or otherwise) regarding EQR, ERPOP and AVB furnished to Seller or its Representatives or made available to Seller or its Representatives in any data room, management presentations or in any other form in expectation of, or in connection with, the Contemplated Transactions, or in respect of any other matter or thing whatsoever, and (ii) that any projections, estimates or forecasts of future results or events provided by or on behalf of EQR, ERPOP and AVB are subject to uncertainty and to the assumptions used in their preparation. Seller has conducted, to its satisfaction, its own

ARTICLE 6 REPRESENTATIONS AND WARRANTIES OF LBHI AND SELLER CONCERNING THE ARCHSTONE ENTITIES

Except as set forth in the Disclosure Schedules, LBHI and Seller, jointly and severally, represent and warrant to the Buyer Parties as follows:

- 6.1 Organization and Good Standing.
- To Seller's Knowledge and LBHI's Knowledge, each Primary Archstone Entity is a legal entity that is duly organized, validly existing and in good standing under the laws of the jurisdiction of its formation and has all requisite corporate, company or other entity power and authority to own, lease and operate its properties and to carry on its business as presently conducted. To Seller's Knowledge and LBHI's Knowledge, each Archstone Entity is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except in such jurisdictions where the failure to be so duly qualified or licensed and in good standing has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- To Seller's Knowledge and LBHI's Knowledge, each Primary Archstone Entity has all requisite entity power and authority necessary to own, lease and operate its properties and assets and to conduct its business as currently conducted, except as has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- 6.2 [Intentionally deleted].
- 6.3 <u>Capitalization of the Archstone Entities.</u>
- 6.3.1 Schedule 6.3.1 sets forth (a) the name of each Primary Archstone Entity, (b) the number or percentage of the Equity Interests and voting interests of each class, series or type outstanding in each Primary Archstone Entity, (c) if an Archstone Entity or Joint Venture is not directly or indirectly wholly-owned by a Primary Archstone Entity, the aggregate ownership interest (not taking into account any distributions (whether priority, promote or otherwise) that are not made pro rata (assuming that all commitments under the applicable Contracts are fully funded)) of any Primary Archstone Entity in any other Archstone Entity or Joint Venture, and (d) if any real property in which any Archstone Entity has an ownership interest is not wholly-owned by

42

the Archstone Entities, the aggregate ownership interest (not taking into account any distributions (whether priority, promote or otherwise) that are not made pro rata) in such property that is not wholly-owned by the Archstone Entities, in each case, as of the date of this Agreement.

6.3.2 To Seller's Knowledge and LBHI's Knowledge, Schedule 6.3.2(a) sets forth, as of October 31, 2012, (i) for each Subsidiary of each Primary Archstone Entity the name and ownership chain of each such Subsidiary, and (ii) for each Person in which a Primary Archstone Entity owns, directly or indirectly, 50% or more of the Equity Interests and that is not otherwise a Subsidiary (such Person, a "Non-Controlled Subsidiary") the ownership interest of the Archstone Entities in such Non-Controlled Subsidiary. Since October 31, 2012, to Seller's Knowledge and LBHI's Knowledge, there has not been any material change to the ownership chain with respect to the Subsidiaries of each of the Primary Archstone Entities or to the ownership of the Archstone Entities in the Non-Controlled Subsidiaries. To Seller's Knowledge and LBHI's Knowledge, all outstanding Equity Interests and/or voting interests of each Subsidiary of a Primary Archstone Entity and each Non-Controlled Subsidiary owned by an Archstone Entity are duly authorized and validly issued and, if such Subsidiary or Non-Controlled Subsidiary is a corporation, are fully paid and non-assessable. Except as set forth in Schedule 6.3.2(b) or in the Organizational Documents of the Primary Archstone Entities, neither the Primary Archstone Entities nor, to Seller's Knowledge and LBHI's Knowledge, any of their Subsidiaries or any Non-Controlled Subsidiaries have (i) granted any options, warrants, rights or other securities convertible into, or exchangeable or exercisable for, Equity Interests and/or voting interests of a Subsidiary or Non-Controlled Subsidiary, or (ii) entered into any Contracts (x) relating to the redemption, exchange, issuance, sale, transfer or other disposition, voting or registration of Equity Interests and/or voting interests of a Subsidiary or Non-Controlled Subsidiary, or options, warrants, debt instruments, or other securities convertible into, or exchangeable or exercisable for, any of the foregoing, or (y) which grant stock appreciation rights, "phantom stock" or similar rights that give any Person the right to receive any benefits or rights similar to any rights enjoyed by or accruing to the holders of any Equity Interests and/or voting interests of a Subsidiary or Non-Controlled Subsidiary, in the case of clause (i) and clause (ii) to any Person other than (A) a Primary Archstone Entity or direct or indirect whollyowned Subsidiary thereof, and (B) the redemption rights and voting rights relating to the Preferred Units. Except for the Voting Agreement, there are no equityholder agreements, voting agreements, voting trusts or proxies with respect to the voting, or registration under the Securities Act of any Equity Interests and/or voting interests of any Subsidiary of a Primary Archstone Entity. The Primary Archstone Entities or the applicable Subsidiary of the Primary Archstone Entities identified in Schedule 6.3.2(a) are the sole record and beneficial owners of, and have good and valid title to, the Equity Interests and/or voting interests in such Subsidiary or Non-Controlled Subsidiary that are held of record by a Primary Archstone Entity or any other Subsidiary of

43

the Primary Archstone Entities (such Equity Interests and/or voting interests, the "<u>Subsidiary Interests</u>"), free and clear of all Liens, other than Permitted Liens. <u>Schedule 6.3.2(c)</u> sets forth as of the date of this Agreement, to Seller's Knowledge and LBHI's Knowledge, the name of each Person (other than any Person holding an interest in any Archstone Entity or Joint Venture to maintain REIT status) that holds any ownership interest in any Archstone Entity or Joint Venture that is not directly or indirectly wholly-owned by a Primary Archstone Entity.

- Books and Records. To Seller's Knowledge and LBHI's Knowledge, the books and records of the Archstone Entities have been maintained to reflect their assets and liabilities and in accordance with sound business practices, including the maintenance of a system of internal controls adequate for the size, operations and business of the Archstone Entities to provide reasonable assurances that (a) all transactions are executed in accordance with management's general or specific authorizations, (b) transactions are recorded as necessary to permit the preparation of financial statements in conformity with GAAP, (c) the recorded accountability for assets is compared with existing assets at reasonable intervals and appropriate action is taken with respect to any differences, (d) accounts, notes and other receivables and inventory are recorded accurately, and proper and adequate procedures are implemented to effect the collection of accounts, notes and other receivables on a current and timely basis, and (e) books and records are maintained in accordance with statutory records retention requirements. To Seller's Knowledge and LBHI's Knowledge, no material weaknesses in the Archstone Entities' internal controls have been identified by the Archstone Entities and their independent auditors in connection with the preparation and audit of the Financial Statements for the year ended December 31, 2011.
- 6.5 No Undisclosed Liabilities. The Archstone Entities do not have any Liabilities, except for (a) Liabilities reflected, disclosed in the notes, or reserved against in any of the Financial Statements; (b) Liabilities incurred since June 30, 2012 in the Ordinary Course of Business; (c) other Liabilities which would not reasonably be expected

to have, individually or in the aggregate, a Material Adverse Effect; and (d) Liabilities described in Schedule 6.5.

- 6.6 <u>Absence of Certain Changes or Events</u>. Since June 30, 2012, no change, event, occurrence or development has occurred that has had a Material Adverse Effect. No change, event, occurrence or development has occurred at any time since June 30, 2012 that, alone or in combination with another change, event, occurrence or development, would reasonably be expected to result in a Material Adverse Effect. Except as described in <u>Schedule 6.6</u>:
 - Between June 30, 2012 and the date of this Agreement (a) no Archstone Entity has declared, set aside or paid any dividend or made any distribution with respect to its Equity Interests (other than, in each case, to any other Archstone Entity); and (b) no Archstone Entity has redeemed, purchased, or otherwise acquired any of its Equity Interests, other than (i) with respect to a wholly-owned Subsidiary of a Primary Archstone Entity (ii) redemptions of Preferred Units of Archstone upon the exercise of put rights by the holders thereof, and (iii) dividends necessary to maintain REIT status with respect to the Archstone Entities that are REITs and are set forth on Schedule 6.6.1.

44

- 6.6.2 Between June 30, 2012 and the date of this Agreement, other than with respect to a direct or indirect wholly-owned Subsidiary, no Archstone Entity has adopted or commenced any plan of liquidation, dissolution, conversion or merger.
- 6.6.3 Between June 30, 2012 and the date of this Agreement, no Archstone Entity has changed in any material respect any of the accounting, reserving, underwriting, claims or actuarial methods, principles or practices, or working capital policies applicable to any Archstone Entity.
- Except as set forth in Schedule 6.6.4 or as described or set forth in any materials in the Data Room, to Seller's Knowledge and LBHI's Knowledge, between September 30, 2012 and the date of this Agreement, no Archstone Entity has taken any action that would have been prohibited by Sections 12.1.1 through 12.1.28 if such Sections 12.1.1 through 12.1.28 had been applicable to Archstone Entities during such period.
- 6.7 Taxes
- 6.7.1 Except as described in Schedule 6.7.1, (a) each Archstone Entity and its predecessors (i) has filed (or has had filed on its respective behalf) all material Tax Returns required to have been filed by or for it, and (ii) has paid (or had paid on its behalf) all material amounts of Taxes (whether or not shown on such Tax Returns) that are required to be paid by it and (b) there are no ongoing or pending material Proceedings, or to Seller's Knowledge and LBHI's Knowledge, material Proceedings threatened in writing, with respect to Taxes of any of the Archstone Entities, except as reflected or disclosed in the Financial Statements.
- 6.7.2 Schedule 6.7.2 sets forth each Archstone Entity that is or has been at any time in the last five (5) years classified as an association taxable as a corporation for U.S. federal income tax purposes; and any entity that has elected to be treated as a real estate investment trust under Section 856 of the Code at any time since 2007.
- 6.7.3 Schedule 6.7.3 sets forth (a) each Tax Protection Agreement pursuant to which any Archstone Entity is a party, (b) the estimated maximum amounts payable as a result of the taxable sale of each tax protected property (subject to the disclaimer included with respect thereto), and (c) the estimated amount of nonrecourse liabilities that must encumber each tax protected property to satisfy any requirements to maintain nonrecourse debt under the applicable Tax Protection Agreements (subject to the disclaimer included with respect thereto). As used herein, "Tax Protection Agreement" shall mean any written agreement to which any Archstone Entity is a party pursuant to which, in connection with the deferral of income Taxes of any party to such agreement (or any intended beneficiary of such agreement), any Archstone Entity (or its predecessor) has agreed to (w) maintain a minimum level of debt or continue

- a particular debt, (x) retain or not dispose of assets for a period of time, whether or not that period now has expired, (y) only dispose of assets in a particular manner, and/or (z) permit any party thereto to guarantee (or have guaranteed) debt of any Archstone Entity.
- 6.7.4 Each Archstone Entity and each of its Subsidiaries and Joint Ventures (other than the Subsidiaries or Joint Ventures listed in <u>Schedule 6.7.4</u>) that is treated as a partnership for U.S. federal income tax purposes has in effect an election pursuant to Section 754 of the Code.
- 6.7.5 Except as listed on <u>Schedule 6.7.5</u>, no sale of assets held by any Archstone Entity would cause any liability for Taxes pursuant to Section 1374 of the Code.
- 6.7.6 Seller has made available to the Buyer Parties true, correct and complete copies of the material federal income Tax Returns for each of the Archstone Entities for the prior three (3) taxable years of each such entity with respect to which Tax Returns have been filed.
- 6.8 <u>Compliance with Laws and Orders.</u>
- To Seller's Knowledge and LBHI's Knowledge, each Archstone Entity is and has been at all times since June 30, 2012, in compliance with all Laws and Orders applicable to it or to its properties, assets or business, except where the failure to be in compliance with such Laws or Orders would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- Neither Seller nor LBHI has, and, to Seller's Knowledge and LBHI's Knowledge, no other Person acting on its behalf has: (a) used funds of any of the Archstone Entities for unlawful contributions, gifts or entertainment or other unlawful payments relating to political activity; (b) made a payment to a foreign or domestic government official or employee, or to a foreign or domestic political party or campaign in violation of any Law on behalf of any Archstone Entity; (c) violated the Foreign Corrupt Practices Act with respect to any Archstone Entity; (d) made to or received from any Person, private or public, regardless of form, whether in money, property or services, a contribution, gift, bribe, rebate, payoff, influence payment, kickback or other payment (i) to unlawfully obtain favorable treatment in securing business, (ii) to unlawfully pay for favorable treatment for business secured, (iii) to unlawfully obtain or provide special concessions or for special concessions already obtained or provided, or (iv) that otherwise was in violation of any Law, in each case for or concerning any Archstone Entity or any Affiliate of the Archstone Entities.

6.9.1

Entity, which seeks to restrain, prohibit or otherwise challenge legality or validity of this Agreement or any of the Contemplated Transactions.

- 6.9.2 To Seller's Knowledge and LBHI's Knowledge, except as set forth on Schedule 6.9.2, (i) no Proceeding is pending to which any Archstone Entity is a party, and (ii) no Proceeding has been threatened in writing against any Archstone Entity or any current managing member or any employee or agent of any Archstone Entity or against any former managing member, employee or agent of any Archstone Entity, with respect to which any Archstone Entity has or would reasonably be expected to have a Liability therefor or any indemnification or expense advancement obligation, in either case, except to the extent such Proceeding has not had and would not reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect.
- 6.9.3 To Seller's Knowledge and LBHI's Knowledge, there are no Proceedings either pending against or threatened in writing against, the Archstone Entities under any bankruptcy or insolvency Law or with respect to any Bankruptcy Event.
- 6.10 Executive Compensation. To Seller's Knowledge and LBHI's Knowledge, Schedule 6.10 sets forth a true, correct and complete list of (a) each Employee Benefit Plan and Executive Employment Arrangement that is applicable to any senior executive officer of any of Archstone, Archstone Communities LLC or any of the Archstone Entities, including any president, chief executive officer, chief financial officer, chief accounting officer, chief investment officer and chief operating officer (or positions with similar responsibilities), and (b) all Contracts relating to employment, severance, change-of-control or retention bonus obligations. Except as set forth in Schedule 6.10, to Seller's Knowledge and LBHI's Knowledge, no Archstone Entity other than Archstone Communities LLC directly employs any Person. Except as set forth in Schedule 6.10, no Archstone Entity is a party to or has granted any long-term incentive plan awards payable during fiscal year 2012 or any subsequent fiscal year, including pursuant to the Development Incentive Plan, Operational EVP Incremental Bonus Program, Acquisition/Disposition Bonus Program, Performance Participation Program, Cash in Lieu of Stock Options program, Annual Target Bonus Program or otherwise.
- 6.11 <u>Brokerage Fees.</u> Except for Persons set forth on <u>Schedule 6.11</u> (whose fees shall be paid by LBHI pursuant to <u>Section 13.10.2</u>), there is no investment banker, broker, finder or other intermediary who has been retained by or is authorized to act on behalf of any Archstone Entity who might be entitled to any fee or commission from an Archstone Entity in connection with the Contemplated Transactions.
- 6.12 No Other Representations and Warranties. Except for the representations and warranties made by Seller and LBHI in this Article 6, Seller and LBHI do not make any other representation or warranty with respect to the Archstone Entities or their respective businesses, operations, assets, liabilities, condition (financial or otherwise) or prospects, notwithstanding the delivery or disclosure to the Buyer Parties or any of their Affiliates or Representatives of any documentation, forecasts or other information with respect to any one of more of the foregoing.

47

ARTICLE 7 REPRESENTATIONS AND WARRANTIES OF THE EQUITY RESIDENTIAL PARTIES

The Equity Residential Parties hereby, jointly and severally, represent and warrant to Seller and the Lehman Entities as follows:

- 7.1 Organization and Good Standing. ERPOP is a limited partnership duly organized, validly existing and in good standing under the laws of the state of Illinois and has all requisite trust power and authority to own, lease and operate its properties and to carry on its business as it is presently being conducted. EQR is a real estate investment trust that is duly organized, validly existing and in good standing under the laws of the state of Maryland and has all requisite trust power and authority to own, lease and operate its properties and to carry on its business as it is presently being conducted. Each of the Equity Residential Parties is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except for those jurisdictions where failure to be so qualified would not reasonably be expected to have, individually or in the aggregate, a Sponsor Material Adverse Effect.
- 7.2 Authorization; Validity of Agreements. Each of the Equity Residential Parties has the requisite trust power and authority to execute and deliver each of the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution, delivery and performance by the Equity Residential Parties of each of the Transaction Documents to which it is a party, and the consummation by the Equity Residential Parties of the Contemplated Transactions, have been duly authorized by, and no other proceedings, actions or authorizations on the part of any Equity Residential Party or any holders of Equity Interest in it are necessary to authorize the execution and delivery by each Equity Residential Party of any Transaction Document to which it is a party or the consummation by the Equity Residential Parties of the Contemplated Transactions. Each of the Transaction Documents to which either of the Equity Residential Parties is a party has been duly executed and delivered by such Equity Residential Party, or will at the Initial Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing, the legal, valid and binding obligation of such Equity Residential Party, enforceable against such Equity Residential Party in accordance with its terms, except that (a) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights generally, and (b) general equitable principles.
- 7.3 <u>Consents and Approvals; No Violations.</u> The execution, delivery and performance by each of the Equity Residential Parties of each Transaction Document to which it is a party, and the consummation by the Equity Residential Parties of the Contemplated Transactions (including delivery of the ERPOP Equity Consideration), do not and will not (a)

the Equity Residential Parties or any of their respective material properties or assets; or (d) except for filings pursuant to the HSR Act, require on the part of any Equity Residential Party any filing or registration with, notification to, or authorization, consent or approval of, any Governmental Authority, except, in the cases of clauses (b) through (d) above, as would not reasonably be expected to have, individually or in the aggregate, a Sponsor Material Adverse Effect.

7.4 <u>Capitalization</u>.

- As of September 30, 2012, the authorized capital stock of EQR consists of (i) 100,000,000 authorized preferred shares, par value \$.01 per share (the "EQR Preferred Shares"), of which, as of the close of business on September 30, 2012, 1,000,000 shares were issued and outstanding, and (ii) 1,000,000,000 authorized common shares, par value \$.01 per share (the "EQR Common Shares") together with the EQR Preferred Shares, the "EQR Shares"), of which, as of the close of business on September 30, 2012, 302,674,716 EQR Common Shares were issued and outstanding, 8,213,332 EQR Common Shares were reserved for issuance under awards previously granted and outstanding pursuant to its compensation plans, an additional 14,289,029 EQR Common Shares were available for new awards to be made subsequent to such date under its compensation plans, and an additional 14,399,790 EQR Common Shares were potentially issuable upon redemptions of limited partnership units in ERPOP. Since September 30, 2012, no EQR Shares have been issued except for EQR Common Shares that were reserved for issuance as of September 30, 2012 under its compensation plans and EQR Common Shares issued upon redemption of limited partnership units in ERPOP. All of the EQR Shares have been duly authorized, and all EQR Shares that are issued and outstanding are validly issued, fully paid and non-assessable, and free of preemptive rights.
- All of the issued and outstanding EQR Shares were issued in compliance with applicable Laws. None of the issued and outstanding EQR Shares were issued in violation of any agreement, arrangement or commitment to which EQR is a party or is subject to, or in violation of any preemptive or similar rights of any Person. Except as set forth in Section 7.4.1 or Schedule 7.4.2, as of the date hereof, (i) EQR has not issued and is not bound by any outstanding subscriptions, options, warrants, calls, convertible or exchangeable securities, rights, commitments or agreements of any character providing for the issuance or disposition of any shares of capital stock, voting securities or Equity Interests of EQR or its Subsidiaries and (ii) there are no outstanding

49

obligations of EQR or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock, voting securities or equity interests (or any options, warrants or other rights to acquire any shares of capital stock, voting securities or equity interests) of EQR or any of its Subsidiaries.

7.5 <u>SEC Reports and Financial Statements.</u>

- 7.5.1 EQR has filed or furnished all forms, documents and reports required to be filed or furnished by it with the SEC since January 1, 2012 (the "EQR SEC Documents"). As of their respective dates, or, if amended, as of the date of such amendment, the EQR SEC Documents complied in all material respects with the requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act, as the case may be, and the applicable rules and regulations promulgated thereunder, and none of the EQR SEC Documents contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 7.5.2 The audited consolidated financial statements and unaudited interim consolidated financial statements (including all related notes and schedules) of EQR included in the EQR SEC Documents complied as to form in all material respects with the rules and regulations of the SEC then in effect, fairly present in all material respects the consolidated financial position of EQR and its consolidated Subsidiaries, as of the respective dates thereof, and the consolidated results of their operations and their consolidated cash flows for the respective periods then ended (subject, in the case of the unaudited statements, to normal recurring year-end audit adjustments that were not or are not expected to be, individually or in the aggregate, materially adverse to EQR or ERPOP), and were prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto).

7.6 <u>Purchase for Investment</u>.

- 7.6.1 ERPOP is an "accredited investor" within the meaning of that term as defined in Rule 501(a) promulgated under the Securities Act.
- 7.6.2 The Transferred Interests will be acquired for investment for ERPOP's own account and not with a view to the distribution of any part thereof in violation of the Securities Act.
- 7.6.3 ERPOP's financial condition is such that it is able to bear the risk of holding the Transferred Interests for an indefinite period of time.

50

- 7.6.4 ERPOP (either alone or together with its advisors) has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of its investment in the Transferred Interests.
- 7.6.5 ERPOP acknowledges that the Transferred Interests and the Equity Interests in the Archstone Entities have not been registered under the Securities Act or under any state or foreign securities laws.

7.7 <u>Financing</u>.

- 7.7.1 ERPOP will have, at or prior to the Initial Closing, sufficient liquid funds (through cash on hand, credit arrangements or otherwise) to (a) pay all amounts payable in cash by ERPOP at the Initial Closing hereunder, and (b) make all other necessary payments of fees and expenses payable by ERPOP at the Initial Closing in connection with the Contemplated Transactions.
- ERPOP has provided to Seller a true, complete and correct copy of an executed commitment letter (the "ERPOP Debt Commitment Letter") from Morgan Stanley Senior Funding Inc. (together with its successors and assigns under the ERPOP Debt Commitment Letter, the "ERPOP Lenders") pursuant to which, and subject to the terms and conditions thereof, the ERPOP Lenders have committed to provide ERPOP with financing in an aggregate amount of \$2,500,000,000 (the "ERPOP Debt Financing"). The ERPOP Debt Commitment Letter is a legal, valid and binding obligation of the ERPOP and, to the knowledge of ERPOP, each of the other parties thereto. The ERPOP Debt Commitment Letter has not been amended or modified prior to the date of this Agreement, and as of the date hereof, the respective commitments contained in the ERPOP Debt Commitment Letter have not been withdrawn or rescinded in any respect. As of the date hereof, the ERPOP Debt Commitment Letter is in full force and effect and is valid and, to

ERPOP's actual knowledge, is enforceable against the parties thereto in accordance with its terms. Except for the payment of customary fees, there are no conditions precedent, side agreements or other contingencies related to the funding of the full amount of the ERPOP Debt Financing, other than as set forth in or contemplated by the ERPOP Debt Commitment Letter. As of the date hereof, no event has occurred which, with or without notice, lapse of time or both, would constitute a default or breach on the part of ERPOP, and to the knowledge of ERPOP, any other parties thereto, under the ERPOP Debt Commitment Letter. As of the date hereof, and assuming the conditions set forth in Section 10.1 hereof are satisfied, ERPOP reasonably believes that it will be able to satisfy the material conditions to the ERPOP Debt Financing contemplated by the ERPOP Debt Commitment Letter and that the ERPOP Debt Financing will be made available to ERPOP on the Initial Closing Date. ERPOP will provide to Seller any amendments to the ERPOP Debt Commitment Letter, or any written notices given pursuant thereto, as promptly as possible. ERPOP has fully paid or caused to be fully paid any and all commitment fees or other fees that are payable by it or EQR under the

51

ERPOP Debt Commitment Letter and the fee letters related thereto on or prior to the date hereof and has otherwise satisfied all other terms and conditions under the ERPOP Debt Commitment Letter required to be satisfied by it or EQR on or prior to the date hereof.

- 7.7.3 ERPOP expressly acknowledges that its obligations hereunder are not subject to any conditions, express or implied, regarding ERPOP's ability to obtain financing (or to obtain financing on terms acceptable to ERPOP) for the consummation of the Contemplated Transactions.
- 7.8 <u>Valid Issuance of Shares</u>. EQR has, and shall have at the Initial Closing, a sufficient number of EQR Common Shares authorized but unissued for the purpose of contributing the ERPOP Equity Consideration to ERPOP so that ERPOP may deliver the ERPOP Equity Consideration to Seller (or, if so directed by Seller, to LBHI and/or any Lehman Designee) at the Initial Closing and any Extension Closing (in each case after giving effect to all anti-dilution adjustments). All EQR Common Shares that will be delivered as the ERPOP Equity Consideration to Seller (or, if so directed by Seller, to LBHI and/or any Lehman Designee) shall have been duly authorized and validly issued and, upon delivery to ERPOP or the applicable Buyer Designees of the Transferred Assets and the Transferred Subsidiary Assets to be acquired by them, shall be fully paid and non-assessable, and shall be free from preemptive rights and free of any Liens or restrictions on transfer other than restrictions under the Transaction Documents, under EQR's Organizational Documents and under applicable state and federal securities Laws. No vote of the holders of the EQR Shares or the board of trustees of EQR is necessary to approve the issuance of the EQR Common Shares for the payment of the ERPOP Equity Consideration or the entry into the Contemplated Transactions.
- 7.9 <u>Brokerage Fees.</u> Other than Morgan Stanley, whose fees will be paid by ERPOP, there is no investment banker, broker, finder or other intermediary who has been retained by or is authorized to act on behalf of any Equity Residential Party who might be entitled to any fee or commission in connection with the Contemplated Transactions.
- 7.10 Non-Reliance. In connection with the due diligence investigation of Seller and the Archstone Entities by the Equity Residential Parties, the Equity Residential Parties have received and may continue to receive from the Lehman Entities and Seller certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Archstone Entities, their business and operations, the Transferred Assets and the Transferred Subsidiary Assets. The Equity Residential Parties hereby acknowledge that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, with which the Equity Residential Parties are familiar. Other than the representations and warranties set forth herein, the Equity Residential Parties acknowledge that (i) the Lehman Entities and Seller make and have made no representation or warranty, including any representation or warranty as to the prospects, financial or otherwise, of the Archstone Entities, including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any Liabilities, the effectiveness or the success of any operations, or the accuracy or completeness of any registration statement filed prior to the date hereof, any confidential information memoranda, documents, projections, material or other information

52

(financial or otherwise) regarding the Archstone Entities, the Transferred Assets and the Transferred Subsidiary Assets furnished to the Equity Residential Parties or their Representatives or made available to such Equity Residential Party and its Representatives in any Data Room, management presentations or in any other form in expectation of, or in connection with, the Contemplated Transactions, or in respect of any other matter or thing whatsoever, and (ii) that any projections, estimates or forecasts of future results or events provided by or on behalf of the Lehman Entities and Seller are subject to uncertainty and to the assumptions used in their preparation. The Equity Residential Parties acknowledge that the Equity Residential Parties are acquiring the Transferred Assets and the Transferred Subsidiary Assets on an "AS IS, WHERE IS" basis, "with all faults," subject only to (x) the specific representations and warranties set forth in Articles 4, 5 and 6, which representations and warranties shall survive the Initial Closing in accordance with Section 14.1, and (y) the indemnification obligations set forth in Article 14 hereof. Notwithstanding anything to the contrary herein, neither Seller nor LBHI are making any representations or warranties with respect to any (x) consent or approval that may be required to be obtained from any Person (other than a Governmental Authority) in connection with this Agreement or the Contemplated Transactions or with respect to any Lien that may be imposed on any asset of Seller or the Archstone Entities as a result of, or in connection with, this Agreement or the Contemplated Transactions, and (y) any Tenant Purchase Option Laws. The lack of any other representations and warranties relating to the Archstone Entities, Transferred Assets and the Transferred Subsidiary Assets was specifically bargained-for among the Equity Residential Parties, Seller and the Lehman Entities and was taken into account by the Equity Residential Parties, Seller and the Lehman Entities in arriving at the Purchase Price. The Equity Residential Parties have conducted, to their satisfaction, their own independent investigation of the condition, operations and business of the Archstone Entities, Transferred Assets and the Transferred Subsidiary Assets and the Equity Residential Parties have been provided access to and an opportunity to review any and all information respecting the Archstone Entities, Transferred Assets and the Transferred Subsidiary Assets requested by the Equity Residential Parties in order for the Equity Residential Parties to make their own determination to proceed with the Contemplated Transactions.

7.11 <u>Solvency</u>. Each of the Equity Residential Parties is solvent for all purposes under federal bankruptcy and applicable state fraudulent transfer and fraudulent conveyance Laws. The Contemplated Transactions will not render the Equity Residential Parties insolvent for such purposes and do not constitute a fraudulent transfer or conveyance under such Laws.

ARTICLE 8 REPRESENTATIONS AND WARRANTIES OF AVB

AVB represents and warrants to Seller and the Lehman Entities as follows:

8.1 Organization and Good Standing. AVB is a corporation that is duly organized, validly existing and in good standing under the laws of the state of Maryland and has all requisite corporate power and authority to own, lease and operate its properties and to carry on its business as it is presently being conducted. AVB is duly qualified or licensed to do business and is in good standing in each jurisdiction in which the property owned, leased or operated by it or the nature of the business conducted by it makes such qualification or licensing necessary, except

for those jurisdictions where failure to be so qualified would not reasonably be expected to have, individually or in the aggregate, a Sponsor Material Adverse Effect.

- 8.2 <u>Authorization; Validity of Agreements.</u> AVB has the requisite corporate power and authority to execute and deliver each of the Transaction Documents to which it is a party and to consummate the Contemplated Transactions. The execution, delivery and performance by AVB of each of the Transaction Documents to which it is a party, and the consummation by AVB of the Contemplated Transactions, have been duly authorized by, and no other proceedings, actions or authorizations on the part of AVB or any holders of Equity Interests in it are necessary to authorize the execution and delivery by AVB of any Transaction Document to which it is a party or the consummation by AVB of the Contemplated Transactions. Each of the Transaction Documents to which AVB is a party has been duly executed and delivered by AVB, or will at the Initial Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to all assets other than any Deferred Closing Assets for which an Extension Closing will occur pursuant to Section 2.5) and at any Extension Closing (with respect to the applicable Deferred Closing Asset subject to such Extension Closing), the legal, valid and binding obligation of AVB, enforceable against AVB in accordance with its terms, except that (a) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights generally, and (b) general equitable principles.
- 8.3 Consents and Approvals; No Violations. The execution, delivery and performance by AVB of each Transaction Document to which it is a party, and the consummation by AVB of the Contemplated Transactions (including delivery of the AVB Equity Consideration), do not and will not (a) violate, contravene or conflict with any provision of any Organizational Documents of AVB; (b) result in a breach of, or constitute (with or without due notice or lapse of time or both) a default or give rise to any right of termination, cancellation or acceleration under, any of the terms, conditions or provisions of any material note, bond, mortgage, indenture, lease, license, Contract, agreement or other instrument or obligation to which AVB is a party or by which AVB, or any of its properties or assets may be bound; (c) violate, contravene or conflict with any material Orders or Laws applicable to AVB or any of its material properties or assets; or (d) except for filings pursuant to the HSR Act, require on the part of AVB any filing or registration with, notification to, or authorization, consent or approval of, any Governmental Authority, except, in the cases of clauses (b) through (d) above, as would not reasonably be expected to have, individually or in the aggregate, a Sponsor Material Adverse Effect.

8.4 <u>Capitalization</u>.

As of September 30, 2012, the authorized capital stock of AVB consists of (a) 50,000,000 shares of authorized preferred stock, par value \$.01 per share (the "AVB Preferred Shares"), of which no shares are issued and outstanding, and (b) 140,000,000 authorized shares of common stock, par value \$.01 per share (the "AVB Common Shares" together with the AVB Preferred Shares, the "AVB

54

Shares") of which, as of the close of business on September 30, 2012, 97,705,713 were issued and outstanding, 1,146,012 AVB Common Shares were reserved for issuance with respect to awards previously made and remaining outstanding under its compensation plans, an additional 3,381,318 AVB Common Shares were reserved for awards to be made subsequent to such date under its compensation plans and an additional 7,500 AVB Common Shares were reserved for issuance upon redemption of units of limited partnership. Since September 30, 2012, no AVB Shares have been issued except for AVB Common Shares that were reserved for issuance as of September 30, 2012 under its compensation plans. All of the AVB Shares have been duly authorized, and the issued and outstanding AVB Shares are validly issued, fully paid and non-assessable, and free of preemptive rights.

All of the issued and outstanding AVB Common Shares were issued in compliance with applicable Laws. None of the issued and outstanding AVB Common Shares were issued in violation of any agreement, arrangement or commitment to which AVB is a party or is subject to, or in violation of any preemptive or similar rights of any Person. Except as set forth in Schedule 8.4.2, as of the date hereof, (i) AVB has not issued and is not bound by any outstanding subscriptions, options, warrants, calls, convertible or exchangeable securities, rights, commitments or agreements of any character providing for the issuance or disposition of any shares of capital stock, voting securities or Equity Interests of AVB or its Subsidiaries and (ii) there are no outstanding obligations of AVB or any of its Subsidiaries to repurchase, redeem or otherwise acquire any shares of capital stock, voting securities or equity interests (or any options, warrants or other rights to acquire any shares of capital stock, voting securities or equity interests) of AVB or any of its Subsidiaries.

8.5 SEC Reports and Financial Statements.

- 8.5.1 AVB has filed or furnished all forms, documents and reports required to be filed or furnished by it with the SEC since January 1, 2012 (the "AVB SEC Documents"). As of their respective dates, or, if amended, as of the date of such amendment, the AVB SEC Documents complied in all material respects with the requirements of the Securities Act, the Exchange Act and the Sarbanes-Oxley Act, as the case may be, and the applicable rules and regulations promulgated thereunder, and none of the AVB SEC Documents contained any untrue statement of a material fact or omitted to state any material fact required to be stated therein or necessary to make the statements therein, in light of the circumstances under which they were made, not misleading.
- 8.5.2 The audited consolidated financial statements and unaudited interim consolidated financial statements (including all related notes and schedules) of AVB included in the AVB SEC Documents complied as to form in all material respects with the rules and regulations of the SEC then in effect,

55

fairly present in all material respects the consolidated financial position of AVB and its consolidated Subsidiaries, as at the respective dates thereof, and the consolidated results of their operations and their consolidated cash flows for the respective periods then ended (subject, in the case of the unaudited statements, to normal recurring year-end audit adjustments that were not or are not expected to be, individually or in the aggregate, materially adverse to AVB), and were prepared in accordance with GAAP applied on a consistent basis during the periods involved (except as may be indicated therein or in the notes thereto).

8.6 <u>Purchase for Investment.</u>

- 8.6.1 AVB is an "accredited investor" within the meaning of that term as defined in Rule 501(a) promulgated under the Securities Act.
- 8.6.2 The Transferred Interests will be acquired for investment for AVB's own account and not with a view to the distribution of any part thereof in violation of the Securities Act.

- 8.6.3 AVB's financial condition is such that it is able to bear the risk of holding the Transferred Interests for an indefinite period of time.
- 8.6.4 AVB (either alone or together with its advisors) has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risks of its investment in the Transferred Interests.
- 8.6.5 AVB acknowledges that the Transferred Interests and the Equity Interests in the Archstone Entities have not been registered under the Securities Act or under any state or foreign securities laws.

8.7 <u>Financing</u>.

- 8.7.1 AVB will have, at or prior to the Initial Closing, sufficient liquid funds (through cash on hand, credit arrangements or otherwise) to (a) pay all amounts payable in cash by AVB at the Initial Closing hereunder, and (b) make all other necessary payments of fees and expenses payable by AVB at the Initial Closing in connection with the Contemplated Transactions.
- 8.7.2 AVB has provided to Seller a true, complete and correct copy of an executed commitment letter (the "AVB Debt Commitment Letter" and, together with the ERPOP Debt Commitment Letter, the "Debt Commitment Letters") from Goldman Sachs Lending Partners LLC (together with its successors and assigns under the AVB Debt Commitment Letter, the "AVB Lenders") pursuant to which, and subject to the terms and conditions thereof, the AVB Lenders have committed to provide AVB with financing in an aggregate amount of up to \$2,200,000,000 (the "AVB Debt Financing" and, together with the ERPOP Debt Financing, the "Debt Financing"). The AVB Debt Commitment Letter is a legal, valid and binding obligation of the AVB and, to

56

the knowledge of AVB, each of the other parties thereto. The AVB Debt Commitment Letter has not been amended or modified prior to the date of this Agreement, and as of the date hereof, the respective commitments contained in the AVB Debt Commitment Letter have not been withdrawn or rescinded in any respect. As of the date hereof, the AVB Debt Commitment Letter is in full force and effect and is valid, and, to AVB's actual knowledge, is enforceable against the parties thereto in accordance with its terms. Except for the payment of customary fees, there are no conditions precedent, side agreements or other contingencies related to the funding of the full amount of the AVB Debt Financing, other than as set forth in or contemplated by the AVB Debt Commitment Letter. As of the date hereof, no event has occurred which, with or without notice, lapse of time or both, would constitute a default or breach on the part of AVB, and to the knowledge of AVB, any other parties thereto, under the AVB Debt Commitment Letter. As of the date hereof, and assuming the conditions set forth in Sections 10.1 hereof are satisfied, AVB reasonably believes that it will be able to satisfy the material conditions to the AVB Debt Financing contemplated by the AVB Debt Commitment Letter and that AVB Debt Financing will be made available to AVB on the Initial Closing Date. AVB will provide to Seller any amendments to the AVB Debt Commitment Letter, or any written notices given pursuant thereto, as promptly as possible. AVB has fully paid or caused to be fully paid any and all commitment fees or other fees that are payable by it under the AVB Debt Commitment Letter and the fee letters related thereto on or prior to the date hereof and has otherwise satisfied all other terms and conditions under the AVB Debt Commitment Letter required to be satisfied by it on or prior to the date hereof.

- 8.7.3 AVB expressly acknowledges that its obligations hereunder are not subject to any conditions, express or implied, regarding AVB's ability to obtain financing (or to obtain financing on terms acceptable to AVB) for the consummation of the Contemplated Transactions.
- 8.8 Valid Issuance of Shares. AVB has, and shall have at the Initial Closing, a sufficient number of AVB Common Shares authorized but unissued for the purpose of issuing the AVB Equity Consideration to Seller (or, if so directed by Seller, to LBHI and/or any Lehman Designee) at the Initial Closing and any Extension Closing (in each case after giving effect to all anti-dilution adjustments). All AVB Common Shares that will be delivered as AVB Equity Consideration to Seller (or, if so directed by Seller, to LBHI and/or any Lehman Designee) shall have been duly authorized and validly issued and, upon delivery to AVB or the applicable Buyer Designees of the Transferred Assets and the Transferred Subsidiary Assets to be acquired by them, shall be fully paid and non-assessable, and shall be free from preemptive rights and free of any Liens or restrictions on transfer other than restrictions under the Transaction Documents, under AVB's Organizational Documents and under applicable state and federal securities Laws. No vote of the holders of the AVB Shares or the board of directors of AVB is necessary to approve the issuance of the AVB Common Shares for the payment of the AVB Equity Consideration or the entry into the Contemplated Transactions.

- 8.9 <u>Brokerage Fees.</u> Other than Greenhill and Goldman Sachs, whose fees will be paid by AVB, there is no investment banker, broker, finder or other intermediary who has been retained by or is authorized to act on behalf of AVB who might be entitled to any fee or commission in connection with the Contemplated Transactions.
- Non-Reliance. In connection with the due diligence investigation of Seller and the Archstone Entities by AVB, AVB has received and may continue to receive from the Lehman Entities and Seller certain estimates, projections, forecasts and other forward-looking information, as well as certain business plan information, regarding the Archstone Entities, their business and operations, the Transferred Assets and the Transferred Subsidiary Assets. AVB hereby acknowledges that there are uncertainties inherent in attempting to make such estimates, projections, forecasts and other forward-looking statements, as well as in such business plans, with which AVB is familiar. Other than the representations and warranties set forth herein, AVB acknowledges that (i) the Lehman Entities and Seller make and have made no representation or warranty, including any representation or warranty as to the prospects, financial or otherwise, of the Archstone Entities, including with respect to merchantability or fitness for any particular purpose of any assets, the nature or extent of any Liabilities, the effectiveness or the success of any operations, or the accuracy or completeness of any registration statement filed prior to the date hereof, any confidential information memoranda, documents, projections, material or other information (financial or otherwise) regarding the Archstone Entities, the Transferred Assets and the Transferred Subsidiary Assets furnished to AVB or its Representatives or made available to AVB and its Representatives in any Data Room, management presentations or in any other form in expectation of, or in connection with, the Contemplated Transactions, or in respect of any other matter or thing whatsoever, and (ii) that any projections, estimates or forecasts of future results or events provided by or on behalf of the Lehman Entities and Seller are subject to uncertainty and to the assumptions used in their preparation. AVB acknowledges that AVB is acquiring the Transferred Assets and the Transferred Subsidiary Assets on an "AS IS, WHERE IS" basis, "with all faults," subject only to (x) the specific representations and warranties set forth in Articles 4, 5 and 6, which representations and warranties shall survive the Initial Closing in accordance with Section 14.1, and (y) the indemnification obligations set forth in Article 14 hereof. Notwithstanding anything to the contrary herein, neither Seller nor LBHI are making any representations or warranties with respect to (x) any consent or approval that may be required to be obtained from any Person (other than a Governmental Authority) in connection with this Agreement or the Contemplated Transactions or with respect to any Lien that may be imposed on any asset of Seller or the Archstone Entities as a result of, or in connection with, this Agreement or the Contemplated Transactions, and (y) any Tenant Purchase Option Laws. The lack of any other representations and warranties relating to the Archstone Entities, Transferred Assets and the Transferred Subsidiary Assets was specifically bargained-for among AVB, Seller and the Lehman Entities and was taken into account by AVB, Seller and the Lehman Entities in arriving at the Purchase Price. AVB has conducted, to its satisfaction, its own independent investigation of the condition, operations and business of the Archstone Entities, Transferred Assets and the Transferred Subsidiary Assets and

58

8.11 <u>Solvency.</u> AVB is solvent for all purposes under federal bankruptcy and applicable state fraudulent transfer and fraudulent conveyance Laws. The Contemplated Transactions will not render AVB insolvent for such purposes and do not constitute a fraudulent transfer or conveyance under such Laws.

ARTICLE 9 CONDITIONS TO EACH PARTY'S OBLIGATION TO CLOSE

The respective obligations of each of the Parties to consummate the Contemplated Transactions at the Initial Closing or any Extension Closing are subject to the satisfaction or (to the extent permitted by applicable Law) waiver in writing by each Party, in their respective sole and absolute discretion, at or prior to the Initial Closing or any Extension Closing, as applicable, of each of the following conditions set forth in this Article 9.

- 9.1 <u>Competition Matters.</u> Any applicable waiting period (and any extension thereof) under any Antitrust Law relating to the Contemplated Transactions shall have expired or been terminated and any necessary approvals under any Antitrust Law shall have been obtained.
- 9.2 No Injunction or Illegality. There must not be in effect any Order or Orders, whether temporary, preliminary or permanent, issued by any Governmental Authority restraining, enjoining, preventing or prohibiting the consummation of (a) with respect to the Initial Closing, a Substantial Portion of the Contemplated Transactions, and (b) with respect to any Extension Closing, the Deferred Closing Assets subject to such Extension Closing. There must not be (i) with respect to the Initial Closing, any Law in effect that makes the consummation of a Substantial Portion of the Contemplated Transactions illegal or (ii) with respect to any Extension Closing, any Law in effect that makes the consummation of the transfer of the applicable Deferred Closing Assets in accordance with Section 2.6 illegal. The Parties acknowledge and agree that, to the extent that (x) any Order or Orders whether temporary, preliminary or permanent, issued by any Governmental Authority exists as of the Initial Closing restraining, enjoining, preventing or prohibiting the consummation of any of the Contemplated Transactions (including the transfer of any assets to the Buyer Parties in accordance with Schedule 2.6), but such Orders do not, in the aggregate, apply to a Substantial Portion of the Contemplated Transactions (including the transfer of any assets to the Buyer Parties in accordance with Schedule 2.6) illegal, but does not make a Substantial Portion of the Contemplated Transactions (including the transfer of any assets to the Buyer Parties in accordance with Schedule 2.6) illegal, but does not make a Substantial Portion of the Contemplated Transactions illegal, then, in each case ((x) and (y)) the assets subject to any such illegal, prohibited, restrained, prevented or enjoined Contemplated Transaction shall be treated as "Deferred Closing Assets" for purposes of this Agreement and shall be subject to the obligations and procedures set forth in Section 2.5 (except that the "Outside Deferred Closing Date" with respect to any such assets

50

ARTICLE 10 CONDITIONS TO THE BUYER PARTIES' OBLIGATION TO CLOSE

10.1 <u>Conditions to the Initial Closing.</u> The Buyer Parties' obligation to consummate the Contemplated Transactions at the Initial Closing and to take the other actions contemplated by this Agreement at the Initial Closing are subject to the satisfaction or (to the extent permitted by applicable Law) waiver in writing by the Buyer Parties, in their sole and absolute discretion, at or prior to the Initial Closing, of each of the following conditions set forth in this Section 10.1.

10.1.1 <u>Accuracy of Representations and Warranties.</u>

Each of the Seller Fundamental Representations shall be true and correct as of the date of this Agreement and as of the Initial Closing Date as if (a) made at and as of such date (unless such representations were expressly made as of an earlier date, in which case such representations shall be true and correct on and as of such earlier date), except for any inaccuracies of the Seller Fundamental Representations that would not reasonably be expected to, individually or in the aggregate, result in Losses in excess of \$10,000,000 (it being acknowledged and agreed that, if the Initial Closing occurs, any such Losses (whether or not in excess of such amount), if actually incurred, would be recoverable by the Buyer Indemnified Parties in accordance with the terms of Article 14 regardless of whether such Losses were known or disclosed prior to the Initial Closing); provided however, that (i) if the condition set forth in this Section 10.1.1(a) would not be satisfied as a result of Losses in excess of \$10,000,000, but less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 10.1.2, 10.1.4 and 10.1.6 to be satisfied), and (ii) the amount of Losses constitute Quantified Losses, then, at Seller's option in its sole discretion, Seller may reduce the Cash Purchase Price by the amount of all such Quantified Losses in excess of \$10,000,000 (except that (A) with respect to a breach of the representation and warranty in Section 5.7, the \$10,000,000 deductible shall not apply and the Cash Purchase Price shall be reduced for all Losses as a result of a breach of Section 5.7; and (B) with respect to a breach of a representation and warranty resulting from the sale of any direct or indirect interests in any apartment project or development site or a material portion thereof, the amount of the Quantified Loss shall be deemed to equal 120% of an amount equal to the difference between (x) the gross asset value for each such apartment project or development site or material portions thereof so conveyed, and (y) the Indebtedness allocated to or directly or indirectly encumbering such assets and the condition set forth in this Section 10.1.1(a) shall be deemed satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the decreased Cash Purchase Price shall be a corresponding decrease in the Purchase Price).

- (b) Each of the representations and warranties of the Lehman Entities and Seller set forth in Article 4 and Article 5 of this Agreement (other than any Seller Fundamental Representations) (disregarding for this purpose all qualifications or exceptions in such representations and warranties relating to materiality, material adverse effect or Material Adverse Effect) shall be true and correct as of the date of this Agreement and as of the Initial Closing Date as if made at and as of such date, except for any breaches of such representations and warranties that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.
- (c) Each of the representations and warranties concerning the Archstone Entities set forth in <u>Article 6</u> of this Agreement (other than any Seller Fundamental Representations) (disregarding for this purpose all qualifications or exceptions in such representations and warranties relating to materiality, material adverse effect or Material Adverse Effect) shall be true and correct as of the date of this Agreement (unless such representations were expressly made as of an earlier date, in which case such representations shall be true and correct on and as of such earlier date), except for any

breaches of such representations and warranties that would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect

Performance of Covenants. LBHI and Seller shall have performed and complied in all material respects with all covenants and obligations in this Agreement (other than the covenants and obligations that are specifically addressed in Sections 10.1.4, 10.1.5 and 10.1.6, which are subject to the conditions contained therein) required to be complied with by LBHI and Seller at or prior to the Initial Closing; provided, however, that if (a) the condition set forth in this Section 10.1.2 would not be satisfied, and (b) the Losses as a result of the failure of such condition to be satisfied constitute Quantified Losses in an amount less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 10.1.1(a), 10.1.4 and 10.1.6 to be satisfied), then, at Seller's option in its sole discretion, Seller may reduce the Cash Purchase Price in the amount of all such Quantified Losses (provided that, with respect to a breach of a covenant resulting from the sale of any direct or indirect interests in any apartment project or development site or a material portion thereof, the amount of the Quantified Loss shall be deemed to equal 120% of an amount equal to the difference between (x) the gross asset value for each such apartment project or development site or material portions thereof so conveyed, and (y) the Indebtedness allocated to or directly or indirectly encumbering such assets) and the condition set forth in this Section 10.1.2 shall be deemed to be satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the decreased Cash Purchase Price shall be a corresponding decrease in the Purchase Price).

61

- 10.1.3 No Material Adverse Effect. From the date of this Agreement, there shall not have occurred any change, event, occurrence or development that has had or would reasonably be expected to result in a Material Adverse Effect.
- Asset Transfer Transactions. Seller shall have taken, or shall have caused the applicable Archstone Entities to have taken, and LBHI shall have caused Seller or the applicable Archstone Entities to have taken, all actions required by Section 2.6 immediately prior to the Initial Closing in all material respects (it being agreed that the actions provided for in Section 2.6 to be taken by Seller, the Lehman Entities and the Archstone Entities shall be deemed to have been performed in all material respects if the applicable Buyer Parties or their respective Buyer Designees, as set forth in Schedule 2.6, receive, in the sequence provided in Schedule 2.6, the benefit of the assets intended to be conveyed to each of them pursuant to Schedule 2.6); provided, however, that if (a) the condition set forth in this Section 10.1.4 would not be satisfied, and (b) the Losses as a result of the failure of such condition to be satisfied constitute Quantified Losses in an amount less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 10.1.1(a), 10.1.2 and 10.1.6 to be satisfied), then, at Seller's option in its sole discretion, Seller may reduce the Cash Purchase Price in the amount of all such Quantified Losses and the condition set forth in this Section 10.1.4 shall be deemed to be satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the decreased Cash Purchase Price shall be a corresponding decrease in the Purchase Price).
- 10.1.5 <u>Closing Mechanics</u>. Seller and LBHI shall have performed and completed the Seller Closing Mechanics in all material respects (it being agreed that the actions provided for in <u>Section 2.6</u> to be taken by Seller, the Lehman Entities and the Archstone Entities shall be deemed to have been performed in all material respects if the applicable Buyer Parties or their respective Buyer Designees, as set forth in <u>Schedule 2.6</u>, receive, in the sequence provided in <u>Schedule 2.6</u>, the benefit of the assets intended to be conveyed to each of them pursuant to <u>Schedule 2.6</u>).
- Other Deliveries. LBHI, Lehman Designees and Seller, as applicable, shall have (a) delivered to the Escrow Agent the Escrowed Deferred Closing Asset Transfer Documents, if applicable, in all material respects (it being agreed that the actions provided for in Section 2.6 to be taken by Seller, the Lehman Entities and the Archstone Entities shall be deemed to have been performed in all material respects if the applicable Buyer Parties or their respective Buyer Designees, as set forth in Schedule 2.6, receive, in the sequence provided in Schedule 2.6, the benefit of the assets intended to be conveyed to each of them pursuant to Schedule 2.6), and (b) delivered to the applicable Buyer Party or Buyer Designee, at the Initial Closing, each of the following documents in a form and substance reasonably satisfactory to the Buyer Parties; provided,

62

however, that if (x) the condition set forth in this Section 10.1.6 would not be satisfied, and (y) the Losses as a result of the failure of such condition to be satisfied constitute Quantified Losses in an amount less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 10.1.1(a), 10.1.2 and 10.1.4 to be satisfied), then, at Seller's option in its sole discretion, Seller may reduce the Cash Purchase Price in the amount of all such Quantified Losses and the condition set forth in this Section 10.1.6 shall be deemed to be satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the decreased Cash Purchase Price shall be a corresponding decrease in the Purchase Price):

- (a) executed counterparts of each of the Transaction Documents to which LBHI, such Lehman Designee and/or Seller is a party (other than any Transaction Documents relating to any Deferred Closing Assets), in each case, executed by a duly authorized Representative of LBHI, such Lehman Designee and/or Seller;
- (b) a certificate executed by a duly authorized Representative of Seller certifying as to the matters set forth in Sections 10.1.1 and 10.1.2;
- (c) if any Transferred Interests or Equity Interests in any Archstone Entity that are to be transferred at or prior to the Initial Closing are certificated, fully endorsed certificates evidencing such Transferred Interests or other Equity Interests, together with stock powers or other necessary transfer documentation; <u>provided</u> that if such certificates have been lost, stolen or destroyed, a lost certificate affidavit will satisfy the requirements of this Section 10.1.6(c):
- (d) one or more bills of sale, assignment and assumption agreements in substantially the form set forth on <u>Exhibit E</u> (the "Bills of Sale"), each executed by a duly authorized Representative of Seller or the applicable Archstone Entity, transferring all Equity Interests held by Seller or such Archstone Entity in certain Archstone Entities (as contemplated by <u>Schedule 2.6</u>) (other than interests in any Archstone Entity relating to any Deferred Closing Assets);
- (e) such other instruments of transfer, assumption, filings or documents, in form and substance reasonably satisfactory to the Buyer Parties, as may be reasonably requested by the Buyer Parties to the extent required to give effect to this Agreement and the Contemplated Transactions to be consummated at the Initial Closing.

any other document required to be delivered, or action to be taken, pursuant to this Agreement by Seller or any Lehman Entity shall be inconsistent with Section 2.6.3 of this Agreement or create any additional representations, warranties, indemnities, agreements or other obligations of Seller (except to the extent that such indemnities, agreements or obligations constitute Assumed Liabilities and except for Excluded Liabilities) or the Lehman Entities not expressly set forth in this Agreement, and (ii) Seller and LBHI shall not be required to execute any documents or other deliverable required by Sections 10.1.4, 10.1.5, 10.1.6 or 13.1 if the form of such document or deliverable was not delivered to Seller a reasonable period in advance of the Initial Closing.

- 10.2 <u>Conditions for Each Extension Closing</u>. The Buyer Parties' obligations to consummate the acquisition of the applicable Deferred Closing Assets and all related transactions at any Extension Closing and to take the other actions contemplated by this Agreement at such Extension Closing are subject to the satisfaction or (to the extent permitted by applicable Law) waiver in writing by the Buyer Parties, in their sole and absolute discretion, at or prior to such Extension Closing, of each of the following conditions set forth in this <u>Section 10.2:</u>
 - 10.2.1 Representations. LBHI and Seller shall have certified that each of the Transaction Documents to which Seller or any Lehman Entity is a party that has been duly executed and delivered at such Extension Closing has been duly executed and delivered, and constitutes a legal, valid and binding obligation, enforceable against the applicable Person in accordance with the terms of the applicable Transaction Document, except that (a) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights generally, and (b) general equitable principles.
 - 10.2.2 New Liens. LBHI shall not have taken (or caused any Lehman Entity, Seller or Archstone Entity to have taken) an affirmative action that shall have subjected a Deferred Closing Asset to a Lien (other than a Permitted Lien) after the Initial Closing Date.
 - 10.2.3 Performance of Covenants. Each of the Lehman Entities and Seller shall have performed and complied in all material respects (it being agreed that the actions provided for in Section 2.6 to be taken by Seller, the Lehman Entities and the Archstone Entities shall be deemed to have been performed in all material respects if the applicable Buyer Parties or their respective Buyer Designees, as set forth in Schedule 2.6, receive, in the sequence provided in Schedule 2.6, the benefit of the assets intended to be conveyed to each of them pursuant to Schedule 2.6) with all covenants and obligations in this Agreement with respect to each of the Deferred Closing Assets to be acquired at such Extension Closing required to be complied with by the Lehman Entities and Seller at or prior to such Extension Closing.
 - 10.2.4 <u>Escrow Release</u>. Each of the Lehman Designees, LBHI and Seller, as applicable, shall have executed and delivered to the Escrow Agent the

64

Consideration and Assumption Escrow Release Instructions and the Asset Document Escrow Release Instructions with respect to such Extension Closing.

ARTICLE 11 CONDITIONS TO SELLER'S OBLIGATION TO CLOSE

11.1 <u>Conditions for Initial Closing.</u> Seller's obligations to consummate the Contemplated Transactions at the Initial Closing and to take the other actions contemplated by this Agreement at the Initial Closing are subject to the satisfaction or (to the extent permitted by applicable Law) the waiver by Seller, in its sole and absolute discretion, at or prior to the Initial Closing, of each of the following conditions set forth in this <u>Section 11.1</u>.

11.1.1 <u>Accuracy of Representations and Warranties.</u>

(a) Each of the AVB Fundamental Representations and ERPOP Fundamental Representations shall be true and correct as of the date of this Agreement and as of the Initial Closing Date as if made at and as of such date (unless such representations were expressly made as of an earlier date, in which case such representations shall be true and correct on and as of such earlier date), except for any inaccuracies of the AVB Fundamental Representations and ERPOP Fundamental Representations that would not reasonably be expected to, individually or in the aggregate, result in Losses in excess of \$10,000,000 (it being acknowledged and agreed that, if the Initial Closing occurs, any such Losses (whether or not in excess of such amount), if actually incurred, would be recoverable by the Seller Indemnified Parties in accordance with the terms of Article 14); provided however, that (i) if the condition set forth in this Section 11.1.1(a) would not be satisfied as a result of Losses in excess of \$10,000,000 but less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 11.1.2, 11.1.4 and 11.1.6 to be satisfied), and (ii) the amount of Losses as a result of the failure of the AVB Fundamental Representations or ERPOP Fundamental Representations, as applicable, to be true and correct constitute Quantified Losses, then, at the applicable Buyer Party's option in its sole discretion, the applicable Buyer Party may increase the AVB Cash Purchase Price Portion or the EQR Cash Purchase Price Portion, as applicable, by the amount of all such Quantified Losses in excess of \$10,000,000 resulting from the failure of the AVB Fundamental Representations or the ERPOP Fundamental Representations, as applicable, and the condition set forth in this Section 11.1.1(a) shall be deemed satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased EQR Cash Purchase Price Portion or AVB Cash Purchase Price Portion shall be a corresponding increase in the Purchase Price).

65

(b) Each of the representations and warranties of the Equity Residential Parties and AVB set forth in Article 7 and Article 8 of this Agreement (other than any Buyer Party Fundamental Representations) (disregarding for this purpose all qualifications or exceptions in such representations and warranties relating to materiality, material adverse effect or Material Adverse Effect) shall be true and correct as of the date of this Agreement (unless such representations were expressly made as of an earlier date, in which case such representations shall be true and correct on and as of such earlier date), except for any breaches of such representations and warranties that would not, individually or in the aggregate, reasonably be expected to have a Sponsor Material Adverse Effect.

Performance of Covenants. EQR, ERPOP and AVB shall have performed and complied in all material respects with all covenants and obligations in this Agreement (other than the covenants and obligations that are specifically addressed in Sections 11.1.4, 11.1.5 and 11.1.6, which are subject to the conditions contained therein) required to be complied with by them, at or prior to the Initial Closing; provided, however, that if the condition set forth in this Section 11.1.2 would not be satisfied and the Losses as a result of the failure of such condition to be satisfied constitute Quantified Losses in an amount less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 11.1.1(a), 11.1.4 and 11.1.6 to be satisfied), then, at the applicable Buyer Party's option in its sole discretion, the applicable Buyer Party may increase the AVB Cash Purchase Price Portion or EQR Cash Purchase Price Portion, as applicable, in the amount of all such Losses and the condition set forth in this Section 11.1.2 shall be deemed to be satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased EQR Cash Purchase Price Portion or AVB Cash Purchase Price Portion shall be a corresponding increase in the Purchase Price).

11.1.3 Sponsor Material Adverse Effect.

- (a) With respect to EQR and ERPOP, from the date of the filing with the SEC of EQR's quarterly report on From 10-Q for the period ended on September 30, 2012, there shall not have occurred any change, event, occurrence or development that has had or would reasonably be expected to result in a Sponsor Material Adverse Effect.
- (b) With respect to AVB, from the date of the filing with the SEC of AVB's quarterly report on Form 10-Q for the period ended on September 30, 2012, there shall not have occurred any change, event, occurrence or development that has had or would reasonably be expected to result in a Sponsor Material Adverse Effect.

66

- 11.1.4 Asset Transfer Transactions. The applicable Buyer Party or Buyer Designee shall have cooperated with Seller, the Lehman Entities or the applicable Archstone Entities with respect to, and to the extent applicable, have taken all actions set forth on Schedule 2.6 immediately prior to the Initial Closing in all material respects provided that all requested actions satisfy the requirements of Section 2.6; provided, however, that if (a) the condition set forth in this Section 11.1.4 would not be satisfied, and (b) the Losses as a result of the failure of such condition to be satisfied constitute Quantified Losses in an amount less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 11.1.1(a), 11.1.2 and 11.1.6 to be satisfied), then, at the Buyer Parties' option in their sole discretion, the Buyer Parties may increase the Cash Purchase Price in the amount of all such Quantified Losses and the condition set forth in this Section 11.1.4 shall be deemed to be satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased EQR Cash Purchase Price Portion or AVB Cash Purchase Price Portion shall be a corresponding increase in the Purchase Price).
- 11.1.5 <u>Closing Mechanics</u>. The Buyer Parties shall have performed and completed the Buyer Closing Mechanics.
- Other Deliveries. The applicable Buyer Party or Buyer Designee shall have (a) paid to Seller the Cash Purchase Price, (b) delivered to the Escrow Agent, if applicable, the Escrowed Deferred Stock Consideration and the Escrowed Deferred Closing Asset Transfer Documents, if applicable, in all material respects, and (c) delivered to Seller, a Lehman Entity and/or a Lehman Designee, at the Initial Closing, each of the following documents and other deliverables in a form and substance reasonably satisfactory to Seller; provided, however, that if (x) the condition set forth in this Section 11.1.6 would not be satisfied and (y) the Losses as a result of the failure of such condition to be satisfied constitute Quantified Losses in an amount less than \$250,000,000 (considered together with the cumulative amount of Losses resulting from the failure of the conditions in Sections 11.1.1(a), 11.1.2 and 11.1.4 to be satisfied), then, at the Buyer Parties' option in their sole discretion, the Buyer Parties may increase the Cash Purchase Price in the amount of all such Quantified Losses and the condition set forth in this Section 11.1.6 shall be deemed to be satisfied (and, for the avoidance of doubt, without any corresponding adjustment to the ERPOP Equity Consideration or the AVB Equity Consideration, but the result of the increased EQR Cash Purchase Price Portion or AVB Cash Purchase Price Portion shall be a corresponding increase in the Purchase Price):
 - (a) certificates or other documentation evidencing the issuance of the ERPOP Equity Consideration (other than any Escrowed ERPOP Equity Consideration:

- (b) certificates or other documentation evidencing the issuance of the AVB Equity Consideration (other than an Escrowed AVB Equity Consideration);
- (c) executed counterparts for each of the Transaction Documents to which a Buyer Party or Buyer Designee is a party (other than any Transaction Documents relating to any Deferred Closing Assets), in each case, duly executed by a duly authorized Representative of the applicable Buyer Party or Buyer Designee;
- (d) a certificate executed by a duly authorized Representative of AVB certifying with respect to the matters set forth in <u>Sections 11.1.1</u> (with respect to the representations and warranties of AVB) and <u>11.1.2</u>;
- (e) a certificate executed by a duly authorized Representative of ERPOP certifying with respect to the matters set forth in <u>Sections 11.1.1</u> (with respect to the representations and warranties of EQR) and <u>11.1.2</u>;
- (f) one or more Assumption Agreements executed by duly authorized Representative of the applicable Buyer Party or Buyer Designee effecting the assumption by the applicable Buyer Party or Buyer Designee of the Assumed Liabilities (other than any Assumed Liabilities relating to any Deferred Closing Assets);
- (g) evidence reasonably satisfactory to Seller and LBHI that the board of directors or board of trustees, as applicable, of EQR and AVB shall have taken all requisite actions to grant Seller or LBHI (or any Lehman Designee) an exemption from the capital stock ownership limitations contained in each of EQR's and AVB's Organizational Documents, to the extent permissible under such Organizational Documents; and
- (h) such other instruments of transfer, assumption, filings or documents, in form and substance reasonably satisfactory to Seller, as may be reasonably requested by Seller to the extent required to give effect to this Agreement and the Contemplated Transactions to be consummated at the Initial Closing.

- 11.2 <u>Conditions for Each Extension Closing.</u> Seller's obligations to consummate the acquisition of the applicable Deferred Closing Assets and all related transactions at any Extension Closing and to take the other actions contemplated by this Agreement at such Extension Closing are subject to the satisfaction or (to the extent permitted by applicable Law) waiver in writing by Seller, in its sole and absolute discretion, at or prior to such Extension Closing, of each of the following conditions set forth in this Section 11.2:
 - 11.2.1 Representation. Each of the Buyer Parties shall have certified that each of the Transaction Documents to which such Buyer Party is a party that has been duly executed and delivered at such Extension Closing has been duly executed and delivered, and constitutes a legal, valid and binding obligation,

68

- enforceable against the applicable Person in accordance with the terms of the applicable Transaction Document, except that (a) such enforcement may be subject to applicable bankruptcy, insolvency, reorganization, moratorium or other similar Laws, now or hereafter in effect, affecting creditors' rights generally, and (b) general equitable principles.
- 11.2.2 Other Deliveries. The Buyer Parties shall have executed and delivered (a) to the Escrow Agent the Consideration and Assumption Escrow Release Instructions and the Asset Document Escrow Release Instructions with respect to such Extension Closing, and (b) to LBHI and Seller such other instruments of transfer, assumption, filings or documents, in form and substance reasonably satisfactory to LBHI and Seller, as may be required to give effect to the acquisition of the applicable Deferred Closing Assets and related transactions to be consummated at such Extension Closing, in each case to the extent not previously delivered to the Escrow Agent.
- 11.2.3 <u>Performance of Covenants.</u> The Buyer Parties shall have performed and complied in all material respects with all covenants and obligations in this Agreement with respect to each of the Deferred Closing Assets to be acquired at such Extension Closing required to be complied with by the Buyer Parties at or prior to such Extension Closing.

ARTICLE 12 CERTAIN PRE-CLOSING COVENANTS OF SELLER AND THE LEHMAN ENTITIES

12.1 Conduct of the Archstone Entities Prior to the Initial Closing. Seller and LBHI agree that, from the date of this Agreement until the consummation of the Initial Closing or the earlier termination of this Agreement, except as required, prohibited or expressly permitted by this Section 12.1, this Agreement or as set forth in Schedule 12.1 (and any subsection thereof) or as requested, directed or prohibited by the Buyer Parties (provided, that nothing in this Section 12.1 shall provide the Buyer Parties with any additional right to direct or prohibit any actions other than in accordance with this Agreement) or with the prior written consent of the Buyer Parties (which shall not be unreasonably withheld, conditioned or delayed), LBHI shall cause Seller to, and Seller shall, and shall cause the Archstone Entities to, use their Commercially Reasonable Efforts to (a) operate and maintain the Transferred Assets and Transferred Subsidiary Assets in the Ordinary Course of Business, and (b) not take any action with respect to the Business, other than in the Ordinary Course of Business or as expressly permitted or required under this Agreement. Except (i) as otherwise required, prohibited or expressly permitted by this Agreement, including as a result of sales of Designated Properties in accordance with Section 12.3 (provided, that nothing in this Section 12.1 shall provide the Buyer Parties with any additional right to direct or prohibit any actions other than in accordance with this Agreement), (ii) as requested or directed by the Buyer Parties, or (iii) as set forth in Schedule 12.1 (and any subsection thereof), Seller and LBHI shall cause the Archstone Entities not to, and Seller and LBHI shall not, between the date of this Agreement and the Initial Closing, take any of the following actions without the prior written consent of the Buyer Parties (which shall not be unreasonably withheld, conditioned or delayed) to:

69

- 12.1.1 amend or otherwise change any provision of any Organizational Document of any Archstone Entity or Joint Venture in any respect other than to correct typographical errors of a de minimis nature;
- (i) authorize for issuance, issue or sell, pledge, dispose of or subject to any Lien (other than Permitted Liens) or agree or commit to any of the foregoing in respect of, any Equity Interest of any Archstone Entity, (ii) repurchase, redeem or otherwise acquire any Equity Interest of any Archstone Entity, other than the redemption of Series O Preferred Units to the extent required by the governing documents of Archstone, (iii) take any action that would result in the triggering of any Liability under a Tax Protection Agreement, other than pursuant to any property sale agreements in the Data Room, (iv) reclassify, combine, split, or subdivide any Equity Interest of any Archstone Entity, or (v) authorize, declare, set aside, make or pay any dividend or other distribution, with respect to any Equity Interest of any Archstone Entity other than, in each case, (x) distributions paid to holders of the Series O Preferred Units in accordance with the terms thereof in connection with the redemption of Series O Preferred Units to the extent required by the governing documents of Archstone, (y) cash dividends or distributions necessary to be paid to maintain real estate investment trust status of any Archstone Entity that is a real estate investment trust, or (z) dividends and distributions paid to Persons who are not Affiliates of LBHI or the Archstone Entities as required pursuant to the Organizational Documents of any Joint Venture existing as of the date hereof;
- 12.1.3 enter into or establish any partnership, joint venture or similar arrangement (including funds or other investment vehicles);
- (i) acquire, divest, transfer, license or sell (by merger, consolidation, acquisition or sale of equity interests or assets, or any other business combination) any corporation, partnership, limited liability company, joint venture or other business organization (or division thereof), including the Archstone Entities, or any personal property, intangible property, trade secrets, Intellectual Property or other asset of Seller or any Archstone Entity, other than pursuant to a contractual provision in the Organizational Documents of an existing Joint Venture disclosed in the Data Room requiring the same (but expressly excluding any contractual provision pursuant to which the applicable Archstone Entity has any discretion to purchase or sell any applicable asset or interest), (ii) acquire (by merger, consolidation or otherwise), divest, or transfer or sell, or enter into any option, commitment or agreement to acquire (by merger, consolidation or otherwise), divest, transfer or sell, any interest in real property or any asset (whether to third parties or from one Archstone Entity to another), or commence any development or construction activity with respect to any real property or any asset, of Seller or any Archstone Entity, other than pursuant to a contractual provision in the Organizational Documents of an existing Joint Venture disclosed in the Data Room requiring the same (but expressly excluding any contractual provision

exceed in the aggregate \$1,000,000, (iii) in respect of development pursuits (including land owned or under option) that have not yet commenced construction, expend an amount in excess of \$250,000 with respect to any particular development pursuit project pursuant to Contracts in existence as of the date hereof and set forth in Section 12.1.4(iii) of Schedule 12.1, other than, in each case, (x) in connection with any construction actually underway as of the date hereof (and set forth in Section 12.1.4(x) of Schedule 12.1) in accordance with any monthly development reports included in the Data Room (including cost overruns not to exceed 5% per property), plus operating expenses related to such properties in the Ordinary Course of Business, or (y) as permitted pursuant to Section 12.3, or (iv) engage in any solicitation or marketing activity with respect to the matters described in clauses (i), (ii) or (iii);

- 12.1.5 enter into any Contract, except for (i) Contracts entered into in the Ordinary Course of Business that: (x) involve total payments of less than \$200,000 per Contract, or (y) are terminable without a fee or penalty on sixty (60) or fewer advance days' notice, and (ii) Contracts otherwise permitted in Sections 12.1.1 through 12.1.27, except that in no event shall such Persons enter into any Contract related to construction that has not yet commenced and for which any such Persons would incur any Liabilities if such construction were never commenced or were commenced on a delayed basis;
- 12.1.6 replace, discontinue, make any material modifications to, allow licensing rights to expire with respect to, or make any material new investment in, any information technology system or software in use by any Archstone Entity as of the date of this Agreement with respect to rent optimization software, lease accounting, financial accounting, payroll accounting, human resources management, collections management, payment of accounts or trade payables or other business functions of any Archstone Entity; provided that this Section 12.1.6 shall not prohibit Seller's employees and independent contractors from continuing work on improvements to existing systems or systems under development as of the date hereof, or prohibit any repairs to existing systems as are necessary;
- 12.1.7 incur Indebtedness or issue any debt securities or other debt instruments or assume, guarantee or endorse, or otherwise as an accommodation become responsible for, the obligations of any Person for Indebtedness, or enter into any material modifications of documents or instruments with respect to such Indebtedness, in each case in a principal amount in excess of \$1,000,000 in the aggregate for all incurrences of Indebtedness, other than (i) intercompany Indebtedness among the Archstone Entities, (ii) advances and drawdowns on construction or development loans that are in existence on the date hereof, or

71

- (iii) drawdowns to pay expenditures otherwise permitted by this <u>Article 12</u> on credit facilities in existence on the date hereof (<u>provided</u> that LBHI and Seller shall be required to provide notices to the Buyer Parties of any such drawdowns in excess of \$20,000,000 in the aggregate);
- 12.1.8 except as required by Law or any existing agreement, arrangement or compensation plan that has been disclosed to the Buyer Parties, as contemplated by Section 13.5 (as such instruments exist as of the date hereof), (i) increase the compensation or benefits payable to its trustees, directors, officers or employees (other than (x) ordinary course increases as year-end annual raises that do not exceed 3.5% in the aggregate of the total compensation or benefits payable to all trustees, directors and employees with respect to the 2012 fiscal year, (y) any employee that fills a new position that pays a higher salary and/or bonus consistent with past practice pursuant to Section 12.1.26, or (z) bonuses with respect to the sale of Designated Properties in accordance with Section 12.3 or the sale of the other properties expressly permitted to be sold pursuant to this Section 12.1 and designated on Schedule 12.1, in each case as such bonuses are described in Section 12.1.8 of Schedule 12.1), (ii) grant to any director, officer, employee or independent contractor any new employment, retention, bonus, severance, change of control or termination pay awards or equity-based cash awards (including cash bonuses or dividend equivalent rights), including any bonuses that vest over time or in the future, (iii) grant any increase in, or otherwise alter or amend, any right to receive any severance, change of control, retention or termination pay or benefits to any employee, (iv) establish, adopt, enter into or amend any collective bargaining (or similar), bonus, profit-sharing, thrift, compensation, stock option, restricted stock, stock unit, dividend equivalent, pension, retirement, deferred compensation, employment, loan, retention, consulting, indemnification, termination, severance or other similar plan, agreement, trust, fund, policy or arrangement with any director, officer, employee or independent contractor, or (v) take any action with respect to salary, compensation, benefits or other terms and conditions of employment that would be reasonably be expected to result in the holder of a change in control or similar agreement having "good reason" to terminate employment and collect severance payments and benefits pursuant to such agreement;
- 12.1.9 repurchase, repay or pre-pay any Indebtedness of Seller or any Archstone Entity in an amount in excess of \$1,000,000 in the aggregate, except (i) repayments of revolving credit facility or other similar lines of credit in the Ordinary Course of Business, (ii) payments made in respect of any termination or settlement of any Derivative Instruments in the Ordinary Course of Business (provided the Parties acknowledge and agree that any such Derivative Instruments so terminated or settled shall adjust the Purchase Price in accordance with Section 2.3.3), (iii) regularly-scheduled payments of Indebtedness in accordance with its terms and any principal and interest payments due at the maturity or earlier par prepayment date of the applicable loan, (iv) intercompany Indebtedness among the Archstone Entities, and (v)

72

Indebtedness associated with a Designated Property sold pursuant to and in accordance with Section 12.3 or a property sale agreement in the Data Room;

- 12.1.10 except as required by changes in GAAP change any financial accounting principles or policies in any material respect or replace or remove the accountants of any Archstone Entity;
- (i) make any loans, advances or investments in any other Person (other than as may be required pursuant to the terms of the Organizational Documents of or any Contract entered into by an Archstone Entity, a Joint Venture, a partnership or similar arrangement in existence as of the date hereof or authorized by the Buyer Parties pursuant to this Section 12.1) in an amount in excess of \$1,000,000 in the aggregate, or (ii) authorize, or enter into any commitment for, any new capital expenditure (expressly excluding turnover and make-ready costs) in an amount in excess of \$250,000 in the aggregate except (x) in the Ordinary Course of Business, provided that such capital expenditures in the Ordinary Course of Business do not exceed \$200,000 per property, (y) for the emergency repair of any real property asset owned by Seller or any of the Archstone Entities, or (z) pursuant to any Contract set forth in Section 12.1.11 of Schedule 12.1 or as permitted by this Section 12.1;
- 12.1.12 subject to any rights or obligations under Section 2.3.3, enter into any new Derivative Instrument or modify any Derivative Instrument except as required by any Contract regarding Indebtedness in effect as of the date hereof;
- 12.1.13 except as set forth in <u>Section 12.1.13</u> of <u>Schedule 12.1</u> or as required pursuant to the terms of any Organizational Documents of any Joint Venture, make, rescind or revoke any material Tax election (whether or not such election is filed with a tax return) or change a material method of Tax accounting, amend any material Tax Return, agree to waive or extend any period of adjustment, assessment or collection of material Taxes, or settle or compromise

any material federal, state, local or foreign income Tax liability, audit, claim or assessment, or enter into any material closing agreement related to Taxes, or knowingly surrender any right to claim any material Tax refund unless in each case such action is required by Law;

- 12.1.14 enter into, amend or modify any Tax Protection Agreement, or take any action that would, or could reasonably be expected to, violate any Tax Protection Agreement except as required pursuant to the terms of the Contemplated Transactions or pursuant to any property sale agreement in the Data Room;
- 12.1.15 amend any term of any Preferred Unit or of any outstanding equity security or Equity Interest of Seller or any Archstone Entity;
- 12.1.16 impose any Lien (other than a Permitted Lien or refinancing permitted in <u>Section 12.1.9</u> above), on any Transferred Asset, any Transferred Subsidiary Asset, or other material asset of Seller or any Archstone Entity other than if

73

specified by the Buyer Parties pursuant to Section 12.3 or pursuant to any Indebtedness permitted pursuant to Section 12.1.7 or Section 12.1.9;

- 12.1.17 adopt a plan of complete or partial liquidation or dissolution or adopt resolutions providing for or authorizing such liquidation or dissolution with respect to Seller or any Archstone Entity;
- 12.1.18 fail to maintain in full force and effect the existing insurance policies or to replace such insurance policies with reasonably comparable insurance policies covering the assets of Seller or the Archstone Entities or the Transferred Assets or Transferred Subsidiary Assets, provided that the existing master property insurance program expiring March 25, 2013 shall not be extended past May 1, 2013;
- 12.1.19 settle any insurance claims relating to casualties affecting any Transferred Asset or Transferred Subsidiary Asset where the damages arising from any single casualty event or series of related casualty events would reasonably be expected to be in excess of \$5,000,000 in the aggregate;
- 12.1.20 settle any condemnation or eminent domain proceedings affecting any Transferred Asset or Transferred Subsidiary Asset where the damages relating to the applicable proceedings would reasonably be expected to be in excess of \$1,000,000 in the aggregate;
- 12.1.21 initiate, waive, release, assign, compromise or settle any action or proceeding, which action or proceeding involves, individually more than \$150,000 or in the aggregate, more than \$1,000,000, other than any compromises or settlements entered into by Seller or LBHI in accordance with Article 14 for which the Archstone Entities, EQR or Buyer Parties are indemnified pursuant to Article 14 (provided, that no payments related to such compromises or settlements may be made using the funds of the Archstone Entities);
- 12.1.22 repatriate funds held in overseas accounts;
- 12.1.23 commit any act that would cause any Archstone Entity to become a registered investment company or registered investment adviser; or
- 12.1.24 pay compensation to underwriters or legal fees of underwriters' counsel incurred in connection with the previously contemplated initial public offering of Archstone, Inc. from the funds of the Archstone Entities (it being acknowledged that this restriction will not prohibit the payment of all actual and documented legal and accounting fees and expenses and other costs (with legal and accounting fees being based on standard hourly billing) (but not compensation paid to underwriters or legal fees of underwriters' counsel) incurred by the Lehman Entities, Archstone or Seller in connection with the previously contemplated initial public offering of Archstone, Inc.);

74

- 12.1.25 voluntarily permit any changes to any board, management committee or similar governing board of any Archstone Entity or Joint Venture, other than replacing any members of any such board, management committee or similar governing board who held such position by reason of appointment or nomination by LBHI or its Affiliates, or agree to any termination or modification of the Information and Board Observation Rights and Confidentiality Agreement dated as of May 19, 2009 between Archstone Multifamily Series I Trust and Pacific Life Insurance Company;
- 12.1.26 fill open employment positions other than any position below the level of vice president;
- 12.1.27 enter into any new retail lease, or renewal of an existing retail lease, with a term of more than two (2) years; or
- 12.1.28 announce any intention, enter into, amend or modify any agreement or otherwise make a commitment, to do any of the foregoing.

No action by Seller or any of the Archstone Entities with respect to matters specifically permitted by <u>Sections 12.1.1</u> through <u>12.1.28</u> (other than <u>Section 12.1.5</u>) shall be deemed a breach of <u>Section 12.1</u> unless such action would constitute a breach of any of <u>Sections 12.1.1</u> through <u>12.1.28</u> (other than <u>Section 12.1.5</u>). Notwithstanding anything to the contrary, this <u>Section 12.1</u> shall not require Seller or LBHI to take any actions that would violate any Law.

12.2 Access and Operational Matters. Seller and LBHI acknowledge that the Buyer Group requires access to the employees of Seller, the Archstone Entities and the Joint Ventures in order to facilitate a smooth transition of the Transferred Assets and Transferred Subsidiary Assets to the Buyer Group. In connection with the continued operation of Seller and the Archstone Entities, Seller shall and shall cause the Archstone Entities to (and LBHI shall cause Seller and the Archstone Entities to), (a) confer in good faith on a regular and frequent basis with one or more Representatives of the Buyer Parties designated by them regarding operational matters and the general status of ongoing operations, including budgeting, and (b) notify the Buyer Parties promptly of any event or occurrence that is material and adverse such that it may reasonably be expected to have (when taken together with future events which may or may not occur) a Material Adverse Effect.

Except as may be necessary to comply with any applicable Laws, and subject to the absolute right of Seller, the Archstone Entities and the Joint Ventures to preserve any applicable privileges (including the attorney-client privilege, subject to entering into joint common defense arrangements where possible) (except that Seller and LBHI shall not, and shall cause the Archstone Entities not to, assert any privilege as grounds for not providing such information described in Section 13.8.5), from the date of this Agreement until the Initial Closing Date or earlier termination of this Agreement, Seller shall cause each of the Archstone Entities and the Joint Ventures to:

- 12.2.1 during normal business hours, give the Buyer Parties and their respective Affiliates and Representatives (the "Buyer Group") reasonable access to the corporate and field offices of the Archstone Entities, together with access to the books, records, software applications, fixed assets, Contracts and other documents of or pertaining to the Archstone Entities and the Joint Ventures (it being acknowledged and agreed that reasonable access may include providing representatives of the Buyer Parties with designated workspace located within such offices);
- 12.2.2 furnish to the Buyer Group financial and operating data and other information relating to the Archstone Entities and the Joint Ventures and the Business in the form currently produced by management of Seller (or in such other form as the Buyer Group may reasonably request and that can be reasonably prepared by management of Seller without material unreimbursed cost), including access to budgets and daily reporting of property operating fundamentals;
- 12.2.3 instruct the employees, counsel, financial advisors and other Representatives of Seller, the Archstone Entities and the Joint Ventures to cooperate with the Buyer Group's reasonable requests in its investigation of the Archstone Entities and the Joint Ventures upon prior reasonable notice;
- 12.2.4 permit the Buyer Group to discuss the Business and the Joint Ventures with members of management, officers, directors and employees of, and advisors to and counsel and accountants for, the Archstone Entities and the Joint Ventures upon reasonable request upon prior reasonable notice;
- 12.2.5 with respect to employees of the Archstone Entities, permit the Buyer Group to conduct interviews and training, employment "onboarding" activities such as benefits pre-enrollment, discuss prospects for future employment, and generally communicate via email and internet systems with such employees regarding all of the foregoing; and
- 12.2.6 permit the Buyer Group to discuss the Business, the Joint Ventures and the Contemplated Transactions with any lenders, ground lessors, other members or partners in the Joint Ventures and any other third parties whose interest in the Archstone Entities may be affected by the Contemplated Transactions; provided, however, that upon reasonable request, the Buyer Group shall keep Seller reasonably informed about such discussions.

Notwithstanding anything to the contrary, nothing contained in this <u>Section 12.2</u> shall require Seller or LBHI to take any actions or refrain from taking any actions that would violate any Law.

12.3 <u>Sales and Marketing of Designated Properties.</u> The Lehman Entities shall cause Seller to, and Seller shall and shall cause the Archstone Entities to, reasonably cooperate with the Buyer Parties in the marketing for sale of the properties designated from time to time by the

76

Buyer Parties (the "<u>Designated Properties</u>") in accordance with the procedures and terms and conditions set forth on <u>Schedule 12.3</u> (it being acknowledged that Seller's obligations pursuant to this <u>Section 12.3</u> shall be consistent with the available resources of the Archstone Entities. In no event shall Seller or the Lehman Entities be required to consummate a sale of the Designated Properties prior to the Initial Closing unless Seller determines to do so in its sole and absolute discretion).

- Acquisition Proposals. From the date of this Agreement until the Initial Closing Date or, if earlier, the termination of this Agreement, LBHI and Seller agree that, except as expressly permitted hereunder (including in connection with a sale pursuant to Section 12.3) or as requested by the Buyer Parties, none of the Lehman Entities, Seller and their respective Representative or Affiliates will, (a) solicit, initiate, seek, entertain, encourage, facilitate, support or induce the making, submission or announcement by any Person of any inquiry, expression of interest, proposal or offer that constitutes, or could reasonably be expected to lead to the acquisition by any Person (other than the Buyer Parties and their respective Affiliates and Representatives) of any of the Transferred Assets or Transferred Subsidiary Assets or any Equity Interests in Seller or the Archstone Entities, (b) enter into, participate in, maintain or continue any communications (except solely to provide written notice as to the existence of these exclusivity provisions) or negotiations regarding, or deliver or make available to any Person any information with respect to, or take any other action regarding, any inquiry, expression of interest, proposal or offer that constitutes, or could reasonably be expected to lead to, the acquisition by any Person (other than the Buyer Parties and their respective Affiliates and Representatives) of any of the Transferred Assets or Transferred Subsidiary Assets or any Equity Interests in Seller or the Archstone Entities, (c) agree to, accept, approve, endorse or recommend any transaction related to the acquisition by any Person (other than the Buyer Parties and their respective Affiliates and Representatives) of the Transferred Assets or Transferred Subsidiary Assets or any Equity Interests in Seller or the Archstone Entities, or (d) enter into any letter of intent or any other contract contemplating or otherwise relating to the acquisition by any Person (other than the Buyer Parties and their respective Affiliates and Representatives) of the Transferred Assets or Transferred Subsidiary Assets or any Equity Interests in Seller or the Archstone Entities. As of the date of this Agreement, except as expressly permitted hereunder (including in connection with a sale pursuant to Section 12.3) or as requested by the Buyer Parties, LBHI and Seller agree to (i) immediately cease and cause to be terminated any existing discussions or negotiations with any Person (other than the Buyer Parties and their respective Affiliates and Representatives) conducted prior to the date hereof with respect to the acquisition by any Person (other than the Buyer Parties and their respective Affiliates and Representatives) of the Transferred Assets or Transferred Subsidiary Assets or any Equity Interests in Seller or the Archstone Entities, and (ii) terminate any negotiations or discussions under any confidentiality, non-disclosure, document and information access and/or other agreements (other than any such agreement with the Buyer Parties and/or their respective Affiliates and Representatives) with respect to the sale of the Transferred Assets or Transferred Subsidiary Assets or any Equity Interests in Seller or the Archstone Entities.
- 12.5 <u>Suspension of Archstone IPO</u>. From the date hereof until the termination of this Agreement, LBHI and Seller shall cause all applicable Archstone Entities, including Archstone Inc., a Maryland corporation, to cease to further pursue or consummate any initial public offering of Archstone Inc., Seller or any other Archstone Entity. For purposes of clarity, Archstone Inc. shall not be required to withdraw the registration statement filed with the SEC.

77

- 12.6 <u>Termination of Confidentiality Agreement.</u> Seller, LBHI, EQR and AVB shall cause (i) the Amended and Restated Confidentiality Agreement, between EQR, LBHI and Seller, dated as of October 19, 2012, and (ii) the Amended and Restated Confidentiality Agreement, between AVB, LBHI and Seller, dated as of October 19, 2012, to be terminated and of no further force or effect, effective as of the Initial Closing
- 12.7 <u>Cure Period.</u> When determining whether there has occurred any breach of the covenants included in this <u>Section 12</u> by any of Seller, LBHI or the Buyer Parties, as applicable, such determination shall be made by taking into account the Seller Cure Period and the Buyer Cure Period, as applicable.

ARTICLE 13 ADDITIONAL COVENANTS

- Subject to the terms and conditions of this Agreement, including Section 13.4.2, the Parties shall use their respective Commercially Reasonable Efforts to take, or cause to be taken (including by using their respective Commercially Reasonable Efforts to cause the Archstone Entities to take actions), all actions and to do, or cause to be done, all things necessary under applicable Laws to consummate the Contemplated Transactions no later than the Outside Closing Date (taking into account the Buyer Parties' right to extend the Outside Closing Date in accordance with and subject to the terms hereof), including (a) preparing and filing as promptly as practicable with any Governmental Authority or other third party all documentation to effect all necessary filings, notices, petitions, statements, registrations, submissions of information, applications and other documents, and (b) obtaining and maintaining all approvals, consents, registrations, permits, authorizations and other confirmations required to be obtained from any Governmental Authority or other third party that are necessary or advisable to consummate the Contemplated Transactions; provided, however, that nothing in this Section 13.1.1 shall be deemed to require any Party to (x) waive compliance by the other Parties and their Affiliates of their respective covenants or obligations under this Agreement, (y) waive any of the conditions set forth in Articles 9, 10 or 11 required to be satisfied by the other Parties or (z) impose any material obligation on any Party not specifically provided for in this Agreement unless such obligation is indemnified by the other Parties (and, in the case of Seller or LBHI, indemnified jointly and severally by ERPOP and AVB).
- The Parties agree to execute and deliver such other reasonably requested documents, certificates, agreements and other writings and to take such other reasonably requested actions as may be proper, desirable or advisable in order to consummate or implement expeditiously the Contemplated Transactions; provided, however, that neither Seller nor any Lehman Entity shall be required to take any action pursuant to this Section 13.1.2 that would

78

reasonably be expected to create any additional representations, warranties, indemnities, agreements or other obligations of Seller (except to the extent that such indemnities, agreements or obligations constitute Assumed Liabilities and except for Excluded Liabilities) or the Lehman Entities not expressly set forth in this Agreement.

- Seller and LBHI shall cooperate with the Buyer Parties and the Buyer Designees to the extent reasonably requested by the Buyer Parties and the Buyer Designees to take all reasonable actions to cause, effective as of the Initial Closing, the transfer, assignment and delivery to the applicable Buyer Party or Buyer Designee of (a) all insurance policies insuring or relating to the Transferred Assets, the Transferred Subsidiary Assets or assets of the Archstone Entities, (b) all open or outstanding rights and claims under any previously effective insurance policies insuring or relating to the Transferred Assets, the Transferred Subsidiary Assets or assets of the Archstone Entities, (c) to the extent requested by the Buyer Parties, all primary and excess liability insurance policies then in effect as of the Initial Closing Date and insuring or relating to any of the Transferred Assets, Transferred Subsidiary Assets or assets of the Archstone Entities, and (d) all insurance policies then in effect as of the Initial Closing Date and insuring or relating to any environmental Liability or matter in connection with the Transferred Assets, Transferred Subsidiary Assets or the Archstone Entities; provided, however, that neither Seller nor any Lehman Entity shall be required to take any action pursuant to this Section 13.1.3 that would reasonably be expected to create any additional representations, warranties, indemnities, agreements or other obligations of Seller (except to the extent that such indemnities, agreements or obligations constitute Assumed Liabilities and except for Excluded Liabilities) or the Lehman Entities not expressly set forth in this Agreement.
- 13.1.4 The Buyer Parties acknowledge and agree that the Archstone Entities have certain obligations under the Indemnity and Release Agreement dated as of March 31, 2009, among Tishman Speyer Properties, L.P. and certain of its Affiliates, certain Archstone Entities, LBHI, Archstone LB Syndication Partner LLC, Bank of America, N.A., Bank of America Strategic Ventures, Inc., Barclays Capital Real Estate Inc., Barclays Capital Real Estate Finance Inc. and BIH ASN LLC (the "Indemnity Agreement").
- 13.1.5 The Parties acknowledge and agree that, based on the information available to them as of the date hereof, the Contemplated Transactions are exempt from all Transfer Taxes by reason of Section 1146(a) of the Bankruptcy Code, none of the Contemplated Transactions require that an offer to purchase be made under any Tenant Purchase Option Law and in no event will any of the Contemplated Transactions be deemed to be illegal under the provisions of any Tenant Purchase Option Laws. Subject to Section 13.10.1(g):
 - (a) If any Proceeding has been initiated regarding the exemption of the Contemplated Transactions from Transfer Taxes or any Tenant Purchase

79

Option Laws prior to or after the Initial Closing, then at any time following the initiation of such Proceeding and within a reasonable time following the reasonable request of the Buyer Parties, LBHI shall file a motion or motions in the United States Bankruptcy Court for the Southern District of New York, in form and substance reasonably acceptable to the Buyer Parties and LBHI, requesting such court to enter an Order as to the following matters: (x) the Contemplated Transactions are being effected pursuant to the Modified Third Amended Joint Chapter 11 Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors, dated December 6, 2011 and that certain Order Confirming Modified Third Amended Joint Chapter 11 Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors, dated December 6, 2011, [Docket No. 23023] and (y) the Contemplated Transactions are, therefore, exempt from any Transfer Taxes by reason of Section 1146(a) of the Bankruptcy Code and qualify for exemptions under the applicable Tenant Purchase Option Law, as the case may be.

- If any Proceeding has been initiated regarding the exemption of the Contemplated Transactions from Transfer Taxes or the requirements of any Tenant Purchase Option Laws prior to or after the Initial Closing, then at any time following the initiation of such Proceeding and within a reasonable time following the reasonable request of the Buyer Parties, LBHI shall and shall cause Seller or any applicable Archstone Entity to, file such motions, complaints, or other pleadings or take such other actions reasonably requested (in each case, in form and substance reasonably acceptable to the Buyer Parties and LBHI) in any court or other tribunal in which Proceedings are pending regarding the exemption of any of the Contemplated Transactions from Transfer Tax or Tenant Purchase Option Laws as are necessary (i) to request that such court or other tribunal enter an Order (x) directing the parties to litigate the dispute in the United States Bankruptcy Court for the Southern District of New York, or (y) to the effect that the Contemplated Transactions are exempt from any requirement to pay Transfer Taxes and any Tenant Purchase Option Law, as applicable, and/or (ii) to request such other relief, including, to the extent necessary, enabling LBHI, Seller or their Affiliates, as the case may be, to join such proceedings as a party or intervenor, as is necessary to obtain a determination that the Contemplated Transactions are exempt from the requirement to pay any Transfer Taxes. In addition, upon the request of the Buyer Parties, LBHI and Seller shall participate in the resolution of any controversy with a Governmental Authority or otherwise regarding the applicability of the exemption of any of the Contemplated Transactions from Transfer Tax or Tenant Purchase Option Laws.
- (c) LBHI shall use Commercially Reasonable Efforts to secure a favorable ruling in any proceeding described in paragraphs (a) and (b) of this <u>Section 13.1.5</u>; <u>provided</u>, <u>however</u>, that none of the Lehman Entities or Seller has made any representation, warranty or guaranty with respect to

- the outcome of any such Proceeding and in no event will any Lehman Entity or Seller have any obligation or Liability with respect to the outcome of such Proceeding.
- (d) Upon the request of the Buyer Parties, subject to Seller's and LBHI's review and approval (not to be unreasonably withheld), LBHI and Seller shall, and shall cause their Affiliates to, execute and deliver to the Buyer Parties transfer tax returns (signed and sworn), prepared substantially in accordance with past practice unless otherwise required by applicable Law, as the Buyer Parties or the Buyer Designees may reasonably require at or after the Initial Closing.
- 13.2 <u>Communications with Governmental Authorities.</u> Between the date hereof and the Initial Closing Date (with respect to all Transferred Assets or Transferred Subsidiary Assets other than any Deferred Closing Assets subject to an Extension Closing in accordance with <u>Section 2.5</u>) and any Extension Closing (with respect to the applicable Deferred Closing Assets subject to any Extension Closing):
 - Subject to compliance with applicable Laws, each Party shall give reasonable prior notice to the other Parties of any communication from any Governmental Authority regarding any of the Contemplated Transactions. If any of the Parties or Affiliates thereof receive a request for additional information or documentary material from any such Governmental Authority with respect to the Contemplated Transactions, then, subject to compliance with applicable Laws, such Party will endeavor in good faith to make, or cause to be made, as soon as reasonably practicable and after consultation with the other Parties, an appropriate response in compliance with such request; provided, that the Buyer Parties and Buyer Designees, on the one hand, and Seller and the Lehman Entities, on the other hand, shall be required to consider in good faith the views of the other Parties in connection with any such response; and provided, further, that when reasonably practicable, the Buyer Parties and Buyer Designees, on the one hand, and Seller and the Lehman Entities, on the other hand, shall not independently participate in any meeting or conversation, or engage in any substantive conversation with any Governmental Authority in respect of any filings, requests for additional information or documentary material, or inquiry without giving the other Parties prior notice of the meeting or conversation and, unless prohibited by such Governmental Authority or applicable Law, the opportunity to attend and/or participate in such meeting or conversation.
 - 13.2.2 The Parties shall, with respect to any threatened or pending Order (including a preliminary or permanent injunction) that would adversely affect the ability of the Parties to consummate the Contemplated Transactions, use their respective Commercially Reasonable Efforts to prevent the entry, enactment or promulgation thereof, as the case may be.

81

13.3 No Control of Operations. Nothing in this Agreement shall give any Buyer Party, directly or indirectly, the right to control or direct the operations of the Archstone Entities or the Joint Ventures prior to the Initial Closing Date, provided that the Parties acknowledge and agree that this Section 13.3 shall not limit or otherwise restrict the rights of the Buyer Parties set forth in this Agreement, including Sections 12.1, 12.2 and 12.3.

13.4 <u>Regulatory Authorizations</u>.

- The Buyer Parties, Seller and LBHI shall reasonably cooperate with one another (a) in determining whether any action by or in respect of, or filing with, any Governmental Authority is required, or any actions, consents, approvals or waivers are required to be obtained from third parties in connection with the consummation of the Contemplated Transactions, and (b) subject to the terms and conditions of this Agreement (and except as otherwise provided in Section 13.4.2), in taking such actions or making any such Filings, furnishing information required in connection therewith and seeking timely to obtain any such actions, consents, approvals or waivers. Except as otherwise provided herein, each Party shall bear its respective costs and expenses incurred in connection with obtaining such actions, consents, approvals and waivers.
- In furtherance, and not in limitation, of the foregoing, the Parties shall use their respective Commercially Reasonable Efforts to (a) make or cause to be made all Filings required of each of them or any of their respective Affiliates under the HSR Act or any other Antitrust Laws with respect to the Contemplated Transactions as promptly as practicable, (b) comply at the earliest practicable date with any request under the HSR Act or other Antitrust Laws for additional information, documents, or other materials received by each of them or any of their respective Subsidiaries or Affiliates from the FTC, the Antitrust Division or any other Governmental Authority in respect of such Filings or the Contemplated Transactions, and (c) cooperate with each other in connection with any such Filing and in connection with resolving any investigation or other inquiry of any of the FTC, the Antitrust Division or other Governmental Authority under any Antitrust Laws with respect to any such Filing or the Contemplated Transactions. Each such Party shall use its Commercially Reasonable Efforts to furnish to the other Parties all information required for any application or other Filing to be made pursuant to any applicable Law in connection with the Contemplated Transactions (unless the furnishing of such information would (i) violate the provisions of any applicable Law or confidentiality agreement, or (ii) cause the loss of attorney-client privilege with respect thereto, provided that each Party shall seek to furnish the other Parties with the substance of any such communication, whether by redacting parts of such material communication or otherwise, so that such communication would not violate applicable Law or cause the loss of the attorney-client privilege with respect thereto). Each Party shall promptly inform the other Parties hereto of any oral communication with, and provide copies of written communications with, any Governmental Authority

82

regarding any such Filings. No Party hereto shall independently participate in any formal meeting with any Governmental Authority in respect of any such Filings, investigation, or other inquiry without giving the other Parties prior notice of the meeting and, to the extent permitted by such Governmental Authority, the opportunity to attend and/or participate. Subject to applicable Law, the Parties will consult and cooperate with one another in connection with any analyses, appearances, presentations, memoranda, briefs, arguments, opinions and proposals made or submitted by or on behalf of any Party relating to Proceedings under the HSR Act or other Antitrust Laws. Notwithstanding anything to the contrary contained in this Agreement, no Party shall be obligated to (1) agree to divest, hold separate or otherwise restrict the use or operation of any material business or assets of Seller, the Archstone Entities, the Equity Residential Parties or AVB, (2) consent to any other material structural or conduct remedy, or (3) agree to any Order regarding Antitrust Laws with respect to the Contemplated Transactions that would have a material impact on the Equity Residential Parties, the Archstone Entities or AVB

13.5 Severance Costs; Bonuses.

From and after the Initial Closing, the Buyer Parties (a) shall assume, comply with and perform, in accordance with their terms, all agreements between any Archstone Entity, on the one hand, and any officer, director or employee of any Archstone Entity, on the other hand, relating to (i) employment, severance, change-of-control or retention bonus obligations pursuant to the Contracts listed on Schedule 6.10 (collectively, the "Severance Obligations"), and (ii) cash incentive or other bonuses required to be paid in the ordinary course payable in 2013 or later for services provided in 2013 or payable in 2012 or 2013 for services provided during or prior to 2012 (together, the "Ordinary Course Bonuses"); provided, however, that the Buyer Parties shall not be responsible for bonus amounts payable in 2012 or 2013 (whether payable before or after the Initial Closing) with respect to services provided during or prior to 2012 that exceed in the aggregate \$28,900,000 for corporate associate annual bonuses, Performance Participation Plan awards, cash in lieu awards (options and RSUs) made before or during 2012, EVP operations bonuses, bonuses for acquisitions or dispositions completed during 2012, DIP and other special bonuses for projects fully completed during 2012, or any other bonuses (other than ordinary course bonuses paid to onsite personnel) payable to employees of any Archstone entity in 2012 or 2013 with respect to services provided during or prior to 2012 (collectively, "2012 Bonuses," the estimated amounts of which bonuses are summarized on Schedule 13.5.1), it being understood, for clarity, that the term "2012 Bonuses" excludes (x) additional compensation payable to Scot Sellers under his employment agreement at the daily rate stated therein in accordance with ordinary payroll practices and (y) DIP awards for projects not fully completed before January 1, 2013 (bonuses described in clauses (x) and (y) "Other Bonuses"), and (b) shall pay, when due, any amounts that become payable

83

under the Severance Obligations as a result of the consummation of the Contemplated Transactions and the Ordinary Course Bonuses.

- 13.5.2 Prior to the Initial Closing, Seller and LBHI shall:
 - (a) pay, or shall provide to the applicable Archstone Entities funds sufficient to pay (following receipt of such funds from the Lehman Entities through capital contribution made at least one (1) Business Day prior to the Initial Closing), (i) all bonus, contractual severance, change-of-control or retention bonus obligations owed to each officer, director or employee of any Archstone Entity, including any payments to be made pursuant to the plan set forth on Schedule 13.5.2, other than (A) the Severance Obligations, (B) Ordinary Course Bonuses payable in 2013 for services provided in 2013, (C) amounts of 2012 Bonuses up to an aggregate of \$28,900,000, and (D) the Other Bonuses; plus (ii) an amount equal to the employer portion of all applicable Taxes and withholdings related to any such payments described in clause (i), including all FICA, workers' compensation amounts, unemployment-related amounts, Medicare-related Taxes and other withholdings required to be paid by any of the Archstone Entities pursuant to applicable Law; and
 - (b) cause, using the funds of the Lehman Entities, to be delivered to Seller hereunder, all Equity Interests in any Archstone Entity that are held by any officer, director or employee of Seller or any Archstone Entity to be redeemed, terminated and discharged without any further rights of any such Person in any of the Archstone Entities. For purposes of clarity, (i) if any amounts that LBHI and Seller are required to pay pursuant to Section 13.5.2(a) shall have been paid by the applicable Archstone Entity prior to the Initial Closing, LBHI and Seller shall nevertheless contribute funds to the applicable Archstone Entities in the amounts so previously paid, and (ii) LBHI acknowledges and agrees that all such obligations set forth in this Section 13.5.2 must be satisfied at least one (1) Business Day prior to the Initial Closing and in no event shall LBHI use the Cash Purchase Price to satisfy the obligations under this Section 13.5.2.
- 13.6 Series O Preferred Units and other Preferred Units.
- 13.6.1 Subject to any redemptions of any Series O Preferred Units in accordance with the terms thereof in compliance with Section 12.1 upon the exercise of put or redemption rights by the holders thereof following the date hereof, each of the Buyer Parties acknowledges and agrees that all of the Preferred Units shall remain outstanding at the Initial Closing and all tax protection obligations arising from the consummation of the Contemplated Transaction, including the transactions specified on Schedule 2.6, and all obligations occurring from and after the Initial Closing with respect to the Series O Preferred Units shall be the responsibility of the Buyer Parties (and neither Seller nor the Lehman Entities shall have any responsibility with respect

- thereto). The Buyer Parties hereby agree to cause Archstone to have sufficient assets to enable it to redeem the Preferred Units in accordance with the terms of the Preferred Units and to cause Archstone to perform when due all obligations that may arise under the certificate of designations of the Preferred Units as a result of the consummation of the Contemplated Transactions or from actions or inactions occurring after the Initial Closing.
- ERPOP and AVB hereby agree, and shall cause their respective Affiliates, (a) to honor in a timely manner any obligation of the Archstone Entities (x) to redeem, pay or otherwise retire any Preferred Units in accordance with Section 13.6.1 and (y) to make any payments required under any Tax Protection Agreement, (b) not to, without the express written consent of any applicable holder of Preferred Units, impede or interfere with the option, to the extent such option exists on the date hereof, of holders of any Preferred Units to (i) request and receive a cash redemption in accordance with the terms of the Preferred Units or (ii) elect that their respective Preferred Units remain outstanding in accordance with the terms of the Preferred Units, (c) not to make any offer, in each case, to less than all of the holders of (i) Series O Preferred Units, (ii) Series P Preferred Units of Archstone, (iii) Series Q Preferred Units of Archstone, and (iv) Series I Preferred Units of Archstone Multifamily Series I Trust, and (d) not to otherwise breach any of the Buyer Parties' obligations with respect to any Preferred Units.
- Affiliates shall issue any press release or public statement concerning this Agreement or the Contemplated Transactions without obtaining the prior written approval of the other Parties (which approval the other Parties shall not unreasonably withhold, condition or delay). Each of the Parties acknowledges and agrees that, on or following the date hereof, the Parties or certain of their Affiliates intend to make a public announcement with respect to the Contemplated Transactions, which shall be subject to this Section 13.7; provided, however, that any such public announcement, press release or other public disclosure shall be made in the form, and include only the content, reasonably agreed upon by the Parties prior to the date hereof. Notwithstanding the foregoing, a Party or its applicable Affiliates may make such disclosure as it determines in good faith is required by applicable Law or Order, or by an obligation pursuant to any agreement with any national securities exchange or national securities association of the United States or any other jurisdiction, including disclosures in connection with an AVB Equity Offering or an EQR Equity Offering; provided that, prior to issuing any such press release or public statement or other disclosure that contains any material information not contained in any prior press release, public statement, or other public disclosure, such Party or its applicable Affiliates shall advise the other Parties of such press release or public statement and shall discuss the contents of the disclosure with the other Parties. Each Party and its Affiliates also may make announcements to its employees that are consistent with the public disclosures made by such Party.

13.8 <u>Tax Matters</u>.

- At the Initial Closing or, if due thereafter, promptly when due, Transfer Taxes applicable to, arising out of or imposed upon (i) the Parties in connection with the Contemplated Transactions, including the transfers specified on <u>Schedule 2.6</u>, and (ii) the Archstone Entities with respect to sales of Designated Properties prior to the Initial Closing, shall be paid 75% by the Buyer Parties and 25% by Seller; <u>provided</u> that in no event shall Seller's payment obligation pursuant to this <u>Section 13.8.1</u> exceed \$20,000,000, which \$20,000,000 cap amount shall include any reasonable legal fees and accounting fees incurred by Seller in connection with such Transfer Taxes, any costs incurred by the Lehman Entities pursuant to <u>Section 13.8.3</u> or any challenge related to whether such Transfer Taxes are due or the defense thereof or action taken by LBHI and Seller pursuant to <u>Section 13.1.5</u>.
- 13.8.2 For purposes of clarity, Seller and the Lehman Entities shall be responsible for and shall pay when due all income Taxes on gain or income recognized by them or otherwise allocable to them in connection with the sale of the Transferred Assets or the Transferred Subsidiary Assets related to the consummation of the Contemplated Transactions.
- 13.8.3 LBHI and Seller hereby agree to reasonably cooperate with the Buyer Parties or any Buyer Designee to (a) implement the Contemplated Transactions in a manner so as to utilize any exemption from the application of any Transfer Taxes to the Contemplated Transactions that may be available under Section 1146 of the Bankruptcy Code and arising from the plan of reorganization of LBHI and its Affiliates, and (b) assist in defending any challenges to such exemption with the costs of such defense to be allocated between Seller and the Buyer Parties as set forth in Section 13.8.1; provided, that except as provided in Section 13.8.1, neither LBHI nor Seller shall be required to incur any risk, Liability, cost or expense pursuant to this Section 13.8.3 unless such risk, Liability, cost or expense is subject to indemnification by the Buyer Parties pursuant to Section 14.2.1.
- 13.8.4 Seller shall prepare and file or cause to be prepared and filed all Tax Returns of the Archstone Entities to the extent that such Tax Returns are due on or prior to the Initial Closing (or, solely with respect to Archstone Entities not sold at the Initial Closing, Extension Closing, as the case may be). All such income Tax Returns filed after the date hereof and on or prior to Initial Closing (or Extension Closing, as the case may be) shall be provided to the Buyer Parties in draft form no later than fifteen (15) days prior to filing and Seller agrees to incorporate reasonable changes requested by the Buyer Parties if such changes to the extent not made reasonably could be expected to result in a material increase in the Buyer Parties or Buyer Designees' Tax liabilities after the Initial Closing (or Extension Closing, as the case may be). The Buyer Parties or Buyer Designees shall prepare and file or cause to be prepared and filed all other Tax Returns of the Archstone Entities that are due after the Initial Closing (or Extension Closing, as the case may be). Seller shall provide the Buyer Parties or Buyer Designees such information and

86

render the Buyer Parties such assistance as may reasonably be requested in order to ensure the proper and timely completion and filing of such Tax Returns. With respect to any income Tax Returns of the Archstone Entities for periods ending at or prior to the Initial Closing Date but due after the Initial Closing Date (or Extension Closing Date, as the case may be), the Buyer Parties shall provide to Seller drafts of such income Tax Returns no later than fifteen (15) days prior to the due date of such return and shall incorporate reasonable changes requested by the Buyer Parties to such drafts so long as such changes are consistent with Law and past practice and applicable Law. To the extent permitted by Law, such income Tax Returns shall be prepared utilizing a "closing of the books" convention as of the Initial Closing Date or applicable Extension Closing Date, as applicable. At the request of Seller, the Buyer Parties shall prepare or cause to be prepared, at no cost to Seller or the Lehman Entities, all income Tax Returns of Seller for the taxable year of Seller ending December 31, 2012, the taxable year of Seller that includes the Initial Closing Date and any taxable year of Seller that includes an Extension Closing Date, provided that the Seller agrees to make all available extension filings requested by the Buyer Parties and that the Seller shall cooperate fully as and to the extent reasonably requested by the Buyer Parties in connection with such preparation. The Buyer Parties shall not have any Liability in respect of the preceding sentence except to the extent caused by the Buyer Parties of the Buyer Parties engage one of the "Big Four" accounting firms to prepare such Tax Returns.

The Buyer Parties and Seller shall cooperate fully as and to the extent reasonably requested by the other Parties, in connection with the filing of Tax Returns and any audit, litigation or other proceeding with respect to Taxes of the Archstone Entities. Such cooperation shall include the retention and the provision of records and information that are reasonably relevant to any such audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided hereunder. Buyer Parties and Seller agree (i) to retain all books and records with respect to Tax matters pertinent to the Archstone Entity relating to any Taxable period beginning before the Initial Closing (or Extension Closing, as the case may be) until the expiration of the applicable statute of limitations (and, to the extent notified by the Buyer Parties or Seller, any extensions thereof) of the respective taxable periods, and to abide by all record retention agreements entered into with any taxing authority, (ii) to give the other parties reasonable written notice prior to transferring, destroying or discarding any such books and records and, to the extent requested by such other party or parties to allow any such requesting Party to take possession of such books and records, and (iii) that from the date hereof, Seller shall (and LBHI shall cause Seller to) provide access and cooperation to the Buyer Parties to all information of Seller, the Archstone Entities or the Lehman Entities related to Taxes, including the Tax Protection Agreements (including

87

underlying calculations, allocations and methodologies related to compliance therewith), requested by the Buyer Parties (and Seller and LBHI shall not, and shall cause the Archstone Entities not to, assert any privilege as grounds for not providing such information related to Taxes to the Buyer Parties).

- 13.8.6 The Parties agree that the Initial Closing Date shall be the treated as the purchase date for all income tax purposes and other purposes for transactions occurring at the Initial Closing.
- 13.9 <u>Asset Acquisition Statement</u>. Each Buyer Party shall prepare and deliver to Seller when available a draft allocation schedule, together with accompanying draft exhibits, if any (collectively, the "<u>Asset Acquisition Statement</u>"), allocating the sum of such Buyer Party's respective portion of the Cash Purchase Price, the fair market value of the ERPOP Equity Consideration as of the Initial Closing (in the case of ERPOP), the fair market value of the AVB Equity Consideration as of the Initial Closing (in the case of AVB), such Buyer Party's respective portion of all other liabilities treated as purchase price for federal income tax purposes among the Transferred Assets and the Transferred Subsidiary Assets acquired by such Buyer Party. The content of the Asset Acquisition Statement shall not be binding on any party and the Buyer Parties shall in no event be prohibited from using different allocations than presented therein. The Buyer Parties shall use

reasonable efforts to provide the Asset Acquisition Statement prior to the due date of any material income Tax Returns of Seller for which such Asset Acquisition Statement is required. Nothing in this Agreement (including the values set forth on Schedule 2.5 or Schedule 2.6, which the Parties acknowledge are merely for the purposes of implementing the closing conditions and transaction steps for the transactions contemplated by this Agreement) shall be treated by the Parties as an agreement or a deemed agreement as to the allocation of the Purchase Price to any of the Transferred Assets or any of the Transferred Subsidiary Assets for income Tax purposes and the Parties shall be free to file any income Tax Return allocating the Purchase Price to any acquired Transferred Assets or Transferred Subsidiary Assets as each determines in its sole discretion.

13.10 <u>Transaction Expenses</u>.

13.10.1 Except as otherwise expressly set forth in this Agreement (including the provision in Section 13.8.1 for sharing Liabilities for Transfer Taxes), the Buyer Parties shall be responsible for the payment of (a) all fees and expenses with respect to the Contemplated Transactions that are incurred by the Buyer Parties or their respective Affiliates and Representatives, including any legal expenses of such Persons and all investment-banking and similar fees and expenses incurred by such Persons, (b) all fees required to obtain any consent to the Contemplated Transactions, (c) all Severance Obligations and Ordinary Course Bonuses, (d) accounting fees and expenses incurred in connection with the preparation of audited financial statements to enable the Buyer Parties to comply with Rule 3-14 under Regulation S-X and other accounting services agreed to be provided by Buyer Parties and Seller (including any accounting fees and expenses incurred in connection with Section 13.12), regardless of whether accruing prior to or after the date of this Agreement but without

88

duplication of the obligations under this Agreement and the Reimbursement Agreements, (e) all costs and expenses incurred by Seller, the Lehman Entities, or any Archstone Entity in connection with the asset transfer transactions set forth on Schedule 2.6 that are in excess of the costs and expenses that would have been incurred by such Persons in connection with an entity level transaction involving the sale of the Equity Interests in Seller by the Lehman Entities (an "Entity Level Transaction") (other than (i) Transfer Taxes, the sharing of liability for which shall be in accordance with Section 13.8.1, and (ii) any legal fees incurred by such Persons in reviewing and commenting on the structure of, or documents implementing, such asset transfers) (and, at the request of Seller, the amount of the costs and expenses that the Buyer Parties are required to reimburse pursuant to this Section 13.10.1(e) shall be advanced to Seller before such costs or expenses are incurred), (f) all actual and documented legal and accounting fees and expenses and other costs (with legal and accounting fees being based on standard hourly billing) (but not compensation paid to underwriters or legal fees of Archstone, Inc., and (g) all reasonable out-of-pocket costs incurred by Seller, the Lehman Entities or any Archstone Entity in connection with Sections 2.5 and 13.1.5 (and, at the request of Seller, the amount of the costs and expenses that the Buyer Parties are required to reimburse pursuant to this Section 13.10.1(g) shall be advanced to Seller before such costs or expenses are incurred); and

- 13.10.2 except as expressly set forth in Section 13.10.1, LBHI and Seller shall be responsible for the payment of (a) all fees and expenses with respect to the Contemplated Transactions that are incurred by any of them, any of their Affiliates, any of the Archstone Entities or any of their respective Representatives, including all legal expenses of such Persons and all investment-banking and similar fees and expenses (including payable to the Persons set forth in Schedules 4.5, 5.5 and 6.11), (b) all costs for which LBHI or Seller are responsible for pursuant to Section 13.8.1, and (c) all costs and expenses incurred in connection with the obligations described in Section 13.5.2.
- 13.11 <u>Confidentiality.</u> From and after the Initial Closing through the second (2nd) anniversary of the date hereof, Seller and LBHI shall, and shall cause their respective Affiliates to, hold, and shall use their respective Commercially Reasonable Efforts to cause their respective Representatives to hold, in confidence any and all information, whether written or oral, concerning the Business or any Archstone Entity, except to the extent (i) that Seller and LBHI can show that such information (a) is generally available to, and known by, the public through no breach of this <u>Section 13.11</u> by Seller, the Lehman Entities, or any of their respective Affiliates or their respective Representatives; (b) is lawfully acquired by Seller, the Lehman Entities, any of their respective Affiliates or their respective Representatives from and after the Initial Closing from sources which are not prohibited from disclosing such information by a legal, contractual or fiduciary obligation; or (c) is from and after the Initial Closing independently conceived or

89

developed by Seller, the Lehman Entities, any of their respective Affiliates or their respective Representatives without use of or reference to any information concerning the Business or any Archstone Entity, or (ii) that such information is disclosed in connection with any litigation arising from or related to this Agreement or the Transaction Documents or the transaction contemplated hereby or thereby (or settlement discussions related thereto). If Seller and/or LBHI, or any of their respective Affiliates or their respective Representatives are compelled to disclose any information by judicial or administrative process or by other requirements of Law, Seller and/or LBHI shall promptly notify the Buyer Parties in writing and shall disclose only that portion of such information which Seller and/or LBHI are advised by their respective counsel in writing is legally required to be disclosed; provided, that Seller and/or LBHI and the Buyer Parties shall use their respective Commercially Reasonable Efforts to preserve the confidentiality of such information, including, by cooperating to obtain an appropriate protective order or other reasonable assurance that confidential treatment will be accorded to such information.

- 13.12 3-14 Financial Statements. Upon reasonable request of a Buyer Party at any time prior to the Initial Closing and at the sole cost and expense of the requesting Buyer Party under this Agreement or in accordance with that certain reimbursement agreement, dated as of October 29, 2012, between EQR and Seller (the "EQR Reimbursement Agreement") and that certain reimbursement agreement, dated as of November 6, 2012, between AVB and Seller (together with the EQR Reimbursement Agreement, the "Reimbursement Agreements"), Seller and LBHI shall, and shall cause the accounting personnel of the Archstone Entities to, use Commercially Reasonable Efforts to as promptly as practicable:
 - prepare statements of revenue and certain expenses with respect to the properties to be acquired by such Buyer Party or any Buyer Designee in the Contemplated Transactions for the year ending December 31, 2012;
 - 13.12.2 request the independent accountants of the Archstone Entities to audit such statements related to the year ending December 31, 2012, so as to enable such Buyer Party to continue to comply with the requirements of Rule 3-14 of Regulation S-X;
 - 13.12.3 take all reasonably requested actions necessary to enable the Buyer Parties or their respective Affiliates or Representatives to conduct a full audit of the books and records of the Archstone Entities and the Joint Ventures relating to the operations and financial results of the Archstone Entities and the Joint Ventures, to the extent required to comply with applicable securities Laws applicable to the Buyer Parties or their respective Affiliates, including Regulation § 210.3-05, Regulation § 210.3-09 or Regulation § 210.3-14 promulgated under the Exchange Act;

13.12.4 take all reasonably requested actions necessary to assist the Buyer Parties and their respective Affiliates in preparing all Filings required to be made by the Buyer Parties with the SEC, including, at the Buyer Parties' request, the delivery of an audit inquiry letter regarding pending litigation and other matters in a form reasonably requested by the Buyer Parties (the "Audit Inquiry Letter") to counsel to the Archstone Entities prior to the Initial

90

Closing and deliver to the Buyer Parties an executed letter from such counsel in response to the Audit Inquiry Letter as soon as reasonably practicable thereafter; and

13.12.5 cooperate reasonably with the Buyer Parties and their respective Affiliates in connection with any audit of the Archstone Entities or the Joint Ventures by the Archstone Entities, the Joint Ventures or their respective Representatives or by Buyer or its Affiliates or Representatives, including, if required by any of the Buyer Parties' registered independent accounting firm in order to complete the audit, (a) causing the applicable employees of the Archstone Entities and the Joint Ventures to answer a standard SAS 99 questionnaire from any Buyer Party's registered independent accounting firm or from the Archstone Entities' registered independent accounting firm and to deliver a representation letter in form and scope reasonably acceptable to Buyer and Buyer's registered independent accounting firm, and (b) causing the auditors of the Archstone Entities to deliver comfort letters, consents and other documentation reasonably requested by any Buyer Party.

13.13 <u>Debt Financing</u>.

Each Buyer Party shall use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to arrange and consummate its respective Debt Financing on the terms and conditions described in its respective Debt Commitment Letter, including using reasonable best efforts to (a) negotiate and enter into definitive agreements with respect thereto on terms and conditions contained therein, (b) satisfy on a timely basis all terms, covenants and conditions applicable to such Buyer Party in such definitive agreements, and (c) consummate the Debt Financing at or prior to the Initial Closing Date; provided, however, that if a Buyer Party has raised through alternative sources sufficient funds to meet its obligations to pay its portion of the Cash Purchase Price and any costs or expenses incurred by such Buyer Party in connection with the consummation of the Contemplated Transactions without any proceeds under such Debt Financing, such Buyer Party shall have no obligation to arrange Debt Financing on the terms and conditions described in its respective Debt Commitment Letter or otherwise. In the event any portion of the Debt Financing of either Buyer Party becomes unavailable on the terms and conditions contemplated in the applicable Debt Commitment Letter, and to the extent such Buyer Party will not have sufficient funds to meet its obligations to pay its portion of the Cash Purchase Price without some or all of the proceeds under such Debt Financing, such Buyer Party shall use its reasonable best efforts to arrange to obtain any such portion from alternative sources on comparable or more favorable terms to such Buyer Party will furnish correct and complete copies of all definitive agreements with respect to the Debt Financing to Seller promptly upon their execution. No Buyer

91

Party shall amend or alter, or agree to amend or alter, the Debt Commitment Letters in any manner that would prevent or materially impair or delay the consummation of the Contemplated Transactions without the prior written consent of Seller.

- 13.13.2 Each Buyer Party shall keep Seller informed on a reasonably current basis in reasonable detail of the status of its efforts to arrange its respective Debt Financing, and shall not permit any material amendment or modification to be made to, or any waiver of any material provision or remedy under, the Debt Commitment Letter if such amendment would or would be reasonably expected to materially and adversely affect or delay in any material respect such Buyer Party's ability to consummate the Contemplated Transactions, without first obtaining Seller's prior written consent (not to be unreasonably withheld or delayed).
- 13.13.3 Notwithstanding anything to the contrary in this Agreement, the Buyer Parties acknowledge and agree that their obligations hereunder are not conditioned in any manner upon their obtaining any financing, and the existence of any conditions contained in their respective Debt Commitment Letters shall not constitute, or be construed to constitute, a condition to the consummation of the Contemplated Transactions; provided, however, that the foregoing shall not limit in any manner the ability of the any Party to terminate this Agreement in accordance with Article 15 and the sole remedy for any such termination shall be as described in Section 15.3.
- 13.13.4 Seller agrees to provide, and shall cause the Archstone Entities and its and their Representatives to use Commercially Reasonable Efforts to provide, all reasonable cooperation in connection with the Debt Financing and any debt, equity or other offering or source of financing by AVB, ERPOP or EQR prior to the Initial Closing as may be reasonably requested by the Buyer Parties (provided that such requested cooperation does not unreasonably interfere with the ongoing operations of the Business of the Archstone Entities), including (a) participation in meetings with potential lenders and underwriters, upon reasonable prior notice by the Buyer Parties, (b) furnishing Buyer Parties with pertinent information available to the Archstone Entities and their Representatives regarding the Archstone Entities and the Transferred Assets and the Transferred Subsidiary Assets, subject to the terms and conditions of the applicable confidentiality agreements, (c) assist the Buyer Parties, to the extent reasonably necessary, in obtaining customary accountants' comfort letters and consents from the Archstone accountants, and (d) assist with the preparation of customary materials in connection with the Debt Financing or any such other debt, equity or other offering or source of financing by AVB or ERPOP to the extent such documents described in this clause (d) contain disclosure reflecting or referring to any of the Archstone Entities or Joint Ventures; provided, that performance of Seller's obligations pursuant to this Section 13.13.4 shall not cause Seller to incur any unreimbursed Liability.

92

13.14 Representative Indemnification.

13.14.1 From and after the Initial Closing, the Buyer Parties shall, and shall cause the Archstone Entities to, honor and perform all obligations of the Archstone Entities to any individual who at the Initial Closing Date is, or at any time prior to the Initial Closing Date was, a Representative of Seller or any Archstone Entity (each, an "Archstone Indemnitee") with respect to all rights to indemnification and exculpation (including the advancement of expenses) from Liabilities for acts or omissions occurring at or prior to the Initial Closing Date as provided in (a) the Organizational Documents of the Archstone Entities (as in effect on the date hereof), and (b) any indemnification agreements or arrangements of any Archstone Entity with respect to the

Archstone Indemnitees existing on the date hereof, which shall survive the consummation of the Contemplated Transactions and continue in full force and effect in accordance with their respective terms; provided, however, that the Buyer Parties and Archstone Entities shall not be required to indemnify the Archstone Indemnitees for claims related to the matters set forth in Sections 14.2.2(c), 14.2.2(d), 14.2.2(e) and 14.2.2(f) to the extent that LBHI is responsible for indemnifying the Buyer Indemnified Parties pursuant to such sections.

- 13.14.2 The provisions of this Section 13.14 are (a) intended to be for the benefit of, and shall be enforceable by, each Archstone Indemnitee, his or her heirs and his or her Representatives, and (b) in addition to, and not in substitution for, any other rights to indemnification or contribution that any such Person may have by contract or otherwise. The obligations of the Buyer Parties and the Archstone Entities under this Section 13.14 shall not be terminated or modified in such a manner as to adversely affect the rights of any Archstone Indemnitee to whom this Section 13.14 applies unless (i) such termination or modification is required by applicable Law or (ii) the affected Archstone Indemnitee shall have consented in writing to such termination or modification (it being agreed by the Parties that the Archstone Indemnitees to whom this Section 13.14 applies shall be third party beneficiaries of this Section 13.14).
- 13.14.3 In the event that any of the Buyer Parties or the Buyer Designees, any Archstone Entity or any of their respective successors or assigns (a) consolidates with or merges into any other Person and is not the continuing or surviving entity of such consolidation or merger or (b) transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each such case, proper provision shall be made so that the successors and assigns of any such Buyer Party or Buyer Designee, Archstone Entity or their respective successors or assigns, as applicable, shall assume all of the obligations thereof set forth in this Section 13.14.
- 13.14.4 The Buyer Parties shall and do hereby, jointly and severally, indemnify the Seller Indemnified Parties against, and hold them harmless, from any and all

93

Losses arising out of, resulting from or related to any breach of this Section 13.14.

13.14.5 The Buyer Parties further agree that with respect to any Archstone Indemnitee who is employed, retained or otherwise associated with, or appointed or nominated by, LBHI or any of its Affiliates and who acts or serves as a director, officer, manager, fiduciary, employee, consultant, advisor or agent of, for or to Seller or any Archstone Entity, that the Archstone Entities shall be primarily liable for all indemnification, reimbursements, advancements or similar payments (the "Indemnity Obligations") afforded to such Archstone Indemnitee acting in such capacity or capacities on behalf or at the request of LBHI or any of its Affiliates, whether the Indemnity Obligations are created by Law, organizational or constituent documents (as in effect on the date hereof), contract or otherwise. Notwithstanding the fact that LBHI or any of its Affiliates may have concurrent Liability to an Archstone Indemnitee with respect to the Indemnity Obligations, the Buyer Parties hereby agree that in no event shall Buyer Parties or any of their Affiliates have any right or claim against LBHI or any of its Affiliates for contribution or have rights of subrogation against LBHI or any of its Affiliates through an Archstone Indemnitee for any payment made by the Buyer Parties or any of their Affiliates with respect to any Indemnity Obligation. In addition, the Buyer Parties hereby agree that in the event that LBHI or any of its Affiliates (including Seller) pay or advance an Archstone Indemnitee any amount with respect to an Indemnity Obligation, the Buyer Parties will promptly reimburse LBHI or its Affiliate for such payment or advance upon request. None of the foregoing provisions shall be deemed to require the Buyer Parties and Archstone Entities to indemnify the Archstone Indemnifying the Buyer Indemnified Parties pursuant to such sections.

13.15 Employee Matters.

- 13.15.1 Each employee of the Archstone Entities (other than any such employees who are covered by collective bargaining or similar agreements) who remains an employee of the Archstone Entities or their successors or assigns (collectively, the "Continuing Employees") will be credited with his or her years of service with the Archstone Entities (and any predecessor entities thereof) before the Closing Date under any employee benefit plan of the Buyer Parties in which the Continuing Employee becomes entitled to participate to the same extent as such employee was entitled, before the Initial Closing Date, to credit for such service under the applicable Employee Benefit Plan (except to the extent such credit would result in the duplication of benefits and except with respect to benefit accrual under a defined benefit plan).
- 13.15.2 The provisions of this Section 13.15 shall be binding upon and inure solely to the benefit of each of the parties to this Agreement, and nothing in this

94

<u>Section 13.15</u>, express or implied, is intended to confer upon any other Person any rights or remedies of any nature whatsoever under or by reason of this <u>Section 13.15</u>.

- 13.15.3 At the Initial Closing, the individuals set forth on <u>Schedule 13.15.3</u> shall be terminated and upon the execution of a release in accordance with each such individual's agreement with the applicable Archstone Entity, each such individual shall be paid the amounts payable pursuant to such individual's agreement.
- 13.15.4 Upon the written request of the Buyer Parties, Seller shall, and shall cause the applicable Archstone Entities to, provide thirty (30) days prior written notice of the termination of employment of each employee of any Archstone Entity that is a party to a change-in-control agreement; provided, however, that such notice shall be conditional on the consummation of the Contemplated Transactions.
- The Parties acknowledge that (i) the Contemplated Transactions will constitute a change in control under the Contracts set forth in Schedule 6.10, and (ii) in exchange for the senior executives set forth on Schedule 13.15.5 agreeing to remain employed by the Archstone Entities for up to 90 days after the Initial Closing Date upon the request of a Buyer Party, the Buyer Parties agree that (x) any termination of the employment of any such senior executive after such period shall constitute a termination for "good reason", which shall not be subject to cure by the Buyer Parties and as to which any 30-day prior notice shall be waived, (y) any such termination for "good reason" shall entitle such senior executive to collect severance payments and benefits pursuant to such senior executive's applicable employment or change-of-control agreement, and (z) on the Initial Closing Date, the Buyer Parties shall deposit the amounts payable to such senior executives set forth on Schedule 13.15.5 who shall have been requested to remain employed following the Initial Closing Date in accordance with the Contracts set forth in Schedule 6.10 into an escrow account with a third party escrow agent mutually agreed upon by Seller and the Buyer Parties and such amounts payable shall be released from escrow to the applicable senior executives upon the written instructions of both of the Buyer Parties, provided that all such amounts deposited in such escrow account shall be subject to the general creditors of the Buyer Parties.

To the extent timely requested by the Buyer Parties or as required by applicable Law, Seller and LBHI shall, or shall cause the applicable Archstone Entities to, make any filings and shall deliver any notices prepared by the Buyer Parties (with assistance of Seller and LBHI) with respect to any employees of the Archstone Entities that are required to be delivered under WARN or any similar applicable Law in any jurisdiction so that neither the Buyer Parties, the Archstone Entities nor their respective Affiliates shall have any liability under WARN or any similar applicable Law in any jurisdiction with respect to any employees for which the Buyer Parties had requested that such WARN filings or notices be made.

95

13.16 <u>Mutual Release</u>. At the Initial Closing, the Archstone Entities, on the one hand, and LBHI, LCPI and Seller, on the other hand, shall execute a mutual release in substantially the form set forth on <u>Exhibit F</u>, and the Buyer Parties shall acknowledge the release as set forth therein. In addition, each of Seller and LBHI (on behalf of itself and each of the Lehman Entities) hereby waives any and all option and preemption rights that it may have pursuant to the Management Regulations, Luxembourg law or other similar Laws to acquire any assets of the Archstone Entities or the Joint Ventures.

13.17 <u>Certain Intellectual Property Matters.</u>

- 13.17.1 It is acknowledged and agreed that the Buyer Parties are obtaining as Transferred Assets all of LBHI's and Seller's right, title and interest in and to and licenses to use the name "Archstone" and any trademarks, logos, trade names or other Intellectual Property rights owned by Seller, LBHI or any other Persons controlled by LBHI and exclusively used in the Business.
- 13.17.2 From and after the Initial Closing, neither Seller nor any other Person controlled by LBHI shall, and shall cause its Affiliates not to, use any trademark, logo or trade name exclusively used in the Business or by the Archstone Entities or any of their respective Affiliates, or any trademarks, logos or trade names that are confusingly similar thereto or that, to Seller's Knowledge and LBHI's Knowledge, are a translation or transliteration thereof.
- 13.17.3 Without limiting the provisions of Section 13.17.1, as soon as reasonably practicable following the Initial Closing, but not later than thirty (30) days after the Initial Closing Date, Seller shall, and LBHI shall cause Seller and each other Person controlled by LBHI to, take any and all actions as reasonably necessary, including making filings with appropriate Governmental Authorities, to change the name of Seller to remove "Archstone" therefrom.
- Without limiting the provisions of Section 13.17.2 as soon as reasonably practicable following the Initial Closing, but not later than thirty (30) days after the Initial Closing Date, Seller shall, and LBHI shall cause Seller and each other Person controlled by LBHI to, use Commercially Reasonable Efforts to destroy, remove, cause to be removed, or change signage, promotional or advertising material in whatever medium, and stationery and packaging, and take all such other steps as may be required or reasonably appropriate to cease use of all trademarks, logos or trade names owned by the Buyer Parties or the Archstone Entities or otherwise constituting a Transferred Asset.
- 13.18 <u>Hedging Arrangements</u>. From the date of this Agreement until the Initial Closing Date or earlier termination of this Agreement in accordance with Article 15, Seller and LBHI shall not, and Seller and LBHI shall cause the Archstone Entities, the Lehman Entities and its other subsidiaries not to, directly or indirectly, enter into any swap or any other agreement,

96

transaction or series of transactions that hedges or transfers, in whole or in part, the economic consequence of ownership of any capital stock of EQR or AVB or any interest therein, whether any such swap, agreement, transaction or series of transactions is to be settled by delivery of securities, in cash or otherwise, including any hedging arrangement or transfer based on the FTSE NAREIT All Residential Capped Index, the FTSE NAREIT Equity Residential Index, the FTSE NAREIT Equity Apartments Index or any other index of which the capital stock of either of EQR or AVB represents at least 5% of the value at the time such hedging arrangement or transfer is entered into.

13.19 Certain Germany Arrangements. Promptly following a request from the Buyer Parties, Seller and LBHI shall, and shall cause the applicable Archstone Entities to, use Commercially Reasonable Efforts to take all actions necessary to terminate and rescind the contemplated equity investment by any Archstone Entity and any other Person with respect to potential investments in each of the international funds or sub-funds formed pursuant to German, Netherlands or Luxembourg law and relating to the potential investment in assets in Germany (including funds or entities in which Archstone Holdings Germany II LLC and Archstone Management Germany II LLC would indirectly have an interest), including by (i) terminating and rescinding all sale and purchase agreements, including with respect to the Dali asset portfolio, (ii) not approving the subscription agreements or other capital commitments related to such potential investment, and (iii) providing notice thereof to all potential investors.

ARTICLE 14 SURVIVAL; INDEMNIFICATION

Representations) or in any certificate delivered pursuant to this Agreement relating to the representations and warranties (other than the Seller Fundamental Representations and the Buyer Party Fundamental Representations and the Buyer Party Fundamental Representations and the Buyer Party Fundamental Representations) and the covenants contained herein that are to be performed at or prior to the Initial Closing shall terminate as of the Initial Closing. The Seller Fundamental Representations and the Buyer Party Fundamental Representations, and the right to make a claim for indemnification with respect to the Seller Fundamental Representations and the Buyer Party Fundamental Representations, shall survive with respect to claims as to which a Claim Notice has been delivered on or before the first (1st) anniversary of the Initial Closing. All covenants and agreements which apply after the Initial Closing shall survive the Initial Closing in accordance with their terms. All claims related to the matters described in Sections 14.2.2(g) through 14.2.2(i) shall survive with respect to claims as to which a Claim Notice has been delivered on or before the fourth (4th) anniversary of the Initial Closing. It is understood and agreed that (a) after the Initial Closing, the sole and exclusive remedy with respect to any breach of this Agreement shall be a claim for Losses made pursuant to (and subject to the limitations of) this Article 14; provided that notwithstanding the foregoing, nothing in this Agreement shall limit the right of any Buyer Party (i) to pursue an action for or to seek remedies with respect to claims for fraud, or (ii) Seller, or LBHI to seek specific performance or other equitable relief in accordance with Section 15.4; and (b) before the Initial Closing, the Parties shall be entitled only to the termination and other remedies set forth in Article 15, and indemnification under this Article 14 shall not apply.

97

14.2 <u>Indemnification</u>.

14.2.1 Effective as of the Initial Closing and subject to the limitations set forth in this <u>Article 14</u>, the Buyer Parties shall and do hereby, jointly and severally, indemnify the Lehman Entities, Seller, their respective Affiliates and their respective Representatives (together, in each case, with their respective

successors and permitted assigns, the "Seller Indemnified Parties") against, and hold them harmless from, any and all Losses arising out of, resulting from or related to:

- (a) any breach or inaccuracy of any Buyer Party Fundamental Representation made in this Agreement or in any certificate delivered pursuant to this Agreement relating to such Buyer Party Fundamental Representations;
- (b) any breach of, or non-fulfillment or failure to satisfy, any covenant, agreement or obligation of the Buyer Parties in this Agreement or any other Transaction Document (x) that is required to be fulfilled following the Initial Closing, including the matters described in Sections 13.5.1, 13.10.1 and 13.14.4, or (y) any breach of the matters described in Sections 2.6, 13.1, 13.6, 13.8, 13.10.1, 13.12, 13.13.4, 13.14.1 and 13.14.5;
- (c) any obligation of the Buyer Parties with respect to Transfer Taxes under <u>Section 13.8.1</u>;
- (d) any Indemnified Archstone Entity Liabilities;
- (e) the Acquisition Litigation to the extent of any Liability that directly results from the breach by any of the Buyer Parties, or after the Initial Closing the Archstone Entities, or their respective Affiliates or Representatives, of any covenant contained in Section 13.6; and
- (f) any Liabilities relating to a breach by the Archstone Entities of their obligations under the Indemnity Agreement from and after the Initial Closing, except for any such Liabilities described in <u>Sections 14.2.2(e)</u>, <u>14.2.2(e)</u>, <u>14.2.2(e)</u> and <u>14.2.2(f)</u> to the extent that LBHI is responsible for indemnifying the Buyer Indemnified Parties pursuant to such sections.
- 14.2.2 Effective as of the Initial Closing and subject to the limitations set forth in this Article 14, LBHI shall indemnify the Buyer Parties, their respective Affiliates (including the Archstone Entities), and their respective Representatives (together, in each case, with their respective successors and permitted assigns, the "Buyer Indemnified Parties") against, and hold them harmless from, any and all Losses incurred by any Buyer Indemnified Party or by any Archstone Entity (including any Losses described in Section 14.4) to the extent arising out of, resulting from or related to:

98

- (a) any breach or inaccuracy of any Seller Fundamental Representation made in this Agreement or in any certificate delivered pursuant to this Agreement relating to such Seller Fundamental Representations;
- (b) any breach of, or non-fulfillment or failure to satisfy, any covenant, agreement or obligation of Seller or LBHI in this Agreement or any other Transaction Document that is required to be fulfilled following the Initial Closing, including the matters described in Sections 13.5.2 and 13.10.2;
- (c) any claim or Proceeding under any Tax Protection Agreement attributable to any action or inaction taken by Seller or any Archstone Entity prior to the Initial Closing (other than in connection with the Contemplated Transactions or any actions set forth on Schedule 2.6);
- (d) except as provided in Section 14.2.1(e), (i) the matters set forth on Schedule 14.2.2, (ii) any Proceedings brought by former holders of Series A-1 units asserting that the LBO Transaction violated any Tax Protection Agreement, any Organizational Documents of the Archstone Entities or fiduciary or other duties owed to such holders, whether alleged prior to or following the Initial Closing (collectively, the "Acquisition Litigation"), and (iii) the Losses (including legal fees and expenses) described in Section 14.4; provided, however, that for the avoidance of doubt, in respect of this Section 14.2.2(d), with respect to any current holder of Preferred Units, Seller shall only be liable for damages payable to such current holder to the extent the sum of (x) (A) the amounts payable to such holder, plus (B) the liquidation preference of any such Preferred Units held by such holder of such Preferred Units immediately following resolution of any applicable Proceeding, exceeds (y) the amount of principal (plus any accrued but unpaid dividends on the date of redemption) required to redeem such Preferred Units, assuming the date of redemption was the date of such resolution:
- (e) any Proceedings related to the recapitalization of the Archstone Entities in December 2010;
- (f) any claim or Proceeding or other costs occurring as a result of any changes to the capital structure of Seller or the Archstone Entities from the date of the closing of the LBO Transaction to the date of this Agreement, including any unpaid Transfer Tax Liability of Seller or any Archstone Entity as a result of the consummation of the transactions contemplated in the Lehman IPAs, but excluding any costs or expenses paid by Seller or any of the Archstone Entities prior to September 30, 2012;
- (g) any obligation of Seller or LBHI with respect to Transfer Taxes under Section 13.8.1;
- (h) any Excluded Liability; and

99

- (i) any Liabilities relating to any "Post-Closing Claims" (as defined in the Indemnity Agreement) or any other matter set forth in the Indemnity Agreement with respect to which LBHI is required to indemnify the Buyer Indemnified Parties under Sections 14.2.2(e), 14.2.2(d), 14.2.2(e) and 14.2.2(f) above, in each case, other than as a result of a breach from and after the Initial Closing of an obligation of the Archstone Entities.
- 14.2.3 For any matters that are indemnifiable by any Indemnifying Party under Section 14.2.1 or Section 14.2.2, (a) if any Losses with respect to such matters are out-of-pocket Losses, then such Losses shall not be payable to the Indemnified Party until such expense has actually been incurred and an invoice related thereto has been submitted for payment, or (b) if any Losses relate to the settlement of any claim or judgment, then such Losses shall not be payable to the Indemnified Party until the earliest of (i) the time when such settlement amount or judgment-related amount has actually been paid, (ii) with respect to any judgment, when the judgment has become final and non-appealable, and (iii) when a surety bond in respect thereof is required to be posted.

14.3 <u>Indemnification Procedures</u>.

14.3.1 If any Buyer Indemnified Party or Seller Indemnified Party, as the case may be, seeks to make a claim for indemnification (an "Indemnified Claim") pursuant to Section 14.2.1 or Section 14.2.2, as applicable, other than in respect of a Third Party Claim, the Party entitled to indemnification (the

"Indemnified Party.") shall notify the party required to provide indemnity hereunder (the "Indemnifying Party.") in writing of such claim specifying in reasonable detail the factual basis of such claim, stating the amount of Losses (or if not known, a good faith estimate of the amount of Losses) and the method of computation thereof, containing a reference to any and all provisions of this Agreement with respect to which indemnification is being sought (each such notice, a "Claim Notice"). The failure by any Indemnified Party to so notify the Indemnifying Party shall not relieve the Indemnifying Party from liability under this Agreement which it may have to such Indemnified Party under Section 14.2 hereof except to the extent the Indemnifying Party shall have been prejudiced with respect to such individual indemnified claim in any material respect as a result of such failure. The Indemnifying Party and the Indemnified Party shall endeavor to negotiate in good faith a resolution of any dispute described in a Claim Notice and, if not resolved through negotiations within sixty (60) days after the delivery of the Claim Notice, such dispute shall be resolved by litigation as provided in Section 16.5.

14.3.2 In the event that any claim is asserted in writing by any Person other than a Buyer Indemnified Party or a Seller Indemnified Party involving a claim for which

100

an Indemnifying Party may be obligated to indemnify an Indemnified Party (a "Third Party Claim"), the Indemnified Party shall give written notice to the Indemnifying Party promptly (and in any event within thirty (30) days) after the Indemnified Party receives written notification of the facts giving rise to such Third Party Claim, specifying in reasonable detail the factual basis of the Third Party Claim, stating the amount of Losses (or if not known, a good faith estimate of the amount of Losses) and the method of computation thereof, containing a reference to any and all provisions of this Agreement with respect to which indemnification is being sought. The failure of an Indemnified Party to provide notice in accordance with this Section 14.3.2, or any delay in providing such notice, shall not constitute a waiver of that Indemnified Party's claims to indemnification pursuant to Section 14.2, except to the extent that (i) the Indemnifying Party shall have been prejudiced in any material respect as a result of such failure, or (ii) such notice is not delivered to the Indemnifying Party prior to the expiration of the applicable survival period set forth in Section 14.1. Any such notice to the Indemnifying Party of a Third Party Claim shall be accompanied by a copy of any papers theretofore served on or delivered to the Indemnified Party in connection with such Third Party Claim.

In the event of receipt of notice of a Third Party Claim from an Indemnified Party pursuant to this Section 14.3.2, the Indemnifying Party will be entitled to assume the defense and control of such Third Party Claim subject to the provisions of this Section 14.3.2, and the Indemnified Party shall be permitted to participate in the defense thereof and to employ separate counsel of its choice for such purpose at its own expense (except as otherwise provided in the last sentence of this Section 14.3.2). After written notice by the Indemnifying Party to the Indemnified Party of its election to assume the defense and control of a Third Party Claim, the Indemnifying Party shall not be liable to such Indemnified Party for any legal fees or expenses subsequently incurred by such Indemnified Party for its own account. Whether the Indemnifying Party or the Indemnified Party is defending and controlling any such Third Party Claim, it shall select counsel, contractors, experts and consultants of recognized standing and competence, shall take reasonable steps necessary in the investigation, defense or settlement thereof, and shall diligently and promptly pursue the resolution thereof. The party conducting the defense thereof shall at all times act as if all Losses relating to the Third Party Claim were for its own account and shall act in good faith and with reasonable prudence to minimize Losses therefrom. The Indemnified Party shall, and shall cause each of its Affiliates and Representatives to, reasonably cooperate with the Indemnifying Party in connection with any Third Party Claim. Notwithstanding anything to the contrary in this Section 14.3.2(a), (i) in any event the Indemnified Party shall have the right to control, pay or settle any Third Party Claim which the Indemnifying Party, shall have undertaken to defend so long as (1) the Indemnified Party shall also waive any right to indemnification therefor by the Indemnifying Party, and (2)

101

the consent to the entry of any judgment, or the entry into any settlement, compromise or discharge, with respect to any such Third Party Claim, (I) does not impose monetary damages, injunctive or other equitable relief against the Indemnifying Party or any of its Affiliates, and (II) does not contain or involve an admission or statement providing for or acknowledging any liability or criminal wrongdoing on behalf of the Indemnifying Party or any of its Affiliates, and (ii) the Indemnifying Party shall be liable for the fees and expenses of counsel employed by the Indemnified Party (x) for any period after notice thereof during which the Indemnifying Party has failed to assume the defense thereof, and (y) if, based on the advice of counsel, a conflict of interest exists between the Indemnifying Party and the Indemnified Party in respect of the matters subject to such Third Party Claim.

- (b) The Indemnifying Party shall not, without the prior written consent of the Indemnified Party, consent to the entry of any judgment, or enter into any settlement, compromise or discharge, with respect to any Third Party Claim, and the Indemnified Party shall be entitled to withhold its consent to a settlement of, or the entry of any judgment arising from, any Third Party Claims that (i) does not involve as the sole remedy monetary damages that are fully borne by the Indemnifying Party, (ii) does not contain an unconditional release of the Indemnified Party from all liabilities with respect to such Third Party Claim, (iii) imposes injunctive or other equitable relief against the Indemnified Party or any of its Affiliates, or (iv) contains or involves an admission or statement providing for or acknowledging any liability or criminal wrongdoing on behalf of the Indemnified Party or any of its Affiliates. Except as expressly provided in this Agreement, the Indemnified Party shall not consent to the entry of any judgment, or enter into any settlement, compromise or discharge of any Third Party Claim without the prior written consent of the Indemnifying Party, which may be conditioned or delayed in its reasonable discretion.
- (c) If an Indemnifying Party makes any payment on an Indemnified Claim, the Indemnifying Party shall be subrogated, to the extent of such payment, to all rights and remedies of the Indemnified Party to any insurance benefits or other claims or benefits of the Indemnified Party with respect to such claim.
- (d) Notwithstanding anything in this <u>Article 14</u> to the contrary, in no event shall the Buyer Indemnified Parties be required to deliver any notice of any Third Party Claim relating to the matters for which the Buyer Parties are indemnified pursuant to <u>Sections 14.2.2(e)</u>, <u>14.2.2(d)</u>, <u>14.2.2(e)</u> and <u>14.2.2(f)</u> and which Third Party Claim had been made or was in existence prior to the Initial Closing Date.
- 14.3.3 The Parties agree that any indemnification payments made with respect to this Agreement, including under this Section 14.3, shall be treated for all Tax

purposes as an adjustment to the purchase price hereunder unless otherwise required by Law (including by a determination of a Tax authority that, under applicable Law, is not subject to further review or appeal).

- Existing Claims. LBHI acknowledges and agrees that, as of the date of this Agreement, (a) the Archstone Entities have suffered Losses in connection with the Acquisition Litigation, including legal fees and related expenses, and (b) LBHI shall be responsible for indemnifying and holding harmless the Buyer Indemnified Parties in accordance with Section 14.2.2 for all Losses of the Archstone Entities relating to the matters set forth in Section 14.2.2 (including all legal fees and related expenses, subject to Section 14.5, relating to such matters, except that LBHI shall not be required to indemnify the Buyer Indemnified Parties for any legal fees and related expenses, settlement payments or other amounts relating to the matters set forth in Section 14.2.2 to the extent that such legal fees and related expenses relate to services performed on or before September 30, 2012 or such settlement payment or other amounts were paid on or prior to September 30, 2012). In the event LBHI fails to indemnify any Buyer Indemnified Party with respect to the matters set forth in Sections 14.2.2(d) or 14.2.2(e) in accordance with this Article 14, then all of the rights and interests of the Lehman Entities to indemnification pursuant to the terms of the Lehman IPAs with respect to the matters set forth in sections 12.2.2(b) and 12.2.2(c) thereof applicable with respect to periods beginning on or after October 1, 2012, shall, to the extent assignable, be automatically assigned to such Buyer Indemnified Party, without any further action by any Party hereto; provided, however, that after the Buyer Indemnified Parties have recovered the Losses for which LBHI failed to pay, the Buyer Indemnified Parties shall pay the Lehman Entities any remaining proceeds recovered by the Buyer Indemnified Parties pursuant to the Lehman IPAs. For the avoidance of doubt, the Lehman Entities (i) do not make any representation or warranty regarding the effectiveness of such assignment, and (ii) shall be entitled to all of the rights of indemnification pursuant to the terms of the Lehman IPAs (and the Buyer Parties and Archstone Entities shall have no rights thereunder) unless LBHI fails to indemnify any Buyer Indemnified Parties with respect to matters set forth in Sections 14.2.2(c), 14.2.2(d), 14.2.2(e) (but in no event shall the Buyer Parties or Archstone Entities have any rights to indemnification under the Lehman IPAs with respect to any legal fees or related expenses incurred during any period prior to October 1, 2012 or any settlement payments or other amounts paid on or prior to September 30, 2012). For the further avoidance of doubt, the assignment of the rights under the Lehman IPAs shall not be deemed to relieve LBHI of any Liability for failure to indemnify any Buyer Indemnified Parties as required hereunder.
- 14.5 <u>Litigation Cooperation</u>. With respect to the matters in <u>Sections 14.2.2(e)</u>, <u>14.2.2(e)</u>, <u>14.2.2(e)</u>, and <u>14.2.2(f)</u>, Proceedings exist as of the date hereof (the "<u>Existing Proceedings</u>"). Seller shall defend and control the Existing Proceedings, and the Buyer Parties acknowledge (x) that counsel currently defending the Existing Proceedings satisfy the requirements of <u>Section 14.3.2(a)</u> and shall continue to defend the Existing Proceedings and any future litigation or Proceeding arising out of or relating to <u>Sections 14.2.2(e)</u>, <u>14.2.2(e)</u>, and <u>14.2.2(f)</u>, and (y) that LBHI shall not be required to indemnify any Buyer Indemnified Party for any of the fees and expenses of counsel employed by any Buyer Indemnified Party other than the reasonable fees and expenses of one set of counsel for all Buyer Indemnified Parties to monitor (but not actively participate in) the Existing Proceedings. Each Buyer Indemnified Party shall, and shall cause each of the Archstone Entities to, at LBHI's sole cost and expense,

103

reasonably cooperate with Seller and the Lehman Entities in the defense of any Third Party Claim relating to any of the matters in Sections 14.2.2(c), 14.2.2(d), 14.2.2(e), and 14.2.2(f), including (a) promptly informing Seller of the receipt of any notices, correspondence, Filings or other communications with respect to the matters described in Sections 14.2.2(c), 14.2.2(d), 14.2.2(e), and 14.2.2(f), (b) by providing Seller or the Lehman Entities with access to records, documents or other information of the Archstone Entities or any of their Affiliates to the extent relevant to the matters described in Sections 14.2.2(c), 14.2.2(d), 14.2.2(e), and 14.2.2(f), and (c) providing Seller and the Lehman Entities access to employees of the Archstone Entities and making such employees available for interviews and testimony as reasonably requested and upon reasonable notice (provided that such employees shall be made so available upon reasonable notice and for a reasonable time without charge other than out-of-pocket expenses). For the avoidance of doubt, Buyer Parties shall, and shall cause each of the other Buyer Indemnified Parties to, cause the Archstone Entities not to enter into any settlement, compromise or discharge of any claims with respect to the matters described in Sections 14.2.2(c), 14.2.2(d), 14.2.2(e) and 14.2.2(f) without the prior written consent of Seller, which may be conditioned or delayed in its sole discretion.

14.6 Third Party Recovery. In calculating the amount of any Losses payable under this Article 14, there shall be deducted from any such Losses payable to the Indemnified Party an amount equal to any payment received from a third party (including an insurer) by such Indemnified Party in respect of any such Loss (or by an Archstone Entity, in respect of any Losses by the Archstone Entities for which the Indemnified Party is indemnified). If an Indemnified Party recovers an amount from a third party (including an insurer) in respect of a Loss that is the subject of indemnification hereunder after all or a portion of such Loss has been paid by the Indemnifying Party hereunder pursuant to this Article 14, the Indemnified Party shall promptly remit to the Indemnifying Party the excess (if any) of (i) the amount paid by the Indemnifying Party in respect of such Loss, plus the amount actually received from the Third Party in respect thereof, over (ii) the full amount of the Loss.

ARTICLE 15 TERMINATION

- 15.1 <u>Termination of Agreement.</u> Notwithstanding anything to the contrary in this Agreement, this Agreement may be terminated and the Contemplated Transactions may be abandoned at any time prior to the Initial Closing as provided in this <u>Article 15</u>.
 - 15.1.1 Consent. ERPOP, AVB, Seller and LBHI may terminate this Agreement by mutual written consent.
 - 15.1.2 Expiration; Illegality; Prohibition. Either the Buyer Parties or LBHI may terminate this Agreement by written notice to the other Parties:
 - (a) if the Initial Closing has not occurred on or before the Outside Closing Date (as such date may be extended in accordance with Section 2.4) (it being understood that the Buyer Parties may not terminate this Agreement pursuant to this Section 15.1.2(a) unless the closing conditions set forth in Sections 9.1, 9.2, or 10.1 (other than those conditions which by their

104

nature are to be satisfied as part of the Initial Closing) have not been satisfied or they simultaneously pay the Seller Liquidated Damages Amount to LBHI or the applicable Lehman Designee in immediately available funds to a bank account designated in writing by LBHI); provided, however, that the right to terminate this Agreement under this Section 15.1.2(a) shall not be available to LBHI if the conditions set forth in Sections 10.1.1 and 10.1.2 are not satisfied;

- (b) if there is any Law that makes consummation of a Substantial Portion of the Contemplated Transactions illegal or otherwise prohibited; or
- (c) if a Governmental Authority of competent jurisdiction having valid enforcement authority issues a final Order, not subject to appeal, permanently restraining, prohibiting or enjoining any Buyer Party, Seller and/or any Lehman Entity from consummating a Substantial Portion of the Contemplated Transactions, <u>provided</u> that the right to terminate this Agreement pursuant to this <u>Section 15.1.2(c)</u> shall not be available to any Party that has initiated any Proceeding that results in such Order, or that has not taken any and all reasonable actions necessary to oppose, contest and resist, and to have lifted, rescinded or vacated, such order, decree or judgment before it became final and non-appealable.

15.1.3 <u>Breach/Failure to Perform.</u>

- (a) Either of the Buyer Parties may terminate this Agreement by written notice to LBHI in the event of a breach by Seller or any Lehman Entity of any representation, warranty, covenant or agreement under this Agreement, where the effect of such breach would be to cause any of the conditions to the Initial Closing set forth in Sections 10.1.1 through 10.1.6 to not be satisfied, and such breach is not cured by Seller or a Lehman Entity, as applicable, within twenty (20) Business Days following receipt of written notice from any Buyer Party of the breach or alleged breach, which written notice shall state that, unless such breach is cured in accordance with this Section 15.1.3(a), a Buyer Party intends to terminate this Agreement (it being understood that such twenty (20) Business Day cure period (the "Seller Cure Period") shall extend the then-current Outside Closing Date (and the other Buyer Party hereby consents to such termination).
- (b) LBHI may terminate this Agreement by written notice to the Buyer Parties in the event of a breach by the Buyer Parties of any representation, warranty, covenant or agreement under this Agreement, where the effect of such breach would be to cause any of the conditions to the Initial Closing set forth in Sections 11.1.1 through 11.1.6 to not be satisfied, and such breach is not cured by the Buyer Parties within twenty (20) Business Days following receipt of written notice from LBHI of the breach or alleged breach, which written notice shall state that, unless such breach is

105

cured in accordance with this Section 15.1.3(b), that LBHI intends to terminate this Agreement (it being understood that such twenty (20)-Business Day cure period (the "Buyer Cure Period") shall not extend the then-current Outside Closing Date, unless (i) the Buyer Parties have certified that the Buyer Parties are ready, willing and able to, and unconditionally obligated to, proceed immediately with the Initial Closing on or prior to the then-current Outside Closing Date, and (ii) Seller and LBHI have indicated that they are not ready, willing and able to, and unconditionally obligated to, proceed immediately with the Initial Closing as a result of the breach by the Buyer Parties of the conditions set forth in Section 11.1.1 or Section 11.1.2, in which case, the then-current Outside Closing Date shall be extended by the Buyer Cure Period).

15.1.4 <u>Satisfaction of Conditions to Closing.</u>

- Either of the Buyer Parties, on the one hand, or LBHI, on the other hand, may terminate this Agreement by written notice to the other Parties if any event shall occur after the date of this Agreement that shall have made it impossible to satisfy a condition precedent to the terminating Party's obligations to perform its obligations hereunder; provided that (i) Buyer Parties shall not have the right to terminate this Agreement pursuant to this Section 15.1.4(a) if any Buyer Party is then in breach of any representations, warranties, covenants or other agreements hereunder (I) that would result in the closing conditions set forth in Sections 9.1, 9.2 and 11.1.1 through 11.1.6 not being satisfied, and (II) that has not been cured following the Buyer Cure Period, and (ii) LBHI shall not have the right to terminate this Agreement pursuant to this Section 15.1.4(a) if LBHI, Seller or any other Lehman Entity is then in breach of representations, warranties, covenants or other agreements hereunder that (x) would result in the closing conditions set forth in Sections 9.1, 9.2 and 10.1.1 through 10.1.6 not being satisfied, and (y) that have not been cured following the Seller Cure Period (and if either Buyer Party terminates pursuant to this Section 15.1.4(a), the other Buyer Party hereby consents to such termination).
- (b) If all of the conditions precedent to the Buyer Parties' obligations to perform their obligations to consummate the Contemplated Transactions hereunder (other than any Contemplated Transactions relating to Deferred Closing Assets for which an Extension Closing will occur and other than those conditions which by their nature are to be satisfied as part of the Initial Closing) have been satisfied and the Buyer Parties fail to timely perform such obligations by the then-current Outside Closing Date (as extended in accordance with Section 2.4), LBHI may terminate this Agreement.
- 15.2 <u>Effect of Termination</u>. In the event of the termination of this Agreement and the abandonment of the Contemplated Transactions pursuant to <u>Section 15.1</u>:

106

- this Agreement shall forthwith terminate, except that <u>Section 13.7</u>, <u>Section 13.10</u>, this <u>Section 15.2</u>, <u>Section 15.3</u> and <u>Article 16</u>, the obligations of the Buyer Parties set forth on <u>Schedule 12.3</u> and all applicable definitions in this Agreement shall survive such termination;
- 15.2.2 each member of the Buyer Group shall return all documents, work papers and other materials (and all copies thereof) obtained from any Lehman Entity, any Archstone Entity, Seller or any of their respective Representatives relating to the Contemplated Transactions, whether so obtained before or after the execution hereof, to Seller and otherwise comply with the terms of the confidentiality agreements described in Section 12.6;
- 15.2.3 Seller and LBHI shall return all documents, work papers and other materials (and all copies thereof) obtained from any member of the Buyer Group or any of its Representatives relating to the Contemplated Transactions, whether so obtained before or after the execution hereof, to ERPOP or AVB, as applicable;
- 15.2.4 all filings, applications and other submissions made to any Person, including any Governmental Authority, in connection with the Contemplated Transactions shall, to the extent practicable, be withdrawn from such Person;
- there shall be no Liability on the part of any Buyer Party, except to the extent, and solely to the extent, provided in <u>Section 15.3.1</u> (except as expressly provided in the last sentence of <u>Section 2.1.1</u>); and
- there shall be no Liability on the part of Seller or LBHI hereunder, except to the extent, and solely to the extent, provided in <u>Section 15.3.2</u> and <u>Section 15.3.3</u> (subject to the proviso appearing at the end of <u>Section 15.3.3</u>).

15.3 Remedies

In the event that this Agreement is terminated by LBHI in accordance Section 15.1.3(b), Section 15.1.4(a) (at a time when any of the conditions set forth in Section 9.1, 9.2, 11.1.1, 11.1.2, 11.1.4, 11.1.5 or 11.1.6 have not been satisfied following the Buyer Cure Period) or Section 15.1.4(b), then the Buyer Parties shall pay to Seller, within five (5) Business Days following the date of such termination, the Seller Liquidated Damages Amount in immediately available funds to a bank account designated in writing by Seller, it being understood that, in no event shall the Buyer Parties, collectively, be required to pay the Seller Liquidated Damages Amount on more than one occasion. Other than as provided in Section 3(e) of Schedule 12.3, the right of Seller and the Lehman Entities, collectively, to receive payment of the Seller Liquidated Damages Amount shall be their sole and exclusive remedy under this

fraud on the part of any Buyer Party); provided, however, if LBHI or Seller commences a suit against any of the Buyer Parties in order to obtain such payment, the prevailing party (after a final, non-appealable judgment of a court of competent jurisdiction) shall be entitled to recover from the other party its documented reasonable out-of-pocket costs and expenses (including attorneys' fees) in connection with such suit (including any and all appeals). Upon payment of such amounts, none of the Buyer Related Parties shall have any further liability or obligation as a result of the failure by any Buyer Party to consummate the Initial Closing (other than as provided in Section 3(e) of Schedule 12.3). Upon payment of the Seller Liquidated Damages Amount (and any amounts due and owing pursuant to Section 3(e) of Schedule 12.3), none of the Buyer Related Parties shall have any further liability or obligation relating to or arising out of this Agreement or the Contemplated Transactions, and no Person shall have any rights or claims against the Buyer Related Parties under this Agreement or otherwise, whether at Law or equity, in contract, in tort or otherwise. For purposes of clarity, the Parties acknowledge and agree that neither Seller nor any of the Lehman Entities shall be entitled to an injunction or injunctions to prevent breaches of this Agreement by the Buyer Parties or otherwise obtain any other relief at Law, equity or otherwise to obtain specific performance (except as specifically provided in Sections 2.1.1 and 2.5.2(h) and except as provided in the proviso appearing in the parenthetical at the end of Section 15.4).

In the event that this Agreement is terminated by the Buyer Parties in accordance with Sections 15.1.3(a) or 15.1.4(a) (at a time when any of the conditions set forth in Section 9.1, 9.2, 10.1.1, 10.1.2, 10.1.4, 10.1.5 or 10.1.6 have not been satisfied following the Seller Cure Period), then LBHI shall pay to the Buyer Parties, within five (5) Business Days following the receipt of a written demand and documentation supporting the amount set forth therein, an amount in cash equal to all actual and documented out-of-pocket costs and expenses (including attorneys' fees and fees payable pursuant to the Reimbursement Agreements referenced in Section 13.12 (with legal and accounting fees being based on standard hourly billing), but, for the avoidance of doubt, excluding any fees payable to financial advisors other than their out-of-pocket costs) incurred by the Buyer Parties and their respective Affiliates and Representatives in connection with the Contemplated Transactions prior to the date of such termination (the "Expenses Amount"). Except as set forth in Section 15.3.3, the right of the Buyer Parties, collectively, to receive payment of the Expenses Amount shall be their sole and exclusive remedy under this Agreement for any Losses suffered by the Buyer Parties or their Affiliates as a result of the termination of this Agreement by the Buyer Parties in accordance with Sections 15.1.3(a) or 15.1.4(a) as a result of a breach of this Agreement by Seller, any of the Lehman Entities or any of the Archstone Entities. Subject to Section 15.3.3, upon payment of such amount, none of LBHI, Seller, any Archstone Entity or any of their Affiliates shall have any rights or claims

108

against Seller, LBHI, any Archstone Entity or any of their Affiliates under this Agreement or otherwise, whether at Law or equity, in contract, in tort or otherwise. The Parties acknowledge and agree that the right of the Buyer Parties to terminate this Agreement in accordance with Sections 15.1.3(a) or 15.1.4(a) as a result of a breach of this Agreement by Seller, any of the Lehman Entities or any of the Archstone Entities and recover the Expenses Amount shall not preclude the exercise of the Buyer Parties' rights under Section 15.4 prior to a termination of this Agreement pursuant to Section 15.1.3(a) or Section 15.1.4(a);

- In addition, in the event that this Agreement is terminated by the Buyer Parties in accordance with Sections 15.1.3(a) or 15.1.4(a) as a result of a breach by Seller or LBHI of their representations, warranties, covenants or agreements hereunder resulting from the willful misconduct or gross negligence of Seller or LBHI that has had or would reasonably be expected to have a material adverse impact on the Contemplated Transactions for the Buyer Parties, then the Buyer Parties shall be entitled to all remedies available at Law or equity arising from such breach; provided that to the extent that the Buyer Parties bring any action seeking specific performance of any of Seller's or LBHI's obligations hereunder, the Buyer Parties shall be deemed to have waived their rights under this Section 15.3.3.
- Seller and the Lehman Entities agree that (a) the Contemplated Transactions are unique and that damages for failure by the Buyer Parties to consummate the Contemplated Transactions will be extremely difficult and impracticable to ascertain, (b) the Seller Liquidated Damages Amount is a reasonable and rational estimate at this time and is an acceptable damages amount to Seller and the Lehman Entities upon the occurrence of the events described in Section 15.3.1, and (c) the Seller Liquidated Damages Amount is not intended as a penalty, but as full liquidated damages under this Agreement and as compensation for Seller's and the Lehman Entities' losses and other expenses associated with the failure of the Buyer Parties to close the Contemplated Transactions pursuant to this Agreement. The agreements contained in this Section 15.3 are an integral part of the Contemplated Transactions and the Parties agree that, without these agreements, the Parties would not enter into this Agreement. In addition, the Parties acknowledge that in the absence of a waiver, a bond or undertaking may be required by a court and the Parties hereby waive any such requirement of such a bond or undertaking.
- 15.4 <u>Specific Performance</u>. The Parties agree that the Buyer Parties would suffer irreparable damage in the event any of the provisions of this Agreement were not performed in accordance with the terms hereof and that, subject to the terms of <u>Section 15.3</u>, each of the Buyer Parties shall be entitled to specific performance of the terms hereof pursuant to this <u>Section 15.4</u>. It is accordingly agreed that, subject to the terms and conditions of <u>Section 15.3</u>, any Buyer Party shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement by LBHI and to enforce specifically the terms and provisions of this Agreement; it being understood and agreed that, notwithstanding the first sentence of this

109

Section 15.4 or anything else in this Agreement to the contrary, a Buyer Party's right to specific performance to require LBHI to consummate the Initial Closing or any Extension Closing shall remain subject to the satisfaction (or waiver by the applicable Party) of the conditions to the Initial Closing or any Extension Closing, as applicable, hereunder. Any requirements for the securing or posting of any bond with such remedy are hereby waived. Except as set forth in Section 15.3.2 and Section 15.3.3 following a termination pursuant to Section 15.1.3(a) or Section 15.1.4(a), the Buyer Parties and their Affiliates shall not be entitled to any monetary damages if LBHI or Seller fail to close the Contemplated Transactions (but this does not limit the obligations of LBHI pursuant to Article 14 after the Initial Closing). For the avoidance of doubt, Seller and the Lehman Entities shall not have any right under this Section 15.4 or otherwise to obtain a temporary restraining order, an injunction, specific performance, or any other equitable relief that may be available from a court of competent jurisdiction to cause the consummation of the Initial Closing or any Extension Closing (except as specifically set forth in Sections 2.1.1 and 2.5.2(h)), provided that Seller and the Lehman Entities do not waive any rights to specific performance of covenants that apply solely after the Initial Closing.

ARTICLE 16 MISCELLANEOUS PROVISIONS

- Notices. All notices, consents and other communications hereunder shall be in writing (including telecopy or similar writing) and shall be deemed to have been duly given (a) when delivered by hand or by Federal Express or a similar overnight courier to (or if that day is not a Business Day, or if delivered after 5:00 p.m., New York, New York time on a Business Day, on the first following day that is a Business Day), (b) five (5) days after being deposited in any United States Post Office enclosed in a postage prepaid, registered or certified envelope addressed to, or (c) when successfully transmitted by fassimile (with a confirming copy of such communication to be sent as provided in clauses (a) or (b) above) to, the Party for whom intended, at the address or facsimile number for such Party set forth below (or to such other address or facsimile number and with such other copies, as such Party may hereafter specify for the purpose by notice to the other Party):
 - (a) if to Seller or any Lehman Entity, to:

Lehman Brothers Holdings Inc. 1271 Avenue of the Americas New York, New York 10020 Facsimile No.: (646) 834-0874 Attention: Joelle Halperin

and

110

Lehman Brothers Holdings Inc. 1271 Avenue of the Americas New York, New York 10020 Facsimile No.: (646) 834-4340 Attention: Douglas W. Sesler

and

Lehman Brothers Holdings Inc. 1271 Avenue of the Americas New York, New York 10020 Facsimile No: (646) 834-4340 Attention: Jeffrey Fitts

with copies (which shall not constitute notice) to:

Weil Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Facsimile No.: (212) 310-8007 Attention: W. Michael Bond Raymond O. Gietz David Herman

(b) if to AVB, to:

> AvalonBay Communities, Inc. 671 N. Glebe Road, Suite 800 Arlington, VA 22203 Facsimile No.: (703) 329-4830 Attention: Kevin P. O'Shea

with a copy (which shall not constitute notice) to:

AvalonBay Communities, Inc. 671 N. Glebe Road, Suite 800 Arlington, VA 22203 Facsimile No.: (703) 329-4830

Attention: Edward M. Schulman

and to:

111

Goodwin Procter LLP Exchange Place Boston, Massachusetts 02109 Facsimile No.: (617) 523-1231 Attention: John T. Haggerty

(c) if to EQR or ERPOP, to:

Equity Residential

Two N. Riverside Plaza, Suite 400 Chicago, Illinois 60606 Facsimile No.: (312) 526-9252

Attention: Mark Parrell, EVP and Chief Financial Officer

with copies (which shall not constitute notice) to:

Equity Residential Two N. Riverside Plaza, Suite 400 Chicago, Illinois 60606 Facsimile No.: (312) 526-0680

Attention: Bruce Strohm, EVP and General Counsel

and to:

Hogan Lovells US LLP 555 13th Street, NW Washington, DC 20004 Facsimile No.: (202) 637-5910 Attention: J. Warren Gorrell, Jr.

and to:

Morrison & Foerster LLP 2000 Pennsylvania Avenue, NW Washington, DC 20006 Facsimile No.: (202) 785-7522 Attention: David P. Slotkin

16.2 <u>Entire Agreement</u>. This Agreement (including the other Transaction Documents, and the other documents and instruments referred to herein and therein) and the Disclosure Schedules sets forth the entire agreement and understanding of the Parties in respect of the Contemplated Transactions and supersedes all prior discussions, negotiations, agreements, arrangements and understandings, whether oral or written, relating to the subject matter hereof and thereof. There are no warranties, representations or other agreements between the Parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement.

112

16.3 Amendments and Waivers.

- 16.3.1 Any provision of this Agreement may be amended or modified only by a written instrument signed by Seller, LBHI, ERPOP and AVB.
- 16.3.2 No waiver hereunder shall be valid or binding unless set forth in writing and duly executed by the Party against whom enforcement of the waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the Party granting such waiver in any other respect or at any other time. Neither the waiver by any of the Parties of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the Parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. Except as otherwise provided herein, no action taken pursuant to this Agreement, including any investigation by or on behalf of any Party, shall be deemed to constitute a waiver by the Party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement.
- 16.4 Governing Law. This Agreement shall be construed, performed and enforced in accordance with the laws of the State of Delaware (without giving effect to its principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of the laws of another jurisdiction). Notwithstanding anything to the contrary contained in this Agreement, each of the Parties agrees that, except as specifically set forth in the Debt Commitment Letters, all claims or causes of action (whether at Law, in equity, in contract, in tort or otherwise) against any of the AVB Lenders or the ERPOP Lenders in any way relating to the Debt Commitment Letters or the performance thereof or the financings contemplated thereby, shall be exclusively governed by, and construed in accordance with, the internal laws of the State of New York, without giving effect to principles or rules or conflict of laws to the extent such principles or rules would require or permit the application of laws of another jurisdiction.

16.5 <u>Jurisdiction and Venue</u>.

16.5.1 Each of the Parties irrevocably agrees that any legal action or Proceeding with respect to this Agreement and the rights and obligations arising hereunder, or for recognition and enforcement of any judgment in respect of this Agreement and the rights and obligations arising hereunder brought by the other Party or its successors or assigns, shall be brought and determined exclusively in the Court of Chancery of the State of Delaware and any state appellate courts therefrom within the State of Delaware. Each of the Parties hereby irrevocably submits with regard to any such action or Proceeding for itself and in respect of its property, generally and unconditionally, to the personal jurisdiction of the aforesaid courts, agrees that it will not bring any Proceeding relating to this Agreement or the Contemplated Transactions in any court

113

other than the aforesaid courts and waives any objection to venue laid therein. Process in any such Proceeding may be served on any Party anywhere in the world, whether within or without the State of Delaware. Without limiting the foregoing, Sellers and the Buyer Parties agree that service of process upon such Party at the address referred to in Section 16.1, together with written notice of such service to such Party, shall be deemed effective service of process upon such Party.

16.5.2 Each of the Parties hereby irrevocably waives, and agrees not to assert as a defense, counterclaim or otherwise, in any Proceeding with respect to this Agreement, (a) any claim that it is not personally subject to the jurisdiction of the above named courts for any reason other than the failure to serve in

accordance with Section 16.1, (b) any claim that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise), and (c) to the fullest extent permitted by the applicable Law, any claim that (i) the Proceeding in such court is brought in an inconvenient forum, (ii) the venue of such Proceeding is improper, or (iii) this Agreement, or the subject matter hereof, may not be enforced in or by such courts

16.5.3 Notwithstanding anything to the contrary contained in this Agreement, each of the Parties hereto agrees that it will not bring or support any person in any action, suit, proceeding, cause of action, claim, cross-claim or third party claim of any kind of description, whether in law or in equity, whether in contract or in tort or otherwise, against any of the AVB Lenders or the ERPOP Lenders in any way relating to this Agreement or any of the transactions contemplated by this Agreement, including, but not limited to, any dispute arising out of or relating in any way to the Debt Commitment Letters or the performance thereof or the financings contemplated thereby, in any forum other than the federal and New York state courts located in the Borough of Manhattan within the City of New York.

16.6 WAIVER OF TRIAL BY JURY.

16.6.1 EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY OR CLAIM WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE CONTEMPLATED TRANSACTIONS WHETHER ARISING IN CONTRACT OR TORT OR OTHERWISE. THIS WAIVER MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING (OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS

114

SECTION 16.6.1) AND EXECUTED BY EACH OF THE PARTIES HERETO). THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT OR ANY OTHER AGREEMENTS OR DOCUMENTS RELATING TO THE CONTEMPLATED TRANSACTIONS. The scope of this waiver is intended to be all-encompassing of any and all disputes that may be filed in any court and that relate to the subject matter of the Contemplated Transactions, including contract claims, tort claims, breach of duty claims and all other common law and statutory claims. In the event of litigation, this Agreement may be filed as a written consent to a trial by the court.

- 16.6.2 EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (I) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE SUCH WAIVER, (II) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVER, (III) IT MAKES SUCH WAIVER VOLUNTARILY, AND (IV) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 16.6.
- 16.7 Binding Effect. This Agreement will be binding upon, inure solely to the benefit of and be enforceable by the Parties and their respective permitted successors and assigns. Notwithstanding anything to the contrary contained in this Agreement, (i) neither Seller nor any of its subsidiaries, affiliates, directors, officers, employees, agents, partners, managers, members or stockholders shall have any right or claim against any of the AVB Lenders or the ERPOP Lenders, in any way relating to this Agreement or any of the transactions contemplated by this Agreement, or in respect of any, or in respect of any oral representations made or alleged to have been made in connection herewith or therewith, including any dispute arising out of or relating in any way to the Debt Commitment Letters or the performance thereof or the financings contemplated thereby, whether at Law or equity, in contract, in tort or otherwise and (ii) the AVB Lenders or the ERPOP Lenders shall not have any liability (whether in contract, in tort or otherwise) to Seller or any of its subsidiaries, affiliates, directors, officers, employees, agents, partners, managers, members or stockholders for any obligations or liabilities of any party hereto under this Agreement or for any claim based on, in respect of, or by reason of, the transactions contemplated hereby and thereby or in respect of any oral representations made or alleged to have been made in connection herewith or therewith, including any dispute arising out of or relating to in any way to the Debt Commitment Letters or the performance thereof or the financings contemplated thereby, whether at Law or equity, in contract, in tort or otherwise. Notwithstanding anything to the contrary in the foregoing, nothing herein shall prohibit, limit or modify any action by the Buyer Parties to enforce the rights or obligations contained in the Debt Commitment Letters to the extent permitted therein.

16.8 Interpretation. As used in this Agreement and required by the context, the singular and plural shall be deemed to include all genders; words importing persons shall include

115

partnerships, corporations and other entities; when reference is made in this Agreement to an Article, Section, Schedule or Exhibit, such reference shall be to an Article, Section, Schedule or Exhibit of this Agreement unless otherwise indicated; and the terms "herein," "hereof" and "hereunder" or other similar terms, refer to this Agreement as a whole and not only to the particular sentence, subsection or section in which any such term may be employed. Whenever in this Agreement the word "including" is used, it shall be deemed to be for purposes of identifying only one or more of the possible alternatives, and the entire provision in which such word appears shall be read as if the phrase "including without limitation" were actually used in the text. The section headings herein are for convenience only and shall not affect the construction hereof. All references to dollars (or the symbol "\$") contained herein shall be deemed to refer to United States dollars. Where any provision in this Agreement refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether the action in question is taken directly or indirectly by such Person. Except when used together with the word "either" or otherwise for the purpose of identifying mutually exclusive alternatives, the term "or" has the inclusive meaning represented by the phrase "and/or". With regard to each and every term and condition of this Agreement, the Parties understand and agree that the same have or has been mutually negotiated, prepared and drafted, and that if at any time the Parties desire or are required to interpret or construe any such term or condition or any agreement or instrument subject thereto, no consideration shall be given to the issue of which Party actually prepared, drafted or requested any term or condition of this Agreement.

16.9 Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other Governmental Authority to be invalid, void or unenforceable, such term, provision, covenant or restriction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Agreement, and the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated so long as the economic or legal substance of the Contemplated Transactions is not affected in any manner materially adverse to any Party. Upon such a determination, the Parties shall use Commercially Reasonable Efforts to negotiate in good faith to modify this Agreement so as to affect the original intent of the Parties as closely as possible in an acceptable manner in order that the Contemplated Transactions shall be consummated as originally contemplated to the fullest extent possible.

16.10 <u>Counterparts</u>. This Agreement may be executed and delivered in multiple counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. It is the express intent of the Parties to be bound by the exchange of signatures on this Agreement via facsimile or electronic mail via the portable document format (PDF). A facsimile or other copy of a signature shall be deemed an original. This Agreement shall become effective when each Party hereto shall have received a counterpart hereof signed by all of the other Parties hereto, this Agreement shall have no effect and no Party shall have any right or obligation hereunder (whether by virtue of any other oral or written agreement or other communication).

116

- 16.11 Third Parties. Except as otherwise expressly provided herein (including Section 13.14.2 in regard to the Archstone Indemnitees), no provision of this Agreement is intended or shall confer on any Person, other than the Parties (and their successors and permitted assigns), any rights under this Agreement and no other Person shall be entitled to rely thereon. Notwithstanding the foregoing, the AVB Lenders and the ERPOP Lenders shall be third party beneficiaries to this Section 16.11 and Sections 15.3, 16.4, 16.5, 16.6 and 16.7.
- 16.12 <u>Assignment</u>. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned, delegated or otherwise transferred by any Party (whether by operation of law or otherwise) without the prior written consent of the other Parties. Notwithstanding the foregoing, any Party shall have the right to assign all or certain provisions of this Agreement, or any interest herein, and may delegate any duty or obligation hereunder, without the consent of the other Parties, (i) to any Affiliate of such Party, or (ii) at any time after the Initial Closing, to any purchaser of all or substantially all of the assets or Equity Interests (whether by merger, recapitalization, reorganization or otherwise) of such Party; <u>provided</u> that, in the case of each of clauses (i) through (ii), no such assignment or delegation shall relieve such Party of any of its obligations hereunder. Any attempted assignment, delegation or transfer in violation of this <u>Section 16.12</u> shall be null and void.
- 16.13 <u>Joint and Several Obligations</u>. For purposes of clarity, (a) all obligations, representations, warranties, covenants and undertakings (to the extent they survive) of the Buyer Parties, the Buyer Designees, ERPOP or AVB set forth in this Agreement, including pursuant to <u>Section 2.2.1</u>, <u>Section 2.3.2</u>, <u>Section 2.3.2</u>, <u>Section 2.3.3</u>, <u>Section 2.3.3</u>, <u>Section 2.3.3</u>, <u>Section 2.3.3</u>, <u>Section 2.3.14</u>, and <u>Section 13.1.4</u>, <u>Section</u>
- 16.14 <u>Cash Payments</u>. All cash amounts payable pursuant to the terms and conditions of this Agreement shall be paid by wire transfer of immediately available funds to an account or accounts designated by the applicable Party to receive such cash amount, which account or accounts shall be designated no later than three (3) days prior to the Initial Closing Date or other date of applicable payment hereunder, unless otherwise agreed to by the applicable Parties.
- 16.15 <u>Schedules and Exhibits</u>. The schedules and exhibits, if any, referenced in this Agreement constitute an integral part of this Agreement and are incorporated herein by reference and made a part hereof. Any information disclosed in a schedule or a section of any such schedule shall be deemed to be disclosed in such other schedules or sections of any such schedule to the extent that the disclosure is reasonably apparent from its face to be applicable to such other schedule or section of any such schedule. Disclosure of any fact or item in any schedule shall not be deemed to constitute an admission that such item or fact is material for the purposes of this Agreement.
- 16.16 <u>Time Periods</u>. Any action required hereunder to be taken within a certain number of days shall, unless otherwise provided herein, be taken within that number of calendar days; <u>provided</u>, <u>however</u>, that if the last day for taking such action falls on a Saturday, a Sunday, or a

117

legal holiday, the period during which such action may be taken shall be extended to the next Business Day. Time shall be of the essence with respect to all obligations under this Agreement.

16.17 No Recourse Against Non-Parties. All claims or causes of action (whether in contract or in tort, at Law or in equity) that may be based upon, arise out of or relate to this Agreement, or the negotiation, execution or performance of this Agreement (including any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement) or the Contemplated Transactions, may be made only against the entities that are expressly identified as Parties hereto. No Person who is not a named Party to this Agreement, including any director, trustee, officer, employee, incorporator, member, partner, stockholder, Affiliate, agent, attorney or Representative of any named Party to this Agreement ("Non-Party Affiliates"), shall have any Liability (whether in contract or in tort, at Law or in equity, or based upon any theory that seeks to impose Liability of an entity Party against its owners or affiliates) for any obligations or Liabilities arising under, in connection with or related to this Agreement or for any claim based on, in respect of, or by reason of this Agreement or its negotiation or execution; and each Party hereto waives and releases all such Liabilities, claims and obligations against any such Non-Party Affiliates.

[Signature Page Follows]

118

IN WITNESS WHEREOF, the Parties have executed and delivered this Agreement as of the date set forth above.

EQR:

EQUITY RESIDENTIAL

By: /s/ Bruce C. Strohm

Name: Bruce C. Strohm

Title: Executive Vice President and General Counsel

ERPOP:

ERP OPERATING LIMITED PARTNERSHIP

By: Equity Residential, its General Partner

By: /s/ Bruce C. Strohm

Name: Bruce C. Strohm

Title: Executive Vice President and General Counsel

AVB:

AVALONBAY COMMUNITIES INC.

By: /s/ Timothy J. Naughton

Name: Timothy J. Naughton

Title: Chief Executive Officer and President

SELLER:

ARCHSTONE ENTERPRISE LP

By: Archstone Enterprise GP LLC, its general partner

By: Archstone Multifamily JV LP, its sole member

By: Archstone Multifamily GP LLC, its general partner

By: Archstone Multifamily (Governance) LLC, its sol member

By: REPE Archstone GP Holding, LLC, its managing member

By: /s/ Jeffrey Fitts

Name: Jeffrey Fitts

Title: Authorized Signatory

2

LBHI:

LEHMAN BROTHERS HOLDINGS INC.

By: /s/ Jeffrey Fitts

Name: Jeffrey Fitts

Title: Authorized Signatory

AVALONBAY COMMUNITIES, INC. REGISTRATION RIGHTS AGREEMENT

THIS REGISTRATION RIGHTS AGREEMENT, dated as of [·] (this "Agreement"), is by and between Lehman Brothers Holdings Inc., a Delaware corporation ("LBHI") and AvalonBay Communities, Inc., a Maryland corporation (the "Company"). LBHI and the Company are sometimes referred to herein as the "Parties" and each, a "Party."

RECITALS

WHEREAS, the Company, ERP Operating Limited Partnership, an Illinois limited partnership, Equity Residential, a Maryland real estate investment trust, LBHI and Archstone Enterprise LP, a Delaware limited partnership have entered into that certain Asset Purchase Agreement, dated as of November [·], 2012 (the "Purchase Agreement");

WHEREAS, as a result of the initial closing of the Contemplated Transactions (as defined in the Purchase Agreement) (the "*Initial Closing*"), LBHI beneficially owns, as of the date of this Agreement, [·] AVB Common Shares acquired from the Company as a portion of the consideration received by Sellers (as defined in the Purchase Agreement) in connection with such transactions (the "*Shares*");

WHEREAS, the execution and delivery of this Agreement is a condition precedent to the obligations of the Parties to consummate the transactions contemplated by the Purchase Agreement; and

WHEREAS, each of the Parties desires to enter into this Agreement in order to establish registration rights of the LBHI Group with respect to the Shares.

NOW, THEREFORE, in consideration of the mutual covenants and agreements set forth herein and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

SECTION 1. DEFINITIONS

As used in this Agreement, the following terms have the respective meanings set forth below:

"Affiliate" means as to any Person, any other Person which, directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with such specified Person. For purposes of this definition, "control" of any Person means possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract, or otherwise.

"Agreement" has the meaning set forth in the Preamble.

"ATM Program" has the meaning set forth in Section 2(h).

- "Automatic Shelf Registration Statement" means an "Automatic Shelf Registration Statement," as defined in Rule 405 under the Securities Act.
- "AVB Common Shares" means the shares of common stock, par value \$0.01 per share, of the Company.
- "beneficial owner" and words of similar import have the meaning assigned to such terms in Rule 13d-3 under the Exchange Act.
- "Black-Out Period" has the meaning set forth in Section 2(1).
- "Board of Directors" means the board of directors of the Company.
- "Business Day" means each day, other than a Saturday, Sunday or other day on which banks in New York, New York are required by Law to close.
- "Commission" means the Securities and Exchange Commission or any other federal agency at the time administering the Securities Act.
- "Company" has the meaning set forth in the Preamble.
- "Demand Notice" has the meaning set forth in Section 2(b).
- "Demand Registration" has the meaning set forth in Section 2(b).
- "Exchange Act" means the Securities Exchange Act of 1934, as amended, or any successor Law thereto, and the rules and regulations issued pursuant to that statute or any successor Law.
 - "FINRA" means the Financial Industry Regulatory Authority, Inc., or any successor thereto.
- "Governmental Authority" means any government or political subdivision, whether federal, state, local or foreign, or any agency or instrumentality of any such government or political subdivision, or any federal, state, local or foreign court, authority, tribunal, department, bureau or commission, in each case having jurisdiction over the applicable matter.
 - "Indemnified Party" has the meaning set forth in Section 4(c).
 - "Indemnifying Party" has the meaning set forth in Section 4(c).
 - "Initial Closing" has the meaning set forth in the Recitals.
- "Issuer Free Writing Prospectus" means an "Issuer Free Writing Prospectus," as defined in Rule 433 under the Securities Act, relating to an offer of Registrable Securities.

public announcement of the Company's intention to conduct such Underwritten Offering, and (z) the public announcement of the pricing of such Underwritten Offering.

"Law" means, with respect to any Person, any federal, state, county, municipal, local, multinational or foreign statute, treaty, law, common law, ordinance, rule, regulation, code, order, writ, stipulation, injunction, judicial decision, decree, ruling, determination, finding, permit, constitutional provision or other legally binding requirement of any Governmental Authority applicable to such Person or any of its respective properties, assets, officers, directors, employees, consultants or agents (in connection with such officer's, director's, employee's, consultant's or agent's activities on behalf of such Person).

"LBHI" has the meaning set forth in the Preamble.

"LBHI Group" means LBHI together with its Subsidiaries and any liquidating trust established pursuant to the Plan that beneficially own Shares, in each case, for so long as LBHI or such Subsidiary or liquidating trust beneficially own any Shares. When this Agreement requires the LBHI Group to provide notice or give consent, such notice or consent shall be given by LBHI and not by any other LBHI Group Member. Notice provided to the LBHI pursuant to Section 5(c) shall constitute valid notice to LBHI and each other LBHI Group Member.

"LBHI Group Member" means any member of the LBHI Group.

"Losses" has the meaning set forth in Section 4(a).

"Parties" and "Party" have the meaning set forth in the Preamble.

"Permitted Assignee" has the meaning set forth in Section 5(d).

"Person" means any individual, a corporation (including any non-profit corporation), a limited liability company, a general partnership, a limited partnership, a limited liability partnership, a trust, a venture, a business, a union, a society, an association, a firm, a Governmental Authority or any other entity or organization.

"Piggyback Registration" has the meaning set forth in Section 2(h).

"Plan" shall mean that certain Modified Third Amended Joint Chapter 11 Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors, dated December 6, 2011 and that certain Order Confirming Modified Third Amended Joint Chapter 11 Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors, dated December 6, 2011, [Docket No. 23023].

"Priced Underwritten Offering Requests" has the meaning set forth in Section 2(a)(ii).

"Prospectus" means the prospectus (including any preliminary, final or summary prospectus) included in any Registration Statement, all amendments and supplements to such prospectus and all other material incorporated by reference in such prospectus.

"Purchase Agreement" has the meaning set forth in the Recitals.

3

"Qualifying Employee Stock" means the rights, options and other securities issued under employee benefit plans of the Company or any predecessor thereof, or otherwise to executives, directors and/or employees in compensation arrangements approved by the Board of Directors or any predecessor thereof, and any securities issued after the date hereof upon exercise of such rights, options and other securities.

"Register," "Registered" and "Registration" means a registration effected by preparing and (a) filing a Registration Statement in compliance with the Securities Act (and any post-effective amendments filed or required to be filed) and the declaration or ordering of effectiveness of such Registration Statement, or (b) filing a Prospectus and/or Prospectus supplement in respect of an appropriate effective Registration Statement.

"Registrable Securities" means all Shares that are beneficially owned by a LBHI Group Member at any time and any AVB Common Shares that may be issued or distributed by way of a share split, recapitalization or reclassification in respect of the Shares; provided, however, that a Share shall cease to be a Registrable Security when (a) it has been effectively Registered under the Securities Act and disposed of in accordance with the Registration Statement covering it, (b) it is transferred in compliance with Rule 144 under the Securities Act (but not Rule 144A under the Securities Act) or any successor provision, such that no restrictive legend is required after giving effect to such transfer, or (c) it has ceased to be outstanding. Notwithstanding anything herein to the contrary, all Shares shall cease to be Registrable Securities upon the Termination Date.

"Registration Expenses" means any and all expenses incurred by the Company and its Subsidiaries in effecting any Registration pursuant to this Agreement, including, all (a) Registration and filing fees, and all other fees and expenses payable in connection with the listing of securities on any securities exchange or automated interdealer quotation system, (b) fees and expenses of compliance with any securities or "blue sky" Laws (including reasonable fees and disbursements of counsel in connection with "blue sky" qualifications of the Registrable Securities), (c) expenses in connection with the preparation, printing, mailing and delivery of any Registration Statements, Prospectuses, Issuer Free Writing Prospectus and other documents in connection therewith and any amendments or supplements thereto, (d) security engraving and printing expenses, (e) internal expenses of the Company (including, all salaries and expenses of its officers and employees performing legal or accounting duties), (f) fees and disbursements of counsel for the Company and fees and expenses for independent certified public accountants retained by the Company (including, the expenses associated with the delivery by independent certified public accountants of any "comfort letters" requested pursuant to the terms hereof), (g) fees and expenses of any special experts retained by the Company in connection with such Registration, (h) fees and expenses in connection with any review by FINRA of any underwriting arrangements or other terms of the offering, and all reasonable fees and expenses of any "qualified independent underwriter," (i) reasonable fees and disbursements of Underwriters customarily paid by issuers or sellers of securities, but excluding any underwriting fees, discounts and commissions attributable to the sale of Registrable Securities and excluding any fees or expenses of counsel to the Underwriters, other than as referred to in clause (b) above, (j) costs of printing and producing any agreements among Underwriters, underwriting ag

Registrable Securities, (k) transfer agents' and registrars' fees and expenses and the fees and expenses of any other agent or trustee appointed in connection with such offering, and (xii) expenses incurred by the Company relating to any analyst or investor presentations or any "road shows" undertaken in connection with the Registration, marketing or selling of the Registrable Securities. Registration Expenses shall not include, and the Company shall not have any obligation to pay, any out-of-pocket expenses of the LBHI Group or Selling Expenses.

- "Registration Statement" means any registration statement of the Company that covers the resale of any Registrable Securities pursuant to the provisions of this Agreement filed with, or to be filed with, the Commission under the rules and regulations promulgated under the Securities Act, including the related Prospectus, amendments and supplements to such registration statement, including pre- and post-effective amendments, and all exhibits, financial information and all other material incorporated by reference in such registration statement.
 - "Resumption Date" has the meaning set forth in Section 2(a)(iii).
- "Securities Act" means the Securities Act of 1933, as amended, or any successor Law thereto, and the rules and regulations issued pursuant to that statute or any successor Law.
 - "security" and "securities" have the meaning set forth in Section 2(a)(1) of the Securities Act.
- "Selling Expenses" means all underwriting discounts, selling commissions and stock transfer taxes applicable to the sale of Registrable Securities and all fees and disbursements of counsel for the LBHI Group, which do not constitute Registration Expenses.
 - "Shares" has the meaning set forth in the Recitals.
- "Shelf Registration Statement" means a "shelf" registration statement of the Company that covers all the Registrable Securities (and may cover other securities of the Company) on Form S-3 and under Rule 415 under the Securities Act or, if the Company is not then eligible to file on Form S-3, on Form S-11 under the Securities Act, or any successor rule that may be adopted by the Commission, including without limitation any such registration statement filed pursuant to Sections 2(a), 2(b) or 2(h), and all amendments and supplements to such "shelf" registration statement, including, post-effective amendments, in each case, including the Prospectus contained therein, all exhibits thereto and any document incorporated by reference therein.
 - "Stand-Down Notice" has the meaning set forth in Section 2(a)(iii).
- "Subsidiary" with respect to any Person (including the Company), means any other Person of which the specified Person, either directly or through or together with any other of its Subsidiaries, owns securities or other ownership interests or equity interests having voting power to elect a majority of the board of directors or other governing body or Persons performing similar functions on behalf of such other Person.
 - "Termination Date" has the meaning set forth in Section 2(o).

5

- "Underwriter" means, with respect any Underwritten Offering, a securities dealer who purchases any Registrable Securities as a principal in connection with a distribution of such Registrable Securities.
- "Underwritten Offering" means a public offering of securities Registered under the Securities Act in which an Underwriter participates in the distribution of such securities, including on a firm commitment basis for reoffer and resale to the public, including any such offering that is a "bought deal" or a block trade.
 - "Underwritten Offering Request" has the meaning set forth in Section 2(a)(ii).

SECTION 2. REGISTRATION RIGHTS

- (a) <u>Shelf Registration; Underwritten Offering.</u>
- (i) Subject to the provisions of Section 2(1), on or prior to the tenth (10th) day following the Initial Closing, the Company will file a Shelf Registration Statement (which Shelf Registration Statement shall be an Automatic Shelf Registration Statement if the Company is then eligible to file an Automatic Shelf Registration Statement) registering for resale the Registrable Securities under the Securities Act. Until such time as all Registrable Securities cease to be Registrable Securities or the Company is no longer eligible to maintain a Shelf Registration Statement, the Company will keep current and effective such Shelf Registration Statement and file such supplements or amendments to such Shelf Registration Statement (or file a new Shelf Registration Statement (which Shelf Registration Statement shall be an Automatic Shelf Registration Statement if the Company is then eligible to file an Automatic Shelf Registration Statement) when such preceding Shelf Registration Statement expires pursuant to the rules of the Commission) as may be necessary or appropriate in order to keep such Shelf Registration Statement continuously effective and useable for the resale of Registrable Securities under the Securities Act. The Company represents and warrants to LBHI that as of the date of this Agreement, the Company is not an "ineligible issuer" and is a "well known seasoned issuer", in each case as defined under the Securities Act.
- (ii) Upon the written request of LBHI from time to time (an "Underwritten Offering Request"), the Company will cooperate with the LBHI Group and any Underwritter in effecting an Underwritten Offering pursuant to a Shelf Registration Statement as promptly as reasonably practicable following receipt of such Underwritten Offering Request; provided, however, that (x) LBHI shall not be entitled to make more than two (2) Underwritten Offering Requests that result in priced Underwritten Offerings in any twelve (12) month period (the "Priced Underwritten Offering Requests"); (y) LBHI will be deemed to have requested the removal of any Registrable Securities held by any LBHI Group Member from any Underwritten Offering Request and to have rescinded the Underwritten Offering Request, with the same effects as provided in Section 2(g), automatically if: (I) the Launch Date in respect of such Underwritten Offering has not occurred by the end of the fifth (5th) Business Day after the following conditions have been met: (I-i) the Company has made available to LBHI a Prospectus under an effective Registration Statement naming the designated member or members of the LBHI Group as selling stockholders and the Underwriter(s) and containing such other information as is required under the Securities Act and the rules of the Commission in a form

ready for filing, assuming that as of such time (X) the Company has otherwise complied with its applicable obligations in Section 2(k) and (Y) LBHI has otherwise complied with its obligations in Section 2(m), and (I-ii) the Company has notified LBHI in writing that the Company has complied with the provisions of this clause (I) and that the five (5) Business Day period referenced in this clause (I) is commencing), or (II) the Underwritten Offering has not been priced by the end of the third (3rd) Business Day after the Launch Date, assuming that the Launch Date occurred before the expiration of the five (5) Business Day period referenced in clause (II) above); provided, further, that in the case of both clause (I) and clause (II) above such Underwritten Offering Request shall not be counted as a utilized Underwritten Offering Request for purposes of the limits in this Section 2(a)(ii) (other than clause (z) hereof) and Section 2(g); and (z) LBHI shall not be entitled to make in the aggregate more than four (4) Underwritten Offering Requests, including two (2) Priced Underwritten Offering Requests and two (2) Underwritten Offering Requests that are deemed rescinded or affirmatively rescinded, in any twelve (12) month period (for the avoidance of doubt, if there are two (2) Priced Underwritten Offering Requests in any twelve (12) month period this clause shall not be interpreted as allowing LBHI to make any further Underwritten Offering requests until the expiration of the then current twelve (12) month period). Each Underwritten Offering Request will specify the number of Registrable Securities proposed by LBHI to be included in such Underwritten Offering, the intended method of distribution and the estimated gross proceeds of such Underwritten Offering at any time prior to the Launch Date of the Underwritten Offering so long as such change would not reduce the estimated gross proceeds of the Underwritten Offering to less than \$100 million.

(iii) The Company will have the right to delay an Underwritten Offering by LBHI following receipt of an Underwritten Offering Request if the Company intends to effect its own Underwritten Offering by giving LBHI written notice of such intent (a "Stand-Down Notice"), whereby the Company's obligation to cooperate with the LBHI Group and any Underwriter in effecting an Underwritten Offering pursuant to Section 2(a)(ii) shall be suspended until the later of the Resumption Date or the date of expiration of any "lock-up" agreement required to be entered into by LBHI pursuant to Section 3(b) with respect to the Company's Underwritten Offering; provided, however, that (w) the Company will not be entitled to give a Stand-Down Notice until after the first (1st) anniversary of the Initial Closing; (x) the Company will not be entitled to deliver a Stand-Down Notice in respect of an Underwritten Offering Request later than 5 p.m. New York time on the next Business Day following receipt of such Underwritten Offering Request; (y) the Company will not be entitled to more than one (1) Stand-Down Notice in any twelve (12) month period; and (z) the Company will be deemed to have rescinded the Stand-Down Notice automatically, whereby the Company's obligation to cooperate with the LBHI Group and any Underwriter in effecting an Underwritten Offering pursuant to Section 2(a)(ii) shall resume, if (I) the Launch Date in respect of the Company's Underwritten Offering has not occurred by the end of the seventh (7th) Business Day after the date of the Underwritten Offering Request or (II) the Company's Underwritten Offering has not been priced by the end of the third (3rd) Business Day after the Launch Date (the date following automatic rescission of a Stand-Down Notice pursuant to either clause (I) and clause (II) above, a "Resumption Date"). LBHI acknowledges and agrees that the receipt of any Stand-Down Notice may constitute material non-public information regarding the Company and shall keep the existence and contents of any Stand-Dow

7

contrary contained herein, if LBHI determines to rescind any prior Underwritten Offering Request or Demand Notice following receipt of a Stand-Down Notice, then LBHI may, at its election, give written notice of such election to the Company; provided, however, that such rescinded Underwritten Offering Request or Demand Notice shall not be counted as an Underwritten Offering Request or Demand Notice for purposes of the limits in Section 2(a)(ii) and Section 2(b)(i), respectively, without any requirement to reimburse the Company for any related expenses incurred by the Company.

(b) <u>Demand Registration Rights</u>. If, at any time while there still remain Registrable Securities, the Company is no longer eligible to use or, notwithstanding its obligations under <u>Section 2(a)(i)</u>, otherwise ceases to maintain an effective Shelf Registration Statement, within ten (10) days after LBHI's written request to Register the resale of a specified amount of the Registrable Securities under the Securities Act (a "*Demand Notice*"), the Company will file a Registration Statement, on an appropriate form which the Company is then eligible to use, to Register the resale of such Registrable Securities, which Registration Statement will (if specified in LBHI's notice) contemplate the ability of the LBHI Group to effect an Underwritten Offering in accordance with <u>Section 2(a)(ii)</u> (each such Registration, a "*Demand Registration*"); <u>provided, however,</u> that LBHI shall not be entitled to request more than two (2) Demand Registrations in any twelve (12) month period. A request that does not result in an effective Registration Statement under the Securities Act shall not be counted as a utilized request for purposes of the limits in the preceding sentence. Each Demand Notice will specify the number of Registrable Securities proposed to be offered pursuant to any Demand Registration, which may not be less than \$100 million. LBHI may change the number of Registrable Securities proposed to be offered pursuant to any Demand Registration at any time prior to the Registration Statement with respect to the Demand Registration being declared effective by the Commission, so long as such change would not reduce the estimated gross proceeds of the Demand Registration to less than \$100 million. The Company shall have the right to satisfy a Demand Notice by filing a Shelf Registration Statement.

(c) Reserved.

Effectiveness. The Company will use its reasonable best efforts to (i) cause any Registration Statement to be declared effective (unless it becomes effective automatically upon filing) as promptly as practicable after the filing thereof with the Commission and (ii) keep such Registration Statement current and effective for a period necessary for the completion of the resale of Registrable Securities Registered thereon. The Company further agrees to supplement or make amendments to each such Registration Statement as may be necessary to keep such Registration Statement effective for the period referred to in clause (ii) above, including (A) to respond to the comments of the Commission, if any, (B) as may be required by the registration form utilized by the Company for such Registration Statement or by the instructions to such registration form, (C) as may be required by the Securities Act, (D) as may be required in connection with an Underwritten Offering, subject to Section 2(a)(iii), or (E) as may be reasonably requested in writing by LBHI or any Underwriter and reasonably acceptable to the Company, subject to Section 2(a)(iii). The Company agrees to furnish to the LBHI Group one electronic copy of any such supplement or amendment no later than the time it is first being used or filed with the Commission.

8

(e) Choice of Underwriter. In the event an Underwritten Offering of Registrable Securities involves one or more Underwriters, such Underwriters will be selected by LBHI; provided, that any Underwriter must be approved by the Company, which approval will not be unreasonably withheld. Any nationally recognized investment banking firm selected by LBHI will be deemed approved by the Company for purposes of the preceding sentence. The Company shall take such actions as the Underwriters may reasonably request in their efforts to sell Registrable Securities pursuant to such Registration Statement. LBHI agrees that it will use its commercially reasonable efforts to direct such Underwriters to engage the Company's existing designated underwriters' counsel (previously identified by the Company to LBHI) as its underwriters' counsel with respect to such Underwritten Offering; provided, however, that the deemed consent referred to in the second sentence of this Section 2(e) shall not be affected if, notwithstanding the use by LBHI of its commercially reasonable efforts, such Underwriters fail to so engage the Company's existing designated underwriters' counsel.

(f) <u>LBHI Determinations</u>. LBHI will be permitted to determine in good faith, matters affecting the structure of Underwritten Offerings undertaken pursuant to <u>Section 2(a)(ii)</u> or Demand Registrations, as applicable, including the price, underwriting discount and other financial terms for the Registrable Securities.

- Securities held by any LBHI Group Member from any Underwritten Offering Request or Demand Registration at any time prior to the pricing of the Underwritten Offering or the effective date of the applicable Registration Statement, as applicable, by providing written notice thereof to the Company; provided, that LBHI reimburses the Company for all reasonable, out-of-pocket expenses incurred by the Company in connection with such Underwritten Offering Request, Demand Notice or Demand Registration; provided, further, that (x) such Underwritten Offering Request or Demand Registration shall not be counted as a utilized Underwritten Offering Request or Demand Registration, as the case may be, for purposes of the limits in Section 2(a)(ii) (other than clause (z) thereof) and Section 2(b), respectively, and (y) LBHI shall not be entitled to make more than four (4) Underwritten Offering Requests and Demand Registrations in the aggregate, regardless of the combination or sequence thereof, whether or not they result in the pricing of an Underwritten Offering or an effective Registration Statement under the Securities Act, in any twelve (12) month period.
- (h) <u>Piggyback Registration</u>. Subject to <u>Section 2(1)</u>, if, at any time while there still remain Registrable Securities, the Company is no longer eligible to use or, notwithstanding its obligations under <u>Section 2(a)(i)</u>, otherwise a Shelf Registration Statement registering the Shares for resale is not effective, the Company proposes to file a new registration statement under the Securities Act with respect to an offering of AVB Common Shares for (i) the Company's own account (other than a registration statement on Form S-4 or S-8 (or any substitute form that may be adopted by the Commission) or with respect to a Company at-the-market offering program ("*ATM Program*") or Company dividend reinvestment plans) or (ii) the account of any holder of AVB Common Shares (other than a LBHI Group Member), then the Company shall give written notice of such proposed filing to LBHI as soon as reasonably practicable (but in no event less than ten (10) days before the anticipated filing date of such new registration statement). Upon a written request, given by LBHI to the Company within five (5)

9

days after delivery of any such notice by the Company, to include Registrable Securities in such Registration (which request shall specify the number of Registrable Securities proposed to be included in such new registration statement if such registration statement is not a "pay as you go" Automatic Shelf Registration Statement), the Company shall, subject to Section 2(i), include all such requested Registrable Securities in such new registration statement on the same terms and conditions as applicable to the Company's or such holder's AVB Common Shares (a "Piggyback Registration"). Notwithstanding the foregoing, if at any time after giving written notice of such proposed filing and prior to the effective date of such new registration statement, the Company or such holders shall determine for any reason not to proceed with the proposed filing of the new registration statement, then the Company may, at its election, give written notice of such determination to the LBHI Group and, thereupon, will be relieved of its obligation to Register any Registrable Securities in connection with such new registration statement.

- (i) Reduction of Size of Underwritten Offering. Notwithstanding anything to the contrary contained herein, if the lead Underwriter or Underwriters of an Underwritten Offering advise LBHI and the Company in writing that, in their reasonable opinion the number of AVB Common Shares (including any Registrable Securities) that the Company, the LBHI Group Members and any other Persons intend to dispose of pursuant to any Underwritten Offering is such that the success of any such Underwritten Offering would be materially and adversely affected, including with respect to the price at which the securities can be sold, then such Underwritten Offering shall include only such securities as the Company and LBHI are advised by such lead Underwriter or Underwriters can be sold without such material and adverse effect, in accordance with the following priorities:
 - (X) priority in the case of an Underwritten Offering initiated by the Company for its own account which gives rise to a Piggyback Registration pursuant to Section 2(h) will be (A) first, AVB Common Shares proposed to be offered by the Company for its own account, (B) second, pro rata among (1) the Registrable Securities requested to be disposed of pursuant to the Underwritten Offering for the account of the LBHI Group Members pursuant to Section 2(h) and (2) any AVB Common Shares requested to be disposed of for the account of other holders of AVB Common Shares pursuant to rights existing on the date hereof and set forth on Schedule I and (C) third pro rata among any other holders of AVB Common Shares requested to be disposed of, so that the total number of AVB Common Shares to be included in any such offering for the account of all such Persons will not exceed the number recommended by such lead Underwriter or Underwriters, if any; and
 - (Y) priority in the case of an Underwritten Offering initiated by holders of AVB Common Shares (other than the LBHI Group Members) which gives rise to a Piggyback Registration pursuant to Section 2(h) will be (A) first, all AVB Common Shares requested to be disposed of for the account of the initiating holder or holders pursuant to rights existing on the date hereof and set forth on Schedule I, (B) second, pro rata among the LBHI Group Members and any other holders of AVB Common Shares requested to be disposed of pursuant to rights existing on the date hereof and set forth on Schedule I, (C) third, pro rata among any other holders of AVB Common Shares requested to be disposed of, and (D)

10

<u>fourth</u> any AVB Common Shares proposed to be offered by the Company for its own account, so that the total number of AVB Common Shares to be included in any such offering for the account of all such Persons will not exceed the number recommended by such lead Underwriter or Underwriters, if any.

- (j) <u>Expenses of Registration</u>. Except as otherwise provided in <u>Section 2(g)</u>, all Registration Expenses incurred in connection with any Underwritten Offering, any Demand Registration or any Piggyback Registration, or any qualification or compliance pursuant to this <u>Section 2</u> shall be borne by the Company, and all Selling Expenses shall be borne by the LBHI Group.
- (k) <u>Registration and Underwritten Offering Procedures</u>. In the case of each Registration or Underwritten Offering, as applicable, effected by the Company pursuant to this <u>Section 2</u>, the Company will keep the LBHI Group advised in writing as to the initiation of each Registration or Underwritten Offering and as to the completion thereof. At its expense, the Company shall:
- (i) before filing a Registration Statement, the Prospectus, any supplement to the Prospectus and any amendments or supplements to any Issuer Free Writing Prospectus containing information pertaining to the LBHI Group, provide the LBHI Group with the opportunity to reasonably object to any information pertaining to the LBHI Group that is contained therein and the Company shall make the corrections reasonably and timely requested by the LBHI Group with respect to such information prior to filing any such Registration Statement, Prospectus, any amendment or supplement thereto or any amendment or supplement to any Issuer Free Writing Prospectus;
- (ii) as promptly as practicable, prepare and file with the Commission such pre- and post-effective amendments to such Registration Statement, supplements to the Prospectus and such amendments or supplements to any Issuer Free Writing Prospectus as may be (A) reasonably and timely requested by any LBHI Group Member or (B) necessary to keep such Registration Statement effective for the period of time required by this Agreement, and comply with provisions of the applicable securities Laws with respect to the sale or other disposition of all Registrable Securities covered by such Registration Statement during such period in accordance with the intended method or methods of disposition by the sellers thereof set forth in such Registration Statement;
- (iii) promptly notify the LBHI Group and the lead Underwriter or Underwriters, if any, and, if requested, confirm such notification in writing and provide copies of the relevant documents, as promptly as reasonably practicable: (A) of the filing or effectiveness, as applicable, of the applicable Registration Statement or any

amendment thereto or the applicable Prospectus or Issuer Free Writing Prospectus or any amendment or supplement thereto, (B) of the receipt of any written comments from the Commission or any request by the Commission or any other Governmental Authority for amendments or supplements to such Registration Statement, Prospectus or Issuer Free Writing Prospectus or for additional information, (C) of the issuance by the Commission of any stop order suspending the effectiveness of such Registration Statement or any order by the Commission or any other regulatory authority preventing or suspending the use of any Prospectus or any Issuer Free

11

Writing Prospectus or the initiation or threatening of any proceedings for such purposes, (D) if, at any time, the representations and warranties of the Company in any applicable underwriting agreement cease to be true and correct in all material respects, and (E) of the receipt by the Company or its legal counsel of any notification with respect to the suspension of the qualification of the Registrable Securities for offering or sale in any jurisdiction or the initiation or threatening of any proceeding for such purpose;

- (iv) promptly notify the LBHI Group and the lead Underwriter or Underwriters, if any: (A) when the Company becomes aware of the occurrence of any event as a result of which the applicable Registration Statement, the Prospectus included in such Registration Statement (as then in effect) or any Issuer Free Writing Prospectus contains any untrue statement of a material fact or omits to state a material fact required to be stated therein or necessary in order to make the statements therein (in the case of such Prospectus or any Issuer Free Writing Prospectus, in light of the circumstances under which they were made) not misleading, and (B) when any Issuer Free Writing Prospectus includes information that may conflict with the information contained in the Registration Statement, or, if for any other reason it shall be necessary during such time period to amend or supplement such Registration Statement, Prospectus or Issuer Free Writing Prospectus in order to comply with the Securities Act and, in either case, as promptly as practicable thereafter, prepare and file with the Commission, and furnish without charge to the LBHI Group and the lead Underwriter or Underwriters, if any, any amendment or supplement to such Registration Statement, Prospectus or Issuer Free Writing Prospectus which shall correct such misstatement or omission or effect such compliance;
- (v) use its reasonable best efforts to prevent, or obtain the withdrawal of, any stop order or other order suspending the use of any Prospectus or any Issuer Free Writing Prospectus;
- (vi) deliver to each LBHI Group Member and each Underwriter, if any, without charge, as many copies of the applicable Prospectus, any Issuer Free Writing Prospectus and any amendment or supplement thereto as such LBHI Group Member or Underwriter may reasonably request (it being understood that the Company consents to the use of such Prospectus, any Issuer Free Writing Prospectus and any amendment or supplement thereto by such LBHI Group Member and Underwriters, if any, in connection with the offering and sale of the Registrable Securities thereby) and such other documents as such LBHI Group Member and each Underwriter, if any, may reasonably request in order to facilitate the disposition of the Registrable Securities by such LBHI Group Member and Underwriters, if any;
- (vii) subject to the terms set forth in Section 2(1), on or prior to the date on which the applicable Registration Statement is declared effective, use its reasonable best efforts to register or qualify the Registrable Securities covered by such Registration Statement under such other securities or "blue sky" Laws of such jurisdictions in the United States as any LBHI Group Member reasonably (in light of its intended plan of distribution) requests and do any and all other acts and things that may be reasonably necessary or advisable to enable such LBHI Group Member to consummate the disposition of the Registrable Securities owned by such LBHI Group Member pursuant to such Registration Statement; provided, however, that the Company shall not be obligated to qualify as a foreign corporation to do business under the Laws

12

of any such state in which it is not then qualified or to file any general consent to service of process in any such state;

- (viii) make such representations and warranties to the LBHI Group and the Underwriters or agents, if any, in form, substance and scope as are customarily made by issuers in underwritten public offerings;
- (ix) enter into such customary agreements (including underwriting and indemnification agreements) and take such other actions as the lead Underwriter, if any, reasonably request in order to expedite or facilitate the Registration and disposition of such Registrable Securities;
- (x) use its reasonable best efforts to obtain for delivery to the lead Underwriter, if any, an opinion or opinions from counsel for the Company dated the effective date of the Registration Statement or, in the event of an Underwritten Offering, the date of the closing under the underwriting agreement, in form and substance as is customarily given to Underwriters in an underwritten secondary public offering;
- (xi) in the case of an Underwritten Offering, use its reasonable best efforts to obtain for delivery to the Company and the lead Underwriter, if any, a "comfort" letter from the Company's independent certified public accountants in form and substance as is customarily given by independent certified public accountants in an underwritten secondary public offering;
- (xii) cooperate with each LBHI Group Member and the Underwriters, if any, of such Registrable Securities and their respective counsel in connection with any filings required to be made with FINRA;
- (xiii) use its reasonable best efforts to cause all Registrable Securities covered by the applicable Registration Statement to be listed or quoted on a national securities exchange or trading system and each securities exchange and trading system, if any, on which similar securities issued by the Company are then listed;
- (xiv) cooperate with the LBHI Group and the Underwriters, if any, to facilitate the timely preparation and delivery of certificates, with requisite CUSIP numbers, representing Registrable Securities to be sold and not bearing any restrictive legends;
- (xv) in the case of an Underwritten Offering, ensure that two (2) senior officers of the Company who are reasonably acceptable to LBHI (A) reasonably participate in good faith in the customary "road show" presentations and other customary marketing efforts that may be reasonably requested by the lead Underwriter or Underwriters in any such Underwritten Offering, and (B) take such actions as the lead Underwriter or Underwriters or the LBHI Group may reasonably request in order to expedite or facilitate the sale of Registrable Securities;
- (xvi) use its reasonable best efforts to procure the cooperation of the Company's transfer agent in settling any offering or sale of Registrable Securities, including with respect to the transfer of physical security instruments into book-entry form in accordance with any procedures reasonably requested by the LBHI Group or any lead Underwriter or Underwriters; and

(xvii) (A) make available for inspection by LBHI if the LBHI Group Members are collectively selling at least five percent (5%) of the Registrable Securities included in such Registration, the lead Underwriter or Underwriters, if any, and any attorneys or accountants retained by the LBHI Group or the lead Underwriter or Underwriters, at the offices where normally kept, during reasonable business hours, financial and other records and pertinent corporate documents of the Company as shall be reasonably necessary to enable them to exercise their due diligence responsibility, (B) cause the officers, directors and employees of the Company to supply all information in each case reasonably requested by any such representative, lead Underwriter, attorney or accountant in connection with such Registration Statement, and (C) make the Company's independent certified public accountants available for any such lead Underwriter's or Underwriters' due diligence if so requested by counsel to the Underwriters or LBHI; provided, however, that the Company shall not be required to provide any information pursuant to this section if (1) the Company believes in good faith based on the advice of counsel that to do so would cause the Company to forfeit an attorney-client privilege that was applicable to such information, (2) the Company has requested and been granted from the Commission confidential treatment of such information or (3) any such information is identified by the Company as being confidential or proprietary, unless, with respect to clause (2) or (3) of this sentence, each Person receiving such information shall take such actions as are reasonably necessary to protect the confidentiality of such information and shall sign customary confidentiality agreements reasonably requested by the Company prior to the receipt of such information; and provided, further that, the foregoing inspection and information gathering shall, to the greatest extent possible, be coordinated on behalf of the parties entitled thereto by one counsel d

(l) Right to Defer Registration.

(i) Notwithstanding anything to the contrary contained in this Agreement, the Company shall be entitled, from time to time, by providing prior written notice to LBHI, to require the LBHI Group Members to suspend the use of the Shelf Registration Statement and the Prospectus included therein for resales of Registrable Securities pursuant to Section 2(a) or Section 2(b) or to postpone the filing or suspend the use of any Registration Statement pursuant to Section 2(a), Section 2(b) or Section 2(b) for a reasonable period of time not to exceed forty-five (45) consecutive days (a "Black-Out Period") if the chief executive officer or chief financial officer of the Company determines in his or her good faith judgment that it would be materially detrimental to the Company and/or its securityholders not to defer the filing, or not to suspend the use by LBHI Group, of a Registration Statement or Prospectus by reason of: (A) the Company being in possession of material non-public information, so long as the chief executive officer or chief financial officer of the Company determines in good faith that the disclosure of such information during the period specified in such notice would be required to be disclosed and that such disclosure would be materially detrimental to the Company and/or its securityholders, (B) a contemplated financing, acquisition, disposition, corporate reorganization, merger, or other similar transaction or other material event or circumstance affecting the Company or its securities, so long as the chief executive officer or chief financial officer of the Company determines in good faith that the disclosure of such transaction, event or circumstance at such time would be required to be disclosed and that such disclosure would be materially

14

detrimental to the Company and/or its securityholders, or (C) a requirement to include pro forma or other information, which requirement the Company is reasonably unable to comply with at such time; provided, that the Company shall not be permitted to (x) require the LBHI Group Members to suspend the use of such Prospectus or (y) postpone the filing, or suspend the use, of such Registration Statement pursuant to this Section 2(1), for more than ninety (90) days, in the aggregate, in any twelve (12) month period. For the avoidance of doubt, the Company shall not be permitted to require the LBHI Group Members to suspend the use of such Prospectus or postpone the filing, or suspend the use, of such Registration Statement pursuant to this Section 2(1) unless the purpose of invoking the Company's rights under this Section 2(1) is separate from, and not related to, the Company's desire or intent to give precedence or confer an advantage (in terms of timing or otherwise) to an Underwritten Offering that the Company might wish to effect, or which it might be contemplating, for its own account or for the account of a holder of AVB Common Shares other than the LBHI Group in connection with an Underwritten Offering of Registrable Securities sought to be effected by LBHI on behalf of the LBHI Group. If, for any reason, the Company files a Registration Statement or, subject to Section 2(a)(iii), commences an Underwritten Offering (on its own behalf or on behalf of a holder of AVB Common Shares) during the Black-Out Period, (1) the Black-Out Period shall immediately cease and no Black-Out Period shall be re-instituted for so long as the Company or another holder of AVB Shares is permitted to sell AVB Common Shares under such Registration Statement or such Underwritten Offering is ongoing, (2) the LBHI Group shall be permitted to resume the use of any effective Shelf Registration Statement for resales of Registrable Securities or to recommence an Underwritten Offering pursuant to a Demand Registration, (3) the obligation of the Company to file a Shelf Registration Statement pursuant to Section 2(a)(ii), or to effect any Underwritten Offering pursuant to Section 2(a)(iii) (subject to Section 2(a)(iii)), or any Demand Registration pursuant to Section 2(b)(i), shall immediately recommence, and (4) the LBHI Group shall be permitted to participate pursuant to Section 2(h) in a new Registration Statement filed by the Company during the Black-Out Period pursuant to Section 2(h). Notwithstanding anything to the contrary contained herein, if LBHI determines to rescind any prior Underwritten Offering Request or Demand Notice following receipt of a notice from the Company pursuant to this Section 2(1), then LBHI may, at its election, give written notice of such election to the Company; provided, however, that such rescinded Underwritten Offering Request or Demand Notice shall not be counted as an Underwritten Offering Request or Demand Notice for purposes of the limits in Section 2(a)(ii) and Section 2(b)(i), respectively, without any requirement to reimburse the Company for any related expenses incurred by the Company.

(ii) In the event of any suspension pursuant to this Section 2(1), the Company shall use its reasonable best efforts to keep LBHI Group apprised of the estimated length of the anticipated delay and such information shall be kept confidential and used by the LBHI Group solely for purposes of planning in connection with the exercise of its rights hereunder. The Company will notify the LBHI Group promptly upon the termination of the Black-Out Period. Upon notice by the Company to the LBHI Group of any determination to commence a Black-Out Period, the LBHI Group Members shall, except as required by applicable Law, including any disclosure obligations under Section 13 of the Exchange Act, keep the fact of any such Black-Out Period strictly confidential, and during any Black-Out Period, promptly halt any offer, sale, trading or transfer of any AVB Common Shares for the duration of the Black-Out Period under the applicable Registration Statement or Shelf Registration Statement until the Company has provided notice to the LBHI Group that the Black-Out Period has been terminated.

15

(iii) After the expiration of any Black-Out Period and without any further request from any member of the LBHI Group, the Company shall as promptly as reasonably practicable prepare a Registration Statement or post-effective amendment or supplement to the applicable Shelf Registration Statement or Prospectus, or any document incorporated therein by reference, or file any other required document so that, as thereafter delivered to purchasers of the Registrable Securities included therein, the Prospectus will not include a material misstatement or omission or be not effective and useable for resale of Registrable Securities.

(m) Obligations of the Members of the LBHI Group.

(i) Each LBHI Group Member shall furnish to the Company such information regarding such LBHI Group Member and its partners and members, and the distribution proposed by such LBHI Group Member, as the Company may reasonably request and as shall be reasonably requested in connection with any Registration, qualification or compliance referred to in this Section 2.

- (ii) In the event that, either immediately prior to or subsequent to the effectiveness of any Registration Statement, any LBHI Group Member shall distribute Registrable Securities to its stockholders, partners or members, such LBHI Group Member shall so advise the Company and provide such information as shall be necessary or advisable to permit an amendment to such Registration Statement or supplement to any Prospectus to provide information with respect to such stockholders, partners or members, in their capacity as selling security holders. As soon as reasonably practicable following receipt of such information, the Company shall file an appropriate amendment to such Registration Statement or supplement to any Prospectus reflecting the information so provided. Any incremental expense to the Company resulting from such amendment shall be borne by such LBHI Group Member.
- (n) Rule 144. With a view to making available the benefits of certain rules and regulations of the Commission which may permit the sale of restricted securities to the public without Registration, the Company agrees to use its reasonable best efforts to file with the Commission in a timely manner all reports and other documents required of the Company under the Securities Act and the Exchange Act at any time after it has become subject to such reporting requirements (or, if the Company is not required to file such reports, it will, upon the reasonable request of LBHI, make publicly available such necessary information for so long as necessary to permit sales pursuant to Rules 144 under the Securities Act).
- (o) <u>Termination</u>. Notwithstanding anything herein to the contrary, this Agreement, other than <u>Section 4</u> and <u>Section 5</u>, shall terminate and cease to be available as to any securities held by a LBHI Group Member upon the earlier of (i) the fifth (5th) anniversary of the date of the Initial Closing and (ii) such time as the LBHI Group Members collectively hold Shares having an aggregate fair market value for any consecutive ten (10) trading-day period (measured by the average daily closing price per share of the AVB Common Shares during such ten (10) trading-day period) of less than \$250 million (the "*Termination Date*").

16

SECTION 3. LOCK-UP AGREEMENTS

- Obligations of the Company. The Company agrees that, if requested by the lead Underwriter or Underwriters in any Underwritten Offering of Registrable Securities contemplated by this Agreement, it will enter into (and will use its reasonable best efforts to cause each of its directors and executive officers to enter into) a customary "lock-up" agreement pursuant to which the Company (or such director or executive officer) will agree not to, directly or indirectly, sell, offer to sell, grant any option for the sale of, or otherwise dispose of any AVB Common Shares or securities convertible into or exchangeable or exercisable for AVB Common Shares for a period of thirty (30) days (or such shorter period to which the LBHI Group is subject) from the pricing date of such Underwritten Offering; provided, however, that (i) any such "lockup" agreement shall not prohibit or in any way restrict the Company from, directly or indirectly, selling, offering to sell, granting any option for the sale of, or otherwise disposing of (A) any Qualifying Employee Stock (or otherwise maintaining its employee benefits plans in the ordinary course of business) or (B) AVB Common Shares upon the redemption or exchange of any securities redeemable or exchangeable for AVB Common Shares (including the redemption of any units of Upper Falls Limited Partnership), (ii) any restrictions on the Company's ability to file registration statements with the SEC shall not apply to the filing and effectiveness of any amendments to the Company's existing resale registration statements or the addition of any subsidiary guarantor registrants thereto, as applicable, any shelf registration statements relating to existing employee or director compensation-related plans or distribution reinvestment plans, or a new universal shelf registration statement, provided that the securities registered under such new universal shelf registration statement shall remain subject to the lockup provisions; (iii) any restrictions on the ability of Affiliates of the Company to enter into transactions relating to the Company's securities shall not apply to (A) transfers by gift, will or intestacy so long as the transferee delivers a similar lock-up, (B) transfers or sales pursuant to contracts, instructions or plans to transfer AVB Common Shares pursuant to Rule 10b5-1 existing on the date of the applicable underwriting agreement, or the amendment or replacement of any such contract, instruction or plan so long as the number of AVB Common Shares subject thereto is not increased, and the exercise of options in connection therewith, and (C) the withholding of securities to pay taxes upon the vesting of certain equity awards granted by the Company.
- (b) Obligations of LBHI. During the period beginning on the date of this Agreement and ending on the Termination Date, if requested by the lead Underwriter or lead Underwriters in any Underwritten Offering effected by the Company (other than on behalf of the LBHI Group) in which any LBHI Group Member participates as a selling holder, so long as the LBHI Group Members collectively beneficially own in excess of five percent (5%) of the AVB Common Shares outstanding, LBHI, on behalf of the LBHI Group, will enter into a customary "lock-up" agreement providing that they will not sell, grant any option for the sale of, or otherwise dispose of any AVB Common Shares outside of such public offering for a period of thirty (30) days (or such shorter period to which the Company is subject (without regard to clause (i) of the proviso contained in Section (3)(a))) from the pricing date of such Underwritten Offering.

17

SECTION 4. INDEMNIFICATION

- Indemnification by the Company. With respect to each Underwritten Offering or Registration which has been effected pursuant to Section 2, the Company agrees to indemnify and hold harmless, to the fullest extent permitted by Law, (i) each of the LBHI Group Members and each of their respective officers, directors, limited or general partners and members, (iii) each member, limited or general partner of each such member, limited or general partner, (iii) each of their respective Affiliates, officers, directors, stockholders, employees, advisors, and agents, and (iv) each Person who controls (within the meaning of the Securities Act or the Exchange Act) such Persons, against any and all claims, losses, damages, penalties, judgments, suits, costs, liabilities and expenses (or actions in respect thereof) (collectively, the "Losses") arising out of or based on (A) any untrue statement (or alleged untrue statement) of a material fact contained in any Registration Statement (including any Prospectus or Issuer Free Writing Prospectus) or any other document incorporated by reference therein, (B) any omission (or alleged omission) to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading (in the case of any Prospectus or Issuer Free Writing Prospectus, in light of the circumstances under which they were made), or (C) any violation by the Company of the Securities Act, the Exchange Act or any state securities or "blue sky" Laws applicable to the Company and relating to action or inaction required of the Company in connection with any such Registration, qualification or compliance, and will reimburse each of the Persons listed above, for any reasonable and documented out-of-pocket legal and any other expenses reasonably incurred in connection with investigating and defending any such Losses; provided, that the Company will not be liable in any such case to the extent that any such Losses arise out of or are based on any untrue statement, Prospectus or Issuer Fre
- (b) Indemnification by the LBHI Group. In connection with a Registration Statement in which any LBHI Group Member participates, each of the LBHI Group Members agrees, jointly and severally, to indemnify and hold harmless, to the fullest extent permitted by Law, (i) the Company, each of its directors, officers and employees, (ii) each of their respective Affiliates, employees, advisors and agents, and (iii) each Person who controls the Company (within the meaning of the Securities Act or the Exchange Act), against any and all Losses arising out of or based on (A) any untrue statement (or alleged untrue statement) of a material fact contained in any Registration Statement (including any Prospectus or Issuer Free Writing Prospectus) or any other document incorporated by reference therein made by such LBHI Group Member or (B) any omission (or alleged omission) to state therein a material fact required to be stated therein or necessary to make the statements by such LBHI Group Member therein not misleading (in the case of any Prospectus or Issuer Free Writing Prospectus, in light of the circumstances under which they were made), and will reimburse the Persons listed above for any reasonable and documented legal or any other expenses reasonably incurred in connection with investigating or defending any such Losses, in each case to the extent, but only to the extent, that such untrue statement (or alleged untrue statement) or omission (or alleged omission) is made in reliance upon and in conformity with written

(after giving effect to any underwriters discounts and commissions) it receives in the disposition of any Registrable Securities.

- (c) Conduct of the Indemnification Proceedings. Each Person entitled to indemnification under this Section 4 (the "Indemnified Party") shall give notice to the Person required to provide indemnification (the "Indemnifying Party") promptly after such Indemnified Party has actual knowledge of any claim as to which indemnity may be sought, and shall permit the Indemnifying Party to assume the defense of any such claim or any litigation resulting therefrom; provided, that counsel for the Indemnifying Party, who shall conduct the defense of such claim or any litigation resulting therefrom, shall be approved by the Indemnified Party (whose approval shall not unreasonably be withheld) and the Indemnified Party may participate in such defense at such Indemnified Party in such action, in which case the fees and expenses of counsel shall be at the expense of the Indemnifying Party); and provided, further, that the failure of any Indemnified Party to give notice as provided herein shall not relieve the Indemnifying Party of its obligations under this Section 4 unless the Indemnifying Party is actually and materially prejudiced thereby. It is understood and agreed that the Indemnifying Party shall not, in connection with any proceeding or related proceeding in the same jurisdiction, be liable for the fees and expenses of more than one separate legal counsel for all Indemnified Parties; provided, however, that where the failure to be provided separate legal counsel would be reasonably likely to result in a conflict of interest on the part of such legal counsel for all Indemnified Parties, separate counsel shall be appointed for the Indemnified Parties to the extent needed to alleviate such potential conflict of interest. No Indemnifying Party, in the defense of any such claim or litigation shall, except with the prior written consent of each Indemnified Party, consent to entry of any judgment or enter into any settlement which does not include as an unconditional term thereof the giving by the claimant or plaintiff to such
- (d) If the indemnification provided for in this Section 4 is held by a court of competent jurisdiction to be unavailable to an Indemnified Party with respect to any Losses, then the Indemnifying Party, in lieu of indemnifying such Indemnified Party hereunder, shall contribute to the amount paid or payable by such Indemnified Party as a result of such Losses in such proportion as is appropriate to reflect the relative fault of the Indemnifying Party, on the one hand, and of the Indemnified Party, on the other, in connection with the statements or omissions (or alleged statements or omissions) which resulted in such Losses, as well as any other relevant equitable considerations. The relative fault of the Indemnifying Party and of the Indemnified Party shall be determined by reference to, among other things, whether the untrue (or alleged untrue) statement of a material fact or the omission (or alleged omission) to state a material fact relates to information supplied by the Indemnifying Party or by the Indemnified Party and such parties' relative intent, knowledge, access to information and opportunity to correct or prevent such statement or omission; provided, however, that the obligations of each of the LBHI Group Members shall be limited to an amount equal to the net proceeds (after giving effect to any underwriters discounts and commissions) it receives in such Registration; and

19

provided, further, that no Person guilty of fraudulent misrepresentation (within the meaning of Section 11(f) of the Securities Act) shall be entitled to contribution from any Person who was not guilty of such fraudulent misrepresentation.

(e) Subject to the limitations on the LBHI Group's liability set forth in Section 4(b) and Section 4(d), the remedies provided for in this Section 4 are not exclusive and shall not limit any rights or remedies which may otherwise be available to any Indemnified Party at Law or equity. The remedies shall remain in full force and effect regardless of any investigation made by or on behalf of the LBHI Group or any Indemnified Party and survive the transfer of such securities by the LBHI Group and the termination of the registration rights set forth in Section 2.

SECTION 5. MISCELLANEOUS

- (a) <u>Adjustments.</u> References to the Shares contained herein will be deemed adjusted or modified to account for any reclassification, exchange, substitution, combination, stock split or reverse stock split of AVB Common Shares.
 - (b) <u>Governing Law; Jurisdiction</u>.
- (i) This Agreement shall be construed, performed and enforced in accordance with the laws of the State of New York (without giving effect to its principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of the Laws of another jurisdiction).
- (ii) The Parties agree that any suit, action or other proceeding seeking to enforce any provision of, or based on any matter arising out of or in connection with, this Agreement or the transactions contemplated hereby shall be brought in any state or federal court located in the State of New York, and each of the Parties hereby irrevocably consents to the exclusive jurisdiction of such courts (and of the appropriate appellate courts therefrom) in any such suit, action or other proceeding and irrevocably waives, to the fullest extent permitted by Law, any objection that it may now or hereafter have to the laying of the venue of any such suit, action or other proceeding in any such court or that any such suit, action or other proceeding brought in any such court has been brought in an inconvenient forum. Process in any such suit, action or other proceeding may be served on any Party anywhere in the world, whether within or without the jurisdiction of any such court. Without limiting the foregoing, each Party agrees that service of process on such Party as provided in Section 5(c) shall be deemed effective service of process on such Party.
- (c) Notices. All notices, consents and other communications hereunder shall be in writing (including facsimile, electronic mail or similar writing) and shall be deemed to have been duly given (i) when delivered by hand or by Federal Express or a similar overnight courier to (or if that day is not a Business Day, or if delivered after 5:00 p.m., New York, New York time on a Business Day, on the first following day that is a Business Day), (ii) five (5) days after being deposited in any United States Post Office enclosed in a postage prepaid, registered or certified envelope addressed to, (iii) when successfully transmitted by facsimile (with a confirming copy of such communication to be sent as provided in clauses (i) or (ii) above) or

20

if to the Company, to:

AvalonBay Communities, Inc. Ballston Tower 671 N. Glebe Road, Suite 800 Arlington, VA 22203

Facsimile No.: (703) 329-4830

E-mail address: ted_schulman@avalonbay.com

Attention: Edward M. Schulman, Executive Vice President — General Counsel and Secretary

with copies (which shall not constitute notice) to:

Goodwin Procter LLP Exchange Place 53 State Street Boston, MA 02109 Facsimile No.: (617) 523-1231

E-mail address: jhaggerty@goodwinprocter.com

Attention: John T. Haggerty

21

if to the LBHI Group, to:

c/o Lehman Brothers Holdings Inc. 1271 Avenue of the Americas New York, New York 10020 Facsimile No.: (646) 834-4340

E-mail address: doug.sesler@lehmanholdings.com

Attention: Douglas Sesler

with copies (which shall not constitute notice) to:

Weil Gotshal & Manges LLP 767 Fifth Avenue New York, New York 10153 Facsimile No.: (212) 310-8007

E-mail address: michael.bond@weil.com

raymond.gietz@weil.com Attention: W. Michael Bond, Esq. Raymond O. Gietz, Esq.

- (d) <u>Successors and Assigns.</u> Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned, delegated or otherwise transferred by any Party (whether by operation of Law or otherwise) without the prior written consent of the other Parties; <u>provided, however</u>, that the rights and obligations hereunder of a LBHI Group Member may be assigned, in whole or in part, to any Person who acquires such Registrable Securities (<u>provided, that any such rights and obligations may be assigned solely with respect to such Registrable Securities) (each such Person, a "*Permitted Assignee*"). Any assignment pursuant to this <u>Section 5(d)</u> shall be effective and any Person shall become a Permitted Assignee only upon receipt by the Company of (A) a written notice from LBHI stating the name and address of the transferee and identifying the number of shares of Registrable Securities with respect to which the rights under this Agreement are being transferred and, if fewer than all of the rights attributable to a LBHI Group Member hereunder are to be so transferred, the nature of the rights so transferred and (B) a written instrument by which the transferee agrees to be bound by all of the terms and conditions applicable to a holder of such Registrable Securities, including, without limitation, pursuant to this Agreement. Any attempted assignment, delegation or transfer in violation of this <u>Section 5(d)</u> shall be null and void.</u>
- (e) <u>Additional Shareholders</u>. The parties hereto acknowledge that certain Persons may become shareholders of the Company and the Company may grant such Persons registration rights with respect to the AVB Common Shares issued to such Persons; <u>provided</u>, <u>however</u>, that such registration rights do not conflict with, and do not impair the registration rights granted to the LBHI Group hereunder.
- (f) Entire Agreement. This Agreement sets forth the entire agreement and understanding of the Parties in respect of the transactions contemplated by this Agreement and supersedes all prior discussions, negotiations, agreements, arrangements and understandings, whether oral or written, relating to the subject matter hereof and thereof. There are no

22

warranties, representations or other agreements between the Parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement.

(g) <u>Amendments and Waivers</u>.

- (i) Any provision of this Agreement may be amended or modified only by a written instrument signed by all of the Parties hereto and any such amendment shall apply to all holders of Registrable Securities and all of their Registrable Securities; provided, that notwithstanding the foregoing, additional Persons may become party hereto upon an assignment of rights and obligations hereunder pursuant to Section 5(d); and provided, further, that other than as set forth in Section 5(d), the Company may not add additional parties hereto without the consent of LBHI.
- (ii) No waiver hereunder shall be valid or binding unless set forth in writing and duly executed by the Party against whom enforcement of the waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the Party granting such waiver in any other respect or at any other tune. Any valid or binding waiver of the observance of any term of this Agreement shall apply to all holders of Registrable

Securities and all of their Registrable Securities only if made by LBHI. Neither the waiver by any of the Parties of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the Parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. Except as otherwise provided herein, no action taken pursuant to this Agreement, including any investigation by or on behalf of any Party, shall be deemed to constitute a waiver by the Party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement.

(h) Specific Performance. The Parties agree that irreparable damage would occur in the event any of the provisions of this Agreement were not performed in accordance with the terms hereof and that each of the Parties shall be entitled to specific performance of the terms hereof pursuant to this Section 4(h), in addition to any other remedy at Law or in equity. It is accordingly agreed that the Company or any holder of Registrable Securities shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement by the other Parties and to enforce specifically the terms and provisions hereof, this being in addition to any other remedy to which they are entitled at Law or in equity, without the necessity of posting bonds or other undertaking in connection therewith. Any requirements for the securing or posting of any bond with such remedy are hereby waived.

(i) WAIVER OF JURY TRIAL.

(i) EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY OR CLAIM WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY

23

LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. THIS WAIVER MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING (OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS SECTION 5(i)(i)) AND EXECUTED BY EACH OF THE PARTIES HERETO). THIS WAIVER SHALL APPLY TO ANY SUBSEQUENT AMENDMENTS, RENEWALS, SUPPLEMENTS OR MODIFICATIONS TO THIS AGREEMENT OR ANY OTHER AGREEMENTS OR DOCUMENTS RELATING TO THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. The scope of this waiver is intended to be all-encompassing of any and all disputes that may be filed in any court and that relate to the subject matter of the transactions contemplated by this Agreement, including contract claims, tort claims, breach of duty claims and all other common law and statutory claims. In the event of litigation, this Agreement may be filed as a written consent to a trial by the court.

- (ii) EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE EITHER OF SUCH WAIVERS, (B) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVERS, (C) IT MAKES SUCH WAIVERS VOLUNTARILY, AND (IV) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 5(i).
- (j) No Conflicting Agreements. The Company represents and warrants to LBHI that it is not currently a party to any agreement which conflicts with or impairs the rights granted to the LBHI Group by this Agreement.
- (k) <u>Binding Effect</u>. This Agreement will be binding upon, inure solely to the benefit of and be enforceable by the Parties and their respective permitted successors and assigns.
- (l) Interpretation of this Agreement. As used in this Agreement and required by the context, the singular shall be deemed to include the plural and the plural shall be deemed to include the singular; all genders shall be deemed to include all other genders; words importing persons shall include partnerships, corporations and other entities; when reference is made in this Agreement to a Section or Schedule, such reference shall be to a Section or Schedule of this Agreement unless otherwise indicated; and the terms "herein," "hereof" and "hereunder" or other similar terms, refer to this Agreement as a whole and not only to the particular sentence, subsection or section in which any such term may be employed. Whenever in this Agreement the word "including" is used, it shall be deemed to be for purposes of identifying only one or more of the possible alternatives, and the entire provision in which such word appears shall be read as if the phrase "including without limitation" were actually used in the text. The section headings herein are for convenience only and shall not affect the construction hereof. Where any provision in this Agreement refers to action to be taken by any Person, or which such Person is prohibited from taking, such provision shall be applicable whether the action in question is taken

24

directly or indirectly by such Person. Except when used together with the word "either" or otherwise for the purpose of identifying mutually exclusive alternatives, the term "or" has the inclusive meaning represented by the phrase "and/or". With regard to each and every term and condition of this Agreement, the Parties understand and agree that the same have or has been mutually negotiated, prepared and drafted, and that if at any time the Parties desire or are required to interpret or construe any such term or condition or any agreement or instrument subject thereto, no consideration shall be given to the issue of which Party actually prepared, drafted or requested any term or condition of this Agreement.

- (m) Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other Governmental Authority to be invalid, void or unenforceable, such term, provision, covenant or restriction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Agreement, and the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated so long as the economic or legal substance of the transactions contemplated by this Agreement is not affected in any manner materially adverse to any Party. Upon such determination, the Parties shall use their commercially reasonable efforts to negotiate in good faith to modify this Agreement so as to affect the original intent of the Parties as closely as possible in an acceptable manner in order that the transactions contemplated by this Agreement shall be consummated as originally contemplated to the fullest extent possible.
- (n) <u>Counterparts</u>. This Agreement may be executed and delivered in multiple counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. It is the express intent of the Parties to be bound by the exchange of signatures on this Agreement via facsimile or electronic mail via the portable document format (PDF). A facsimile or other copy of a signature shall be deemed an original. This Agreement shall become effective when each Party hereto shall have received a counterpart hereof signed by all of the other Parties hereto. Until and unless each Party has received a counterpart hereof signed by the other Parties hereto, this Agreement shall have no effect and no Party shall have any right or obligation hereunder (whether by virtue of any other oral or written agreement or other communication).
- (o) Third Parties. Except as otherwise expressly provided herein, no provision of this Agreement is intended or shall confer on any Person, other than the Parties (and their successors and permitted assigns), any rights under this Agreement and no other Person shall be entitled to rely thereon.

observed.	(p) <u>Further Assurances</u> . The parties hereto will do such further acts and things necessary to assure that the terms of this Agreement are carried out and								
	[The remainder of this page is intentionally left blank.]								
				25					
	IN WITNE	SS WHEREOF, th	ne undersigned have		n Rights Agreement as of the date first set forth above. LONBAY COMMUNITIES, INC.				
				Ву:	Name: Title:				
				LEF	IMAN BROTHERS HOLDINGS INC.				
				Ву:	Name: Title:				
AVALONBAY COMMUNITIES, INC. REGISTRATION RIGHTS AGREEMENT									

AVALONBAY COMMUNITIES, INC. SHAREHOLDERS AGREEMENT

This Shareholders Agreement (this "Agreement") is entered into as of [], by and among AvalonBay Communities, Inc., a Maryland corporation ("AVB"), Archstone Enterprise LP, a Delaware limited partnership ("Archstone") and Lehman Brothers Holdings Inc., a Delaware corporation ("LBHI"). AVB, Archstone and LBHI are sometimes referred to herein as the "Parties" and each, a "Party."

WHEREAS, on November 26, 2012, Archstone, LBHI, AVB, ERP Operating Limited Partnership, an Illinois limited partnership ("ERPOP"), and Equity Residential, a Maryland real estate investment trust, entered into an Asset Purchase Agreement (as it may be amended from time to time, the "Asset Purchase Agreement") pursuant to which, among other things, (i) ERPOP and AVB or one or more Buyer Designees (as defined in the Asset Purchase Agreement) have agreed to acquire all of the assets and assume all of the liabilities of Archstone (other than as provided therein), (ii) ERPOP has agreed to deliver, or cause its Buyer Designees to deliver, to Archstone (or, if so directed by Archstone, to LBHI and/or a Lehman Designee (as defined in the Asset Purchase Agreement)) the ERPOP Equity Consideration (as defined in the Asset Purchase Agreement), and (iii) AVB has agreed to deliver, or cause its Buyer Designees to deliver, to Archstone (or, if so directed by Archstone, to LBHI and/or a Lehman Designee) the AVB Equity Consideration (as defined in the Asset Purchase Agreement); and

WHEREAS, pursuant to and in accordance with Section 10.1.6(a) of the Asset Purchase Agreement, as a condition precedent to receiving the AVB Equity Consideration, each of the LBHI Parties have agreed to be bound by the terms and conditions of this Agreement.

NOW, THEREFORE, in consideration of the foregoing and the respective representations, warranties, covenants and agreements set forth herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereto, intending to be legally bound hereby, agree as follows:

ARTICLE 1 DEFINITIONS

1.1 <u>Definitions</u>. When used in this Agreement, the following terms in all of their tenses, cases and correlative forms shall have the meanings assigned to them in this <u>Section 1.1</u> or elsewhere in this Agreement. Capitalized terms used herein and not otherwise defined have the meanings assigned to them in the Asset Purchase Agreement. "<u>Beneficial Ownership</u>" and "<u>Beneficially Own</u>" shall have the meaning given to such term in Rule 13d-3 under the Exchange Act (disregarding the reference to "within 60 days" in Rule 13d-3(d)(1)(i)).

"Extraordinary Transactions" means (a) any merger, consolidation, sale of all or substantially all of the assets, other business combination, liquidation, reclassification, recapitalization, restructuring or other similar action to which AVB is a constituent party, or (b) any issuance of securities to any Person or Group (within the meaning of Section 13(d)(3) of the Exchange Act) requiring approval of the common shareholders of AVB in accordance with any

Law or the rules and regulations of the New York Stock Exchange as to such matters, as in effect from time to time.

"LBHI Party" means Archstone, LBHI and any Lehman Designee holding AVB Common Shares.

"Permitted Transfers" shall mean, in each case, so long as (x) such Transfer is in accordance with applicable Law and (y) each LBHI Party is and at all times has been in compliance with Article 3 of this Agreement (other than any unintentional noncompliance that was cured as promptly as practicable upon the applicable LBHI Parties becoming aware of such noncompliance): (i) any Transfer to an Affiliate of the applicable LBHI Party or a liquidating trust established pursuant to the Plan, so long as such Person, in connection with such Transfer, executes a joinder to this Agreement in the form attached hereto as Exhibit A, pursuant to which such Person or Group (within the meaning of Section 13(d)(3) of the Exchange Act) agrees to become a Party to this Agreement and subject to the restrictions applicable to a LBHI Party and otherwise become a Party for all purposes of this Agreement; provided that no such Transfer(s) shall relieve the transferring LBHI Party or LBHI Parties from the obligations of such LBHI Party or LBHI Parties under this Agreement, and (ii) any Transfer solely to tender any of the AVB Equity Consideration into a tender or exchange offer commenced by AVB or a third party if the board of directors of AVB has affirmatively publicly recommended to the AVB shareholders acceptance of such tender offer or exchange offer pursuant to Rule 14d-9 under the Exchange Act with respect to a third party tender or exchange offer or has determined not to oppose (as evidenced by its filings pursuant to such Rule 14d-9) the tender or exchange offer.

"Plan" shall mean that certain Modified Third Amended Joint Chapter 11 Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors, dated December 6, 2011 and that certain Order Confirming Modified Third Amended Joint Chapter 11 Plan of Lehman Brothers Holdings Inc. and its Affiliated Debtors, dated December 6, 2011, [Docket No. 23023].

"Qualified Successor Transfer" shall mean any Transfer of common equity securities of AVB: (i) through an underwritten offering, (ii) to an underwriter that participates in a public offering (including a "bought deal" or a registered block trade) of common equity securities of AVB, but only to the extent necessary to facilitate such public offering, (iii) to Persons that are and will remain (for so long as such Persons own a sufficient number of common equity securities of AVB to require a filing on Schedule 13G under the Exchange Act) eligible to file a Schedule 13G with the Securities and Exchange Commission with respect to such common equity securities of AVB, (iv) pursuant to Rule 144 under the Securities Act, (v) to any bona fide financing source pursuant to a pledge by any LBHI Party of any of the AVB Equity Consideration as collateral securing indebtedness of such LBHI Party or any of its Affiliates, (vi) pursuant to a sale conducted by a bona fide financing source described in the preceding clause (v) in connection with a foreclosure with respect to any of the AVB Equity Consideration or a conveyance in lieu of foreclosure to any such financing source or its Affiliates, (vii) in a transaction described in clause (ii) of the definition of Permitted Transfers or (viii) during a "subsequent offering period" (as referenced in Rule 14d-11 of the Exchange Act) to any Person who has made a tender or exchange offer for all outstanding AVB Common Shares and who has purchased a number of AVB Common Shares tendered pursuant to such tender or exchange offer such that such Person, together with its Affiliates, Beneficially Owns in excess of 50% of the

2

outstanding AVB Common Shares following the expiration of the initial offering period for such tender or exchange offer.

"Standstill Period" shall mean the period beginning on the date hereof and ending on the first date on which the LBHI Parties (and any Affiliate transferees) in the aggregate cease to Beneficially Own 5% or more of the outstanding common equity securities of AVB.

"Transfer" means (i) any direct or indirect offer, sale, assignment, encumbrance, pledge, hypothecation, disposition, loan or other transfer (by operation of law or otherwise), either voluntary or involuntary, or entry into any contract, option or other arrangement or understanding with respect to any offer, sale, assignment, encumbrance, pledge, hypothecation, disposition, loan or other transfer (by operation of law or otherwise), of any capital stock (or any security convertible or exchangeable into capital stock) or interest in any capital stock or (ii) during the Lock-Up Period only, in respect of any capital stock or interest in any capital stock, to enter into any swap or any other agreement, transaction or series of transactions that hedges or transfers, in whole or in part, directly or indirectly, the economic consequence of ownership of such capital stock or interest in capital stock, whether any such swap, agreement, transaction or series of transactions is to be settled by delivery of securities, in cash or otherwise (and with respect to the AVB Equity Consideration, this clause (ii) shall be deemed to include, without limitation, any hedging arrangement or transfer based on the FTSE NAREIT All Residential Capped Index, the FTSE NAREIT Equity Residential Index, the FTSE NAREIT Equity Apartments Index or any other index of which the capital stock of AVB represents at least 5% of the value at the time such hedging arrangement or transfer is entered into).

ARTICLE 2 LOCK-UP AGREEMENT

- 2.1 Other than in the case of a Permitted Transfer, no LBHI Party shall Transfer any of the AVB Equity Consideration for the period beginning on the date hereof and ending on April 26, 2013 (such period, the "Lock-Up Period"). For the avoidance of doubt, nothing in this Section 2.1 shall prohibit any LBHI Party, from and after the expiration of the Lock-Up Period, from entering into any swap, derivative or any other agreement, transaction or series of transactions that hedges or transfers, in whole or in part, directly or indirectly, the economic consequence or ownership of the AVB Equity Consideration or interest in the AVB Equity Consideration, whether any such swap, derivative, agreement, transaction or series of transactions is to be settled by delivery of securities, in cash or otherwise, including, without limitation, any hedging arrangement or transfer based on any index referenced in the definition of "Transfer".
- 2.2 Any Transfer or attempted Transfer of any of the AVB Equity Consideration in violation of this Article 2 shall, to the fullest extent permitted by Law, be null and void <u>ab initio</u>, and AVB shall not, and shall instruct its transfer agent and other third parties not to, record or recognize any such purported Transfer on the share register of AVB.

3

ARTICLE 3 STANDSTILL AGREEMENT

- 3.1 During the Standstill Period, the LBHI Parties shall not, directly or indirectly, without the prior written consent of AVB:
- 3.1.1 Acquire, agree to acquire or make any public proposal to acquire, directly or indirectly, Beneficial Ownership of common equity securities of AVB or any other securities of AVB entitled to vote generally in the election of directors of AVB (collectively, "Voting Securities"), or securities of the Company that are convertible, exchangeable or exercisable into Voting Securities (other than (i) the receipt of common equity securities of AVB pursuant to the Asset Purchase Agreement, (ii) the acquisition of common equity securities of AVB or other Voting Securities as a result of any stock splits, stock dividends or other distributions or recapitalizations, reclassifications, reorganizations or similar transactions or offerings made available by AVB to holders of Voting Securities, including rights offerings, and (iii) from an Affiliate of any LBHI Party in a Permitted Transfer);
- 3.1.2 Deposit any Voting Securities in a voting trust or similar arrangement or subject any Voting Securities to any voting agreement, pooling arrangement or similar arrangement (other than (x) with another LBHI Party or any direct or indirect subsidiary of the LBHI Parties or (y) in connection with any transaction contemplated by the Asset Purchase Agreement or this Agreement), or grant any proxy with respect to any Voting Securities (other than (x) to AVB or a person specified by AVB, in a proxy card provided to shareholders of AVB by or on behalf of AVB, (y) to another LBHI Party or any direct or indirect subsidiary of the LBHI Parties, or (z) as otherwise necessary to permit any LBHI Party to vote Voting Securities as expressly permitted by Section 4.2);
- 3.1.3 Enter, agree to enter, propose or offer to enter into or facilitate any merger, business combination, tender offer, recapitalization, restructuring, change in control transaction or other similar extraordinary transaction involving AVB or any of its subsidiaries (excluding voting as a shareholder with respect to such a transaction, to the extent permitted by Section 4.2, or tendering shares in a tender offer);
- 3.1.4 Make, or in any way participate or engage in, any "solicitation" of "proxies" (as such terms are used in the proxy rules of the Securities and Exchange Commission) to vote, or advise or knowingly influence any Person (other than a controlled Affiliate of any LBHI Party) with respect to the voting of, any voting securities of AVB or any of its subsidiaries;
 - 3.1.5 Call, or seek to call, a meeting of the shareholders of AVB or initiate any shareholder proposal for action by the shareholders of AVB;
- 3.1.6 Form, join or in any way participate in a Group (within the meaning of Section 13(d)(3) of the Exchange Act) (other than with another LBHI Party or an Affiliate of any LBHI Party, or any direct or indirect subsidiary, of any LBHI Party), with respect to any voting securities of AVB;

4

- 3.1.7 Otherwise act, alone or in concert with others, to seek to control or influence the board of directors of AVB, or the management or policies of AVB (including, without limitation, the submission of nominees for election to the board of directors of AVB);
 - 3.1.8 Publicly disclose any intention, plan or arrangement prohibited by, or inconsistent with, the foregoing;
- 3.1.9 Advise or knowingly assist or encourage or enter into any discussions, negotiations, agreements or arrangements with any other Person or Group (within the meaning of Section 13(d)(3) of the Exchange Act) in connection with the foregoing; or
- 3.1.10 Propose, seek or request permission to do any of the foregoing, request to amend or waive any provision of this <u>Article 3</u> (including, without limitation, this clause 3.1.10), make or seek permission to make any public announcement with respect to any of the foregoing or take any action that such Person reasonably believes will require AVB to make a public announcement regarding the possibility of a business combination, merger or other type or transaction described above.

- 4.1 From and after the date hereof through and including the first anniversary of the date hereof (but in any event only so long as the Standstill Period is continuing), the LBHI Parties shall vote (including, through the execution of one or more written consents, if applicable) all common equity securities of AVB with respect to which the LBHI Parties have the power to vote (including the AVB Equity Consideration), in accordance with the recommendations of the board of directors of AVB with respect to any action, proposal or other matter to be voted on by the respective common equity holders of AVB; provided that, so long as each of the LBHI Parties is and has at all times been in compliance with the provisions of Article 3 of this Agreement (other than any unintentional noncompliance that was cured as promptly as practicable upon the applicable LBHI Parties becoming aware of such noncompliance), the LBHI Parties may vote their common equity securities of AVB in their sole discretion with respect to Extraordinary Transactions if and to the extent submitted to a vote of AVB common equity holders.
- 4.2 Following the first anniversary of the date hereof and continuing until the termination of the Standstill Period, the LBHI Parties shall vote (including, through the execution of one or more written consents, if applicable) all common equity securities of AVB held by the LBHI Parties (including the AVB Equity Consideration): (i) in accordance with the recommendation of the board of directors of AVB with respect to (A) any election of directors, (B) compensation matters and matters relating to equity or other incentive plans (including the adoption of any new plan or the amendment of any existing plan, including to increase the amount of equity or other compensation issuable thereunder), and (C) any amendment to AVB's articles of incorporation to increase the authorized capital stock, (ii) on all shareholder proposals, in one of the following two manners, at the election of the LBHI Parties (x) proportionally in accordance with the votes of other shareholders of AVB or (y) in accordance with the

5

recommendation of the board of directors of AVB, and (iii) on all other matters, in the sole and absolute discretion of the LBHI Parties.

ARTICLE 5 TRANSFERS

5.1 Until the one-year anniversary of the last day of the Standstill Period, prior to and as a condition to any Transfer (other than solely in the case of a Qualified Successor Transfer) by a LBHI Party of AVB Common Shares (including the AVB Equity Consideration) to any Person or Group (within the meaning of Section 13(d)(3) of the Exchange Act) that would result in such Person or Group acquiring more than 5% of AVB's outstanding common equity securities from the LBHI Parties or their Affiliates, such Person or Group shall be required to execute a joinder to this Agreement, in the form attached hereto as Exhibit A, pursuant to which such Person or Group agrees to become a Party to this Agreement and subject to the restrictions applicable to a LBHI Party and otherwise become a Party for all purposes of this Agreement; provided that no such Transfer(s) shall relieve the transferring LBHI Party or LBHI Parties from their obligations under this Agreement.

ARTICLE 6 REPRESENTATIONS AND WARRANTIES OF AVB

- 6.1 AVB hereby represents and warrants to the other Parties as follows:
- 6.1.1 AVB has been duly formed, is validly existing, and, where such concept is applicable, is in good standing under the laws of its jurisdiction of organization. AVB has all requisite power and authority to execute and deliver this Agreement and to perform its obligations under this Agreement.
- 6.1.2 The execution and delivery by AVB of this Agreement and the performance by AVB of its obligations under this Agreement do not and will not conflict with or violate any provision of, or require the consent or approval of any Person (except for any such consents or approvals which have been obtained) under, (i) applicable Law, or (ii) the organizational documents of AVB.
- 6.1.3 The execution and delivery by AVB of this Agreement and the performance by AVB of its obligations under this Agreement have been duly authorized by all necessary corporate or other analogous action on the part of AVB. This Agreement has been duly executed and delivered by AVB and, assuming the due authorization, execution and delivery by the other Parties hereto, constitutes a legal, valid and binding obligation of AVB, enforceable against AVB in accordance with its terms, subject to bankruptcy, insolvency and other Laws of general applicability relating to or affecting creditors' rights and to general principles of equity.

ARTICLE 7 REPRESENTATIONS AND WARRANTIES OF LBHI AND ARCHSTONE

7.1 Each of LBHI and Archstone hereby severally, and not jointly and severally, represents and warrants to the other Parties as follows:

6

- 7.1.1 Each LBHI Party has been duly formed, is validly existing, and, where such concept is applicable, is in good standing under the laws of its jurisdiction of organization. LBHI and Archstone have all requisite power and authority to execute and deliver this Agreement and each LBHI Party has all requisite power and authority to perform its obligations under this Agreement.
- 7.1.2 The execution and delivery by LBHI and Archstone of this Agreement and the performance by each of the LBHI Parties of their obligations under this Agreement do not and will not conflict with or violate any provision of, or require the consent or approval of any Person (except for any such consents or approvals which have been obtained) under, (i) applicable Law, or (ii) the organizational documents of any LBHI Party.
- 7.1.3 The execution and delivery by LBHI and Archstone of this Agreement and the performance by LBHI and Archstone of their obligations under this Agreement have been duly authorized by all necessary corporate or other analogous action on the part of LBHI and Archstone. This Agreement has been duly executed and delivered by LBHI and Archstone and, assuming the due authorization, execution and delivery by the other parties hereto, constitutes a legal, valid and binding obligation of LBHI and Archstone, enforceable against LBHI and Archstone in accordance with its terms, subject to bankruptcy, insolvency and other laws of general applicability relating to or affecting creditors' rights and to general principles of equity.
- 7.1.4 Each of LBHI and Archstone acknowledges that the securities included in the AVB Equity Consideration have not been registered under the Securities Act or any state securities laws.

ARTICLE 8 MISCELLANEOUS PROVISIONS

8.1 Notices. All notices, consents and other communications hereunder shall be in writing (including telecopy or similar writing) and shall be deemed to have been duly given (a) when delivered by hand or by Federal Express or a similar overnight courier to (or if that day is not a Business Day, or if delivered after 5:00 p.m., New

York, New York time on a Business Day, on the first following day that is a Business Day), (b) five (5) days after being deposited in any United States Post Office enclosed in a postage prepaid, registered or certified envelope addressed to, or (c) when successfully transmitted by facsimile to, the Party for whom intended, at the address or facsimile number for such Party set forth in Section 16.1 of the Asset Purchase Agreement.

- 8.2 <u>LBHI Parties.</u> LBHI shall cause each LBHI Party to perform and observe all obligations applicable to such LBHI Party hereunder.
- 8.3 Entire Agreement. This Agreement sets forth the entire agreement and understanding of the Parties in respect of the provisions contained herein and supersedes all prior discussions, negotiations, agreements, arrangements and understandings, whether oral or written, relating to the subject matter hereof and thereof. There are no warranties, representations or other agreements between the Parties in connection with the subject matter of this Agreement, except as specifically set forth in this Agreement.

7

8.4 <u>Amendments and Waivers</u>.

- 8.4.1 Any provision of this Agreement may be amended or modified only by a written instrument signed by all of the Parties hereto.
- 8.4.2 No waiver hereunder shall be valid or binding unless set forth in writing and duly executed by the Party against whom enforcement of the waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the Party granting such waiver in any other respect or at any other time. Neither the waiver by any of the Parties of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the Parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. Except as otherwise provided herein, no action taken pursuant to this Agreement, including any investigation by or on behalf of any Party, shall be deemed to constitute a waiver by the Party taking such action of compliance with any representations, warranties, covenants or agreements contained in this Agreement.
- 8.5 Governing Law. This Agreement shall be construed, performed and enforced in accordance with the laws of the State of Maryland (without giving effect to its principles or rules of conflict of laws to the extent such principles or rules would require or permit the application of the laws of another jurisdiction).
- 8.6 Remedies; Specific Performance. The Parties hereto agree that monetary damages would not be an adequate remedy in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms. It is expressly agreed that the Parties hereto shall be entitled to equitable relief, including injunctive relief and specific performance of the terms hereof, this being in addition to any other remedies to which they are entitled at law or in equity.
- 8.7 WAIVER OF TRIAL BY JURY. EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY OR CLAIM WHICH MAY ARISE UNDER THIS AGREEMENT IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE IT HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT. THIS WAIVER MAY NOT BE MODIFIED EITHER ORALLY OR IN WRITING (OTHER THAN BY A MUTUAL WRITTEN WAIVER SPECIFICALLY REFERRING TO THIS SECTION 8.7 AND EXECUTED BY EACH OF THE PARTIES HERETO). The scope of this waiver is intended to be all-encompassing of any and all disputes that may be filed in any court and that relate to the subject matter herein, including contract claims, tort claims, breach of duty claims and all other common law and statutory claims. In the event of litigation, this Agreement may be filed as a written consent to a trial by the court.

8

EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (I) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE EITHER OF SUCH WAIVERS, (II) IT UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF SUCH WAIVERS, (III) IT MAKES SUCH WAIVERS VOLUNTARILY, AND (IV) IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 8.7.

- 8.8 <u>Binding Effect</u>. This Agreement will be binding upon, inure solely to the benefit of and be enforceable by the Parties and their respective permitted successors and assigns.
- 8.9 Severability. If any term, provision, covenant or restriction of this Agreement is held by a court of competent jurisdiction or other Governmental Authority to be invalid, void or unenforceable, such term, provision, covenant or restriction shall, as to such jurisdiction, be ineffective to the extent of such prohibition or unenforceability without invalidating the remaining provisions of this Agreement, and the remainder of the terms, provisions, covenants and restrictions of this Agreement shall remain in full force and effect and shall in no way be affected, impaired or invalidated.
- 8.10 <u>Counterparts.</u> This Agreement may be executed and delivered in multiple counterparts, each of which shall be deemed an original, and all of which together shall constitute one and the same instrument. It is the express intent of the Parties to be bound by the exchange of signatures on this Agreement via facsimile or electronic mail via the portable document format (PDF). A facsimile or other copy of a signature shall be deemed an original. This Agreement shall become effective when each Party hereto shall have received a counterpart hereof signed by all of the other Parties hereto, this Agreement shall have no effect and no Party shall have any right or obligation hereunder (whether by virtue of any other oral or written agreement or other communication).
- 8.11 Third Parties. Except as otherwise expressly provided herein, no provision of this Agreement is intended or shall confer on any Person, other than the Parties (and their successors and permitted assigns), any rights under this Agreement and no other Person shall be entitled to rely thereon.
- 8.12 Assignment. Neither this Agreement nor any of the rights, interests or obligations hereunder shall be assigned, delegated or otherwise transferred by any Party (whether by operation of law or otherwise) without the prior written consent of the other Parties. Notwithstanding the foregoing, (i) AVB shall have the right to assign all or certain provisions of this Agreement, or any interest herein, and may delegate any duty or obligation hereunder, without the consent of the other Parties, to any Affiliate of AVB, and (ii) the LBHI Parties shall have the right to assign certain provisions of this Agreement, without the consent of the other Parties, only as specifically permitted by Articles 2 and 5 of this Agreement; provided that, in the case of each of clauses (i) and (ii), no such assignment or delegation shall relieve such Party of

any of its obligations hereunder. Any attempted assignment, delegation or transfer in violation of this Section 8.12 shall be null and void.

[Signature Page Follows]

10

IN WITNESS WHEREOF, the Parties have executed and delivered this Agreement as of the date set forth above.

A	VALONBAY COMMUNITIES, INC.
В	By: Name: Title:
Α	ARCHSTONE ENTERPRISE LP
В	By: Name: Title:
I	EHMAN BROTHERS HOLDINGS INC.
В	Name: Title:
[SIGNATURE PAGE TO AVALONBAY COMMUN	NITIES, INC. SHAREHOLDERS AGREEMENT]

EXHIBIT A

FORM OF JOINDER AGREEMENT

The undersigned is acquiring, simultaneously with the execution of this Joinder Agreement, [•] shares of [•] (the "Securities"), of AvalonBay Communities, Inc., a Maryland corporation ("AVB").

WHEREAS, as a condition to the acquisition of the Securities, the undersigned has agreed to join in a certain Shareholders Agreement (as it may be amended from time to time the "Shareholders Agreement"), dated as of [•], 2012 among AVB, Archstone Enterprise LP, a Delaware limited partnership ("Archstone"), Lehman Brothers Holdings, Inc., ("LBHI") and the other parties signatory thereto; and

WHEREAS, the undersigned understands that the execution of this Joinder Agreement is a condition precedent to the acquisition of the Securities.

NOW, THEREFORE, as an inducement to AVB and to the holder from whom or which the undersigned is acquiring the Securities to consummate the transfer of the Securities to the undersigned, the undersigned hereby agrees to join in the Shareholders Agreement and agrees to be bound by all of the terms, restrictions, obligations and agreements thereof applicable to an "LBHI Party" thereunder, and the undersigned hereby represents and warrants that:

The undersigned has been duly formed, is validly existing and, where such concept is applicable, is in good standing under the laws of its jurisdiction of organization. The undersigned has all requisite power and authority to execute and deliver this Joinder Agreement and the Shareholders Agreement and to perform its obligations hereunder and thereunder.

The execution and delivery by the undersigned of this Joinder Agreement and the Shareholders Agreement and the performance by the undersigned of its obligations hereunder and thereunder do not and will not conflict with or violate any provision of, or require the consent or approval of any Person (except for any such consents or approvals which have been obtained) under, (i) applicable Law (as such term is used in the Shareholders Agreement), or (ii) the organizational documents of the undersigned.

The execution and delivery by the undersigned of this Joinder Agreement and the performance by the undersigned of its obligations hereunder and under the Shareholders Agreement have been duly authorized by all necessary corporate or other analogous action on the part of the undersigned. This Agreement has been duly executed and delivered by the undersigned, and assuming the due authorization, execution and delivery by the other parties hereto, each of this Joinder Agreement and the Shareholders Agreement constitutes a legal, valid and binding obligation of the undersigned, enforceable against the undersigned in accordance with its terms, subject only to bankruptcy, insolvency and other laws of general applicability relating to or affecting creditors' rights and to general principles of equity.

			a [·]		
			Ву:	Name: Title:	
Accepted by:					
AvalonBay Commun	nities, Inc., a Maryland corpor	ation			
By: Name: Title:					
	[SIGNATURE PAGE TO JO	DINDER AGREEMENT TO AVALO	NBAY (COMMUNITIES, INC. SHAREHOLDER AGREEMENT]	

Consent of Independent Auditors

We consent to the incorporation by reference in the registration statements (Nos. 333-179720, 333-87063 and 333-107413) on Form S-3 and (Nos. 333-161258, 333-16837, and 333-115290) on Form S-8 of AvalonBay Communities, Inc. of our report dated November 26, 2012, with respect to the combined statement of revenue and certain expenses of the Archstone Portfolio for the year ended December 31, 2011, which report appears in the Form 8-K of AvalonBay Communities, Inc. dated November 26, 2012.

Our report on the combined statement of revenue and certain expenses of the Archstone Portfolio contains a paragraph that states that the combined statement of revenue and certain expenses was prepared for the purpose of complying with the rules and regulations of the Securities and Exchange Commission, as described in Note 1 to the combined statement of revenue and certain expenses, and it is not intended to be a complete presentation of the Archstone Portfolio's revenue and expenses.

/s/KPMG LLP Denver, Colorado November 26, 2012



PRESS RELEASE

Contact: Timothy J. Naughton

Chief Executive Officer and President AvalonBay Communities, Inc. 703-317-4620

For Immediate Release

AVALONBAY ANNOUNCES AGREEMENT TO ACQUIRE 40% OF ARCHSTONE; AFFIRMS 2012 FULL YEAR OUTLOOK; AND ANNOUNCES EXPECTED 2013 DIVIDEND INCREASE

November 26, 2012 — ARLINGTON, VA — AvalonBay Communities, Inc. (NYSE: AVB) along with Equity Residential (NYSE: EQR) announced today that the companies entered into an agreement ("Agreement") with Lehman Brothers Holdings, Inc. ("Lehman") to acquire the assets and liabilities of Archstone Enterprise LP ("Archstone"), which consist principally of a portfolio of high quality apartment communities in major markets in the United States, for approximately \$16 billion in the aggregate. Under the terms of the Agreement, AvalonBay will acquire approximately 40% of Archstone's assets and liabilities. The transaction is expected to close during the first quarter of 2013.

The combined purchase price for the assets consists of (i) \$2.7 billion in cash, (ii) a stipulated fixed number of shares of AvalonBay and Equity Residential common stock valued at \$3.8 billion as of the market's close on Friday, November 23, 2012, (iii) the assumption of approximately \$9.5 billion of consolidated and unconsolidated debt at Archstone's share, and (iv) preferred equity units with a redemption value of \$330 million as of September 30, 2012. Of the debt to be assumed, approximately \$8.6 billion is held by the Federal National Mortgage Association ("Fannie Mae") and the Federal Home Loan Mortgage Corporation ("Freddie Mac"), each of which has provided written consent to the assumption of this debt by AvalonBay and Equity Residential, subject to satisfactory documentation. The debt from Fannie Mae and Freddie Mac, as well as the other debt to be assumed, includes certain tax exempt bonds and other single asset loans for which additional consents will be needed from parties such as bond issuers, joint venture partners and debt servicers. The weighted average Capitalization Rate (as defined later in this press release) for the entire portfolio is projected to be in the high 4% range.

AvalonBay's portion of the purchase price consists of (i) \$669 million in cash, (ii) 14,889,706 shares of AvalonBay common stock, valued at \$1.9 billion as of the market's close on Friday, November 23, 2012 of \$128.54 per share, (iii) the assumption of indebtedness with a face value of approximately \$3.9 billion, (iv) obligations related to preferred equity units with a redemption value of \$132 million as of September 30, 2012, and (v) the assumption of 40% of all other liabilities, known and unknown, of Archstone, other than certain excluded liabilities. The \$3.9 billion debt assumption consists of (i) \$3.7 billion in principal amount for consolidated borrowings and (ii) \$238 million in principal amount for our proportionate share of debt related to unconsolidated joint ventures. Contemporaneously with entering into the Agreement with the seller, AvalonBay has also obtained a commitment from Goldman Sachs Lending Partners LLC

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

to provide a \$2.2 billion bridge loan facility, which has been sized appropriately should the additional lender consents, noted above, not be received.

Commenting on the Agreement, Tim Naughton, AvalonBay's CEO and President said, "This acquisition accelerates our strategic growth vision of more deeply penetrating our core, high barrier-to-entry coastal markets. This is a rare opportunity to acquire a high quality portfolio of apartment communities concentrated in our markets, to better achieve our geographic portfolio allocation goals, to enhance operating efficiencies and to further advance our multi-brand strategy."

Transaction Details

Under the agreement, AvalonBay will purchase the following:

• 66 apartment communities, containing 22,222 apartment homes, of which six communities are under construction and are expected to contain 1,666 apartment homes upon completion, as detailed in the following table:

Market	Properties	Number of Homes		
New England	2	548		
Metro NY/NJ	4	1,573		
Mid-Atlantic	18	5,311		
Pacific Northwest	2	588		
Northern California	9	3,251		
Southern California	23	8,507		
Non-Core	2	778		
Development	6	1,666		
Total	66	22,222		

- Three parcels of land, which if developed as anticipated, are expected to contain a total of 968 apartment homes;
- · Interests in joint ventures in which we expect to be the general partner or managing member, which own 10 apartment communities, containing 2,040 apartment homes, of which one community is under construction and is expected to contain 157 apartment homes upon completion;

· Through a 40% ownership interest in joint venture arrangements with Equity Residential, certain other assets that will be jointly managed until sold or subsequently transferred to AvalonBay or Equity Residential.

The following table depicts the approximate allocation of AvalonBay's investment:

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

	_	Acquisition Value (1) \$'s in (000)s
Consolidated stabilized assets	\$	6,140,323
Development communities under construction		308,819
Land held for future development		49,800
Net equity in unconsolidated joint ventures plus allocable venture debt		410,112
Total	\$	6,909,054

⁽¹⁾ Value is based on the closing price of our common stock on November 23, 2012 and the debt to be assumed after a \$198 million adjustment to the face value of debt to reflect debt at fair market value.

The following table provides the percentage of Net Operating Income ("NOI" as defined later in this press release) generated during the nine months ended September 30, 2012, by region, for consolidated operating communities, as well as communities that were under construction, purchased or sold or for which substantial redevelopment is planned or that occurred during 2012, pro forma for the Archstone Portfolio (dollars in thousands):

		AvalonBay C	Communities	Archstone	e Portfolio (1)	Pro Forma		
Region/Portfolio (2) \$'s in (000)s	_	NOI YTD 2012 (3)	% of Total NOI	NOI YTD 2012 (3)	% of Total NOI	NOI YTD 2012 (3)	% of Total NOI	
New England	\$	99,277	19%	\$ 7,433	3%	\$ 106,710	14%	
Metro NY/NJ		156,135	29%	31,336	15%	187,47	25%	
Mid-Atlantic		71,401	13%	61,727	29%	133,128	3 18%	
Pacific Northwest		27,241	5%	5,471	3%	32,712	2 4%	
Northern California		102,335	19%	31,544	15%	133,879	18%	
Southern California		74,287	14%	72,910	34%	147,19	7 20%	
Non-Core and Other		447	0%	5,227	2%	5,674	1%	
Total NOI	\$	531,123	100%	\$ 215,648	100%	\$ 746,77	100%	

⁽¹⁾ NOI for the nine months ended September 30, 2012 for the Archstone Portfolio includes \$26 for communities under construction.

2012 Outlook Reaffirmed

AvalonBay reaffirms its previously disclosed fourth quarter and full year 2012 Earnings per Share — diluted ("EPS") and Funds from Operations attributable to common stockholders — diluted ("FFO") (as defined later in this press release) outlook.

• For the fourth quarter of 2012, including the impact of Hurricane Sandy, AvalonBay expects EPS of \$1.34 to \$1.39. AvalonBay expects EPS for the full year 2012 to be in the range of \$4.47 to \$4.52.

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

For the fourth quarter of 2012, including the impact of Hurricane Sandy, AvalonBay expects Projected FFO of \$1.40 to \$1.45. AvalonBay expects Projected FFO for the full year 2012 to be in the range of \$5.45 to \$5.50.

2013 Dividend Increase

AvalonBay expects to increase the common stock dividend for the first quarter of 2013 by a range of between 8% and 12%. This increase is supported by AvalonBay's current operating platform and recent operating trends, the expectation for continued favorable apartment fundamentals in 2013 and the acquisition of the Archstone portfolio described in this press release. The payment of any dividend, however, is subject to the approval of the Board of Directors and there can be no assurance that economic conditions generally and/or their impact on our operating results and payout ratio will not lead us to change our expectations.

Joint Venture Arrangements with Equity Residential

As previously discussed in this press release, some Archstone assets will be placed in joint venture arrangements co-owned by AvalonBay and Equity Residential. These assets include Archstone's interests in unconsolidated joint ventures that own properties in various U.S. markets as well as Archstone's interest in a portfolio of apartment communities in Germany. AvalonBay will maintain a 40% interest and Equity Residential will maintain a 60% interest in these joint venture arrangements. The companies will

⁽²⁾ NOI for the nine months ended September 30, 2012 for consolidated apartment communities, excluding NOI from apartment communities in joint ventures.

⁽³⁾ GAAP net income for AvalonBay, the Archstone Portfolio Acquisition, and on a pro forma basis, assuming the acquisition of the portfolio was \$301,178, \$43,425 and \$344,603, respectively, for the nine months ended September 30, 2012.

co-manage these assets until they are sold or assigned out to AvalonBay or Equity Residential. The combined gross value for both companies is approximately \$500 million, with a net value (after considering debt at share) of approximately \$170 million.

Other Matters

Lehman has agreed that, as of today, it will not sell or hedge shares of the Company's common stock until April 26, 2013. AvalonBay has agreed that it will enter into a Registration Rights Agreement with Lehman to help facilitate its sale of the shares acquired in the transaction.

In the event that the Agreement is terminated by Lehman due to the failure of AvalonBay and Equity Residential to perform their obligations under the Agreement, AvalonBay and Equity Residential will be required to pay to Lehman liquidated damages of \$800 million (or \$650 million if the agreement is terminated in the first 60 days). AvalonBay and Equity Residential have agreed that they will share any such liquidated damages in accordance with their 40% and 60% interests in the Agreement, unless one party is determined to be solely responsible for the termination.

Conference Call

There will be a conference call hosted by Timothy J. Naughton, CEO and President of AvalonBay, and David J. Neithercut, CEO and President of Equity Residential, at 4:30 pm Eastern this afternoon to discuss the transaction. To listen to the call, dial 1-866-589-2723 domestically and 1-706-645-4945 internationally, and use Conference ID: 74790864. This call will also be web cast. Links for the web cast can be found in the investor section of each company's website.

Advisors

Greenhill & Co., LLC acted as financial advisor to AvalonBay and also provided a fairness opinion. Goodwin Procter LLP served as AvalonBay's legal advisor on these transactions. Morgan Stanley & Co. LLC served as Equity Residential's exclusive financial advisor and Hogan Lovells US LLP and

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

Morrison & Foerster LLP as its legal advisors on these transactions. Gleacher & Company served as Lehman's lead financial advisor and Citigroup and J.P. Morgan Securities LLC also advised; Weil, Gotshal & Manges served as Lehman's legal advisor.

About AvalonBay Communities

AvalonBay Communities, Inc. is a real estate investment trust (a "REIT") focused on developing, redeveloping, acquiring and managing high-quality apartment communities in high barrier-to-entry markets of the United States. These markets are in the Northeast, Mid Atlantic, Pacific Northwest, Northern California and Southern California. As of September 30, 2012, the Company owned or held a direct or indirect ownership interest in 205 apartment communities containing 60,101 apartment homes in nine states and the District of Columbia of which 22 communities were under construction and seven communities were under reconstruction.

Forward-Looking Statements

This release contains forward-looking statements within the meaning of Section 27A of the Securities Act of 1933, as amended, and Section 21E of the Securities Exchange Act of 1934, as amended, with respect to the financial condition, results of operations and businesses of AvalonBay and the Archstone acquisition. These forward-looking statements, which can be identified by the use of words such as "expects," "plans," "estimates," "anticipates," "projects," "intends," "believes," outlook" and similar expressions that do not relate to historical matters, are based on current expectations, forecasts and assumptions which may not be realized and involve risks and uncertainties, many of which cannot be predicted with accuracy and some of which might not even be anticipated, that could cause actual outcomes and results, financial and otherwise, to differ materially, including statements related to the closing of the Archstone acquisition. Risks and other factors related to the Company that might cause such differences include, among others, the following: we may abandon development or redevelopment opportunities for which we have already incurred costs; adverse capital market conditions may affect our access to various sources of capital and/or cost of capital, which may affect our business activities, earnings and common stock price, among other things; changes in local employment conditions, demand for apartment homes, supply of competitive housing products, and other economic conditions may result in lower than expected occupancy and/or rental rates and adversely affect the profitability of our communities; delays in completing development, redevelopment and/or lease-up may result in increased financing and construction costs and may delay and/or reduce the profitability of a community; debt and/or equity financing for development, redevelopment or acquisitions of communities may not be available or may not be available on favorable terms; we may be unable to obtain, or experience delays in obtaining, necessary governmental permits and authorizations; and increases in costs of materials, labor or other expenses may result in communities that we develop or redevelop failing to achieve expected profitability. Any forward-looking statements or forecasts relating to the business, prospects, operating statistics or financial results relating to the Archstone acquisition are based on assumptions and are inherently speculative, are subject to substantial uncertainty, and the actual operating statistics and financial results may differ materially from AvalonBay's forecasts. Risks and other factors related to the Archstone acquisition that might cause such differences include, among other things the following, the Archstone acquisition may not close at the time or on the terms currently expected; assumptions concerning the availability and/or terms of financing, including among other things, obtaining lender consents to the assumption of indebtedness, related to the Archstone acquisition may not be realized; we may not be able to integrate the assets and operations acquired in the Archstone acquisition in a manner consistent with our assumptions and/or we may fail to achieve expected efficiencies and synergies; we may encounter liabilities for which we are responsible that were unknown at the time we agreed to the Archstone acquisition; our assumptions concerning risks relating to our lack of control of joint ventures and our ability to successfully dispose of certain assets may not be realized. In addition, AvalonBay's

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

forecasts are subject to a variety of additional factors and risks, including the risks set forth under "Risk Factors" in AvalonBay's Form 10-K, and in AvalonBay's other periodic and Form 8-K filings with the Securities and Exchange Commission. AvalonBay undertakes no obligation to publicly update any forward-looking statements, whether as a result of new information, future events or the occurrence of unanticipated events except as required by applicable law.

Definitions and Reconciliations

This release includes the definitions and reconciliations of the following non-GAAP financial measures and other capitalized terms:

<u>Capitalization Rate</u> is defined by AvalonBay as NOI as a percentage of the purchase price after making adjustments to historical NOI to reflect anticipated growth rates in rental rates as well as reserves for capital expenditures, property management fees, and other market conventions to estimate future stabilized NOI.

FFO is determined based on a definition adopted by the Board of Governors of the National Association of Real Estate Investment Trusts ("NAREIT"). FFO is calculated by AvalonBay as net income or loss attributable to common stockholders computed in accordance with GAAP ("Net Income"), adjusted for gains or losses on sales of previously depreciated operating communities, extraordinary gains or losses (as defined by GAAP), cumulative effect of a change in accounting principle, impairment write-downs of depreciable real estate assets, write-downs of investments in affiliates which are driven by a decrease in the value of depreciable real estate assets held by the affiliate and depreciation of real estate assets, including adjustments for unconsolidated partnerships and joint ventures. Management generally considers FFO to be an appropriate supplemental measure of operating performance because, by excluding gains or losses related to dispositions of previously depreciated operating communities and excluding real estate depreciation (which can vary among owners of identical assets in similar condition based on historical cost accounting and useful life estimates), FFO can help one compare the operating performance of a company's real estate between periods or as compared to different companies.

<u>Projected FFO</u> as provided within this release in AvalonBay's outlook, is calculated on a basis consistent with historical FFO, and is therefore considered to be an appropriate supplemental measure to projected Net Income from projected operating performance. A reconciliation of the range provided for Projected FFO per share (diluted) for the third quarter and full year 2012 to the range provided for projected EPS (diluted) is as follows:

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

	Low range			High range
Projected EPS (diluted) - Q4 2012	\$	1.34	\$	1.39
Projected depreciation (real estate related)		0.67		0.72
Projected gain on sale of operating communities		(0.61)		(0.66)
Projected FFO per share (diluted) - Q4 2012	\$	1.40	\$	1.45
Projected EPS (diluted) - Full Year 2012	\$	4.47	\$	4.52
Projected depreciation (real estate related)		2.74		2.79
Projected gain on sale of operating communities		(1.76)		(1.81)
Projected FFO per share (diluted) - Full Year 2012	\$	5.45	\$	5.50

NOI is defined by AvalonBay as total property revenue less direct property operating expenses (including property taxes), and excludes corporate-level income (including management, development and other fees), corporate-level property management and other indirect operating expenses, investments and investment management expenses, expensed development and other pursuit costs, net interest expense, gain (loss) on extinguishment of debt, general and administrative expense, joint venture income (loss), depreciation expense, impairment loss on land holdings, gain on sale of real estate assets and income from discontinued operations. AvalonBay considers NOI to be an appropriate supplemental measure to Net Income of operating performance of a community or communities because it helps both investors and management to understand the core operations of a community or communities prior to the allocation of corporate-level property management overhead or general and administrative costs. This is more reflective of the operating performance of a community, and allows for an easier comparison of the operating performance of single assets or groups of assets. In addition, because prospective buyers of real estate have different overhead structures, with varying marginal impact to overhead by acquiring real estate, NOI is considered by many in the real estate industry to be a useful measure for determining the value of a real estate asset or groups of assets. A reconciliation of NOI (from continuing operations) to Net Income is as follows (dollars in thousands):

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved

	For the nine months ended September 30, 2012					
	Company Historical		Archstone Acquisition/ Pro Forma		Company Pro Forma	
Income from continuing operations	\$	203,259	\$	43,425	\$	246,684
Discontinued operations		97,919		_		97,919
Net income	\$	301,178	\$	43,425		344,603
		24.040				24.040
Indirect operating expenses, net of corporate income		24,049		_		24,049
Investment and investment management expense		4,526		_		4,526
Expensed development and other pursuit costs		1,749		_		1,749
Interest expense, net		100,804		57,139		157,943
Loss on extinguishment of debt, net		1,179		_		1,179
General and administrative expense		26,398		1,605		28,003
Equity in income of unconsolidated entities		(9,801)		7,274		(2,528)
Depreciation expense		193,434		106,206		299,640
Impairment loss - land holdings		_		_		_
Gain on sale of real estate assets		(95,329)		_		(95,329)
Income from discontinued operations		(2,870)		_		(2,870)
Gain on acquisition of unconsolidated entity		(14,194)		_		(14,194)
Net operating income	\$	531,123	\$	215,648	\$	746,771

Copyright © 2012 AvalonBay Communities, Inc. All Rights Reserved